# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

# **FORM 10-Q**

(Mark One)

x QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

FOR THE QUARTERLY PERIOD ENDED SEPTEMBER 30, 2024

OR

0 TRANSITION REPORT PURSUANT SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

FOR THE TRANSITION PERIOD FROM \_\_\_\_\_\_ TO \_\_\_\_\_ COMMISSION FILE NUMBER 001-41261

# DIRECT DIGITAL HOLDINGS, INC.

(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation or organization)

1177 West Loop South, Suite 1310

Houston, Texas

(Address of principal executive offices)

(832) 402-1051

(Registrant's telephone number, including area code)

Securities registered pursuant to Section 12(b) of the Act:							
Title of Each Class:         Trading symbol(s)         Name of Each Exchange on Which Registered:							
Class A Common Stock, par value \$0.001 per share	DRCT	Nasdaq Capital Market					

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes x No o

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes x No o

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company or an emerging growth company. See the definitions of "large accelerated filer", "accelerated filer", "smaller reporting company" and "emerging growth company" in Rule 12b-2 of the Exchange Act. (Check one):

Large Accelerated Filer	0	Accelerated filer	0
Non-accelerated filer	x	Smaller reporting company	х
Emerging growth company	x		

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act .0

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes o No x

As of November 12, 2024, there were 3,799,901 shares of the registrant's Class A Common Stock outstanding, par value \$0.001 per share, and 10,868,000 shares of the registrant's Class B Common Stock outstanding, par value \$0.001 per share.

87-2306185 (I.R.S. Employer Identification No.)

> 77027 (Zip code)

#### TABLE OF CONTENTS

		PAGE
ITEM		
	Part I. Financial Information	3
<u>1.</u>	FINANCIAL STATEMENTS (UNAUDITED)	
	Condensed Consolidated Balance Sheets as of September 30, 2024 and December 31, 2023	3
	Condensed Consolidated Statements of Operations for the Three and Nine Months Ended September 30, 2024 and 2023	4
	Condensed Consolidated Changes in Stockholders' Deficit for the Three and Nine Months Ended September 30, 2024 and 2023 Condensed Consolidated Statements of Cash Flows for the Nine Months Ended September 30, 2024 and 2023	5 6
	Notes to Condensed Consolidated Financial Statements	7
<u>2.</u>	Management's Discussion and Analysis of Financial Condition and Results of Operations	28
<u>3.</u>	Quantitative and Qualitative Disclosures About Market Risk	42
<u>4.</u>	Controls and Procedures	42
	Part II. Other Information	43
<u>1.</u>	Legal Proceedings	43
<u>1A.</u>	Risk Factors	43
<u>2.</u>	Unregistered Sales of Equity Securities and Use of Proceeds	44
<u>3.</u>	Defaults Upon Senior Securities	44
<u>4.</u>	Mine Safety Disclosures	44
<u>5.</u>	Other Information	44
<u>6.</u>	Exhibits	45
<u>Signatures</u>		47

#### PART 1. FINANCIAL INFORMATION

#### ITEM 1. FINANCIAL STATEMENTS DIRECT DIGITAL HOLDINGS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED BALANCE SHEETS (in thousands, except share and par value amounts)

CURRENT ASSETSS0.4080.50.5.116Accounts receivable, net of provision for eredit losses of \$385 and \$3416.2877.207Propia degenees and other current assets11.1287.890Total current assets11.0807.007Propit sequipment and software, net4.0009.900Goodwill6.5016.5026.502Diard burrent assets10.21911.048Defered tax asset, net		Septe	mber 30, 2024	December 31, 2023
CURRENT ASSETSS0.4080.50.5.116Accounts receivable, net of provision for eredit losses of \$385 and \$3416.2877.207Propia degenees and other current assets11.1287.890Total current assets11.0807.007Propit sequipment and software, net4.0009.900Goodwill6.5016.5026.502Diard burrent assets10.21911.048Defered tax asset, net		J)	Jnaudited)	
Cash and cash equivalentsS40.03S5.116Accounts receivable, net of provision for erdit losses of 535 and 534	ASSETS			
Account receivable, net of provision for circli losses of \$385 and \$3446.03777.207Prepaid expenses and other current assets1.112789Total current assets4.04824.0482Property, expipment and software, net4.0996.050Gado strutt assets6.506.503Defend as asset, net6.12191.1044Defend as asset, net6.1219Operating loss reflocif-base assets7.1737.1737Total asset1.12191.1044Defend asset, net-7.17377.1737Total assets-7.4737.1737Total assets-7.4737.1737Total assets-7.4737.1737Total assets-7.4737.1737Total assets-7.4747.1337Total assets-7.4747.1337Chrest Instruction Structure Instruction8.813.816Accrued liabilities-4.813.816Lability related to tax receivable agreement, current portion1.831.124Current labilities, current portion-4.52993.9302Total current labilities, current portion-4.5337.537Total current labilities, current portion-4.5393.9302Total current labilities, current portion-4.5393.9302Total current labilities, net of current portion-4.5393.9302Total current labilities, net of current portion-4.5393.930	CURRENT ASSETS			
Preprior1.112799Total current assets1.112799Total current assets40043.082Property, equipment and software, net4006.520Godovill6.5206.520Intang ble assets, net-6.612Operating less right-Gine assets873788Related party receivable1.7277.737Other less right-Gine assets873708Related party receivable1.7277.1373Other less right-Gine assets83.628CURRENT LIABILITIES56.4228.33,262CURRENT LIABILITIES83.6163.616Current assets966.4383.616Liability related to tax receivable agreement, current portion83.616Liability related to tax receivable agreement, net of current portion1.6331.026Current lashilities9763.8403.620Income taxes payable9963.443.620Concent asset payable9963.4203.620Income taxes payable9963.4203.620Income taxes payable9963.4203.620Income taxes payable9973.6203.620Income taxes payable9963.4203.620Income taxes payable9963.4203.620Income taxes payable9963.6203.620Income taxes payable9963.6203.620Income taxes payable9973.6203.620 <tr< td=""><td>Cash and cash equivalents</td><td>\$</td><td>4,087</td><td>\$ 5,116</td></tr<>	Cash and cash equivalents	\$	4,087	\$ 5,116
Total current assets11.46643.082Property, expirament and software, net409599Goodwill6.5206.520Inanghie assets, net	Accounts receivable, net of provision for credit losses of \$385 and \$344		6,287	37,207
Poperty, equipment and software, net	Prepaid expenses and other current assets		1,112	759
Goodwill         6.520         6.520           Intangibe assets, net         10,219         11,648           Defered tas asset, net         -         6,132           Operating lasse right-of-lass assets         873         788           Related party receivable         1737         1737           Other long-term assets         47         130           Total assets         \$         3.1,201         \$         70,672           LABLITIES AND STOCKIOLDERS DEFICIT         ************************************	Total current assets		11,486	43,082
Intagble asets, net         10,219         11,684           Deferring lases right-fo-use asets         737         738           Related party receivable         1,737         71,737           Other long-term asets         1,737         71,737           Total assets         \$ 31,201         \$ 70,000           Total assets         \$ 31,201         \$ 70,000           CURRENT LABILITIES         \$ 0,642         \$ 33,202           Accounts payable         818         3,816           Lability related to tax receivable agreement, current portion         41         41           Current Mainfulfies         976         381           Operating lases rights         976         381           Operating lases rights, current portion         1183         1265           Liability related to tax receivable agreement, net of current portion         150         28,578           Liability related to tax receivable agreement, net of current portion         150         28,578           Liability related to tax receivable agreement, net of current portion         822         773           Total current liabilities, eut of current portion         823         743           Long-term portion         823         743           Lopatrem debt, net of current portion         45	Property, equipment and software, net		409	599
Deferred tax asset, net-6.132Operating lease right-of us assets873778Related party receivable1.73771.737Other long-term assets4.771.730Total assets3.1299.070.672IARLIFLES AND STOCKIDDERS' DEFICITCURRENT LIABILITIES83.3292Accound payable\$6.53.5292Liability related to tax receivable agreement, current portion4.14.141Current inabilities, current portion4.1631.262Deferred revenues9.993.431.262Deferred revenues9.993.431.262Liability related to tax receivable agreement, eurrent portion1.632.85.78Liability related to tax receivable agreement, net of current portion1.632.85.78Liability related to tax receivable agreement, net of current portion3.8123.816Courter tabilities, net of current portion1.902.85.78Liability related to tax receivable agreement, net of current portion3.8123.812Courter tabilities, net of current portion3.8227.733Total Liabilities, net of current portion4.3293.823Courter tabilities, net of current portion3.8227.733<	Goodwill		6,520	6,520
Operating lease right-of-use assets         873         788           Related party receivable         1,737         1,737           Total assets         8         31,201         \$         70,672           LIABILITIES AND STOCKHOLDERS' DEFICIT           33,926           Accounts payable         \$         6,652         \$         33,926           Accounts payable         \$         6,657         1,478           Defered revenues         \$         6,667         1,478           Defered revenues         \$         967         3,810           Income taxes payable         \$         99         34           Total current liabilities         \$         3,802         \$           Income taxes payable         \$         9         34         \$           Total current trabilities         \$         \$         \$         \$ <td< td=""><td>Intangible assets, net</td><td></td><td>10,219</td><td>11,684</td></td<>	Intangible assets, net		10,219	11,684
Acta ad party receivable         1,737         1,737           Other long-term assets         4         130           Total assets         5         31.201         5         70.607           LIMBILITIES AND STOCKIOLDERS' DEFICT         UCREANT LIABILITIES         5         6.452         \$         33.926           Accruad liabilities         881         3.816         3.816         3.816         3.816           Liability related to tax receivable agreement, current portion         41         414           Current inabilities of long-term debt         36.667         1.478           Deferred revenues         976         381         126           Income taxes payable         999         34         126           Income taxes payable         991         34         126           Income taxes payable         991         34         126         128         743           Total current portion         150         28.578         128         743         126         7435	Deferred tax asset, net		_	6,132
Other long-term assets         47         130           Total assets         \$         31,291         \$         70,672           LIABLITIES AND STOCKHOLDERS' DEFICT           33,926           Accounts payable         \$         6,452         \$         33,926           Current matritics of long-term debt         \$         6,657         1,478           Deferred revenues         \$         76         381           Operating lease liabilities, eurrent portion         \$         126         126           Incone taxes payable         \$         99         94           Total abilities         \$         150         28,578           Liability related to tax receivable agreement, et of current portion         \$         5,201           Operating lease liabilities, net of current portion	Operating lease right-of-use assets		873	788
S         31,291         S         70,672           LABILITIES AND STOCKHOLDERS' DEFICIT               33,926          33,926          33,926          33,926          33,926          38,816         3,816          3,816	Related party receivable		1,737	1,737
LABLITTIES AND STOCKHOLDERS' DEFICIT         Image: margin and state and s	Other long-term assets		47	130
CURRENT LIABILITIES       \$ 6,52 \$ 33,926         Accrued liabilities       881       3,816         Accrued liabilities       811       41         Liability related to tax receivable agreement, current portion       41       41         Current maturities of long-term debt       36,667       11,478         Deferred revenues       976       381         Operating lease liabilities, current portion       183       126         Income taxes payable       99       34         Total current liabilities       150       28,578         Liability related to tax receivable agreement, net of current portion	Total assets	\$	31,291	\$ 70,672
Accounts payable         \$         6.452         \$         33.926           Accounts payable         881         3.816           Liability related to tax receivable agreement, current portion         41         44           Current maturities of long-term debt         36.667         1.478           Deferred revenues         976         381           Operating lease liabilities, current portion         183         126           Inome taxes payable         999         34           Total current liabilities         45.299         93.9802           Long-term debt, net of current portion         150         25.578           Liability related to tax receivable agreement, net of current portion         -         5.201           Operating lease liabilities, net of current portion         832         773           Total liability related to tax receivable agreement, net of current portion         -         5.201           Operating lease liabilities, net of current portion         832         773           Total liabilitie         46.281         74.354           COMMITMENTS AND CONTINGENCIES (Note 9)         -         -           STOCKHOLDERS' DEFICIT         -         -           Class D Common Stock, \$0.001 par value per share, 160,000,000 shares authorized, 3,795,199 and 3,478,776 share	LIABILITIES AND STOCKHOLDERS' DEFICIT			
Accrued liabilities         881         3,816           Liability related to tax receivable agreement, current portion         41         44           Current maturities of long-term debt         36,667         1,478           Deferred revenues         976         381           Operating lease liabilities, current portion         183         126           Income taxes payable         99         34           Total current liabilities         45,299         39,800           Long-term debt, net of current portion         150         28,578           Liability related to tax receivable agreement, net of current portion         52,010         733           Operating lease liabilities, net of current portion         832         7733           Total liabilities         46,281         74,354           COMMITMENTS AND CONTINGENCIES (Note 9)         4         74,354           STOCKHOLDERS' DEFICIT         1         11           Class B Common Stock, \$0,001 par value per share, 160,000,000 shares authorized, 3,795,199 and 3,478,776 shares issued and outstanding, respectively         4         3           Class B Common Stock, \$0,001 par value per share, 20,000,000 shares authorized, 3,795,199 and 3,478,776 shares issued and outstanding, respectively         4         3           Class B Common Stock, \$0,001 par value per share, 160,000,000 shares autho	CURRENT LIABILITIES			
Liability related to tax receivable agreement, current portion4141Current maturities of long-term debt36,6671.478Deferred revenues976381Operating lease liabilities, current portion183126Income taxes payable9934Total current liabilities45,29939,802Long-term debt, net of current portion15028,578Liability related to tax receivable agreement, et of current portion15028,578Liability related to tax receivable agreement, net of current portion832773Total liabilities, net of current portion832774354COMMITMENTS AND CONTINGENCIES (Note 9)434STOCKHOLDERS' DEFICIT111Class B Common Stock, \$0.001 par value per share, 160,000,000 shares authorized, 3,795,199 and 3,478,776 shares issued and outstanding, respectively43Class B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 1,96,80,000 shares issued and outstanding, respectively3,4813,067Accumulated deficit(6,593)(2,538)(2,538)Noncontrolling interest(11,893)(4,225)Total stockholders' deficit(14,990)(3,682)	Accounts payable	\$	6,452	\$ 33,926
Current maturities of long-term debt         36,667         1,478           Deferred revenues         976         381           Operating lease liabilities, current portion         183         126           Income taxes payable         99         34           Total current liabilities         45,299         39,802           Long-term debt, net of current portion         150         28,578           Liability related to tax receivable agreement, net of current portion         -         5,201           Operating lease liabilities, net of current portion         832         773           Total current liabilities         46,281         74,354           COMMITMENTS AND CONTINGENCIES (Note 9)         -         520           STOCKHOLDERS' DEFICIT         -         4         3           Class A Common Stock, \$0.001 par value per share, 160,000,000 shares authorized, 3,795,199 and 3,478,776 shares issued and outstanding, respectively         4         3           Class A Common Stock, \$0.001 par value per share, 160,000,000 shares authorized, 10,868,000 shares issued and outstanding, respectively         4         3           Class A Common Stock, \$0.001 par value per share, 160,000,000 shares authorized, 10,868,000 shares issued and outstanding, respectively         4         3           Class A Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 share	Accrued liabilities		881	3,816
Deferred revenues         976         381           Operating lease liabilities, current portion         183         126           Income taxes payable         99         34           Total current liabilities         45,299         39,802           Long-term debt, net of current portion         150         28,578           Liability related to tax receivable agreement, net of current portion         50         28,578           Liability related to tax receivable agreement, net of current portion	Liability related to tax receivable agreement, current portion		41	41
Operating lease liabilities, current portion         183         126           Income taxes payable         99         34           Total current liabilities         45,299         39,802           Long-term debt, net of current portion         150         28,578           Liability related to tax receivable agreement, net of current portion         -         5,201           Operating lease liabilities, net of current portion         832         773           Total liabilities         46,281         74,354           COMMITMENTS AND CONTINGENCIES (Note 9)         -         -           STOCKHOLDERS' DEFICIT         -         -           Class A Common Stock, \$0.001 par value per share, 160,000,000 shares authorized, 3,795,199 and 3,478,776 shares issued and outstanding, respectively         -         4           Additional paid-in capital         -         11         11           Additional paid-in capital         3,481         3,067         3.481         3,067           Accumulated deficit         (6,593)         (2,538)         3.481         3,067           Accumulated deficit         (6,593)         (2,538)         3.481         3,067           Accumulated deficit         (6,593)         (2,538)         3.481         3,067           Accumulated deficit	Current maturities of long-term debt		36,667	1,478
Income taxes payable9934Total current liabilities45,29939,802Long-term debt, net of current portion15028,578Liability related to tax receivable agreement, net of current portion-5,201Operating lease liabilities, net of current portion832773Total liability46,28174,354COMMITMENTS AND CONTINGENCIES (Note 9)43STOCKHOLDERS' DEFICITClass A Common Stock, \$0.001 par value per share, 160,000,000 shares authorized, 3,795,199 and 3,478,776 shares issued and outstanding, respectively43Additional paid-in capital3,4813,067Accumulated deficit(6,593)(2,538)Noncontrolling interest(11,893)(4,225)Total stockholders' deficit(14,990)(3,682)	Deferred revenues		976	381
Total current liabilities45,29939,802Long-term debt, net of current portion15028,578Liability related to tax receivable agreement, net of current portion-5,201Operating lease liabilities, net of current portion832773Total liabilities46,28174,354COMMITMENTS AND CONTINGENCIES (Note 9)STOCKHOLDERS' DEFICITClass A Common Stock, \$0.001 par value per share, 160,000,000 shares authorized, 3,795,199 and 3,478,776 shares issued and outstanding, respectively43Class B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding1111Additional paid-in capital3,4813,067-Accumulated deficit(6,593)(2,538)-Noncontrolling interest(11,893)(4,225)-Total stockholders' deficit(14,990)(3,682)	Operating lease liabilities, current portion		183	126
Long-term debt, net of current portion15028,578Liability related to tax receivable agreement, net of current portion—5,201Operating lease liabilities, net of current portion832773Total liabilities46,28174,354COMMITMENTS AND CONTINGENCIES (Note 9)7STOCKHOLDERS' DEFICITClass A Common Stock, \$0.001 par value per share, 160,000,000 shares authorized, 3,795,199 and 3,478,776 shares issued and outstanding, respectively43Class B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding1111Additional paid-in capital3,4813,0673,4813,067Accumulated deficit(6,593)(2,538)(2,538)14,25514,255Total stockholders' deficit(11,893)(4,225)15,382Total stockholders' deficit(14,990)(3,682)3,481	Income taxes payable		99	34
Liability related to tax receivable agreement, net of current portion–5,201Operating lease liabilities, net of current portion832773Total liabilities46,28174,354COMMITMENTS AND CONTINGENCIES (Note 9)STOCKHOLDERS' DEFICIT22Class A Common Stock, \$0.001 par value per share, 160,000,000 shares authorized, 3,795,199 and 3,478,776 shares issued and outstanding, respectively43Class B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding1111Additional paid-in capital3,4813,0673,4813,067Accumulated deficit(6,593)(2,538)(2,538)14,225Total stockholders' deficit(11,893)(4,225)14,900(3,682)	Total current liabilities		45,299	39,802
Operating lease liabilities, net of current portion832773Total liabilities46,28174,354COMMITMENTS AND CONTINGENCIES (Note 9)STOCKHOLDERS' DEFICITClass A Common Stock, \$0.001 par value per share, 160,000,000 shares authorized, 3,795,199 and 3,478,776 shares issued and outstanding, respectively43Class B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding1111Additional paid-in capital3,4813,067Accumulated deficit(6,593)(2,538)Noncontrolling interest(11,893)(4,225)Total stockholders' deficit(14,900)(3,682)	Long-term debt, net of current portion		150	28,578
Total liabilities       46,281       74,354         COMMITMENTS AND CONTINGENCIES (Note 9)	Liability related to tax receivable agreement, net of current portion		_	5,201
COMMITMENTS AND CONTINGENCIES (Note 9) STOCKHOLDERS' DEFICIT Class A Common Stock, \$0.001 par value per share, 160,000,000 shares authorized, 3,795,199 and 3,478,776 shares issued and outstanding, respectively Class A Common Stock, \$0.001 par value per share, 160,000,000 shares authorized, 10,868,000 shares issued and outstanding (Lass B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding (Lass B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding (Lass B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding (Lass B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding (Lass B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding (Lass B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding (Lass B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding (Lass B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding (Lass B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding (Lass B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding (Lass B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding (Lass B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding (Lass B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 20,868,000 shares authorized, 20	Operating lease liabilities, net of current portion		832	773
STOCKHOLDERS' DEFICIT Class A Common Stock, \$0.001 par value per share, 160,000,000 shares authorized, 3,795,199 and 3,478,776 shares issued and outstanding, respectively			46,281	74,354
Class A Common Stock, \$0.001 par value per share, 160,000,000 shares authorized, 3,795,199 and 3,478,776 shares issued and outstanding, respectively       4       3         Class B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding       11       11         Additional paid-in capital       3,481       3,067         Accumulated deficit       (6,593)       (2,538)         Noncontrolling interest       (11,893)       (4,225)         Total stockholders' deficit       (14,900)       (3,682)	COMMITMENTS AND CONTINGENCIES (Note 9)			
respectively         4         3           Class B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding         11         11           Additional paid-in capital         3,481         3,067           Accumulated deficit         (6,593)         (2,538)           Noncontrolling interest         (11,893)         (4,225)           Total stockholders' deficit         (14,900)         (3,682)	STOCKHOLDERS' DEFICIT			
Class B Common Stock, \$0.001 par value per share, 20,000,000 shares authorized, 10,868,000 shares issued and outstanding       11       11         Additional paid-in capital       3,481       3,067         Accumulated deficit       (6,593)       (2,538)         Noncontrolling interest       (11,893)       (4,225)         Total stockholders' deficit       (14,990)       (3,682)			Α	3
Additional paid-in capital         3,481         3,067           Accumulated deficit         (6,593)         (2,538)           Noncontrolling interest         (11,893)         (4,225)           Total stockholders' deficit         (14,990)         (3,682)				
Accumulated deficit         (6,593)         (2,538)           Noncontrolling interest         (11,893)         (4,225)           Total stockholders' deficit         (14,990)         (3,682)				
Noncontrolling interest         (11,893)         (4,225)           Total stockholders' deficit         (14,900)         (3,682)			,	,
Total stockholders' deficit (14,990) (3,682)				
	-			
	Total liabilities and stockholders' deficit	\$		

See accompanying notes to the unaudited condensed consolidated financial statements.

# DIRECT DIGITAL HOLDINGS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (Unaudited) (in thousands, except per-share data)

	Three Months Ended September 30,			Nine Months Ended September 30,		
	 2024		2023	 2024		2023
Revenues	 					
Sell-side advertising	\$ 2,202	\$	51,622	\$ 33,001	\$	89,006
Buy-side advertising	 6,873		7,850	 20,204		27,093
Total revenues	 9,075		59,472	 53,205		116,099
Cost of revenues						
Sell-side advertising	2,654		44,606	30,670		77,190
Buy-side advertising	2,907		3,113	8,091		10,650
Total cost of revenues	5,561		47,719	38,761		87,840
Gross profit	 3,514		11,753	 14,444		28,259
Operating expenses						
Compensation, taxes and benefits	3,526		4,747	12,216		12,934
General and administrative	3,646		2,512	10,757		8,718
Total operating expenses	7,172		7,259	22,973		21,652
(Loss) income from operations	 (3,658)		4,494	(8,529)		6,607
Other income (expense)						
Other income	99		83	190		175
Loss on early termination of line of credit	_		_	_		(300
Derecognition of tax receivable agreement liability	5,201		—	5,201		_
Interest expense	 (1,413)		(1,060)	 (4,068)		(3,104
Total other income (expense), net	 3,887		(977)	 1,323		(3,229
Income (loss) before income taxes	229		3,517	(7,206)		3,378
Income tax expense	6,606		166	6,132		166
Net (loss) income	 (6,377)		3,351	(13,338)		3,212
Net (loss) income attributable to noncontrolling interest	(3,687)		2,780	(9,283)		2,663
Net (loss) income attributable to Direct Digital Holdings, Inc.	\$ (2,690)	\$	571	\$ (4,055)	\$	549
Net (loss) income per common share:						
Basic	\$ (0.71)	\$	0.09	\$ (1.11)	\$	0.09
Diluted	\$ (0.71)	\$	0.09	\$ (1.11)	\$	0.09
Weighted-average number of shares of common stock outstanding:						
Basic	3,793		2,990	3,667		2,93
Diluted	 3,793		3,044	 3,667		3,08
Diluttu	 5,175		5,011	 5,007		5,00

See accompanying notes to the unaudited condensed consolidated financial statements.

# DIRECT DIGITAL HOLDINGS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CHANGES IN STOCKHOLDERS' DEFICIT (Unaudited) (in thousands except share data)

### Nine Months Ended September 30, 2024

	Common Stock								
	Clas	is A	Cla	ss B			Accumulated		Stockholders'
	Units	Amount	Units	А	mount	APIC	Deficit	Noncontrolling Interest	Deficit
Balance, December 31, 2023	3,478,776	\$ 3	10,868,000	\$	11	\$ 3,067	\$ (2,538)	\$ (4,225)	\$ (3,682)
Stock-based compensation	_	_	—		_	811	—	—	811
Issuance related to vesting of restricted stock units, net of tax withholdings	195,088	1	_		_	(1)	_	_	_
Warrants exercised	39,101	—	—		_	215	_	_	215
Stock options exercised	12,557	_	_		_	92	_	-	92
Issuance of stock in lieu of cash bonus, net of tax withholdings	69,677	_	—		—	912	—	—	912
Net loss	_		—		_	—	(4,055)	(9,283)	(13,338)
Noncontrolling interest rebalancing	_	—	—		_	(1,615)	_	1,615	_
Balance, September 30, 2024	3,795,199	\$ 4	10,868,000	\$	11	\$ 3,481	\$ (6,593)	\$ (11,893)	\$ (14,990)

#### Three Months Ended September 30, 2024

		Comm	on Stock					
	Cla	ss A	Cla	ss B		Accumulated		Stockholders'
	Units	Amount	Units	Amount	APIC	Deficit	Noncontrolling Interest	Deficit
Balance, June 30, 2024	3,788,446	\$ 4	10,868,000	\$ 11	\$ 3,444	\$ (3,903)	\$ (8,328)	\$ (8,772)
Stock-based compensation	_	—	—	_	149	_	_	149
Issuance related to vesting of restricted stock units, net of tax withholdings	2,950	_	_	_	_	_	_	_
Stock options exercised	3,803	_	_		10	_	_	10
Net loss	_	—	_	_	_	(2,690)	(3,687)	(6,377)
Noncontrolling interest rebalancing	—	—	—	_	(122)	—	122	—
Balance, September 30, 2024	3,795,199	\$ 4	10,868,000	\$ 11	\$ 3,481	\$ (6,593)	\$ (11,893)	\$ (14,990)

#### Nine Months Ended September 30, 2023

	Common Stock									
	Cla	Class A Class B			Accumulated				Stockholders'	
	Units	Amount	Units	Amount	APIC		Deficit	Noncontrolling Interest	Equity	
Balance, December 31, 2022	2,900,000	\$ 3	11,278,000	\$ 11	\$ 2,61	1 \$	(344)	\$ 3,314	\$ 5,595	
Stock-based compensation	_	_	_	_	54	6	_	—	546	
Issuance related to vesting of restricted stock units, net of tax withholdings	89,459	_	_	_	-	_	_	_	_	
Warrants exercised	2,200	_	_	_	1	2	_	—	12	
Warrant redemption accrual	_	_	_	_	(3,54	0)	_	—	(3,540)	
Stock options exercised	133	—	—	_	-	_	—	—	—	
Distributions to holders of LLC Units	_	_	—	_	-	_	_	(1,968)	(1,968)	
Net income	_	_	_	_	-	_	549	2,663	3,212	
Balance, September 30, 2023	2,991,792	\$ 3	11,278,000	\$ 11	\$ (37	1) \$	205	\$ 4,009	\$ 3,857	

#### Three Months Ended September 30, 2023

		Commo	on Stock						
	Clas	ss A	Cla	ss B			Accumulated		Stockholders'
	Units	Amount	Units	Amount	_	APIC	Deficit	Noncontrolling Interest	Equity
Balance, June 30, 2023	2,988,916	\$ 3	11,278,000	\$ 1	1 \$	2,927	\$ (366)	\$ 2,445	\$ 5,020
Stock-based compensation	_	—	_	-	-	242	_	_	242
Issuance related to vesting of restricted stock units, net of tax withholdings	2,743	_	_	-	_	_	_	_	
Warrant redemption accrual	_	—	_	-	-	(3,540)	_	_	(3,540)
Stock options exercised	133	_	_	-	-	_	_	_	_
Distributions to holders of LLC Units	_	_	_	-	-	—	—	(1,216)	(1,216)
Net income	_	_	_	-	-	_	571	2,780	3,351
Balance, September 30, 2023	2,991,792	\$ 3	11,278,000	\$ 1	1 \$	(371)	\$ 205	\$ 4,009	\$ 3,857

See accompanying notes to the unaudited condensed consolidated financial statements.

#### DIRECT DIGITAL HOLDINGS, INC. AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited) (in thousands)

(in thousands)			
		ne Months Ended Sept	
	202	24	2023
Cash Flows (Used In) Provided By Operating Activities:	0	(12.220)	2 212
Net (loss) income	\$	(13,338) \$	3,212
Adjustments to reconcile net (loss) income to net cash (used in) provided by operating activities:			10.5
Amortization of deferred financing costs		558	435
Amortization of intangible assets		1,465	1,465
Reduction in carrying amount of right-of-use assets		115 205	124
Depreciation and amortization of property, equipment and software		811	546
Stock-based compensation		6,132	546 82
Deferred income tax expense		0,132	
Loss on early termination of line of credit		(5,201)	300
Derecognition of tax receivable agreement liability		(5,201)	
Provision for credit losses/bad debt expense, net of recoveries		30	98
Changes in operating assets and liabilities:		30,884	(29.291)
Accounts receivable			(28,381)
Prepaid expenses and other assets		(394)	(524)
Accounts payable		(27,474)	27,326
Accrued liabilities and tax receivable agreement payable		(1,471) 65	(753)
Income taxes payable		595	(61)
Deferred revenues		(83)	497
Operating lease liability			(70)
Net cash (used in) provided by operating activities		(7,095)	4,481
Cash Flows Used In Investing Activities:			
Cash paid for capitalized software and property and equipment		(17)	(137)
Net cash used in investing activities		(17)	(137)
Cash Flows Provided by (Used In) Financing Activities:			
Payments on term loan		(373)	(491)
Proceeds from lines of credit		6,700	—
Payments on shares withheld for taxes		(551)	_
Payment of deferred financing costs		—	(442)
Proceeds from options exercised		92	_
Proceeds from warrants exercised		215	12
Distributions to holders of LLC Units			(1,988)
Net cash provided by (used in) financing activities		6,083	(2,909)
Net (decrease) increase in cash and cash equivalents		(1,029)	1,435
Cash and cash equivalents, beginning of the period		5,116	4,047
Cash and cash equivalents, end of the period	\$	4,087 \$	5,482
Supplemental Disclosure of Cash Flow Information:			
Cash paid for taxes	\$	263 \$	349
Cash paid for interest	\$	3,472 \$	2,667
Non-cash Financing Activities:			
Accrual of warrant redemption liability	\$	— \$	3,540
roomaa or waraan todomphon naonny	φ	ş	5,540

See accompanying notes to the unaudited condensed consolidated financial statements.

#### DIRECT DIGITAL HOLDINGS, INC. AND SUBSIDIARIES NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

#### Note 1 — Organization and Description of Business

Direct Digital Holdings, Inc., incorporated as a Delaware corporation on August 23, 2021 and headquartered in Houston, Texas, together with its subsidiaries, operates an end-to-end, programmatic advertising platform primarily focused on providing advertising technology, data-driven campaign optimization and other solutions intended for underserved and less efficient markets on both the sell- and buy-side of the digital advertising ecosystem. Direct Digital Holdings, Inc. is the holding company for Direct Digital Holdings, LLC ("DDH LLC"), which is, in turn, the holding company for the business formed by DDH LLC's founders in 2018 through the acquisition of Colossus Media, LLC ("Colossus Media") and Huddled Masses, LLC ("Huddled Masses®" or "Huddled Masses"). Colossus Media operates the Company's proprietary sell-side programmatic platform operating under the trademarked banner of Colossus SSP<sup>TM</sup> ("Colossus SSP"). In late September 2020, DDH LLC acquired Orange142, LLC ("Orange 142") to further bolster its overall programmatic buy-side advertising platform and to enhance its offerings across multiple industry verticals such as travel, healthcare, education, financial services, consumer products, and other sectors with particular emphasis intended for small and mid-sized businesses transitioning into digital with growing digital media budgets. In February 2022, Direct Digital Holdings, Inc. completed an initial public offering of its securities and, together with DDH LLC, effected a series of transactions (together, the "Organizational Transactions") whereby Direct Digital Holdings, Inc. became the sole managing member of DDH LLC, the holder of 100% of the voting interests of DDH LLC and the holder of 19.7% of the economic interests of DDH LLC, commonly referred to as an "Up-C" structure. (See Note 6 — Related Party Transactions). In these condensed consolidated financial statements, the "Company," "Direct Digital Holdings, "DDH," "we," "us" and "our" refer to Direct Digital Holdings, Inc., and, unless otherwis

Direct Digital Holdings, Inc. owns 100% of the voting interest in Direct Digital Holdings, LLC. As of September 30, 2024, DDH owns 25.9% of the economic interest in Direct Digital Holdings, LLC. See further discussion of the Up-C structure in Note 6 of our condensed consolidated financial statements. Direct Digital Holdings, LLC was formed on June 21, 2018 and acquired by DDH on February 15, 2022 in connection with the Organizational Transactions. Direct Digital Holdings, LLC's wholly-owned subsidiaries are as follows:

Subsidiary	Current % Ownership	Business Segment	Date of Formation	Date of Acquisition
Colossus Media, LLC	100%	Sell-side	September 8, 2017	June 21, 2018
Orange142, LLC	100%	Buy-side	March 6, 2013	September 30, 2020
Huddled Masses, LLC	100%	Buy-side	November 13, 2012	June 21, 2018

Colossus SSP is a stand-alone platform intended to deliver targeted advertising to diverse and multicultural audiences as well as to general audiences. Both buy-side subsidiaries, Orange 142 and Huddled Masses, offer technology-enabled advertising solutions and consulting services to clients through demand side platforms ("DSPs").

Providing both the front-end, buy-side operations coupled with the Company's proprietary sell-side operations enables the Company to curate the first through the last mile in the ad tech ecosystem execution process to drive higher results.

#### Note 2 — Basis of Presentation and Consolidation and Summary of Significant Accounting Policies

#### Basis of presentation and consolidation

The accompanying condensed unaudited consolidated financial statements have been prepared by the Company in accordance with accounting principles generally accepted in the United States ("GAAP") for interim financial reporting and as required by Rule 8-03 of Regulation S-X. Accordingly, the condensed unaudited consolidated financial statements may not include all of the information and notes required by GAAP for audited financial statements. The condensed consolidated balance sheet as of December 31, 2023 included herein was derived from audited financial statements but does not include all disclosures required by GAAP for complete financial statements. In the opinion of the Company's management, the accompanying condensed unaudited consolidated financial statements contain all adjustments, consisting of items of a normal and recurring nature, necessary to present fairly the Company's financial position as of September 30, 2024, the results of its operations for the three and nine months ended September 30, 2024 and 2023, and stockholders' deficit for the three and nine months ended

September 30, 2024 and 2023. The results of operations for the three and nine months ended September 30, 2024, respectively, are not necessarily indicative of the results to be expected for the full year. The preparation of condensed consolidated financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the amounts of assets and liabilities, and related disclosures, as of the date of the financial statements, and the amounts of revenues and expenses reported during the period. Actual results could differ from estimates. The accompanying condensed unaudited consolidated financial statements should be read in conjunction with the Company's audited financial statements and the accompanying notes for the year ended December 31, 2023, which was included in Form 10-K filed with the SEC on October 15, 2024.

The condensed consolidated financial statements include the accounts of Direct Digital Holdings, Inc. and its wholly owned subsidiaries. All material intercompany accounts and transactions have been eliminated in consolidation.

The Company is an emerging growth company, as defined in the Jumpstart Our Business Startups Act of 2012 (the "JOBS Act"). Under the JOBS Act, emerging growth companies can delay adopting new or revised accounting standards otherwise applicable to public companies until such time as those standards apply to private companies. The Company has elected to use this extended transition period for complying with new or revised accounting standards that have different effective dates for public and private companies until the earlier of the date that it (i) is no longer an emerging growth company or (ii) it affirmatively and irrevocably opts out of the extended transition period provided in the JOBS Act. As a result, these condensed consolidated financial statements may not be comparable to companies that comply with the new or revised accounting pronouncements as of public company effective dates. The adoption dates discussed below reflect this election.

#### **Revenue** recognition

The Company recognizes revenue using the following five steps: 1) identification of a contract with a customer; 2) identification of the performance obligation(s) in the contract; 3) determination of the transaction price; 4) allocation of the transaction price to the performance obligation(s) in the contract; and 5) recognition of revenue when, or as, the performance obligation(s) are satisfied. The Company's revenues are derived primarily from two sources: sell-side advertising and buy-side advertising. Thus, the Company disaggregates the revenue earned into these two segments. For additional segment disclosures, refer to Note 7 — Segment Information of our condensed consolidated financial statements. The Company maintains agreements with its customers in the form of written service agreements, which set out the terms of the relationship, including payment terms (typically 30 to 90 days).

For the sell-side advertising segment, the Company generates revenue by selling advertising inventory (digital ad units) that the Company purchases from publishers to advertisers through a process of monetizing ad impressions on the Company's proprietary sell-side programmatic platform operating under the trademarked banner Colossus SSP. For the buy-side advertising segment, the Company generates revenue from customers that enter into agreements with the Company to provide managed advertising campaigns, which include digital marketing and media services to purchase digital advertising space, data and other add-on features.

In connection with the Company's analysis of principal vs agent considerations, the Company has evaluated the specified goods or services and considered whether the Company controls the goods or services before they are provided to the customer, including the three indicators of control. Based upon this analysis and the Company's specific facts and circumstances, the Company concluded that it is a principal for the goods or services sold through both the Company's sell-side advertising segment and buy-side advertising segment because the Company controls the specified good or service before it is transferred to the customer and the Company is the primary obligor in the agreement with customers. Therefore, the Company reports revenue on a gross basis inclusive of all supplier costs and pays suppliers for the cost of digital media, advertising inventory, data and any add-on services or features.

In the advertising industry, companies commonly experience seasonal fluctuations in revenue. For example, in our sell-side advertising segment, many advertisers allocate the largest portion of their budgets to the fourth quarter of the calendar year in order to coincide with increased holiday purchasing while, in our buy-side segment, the second and third quarters of the year reflect our highest levels of advertising activity and the first quarter reflects the lowest level of such activity.

#### Sell-side advertising

The Company partners with publishers to sell advertising inventory to the Company's Colossus Media-curated clients and the open markets (collectively referred to as "buyers") seeking to access the general market as well as unique multi-cultural audiences. The Company generates revenue from the delivery of targeted digital media solutions, enabling

advertisers to connect intelligently with their audiences across online display, video, social and mobile mediums using its proprietary programmatic sell-side platform ("SSP"). The Company refers to its publishers, app developers, and channel partners collectively as its "publishers." The Company's platform allows the Company to sell, in real time, ad impressions from publishers to buyers and provides automated inventory management and monetization tools to publishers across various device types and digital ad formats. The Company recognizes revenue at a point in time when an ad is delivered or displayed in response to a winning bid request from ad buyers.

#### Buy-side advertising

The Company purchases media based on the budget established by its customers with a focus on leveraging data services, customer branding, real-time market analysis and micro-location advertising. The Company offers its services on a fully managed basis, which is recognized over time using the output method when the performance obligation is fulfilled. An "impression" is delivered when an advertisement appears on pages viewed by users. The performance obligation is satisfied over time as the volume of impressions are delivered up to the contractual maximum. Many customers run several different campaigns throughout the year to capitalize on different seasons, special events and other happenings at their respective regions and localities. The Company provides digital advertising and media buying capabilities with a focus on generating measurable digital and financial life for its customers.

Revenue arrangements are evidenced by a fully executed insertion order ("IO") and/or a master service agreement ("MSA") covering a combination of marketing tactics. Generally, IOs specify the number and type of advertising impressions to be delivered over a specified time at an agreed upon price and performance objectives for an ad campaign. Performance objectives are generally a measure of targeting, as defined by the parties in advance, such as number of ads displayed, consumer clicks on ads or consumer actions (which may include qualified leads, registrations, downloads, inquiries or purchases). These payment models are commonly referred to as CPM (cost per impression), CPC (cost per click) and CPA (cost per action). The majority of the Company's contracts are flat-rate, fee-based contracts.

Cash payments received prior to the Company's delivery of its services are recorded to deferred revenue until the performance obligation is satisfied. The Company recorded deferred revenue (contract liabilities) to account for billings in excess of revenue recognized, primarily related to contractual minimums billed in advance and customer prepayment, of \$1.0 million and \$0.4 million as of September 30, 2024 and December 31, 2023, respectively. Revenue recognized during the nine months ended September 30, 2024 and 2023 from amounts included within the deferred revenue balances at the beginning of each respective period amounted to \$0.4 million and \$0.5 million, respectively.

Accounting Standards Codification ("ASC") 606 provides various optional practical expedients. The Company elected the use of the practical expedient relating to the disclosure of remaining performance obligations within a contract and will not disclose remaining performance obligations for contracts with an original expected duration of one year or less.

#### Goodwill

As of September 30, 2024 and December 31, 2023, goodwill was \$6.5 million, which includes \$2.4 million as a result of the acquisition of Huddled Masses and Colossus Media in 2018 and \$4.1 million from the acquisition of Orange 142 in 2020. The Company expects to deduct goodwill for tax purposes in future years. Goodwill is attributable to entry into new markets not previously accessible and generation of future growth opportunities. Goodwill is assessed for impairment at least annually (December 31) starting with a qualitative assessment to determine whether it is more likely than not that the fair value of a reporting unit containing goodwill is less than its carrying value. This qualitative assessment may include, but is not limited to, reviewing factors such as macroeconomic conditions, industry and market considerations, cost factors, entity-specific financial performance and other events, such as changes in our management, strategy and primary user base. If the Company determines that it is more likely than not that the fair value of a reporting unit is performed. Depending upon the results of the quantitative measurement, the recorded goodwill may be written down and an impairment expense is recorded in the condensed consolidated statements of operations when the carrying amount of the reporting unit exceeds the fair value of the reporting unit. Goodwill is reviewed annually and tested for impairment upon the occurrence of a triggering event. The Company determined that there was no impairment of goodwill during the nine months ended September 30, 2024 and 2023.

#### Intangible assets, net

Intangible assets consist of customer relationships, trademarks and non-compete agreements. Intangible assets are recorded at fair value at the time of their acquisition and are stated within the condensed consolidated balance sheets net of accumulated amortization. Intangible assets are amortized on a straight-line basis over their estimated useful lives and recorded as amortization expense within general and administrative expenses in the condensed consolidated statements of

operations. The Company's intangible assets are being amortized over their estimated useful lives, using the straight-line method with non-compete agreements over 5 years and other intangibles over 10 years.

#### Impairment of long-lived assets

The Company evaluates the recoverability of long-lived assets, including property, equipment and software costs and intangible assets if facts or circumstances indicate that any of those assets might be impaired. ASC 360-10-15 requires the Company to group assets and liabilities at the lowest level for which identifiable cash flows are largely independent of the cash flows of other assets and liabilities and evaluate the asset group against the sum of the undiscounted future cash flows to determine if a write-down to fair value is necessary. No impairment loss was recognized during the nine months ended September 30, 2024 and 2023.

#### Stock-based compensation

Stock-based compensation cost for options and restricted stock units ("RSU") awarded to employees and directors is measured at the grant date based on the calculated fair value of the award and is recognized as an expense over the requisite service period (generally the vesting period of the equity grant). Contingently issued awards with a requisite service period that precedes the grant date are measured and recognized at the start of the requisite service period and remeasured each reporting period until the grant date.

The Company estimates the fair value of RSU's based on the closing price of the Company's common stock on the date of the grant. The Company estimates the fair value of stock options using the Black-Scholes valuation model. Key input assumptions used to estimate the fair value of stock options include the fair value of the Company's common stock, as well as assumptions regarding the expected common stock price volatility over the term of the stock options, the expected term of the stock options, risk-free interest rates and the expected dividend yield. Given the Company's short history as a public company, the expected volatility is determined based on the trading history of several unrelated public companies within the industry that the Company considers to be comparable and the expected term is determined based on a combination of the terms of stock options and peer data. The risk-free interest rate is derived using the U.S. Treasury yield curve in effect at date of grant. Other assumptions are based on historical experience and activity. The Company considers an estimated forfeiture rate for stock options based on historical experience and the anticipated forfeiture rates during the future contract life.

#### Income taxes

In February 2022, concurrent with the Organizational Transactions, the Company entered into a tax receivable agreement ("Tax Receivable Agreement" or "TRA") with DDH LLC and Direct Digital Management, LLC ("DDM"). The TRA provides for certain income (loss) allocations between the Company and DDH LLC under the agreement. DDH LLC is a limited liability company, is treated as a partnership for federal income tax purposes and generally is not subject to any entity-level U.S. federal income tax and certain state and local income taxes. Any taxable income or loss generated by the Company is allocated to holders of LLC units ("LLC Units") in accordance with the Second Amended and Restated Limited Liability Company Agreement ("LLC Agreement"), and distributions to the owners of LLC Units in an amount sufficient to fund their tax obligations. The Company is subject to U.S. federal income taxes, in addition to state and local income taxes with respect to its allocable share of any taxable income or loss under the LLC Agreement. Pursuant to the Company's election under Section 754 of the Internal Revenue Code (the "Code"), the Company expects to obtain an increase in its share of the tax basis in the net assets of DDH, LLC Units are redeemed or exchanged by the members of DDH, LLC. The Company and an election under Section 754 of the Code for each taxable year in which a redemption or exchange of LLC interest occurred. During the nine months ended September 30, 2024 and 2023, members of DDM exchanged no shares of Class B Common Stock into shares of Class A Common Stock.

Deferred tax assets and liabilities are determined based on differences between financial reporting and tax bases of assets and liabilities and are measured using the enacted tax rates and laws that will be in effect when the differences are expected to reverse. A valuation allowance is established when it is more likely than not that some portion or all of the deferred tax assets will not be realized. The establishment of a valuation allowance requires significant judgment and is impacted by various estimates. Both positive and negative evidence, as well as the objectivity and verifiability of that evidence, is considered in determining the appropriateness of recording a valuation allowance on deferred tax assets. As of September 30, 2024 and December 31, 2023, the Company recorded a valuation allowance of \$7.3 million and \$0.5 million, respectively.

#### Accounts receivable, net

Accounts receivable primarily consists of billed amounts for products and services rendered to customers under normal trade terms. The Company performs credit evaluations of its customers' financial condition and generally does not require collateral. Accounts receivable are stated at net realizable value. The Company insures a significant portion of its accounts receivable with unrelated third-party insurance companies in an effort to mitigate any future write-offs and establishes provision for credit losses as deemed necessary for accounts not covered by this insurance. Management periodically reviews outstanding accounts receivable for reasonableness. If warranted, the Company processes a claim with the third-party insurance company to recover uncollected balances, rather than writing the balances off to bad debt expense. The guaranteed recovery for the claim is approximately 90% of the original balance, and if the full amount is collected by the insurance company, the remaining 10% is remitted to the Company. If the insurance company is unable to collect the full amount, the Company records the remaining 10% to bad debt expense. The Company's provision for credit losses reflects the current expected credit loss inherent in the accounts receivable considering the Company's aging analysis, historical collection experience, customer creditworthiness, current and future economic conditions and market conditions. Accounts receivable balances are written off against the provision for credit losses net of recovered. For the nine months ended September 30, 2024 and 2023, the Company's provision for credit losses net of recoveries, as reflected in the condensed consolidated statements of cash flows was less than \$0.1 million.

#### Concentrations of customers and suppliers

There is an inherent concentration of credit risk associated with accounts receivable arising from revenue from major customers on both the sell-side and buy-side of the business. For the three months ended September 30, 2024 and 2023, one buy-side customer represented 12% and one sell-side customer represented 82% of revenues, respectively. For the nine months ended September 30, 2024 and 2023, one sell-side customer represented 52% and 72% of revenues, respectively. As of September 30, 2024 and 2023, and 83%, respectively, of accounts receivable.

As of September 30, 2024 and December 31, 2023, one vendor and three sellers of advertising inventory each accounted for at least 10%, and collectively accounted for 17% and 77%, respectively, of consolidated accounts payable.

#### Accrued Liabilities

The components of accrued liabilities on the balance sheet as of September 30, 2024 and December 31, 2023 are as follows (in thousands):

	ember 30, 2024	December 31, 2023	
Accrued compensation and benefits	\$ 277	\$	2,789
Accrued expenses	534		631
Accrued severance	—		189
Accrued litigation settlement <sup>(1)</sup>			171
Accrued interest	 70		36
Total accrued liabilities	\$ 881	\$	3,816

(1) In July 2022, the Company entered into a litigation settlement agreement with a vendor of Huddled Masses related to a delinquent balance from 2019 and agreed to pay a total of \$0.5 million with monthly installment payments over 24 months beginning September 1, 2022.

#### Cash and cash equivalents

Cash and cash equivalents consist of funds deposited with financial institutions and highly liquid instruments with original maturities of three months or less. Such deposits may, at times, exceed federally insured limits. The risk of loss attributable to any uninsured balances is mitigated by depositing funds only in high credit quality financial institutions. The Company has not experienced any losses in such amounts and believes it is not exposed to any significant credit risk to cash.



#### Deferred offering costs

The Company records certain legal, accounting and other third-party fees that are directly associated with an offering to stockholders' equity or debt in the event that the Company completes an offering. Costs associated with debt offerings are amortized to interest expense using the straight-line method over the life of the debt. As of September 30, 2024 and December 31, 2023, \$1.3 million and \$1.7 million, respectively, of unamortized deferred financing costs are netted against debt in the condensed consolidated balance sheets. As of September 30, 2024 and December 31, 2023, \$0.1 million and \$0.2 million, respectively, of unamortized deferred issuance costs are classified as prepaid expenses and other current assets in the condensed consolidated balance sheets.

#### Fair value measurements

The Company employs a hierarchy which prioritizes the inputs used to measure recurring fair value into three distinct categories based on the lowest level of input that is significant to the fair value measurement. The methodology for categorizing assets and liabilities that are measured at fair value pursuant to this hierarchy gives the highest priority to unadjusted quoted prices in active markets and the lowest levels to unobservable inputs, summarized as follows:

- Level 1 Quoted prices in active markets for identical assets or liabilities.
- Level 2 Other significant observable inputs (including quoted prices in active markets for similar assets or liabilities).
- Level 3 Significant unobservable inputs (including our own assumptions in determining fair value).

We use the cost, income or market valuation approaches to estimate the fair value of our assets and liabilities when insufficient market-observable data is available to support our valuation assumptions.

#### Fair value of financial instruments

The Company considers the fair value of all financial instruments, including cash, accounts receivable and accounts payable to approximate their carrying values at yearend due to their short-term nature. The carrying value of the Company's debt approximates fair value due to the market rates of interest.

#### Net income (loss) per share

Basic net income (loss) per share excludes dilution and is determined by dividing net income (loss) by the weighted average number of common shares outstanding including participating securities during the period. Diluted net income (loss) per share reflects the potential dilution that could occur if securities and other contracts to issue common stock were exercised or converted into common stock.

#### **Recent Accounting Pronouncements**

#### Accounting pronouncements adopted

No standards have been adopted which have had a material impact on the Company's condensed consolidated financial statements.

#### Accounting pronouncements not yet adopted

In December 2023, the Financial Accounting Standards Board (the "FASB") issued Accounting Standards Update ("ASU") 2023-09, *Improvements to Income Tax Disclosures*, a final standard on improvements to income tax disclosures. The standard requires disaggregated information about a reporting entity's effective tax rate reconciliation as well as information on income taxes paid. The standard is intended to benefit investors by providing more detailed income tax disclosures that would be useful in making capital allocation decisions and applies to all entities subject to income taxes. The new standard is effective for emerging growth companies for annual periods beginning after December 15, 2025. This accounting standard is effective in the first quarter of the Company's fiscal year ending December 31, 2026. The Company is currently evaluating the impact of adoption on our financial disclosures.

In November 2023, the FASB issued ASU 2023-07, Segment Reporting (Topic 280) - Improvements to Reportable Segment Disclosures. The ASU requires that an entity disclose significant segment expenses impacting profit and loss that are regularly provided to the Company's chief operating decision maker ("CODM"). The update is required to be applied



retrospectively to prior periods presented, based on the significant segment expense categories identified and disclosed in the period of adoption. The amendments in this ASU are required to be adopted for fiscal years beginning after December 15, 2023, and interim periods within fiscal years beginning after December 15, 2024. Early adoption is permitted. The Company is currently evaluating the impact of adoption on our financial disclosures.

Other accounting standards that have been issued or proposed by the FASB or other standards-setting bodies that do not require adoption until a future date are not expected to have a material impact on the Company's condensed consolidated financial statements.

#### Liquidity and capital resources

#### Going Concern

The Company evaluated whether relevant conditions or events, considered in the aggregate, raise substantial doubt about the Company's ability to continue as a going concern. Substantial doubt exists when conditions and events, considered in the aggregate, indicate it is probable that a company will not be able to meet its obligations as they become due within one year after the issuance date of its financial statements. Management's assessment is based on the relevant conditions that are known or reasonably knowable as of the date these condensed consolidated financial statements were issued or were available to be issued.

As discussed in Note 9, one of the Company's sell-side customers paused its connection to the Company for a couple of weeks in May 2024, which reduced sell-side sales volumes. As of the date of this report, sell-side volumes related to this customer have resumed but not yet at the levels experienced prior to the pause in May 2024 which has created significant disruption in the Company's sell-side business. The Company is actively working with its partners to achieve prior volume levels. However, there can be no assurance that the Company will be able to achieve prior volume levels with its partners or on the timing of achieving such volume levels. Additionally, the Company (1) incurred a net loss of \$13.3 million for the nine months ended September 30, 2024 reflecting the impact of the sell-side disruption described above and a decrease in customer spend on the buy-side, (2) reported an accumulated deficit of \$6.6 million as of September 30, 2024, (3) reported cash and cash equivalents of \$4.1 million as of September 30, 2025, (5) was notified on April 17, 2024 that the Company's additor had resigned, (6) was unable to timely file its 2023 annual report and quarterly requirements. The delay in filing the Company's annual and quarterly reports disrupted existing capital-raising efforts and created additional audit, legal and other expenses. These factors raise substantial doubt about the Company's ability to continue as a going concern over the next twelve months.

The Company anticipates sources of liquidity to include cash on hand and cash flow from operations and has taken several actions to address liquidity concerns. These actions include (1) a plan to reduce expenses through a staff reduction, a pause on hiring and cost savings measures that were executed on July 1, 2024, (2) working with lenders to provide temporary various relief from debt covenants (see Note 3 — Long-Term Debt) while rebuilding sell-side volumes, (3) putting in place a program to raise capital through an equity reserve facility (see Note 4 — Stockholders' Deficit and Stock-Based Compensation), (4) regaining compliance with respect to delinquent SEC filings on October 15, 2024 which will allow the Company to access the capital markets as well as other financing sources and (5) a plan to achieve compliance with Nasdaq's minimum stockholders' equity requirements. There can be no assurance that the Company's actions will be successful or that additional financing will be available when needed or on acceptable terms.

The accompanying condensed consolidated financial statements have been prepared assuming the Company will continue to operate as a going concern, which contemplates the realization of assets and settlement of liabilities in the normal course of business, and do not include any adjustments to reflect the possible future effects on the recoverability and classification of assets or the amounts and classifications of liabilities that may result from uncertainty related to its ability to continue as a going concern.



#### Note 3 — Long-Term Debt

At September 30, 2024 and December 31, 2023, long-term debt consisted of the following (in thousands):

	Sept	ember 30, 2024	December 31, 2023			
2021 Credit Facility	\$	28,221	\$	28,594		
Credit Agreement		9,700		3,000		
Economic Injury Disaster Loan		150		150		
Total debt		38,071		31,744		
Less: deferred financing costs		(1,254)		(1,688)		
Total debt, net of deferred financing costs		36,817		30,056		
Less: current portion		(36,667)		(1,478)		
Total long-term debt, net of current portion	\$	150	\$	28,578		

The components of interest expense and related fees for long-term debt is as follows (in thousands):

	Three Months Ended September 30,			Nine Months Ended September 30,			
		2024		2023	 2024		2023
Interest expense – Lafayette Square	\$	1,020	\$	896	\$ 2,975	\$	2,665
Interest expense – East West Bank		206		—	531		_
Interest expense – other		1		1	4		4
Amortization of deferred financing costs		186		163	558		435
Total interest expense and amortization of deferred financing costs	\$	1,413	\$	1,060	\$ 4,068	\$	3,104

#### Lafayette Square

On December 3, 2021, the Company entered into the Term Loan and Security Agreement (the "2021 Credit Facility") with Lafayette Square Loan Services, LLC ("Lafayette Square") as administrative agent, and the various lenders thereto. The term loan under the 2021 Credit Facility initially provided for a term loan in the principal amount of up to \$32.0 million, consisting of a \$22.0 million closing date term loan (the "Term Loan") and an up to \$10.0 million delayed draw term loan (the "Delayed Draw Loan"). The loans under the 2021 Credit Facility originally bore interest at LIBOR plus the applicable margin minus any applicable impact discount. The applicable margin under the 2021 Credit Facility was determined based on the consolidated total net leverage ratio of the Company and its consolidated subsidiaries, at a rate of 6.50% per annum if the consolidated total net leverage ratio is less than 2.00 to 1.00 and up to 9.00% per annum if the consolidated total net leverage ratio was greater than 4.00 to 1.00. On June 1, 2023, as originally contemplated under the 2021 Credit Facility, the Company entered into an agreement with Lafayette Square to convert the existing LIBOR based rate to a Term Secured Overnight Financing Rate ("SOFR") with a credit spread of 0.15% per annum for the interest periods of three months and providing for a credit spread adjustment of 0.10%, 0.15% or 0.25% per annum for interest periods of one month, three months or six months, respectively. The loans under the 2021 Credit Facility was based on the consolidated total net leverage ratio of the Company at a rate of 7.00% per annum if the consolidated total net leverage ratio of the Company at a rate of 7.00% per annum if the consolidated total net leverage ratio of the Company at a rate of 7.00% per annum if the consolidated total net leverage ratio of the Company at a rate of 7.00% per annum if the consolidated total net leverage ratio of the Company at a rate of 7.00% per annum if the consolidated total net leverage ratio of the Company at a

On July 28, 2022, the Company entered into the Second Amendment and Joinder to Term Loan and Security Agreement and received proceeds of \$4.3 million borrowed under the Delayed Draw Loan to pay the balance owed on the common unit redemption as well as costs associated with the transaction.

Subsequently, on October 3, 2023, the Company entered into the Fourth Amendment to the 2021 Credit Facility (the "Fourth Amendment") and received proceeds of \$3.6 million borrowed under the Delayed Draw Loan to make payments in connection with the consummation of the 2023 warrant tender offer and fees and expenses incurred as described in Note 4 — Stockholders' Deficit and Stock-Based Compensation in the notes to the condensed consolidated financial statements. In

connection with the Fourth Amendment, the Company agreed it would not be permitted to request any additional funds under the Delayed Draw Loan, and Lafayette Square would not be obligated to fund any such requests.

Quarterly installment payments on the Term Loan and the Delayed Draw Loan, due on the last day of each fiscal quarter, began March 31, 2022 with a final installment due December 3, 2026 for remaining balances outstanding under each loan. Each quarterly installment payment under the closing date term loan was \$0.1 million from January 1, 2022 through December 31, 2023, and each installment payment thereafter until maturity is \$0.3 million. Each quarterly installment payment under the Delayed Draw Loan was 0.625% of the amount of the Delayed Draw Loan through December 31, 2023, and each installment payment thereafter until maturity is 1.25% of the amount of the Delayed Draw Loan.

Under the 2021 Credit Facility, dividends and distributions by DDH LLC to the Company and any shareholders of the Company are permitted so long as (i) no default or event of default is continuing or would occur after giving pro forma effect to such dividends and distributions under the 2021 Credit Facility, (ii) the Company, on a pro forma basis, maintains a consolidated senior net leverage ratio of not greater than 1.5 to 1.0, and (iii) the Company, on a pro forma basis, maintains liquidity of not less than \$15.0 million. The Company did not meet these conditions as of September 30, 2024 and therefore DDH LLC assets are considered restricted and no dividends or distributions to the Company and any shareholders is permitted while these conditions are not met.

The obligations under the 2021 Credit Facility are secured by senior, first-priority liens on all or substantially all assets of the Company. As of September 30, 2024, the Company owed a balance on the 2021 Credit Facility of \$28.2 million. No additional deferred financing costs were incurred during the nine months ended September 30, 2024 and less than \$0.1 million of additional deferred financing costs were incurred during the nine months ended September 30, 2023. Unamortized deferred financing costs as of September 30, 2024 and December 31, 2023 were \$1.3 million and \$1.7 million respectively. Accrued and unpaid interest was less than \$0.1 million as of September 30, 2024 and December 31, 2023. The 2021 Credit Facility contains customary affirmative and negative covenants. Prior to entering into the Fifth Amendment, the Company was required to maintain a net leverage ratio of no more than 3.50 to 1.00 as of December 31, 2025 and September 30, 2025, with incremental tightening of the ratio to 2.50 to 1.00 as of June 30, 2026 and thereafter through March 31, 2025, 3.00 to 1.00 as of June 30, 2025 and September 30, 2025, with incremental tightening of the ratio to 2.50 to 1.00 as of June 30, 2026 and thereafter through maturity. Prior to entering into the Fifth Amendment, the Company to maintain a fixed charge coverage ratio of not less than 1.50 to 1.00 as of the last day of each fiscal quarter, as well as restrictions on the ability to incur indebtedness, create certain liens, make certain investments, make certain dividends and other types of distributions, and enter into or undertake certain mergers, consolidations, acquisitions and sales of certain assets and subsidiaries.

On October 15, 2024, with an effective date of June 30, 2024, the Company and Lafayette Square entered into the Fifth Amendment to the Term Loan and Security Agreement (the "Fifth Amendment") which among other things, (1) defers quarterly installment payments on the Term Loan and the Delayed Draw Loan for the periods from June 30, 2024 through December 31, 2025, (2) requires that the Company pay a commitment fee of 50 basis points or an amount of \$0.1 million to Lafayette Square, (3) allows proceeds from future equity raises by the Company, if any, to cure potential financial covenant noncompliance, (4) provides for one-month and three-month interest periods, (5) replaces the calculation of the consolidated total net leverage ratio with a consolidated total leverage ratio for purposes of calculating the applicable margin and the financial covenant and (6) replaces the financial covenants under the 2021 Credit Facility (effective as of June 30, 2024) with the following:

As of	Minimum TTM* EBITDA (\$ in millions)	Minimum Liquidity (\$ in millions)	Maximum Consolidated Total Leverage Ratio	Minimum Fixed Charge Coverage Ratio
June 30, 2024	n/a	n/a	n/a	n/a
September 30, 2024	\$5.0	\$1.5	n/a	n/a
December 31, 2024	\$3.5	\$1.5	n/a	n/a
March 31, 2025	\$5.5	\$2.0	n/a	n/a
June 30, 2025	\$7.5	\$2.0	n/a	1.50 to 1.00
September 30, 2025	n/a	\$2.0	4.25 to 1.00	1.50 to 1.00
December 31, 2025	n/a	\$2.0	4.00 to 1.00	1.50 to 1.00
March 31, 2026	n/a	\$2.0	3.75 to 1.00	1.50 to 1.00
June 30, 2026	n/a	\$2.0	3.50 to 1.00	1.50 to 1.00
September 30, 2026	n/a	\$2.0	3.25 to 1.00	1.50 to 1.00

\* TTM = Trailing Twelve Months

The Company was in compliance with all the financial covenants under the 2021 Credit Facility as of September 30, 2024 after giving effect to the amendments to the 2021 Credit Facility under the Fifth Amendment, except for the minimum trailing twelve months EBITDA covenant. The Company is in discussion with Lafayette Square to waive the current non-compliance with the financial covenant. While the Company expects to negotiate an acceptable resolution, there can be no assurance that it will be able to negotiate a waiver or an amendment, or that such waiver or amendment will be on terms acceptable to the Company. If the Company is unable to obtain a waiver from or enter into an amendment with Lafayette Square, it could have a material adverse effect on our financial position and our ability to execute our business plan. Because of the non-compliance, the debt under the 2021 Credit Facility is classified as current as of September 30, 2024.

#### 2023 Revolving Line of Credit - East West Bank

On July 7, 2023, the Company entered into a Credit Agreement (as amended, the "Credit Agreement"), with East West Bank ("EWB"), as lender. The Credit Agreement provides for a revolving credit facility in the principal amount of up to \$10 million, subject to a borrowing base determined based on eligible accounts, and an up to \$5 million uncommitted incremental revolving facility. Loans under the Credit Agreement mature on July 7, 2025 (the "Maturity Date"), unless the Credit Agreement is otherwise terminated pursuant to the terms of the Credit Agreement.

Borrowings under the Credit Agreement bear interest at a rate per annum equal to the one-month Term SOFR rate and as determined by EWB on the first day of the applicable interest period, plus 0.10% (10 basis points), plus 3.00% per annum (the "Loan Rate"); provided, that, in no event shall the Loan Rate be less than 0.50% of the Loan Rate effective as of the date of the Credit Agreement nor more than the maximum rate of interest allowed under applicable law. Upon an event of default under the Credit Agreement, the outstanding principal amounts of any advances will accrue interest at a rate per annum equal to the Loan Rate plus five percent (5%), but in no event in excess of the maximum rate of interest allowed under applicable law.

At the Company's option, the Company may at any time prepay the outstanding principal balance of the Credit Agreement in whole or in part, without fee, penalty or premium. All accrued but unpaid interest on outstanding advances under the Credit Agreement are payable in monthly installments on the last day of each monthly interest period until the Maturity Date when the then-outstanding principal balance of the advances and all accrued but unpaid interest thereon becomes due and payable. The obligations under the Credit Agreement are secured by all or substantially all of the borrowers' assets.

Prior to entering into the Third Amendment as defined below, the Company was required to maintain compliance at all times with the following financial covenants on a consolidated basis: (i) a fixed charge coverage ratio of not less than 1.25 to 1.0, beginning with the fiscal quarter ended on June 30, 2023 and at the end of each fiscal quarter thereafter; (ii) a total funded debt-to-EBITDA ratio of no more than 3.50 to 1.00 as of June 30, 2023 and the last day of each fiscal quarter through December 31, 2023, 3.25 to 1.00 as of March 31, 2024 and the last day of each fiscal quarter through March 31, 2025 and 3.00 to 1.00 as of June 30, 2025 and thereafter through maturity; and (iii) a liquidity covenant requiring the Company to maintain minimum liquid assets at all times (calculated in the manner provided for in the Credit Agreement), in one or more accounts held with EWB plus Revolving Credit Availability in the amount of \$1.0 million. Revolving Credit Availability is defined as an amount such that the ratio of the value of eligible accounts to the aggregate amount of all outstanding advances under the credit agreement at such time is not less than 2.0 to 1.0. Additionally, the amounts

outstanding under the Credit Agreement exceeded the Company's borrowing base as of June 30, 2024 by \$0.5 million which was addressed in the Third Amendment, requiring a \$1.0 million principal payment on the outstanding loans under the Credit Agreement as of the date of the Third Amendment.

On October 15, 2024, with an effective date of June 30, 2024, the Company and EWB entered into the Third Amendment to the Credit Agreement (the "Third Amendment") which, among other things, (1) provides that the Company will make prepayments of the outstanding principal balance of the Credit Agreement of \$1.0 million upon execution of the Third Amendment, \$1.0 million on or before January 15, 2025 and \$2.0 million on or before April 15, 2025, (2) requires the Company to file a registration statement with the SEC to establish an equity line of credit offering on or before October 31, 2024 and to use commercially reasonable efforts to cause such registration statement to become effective, (3) requires the net proceeds of a potential equity line of credit to be applied to the outstanding principal balance under the Credit Agreement in an amount that would cause the ratio of the value of eligible accounts to the aggregate amount of revolving credit advances to be not less than 1.00 to 1.00, (4) requires the company to make certain restricted payments, including cash dividends, (5) requires the Company to make additional prepayments in the amount by which the outstanding loans under the Credit Agreement exceed the borrowing base between the calendar months ending November 30, 2024 and April 15, 2025, and (6) replaces the financial covenants under the Credit Agreement, effective as of June 30, 2024, with the following:

	Minimum TTM <sup>(1)</sup> EBITDA (\$ in	Minimum Liquid	Maximum Total Funded Debt to EBITDA Leverage	Minimum Fixed	Revolving Credit Availability (as of each
As of	millions)	Assets (\$ in millions)		Charge Coverage Ratio	
June 30, 2024	n/a	\$1.0	n/a	n/a	n/a
September 30, 2024	\$5.0	\$1.5	n/a	n/a	n/a
December 31, 2024	\$3.5	\$1.5	n/a	n/a	1.00 to 1.00 <sup>(2)</sup>
March 31, 2025	\$5.5	\$2.0	n/a	n/a	1.50 to 1.00 <sup>(3)</sup>
June 30, 2025	\$7.5	\$2.0	n/a	1.25 to 1.00	2.00 to 1.00 <sup>(4)</sup>

<sup>(1)</sup> TTM = Trailing Twelve Months

(2) Beginning November 30, 2024

(3) Beginning January 31, 2025(4) Beginning April 15, 2025

The Company was in compliance with all the financial covenants under the Credit Agreement as of September 30, 2024 after giving effect to the amendments to the Credit Agreement under the Third Amendment, except for the minimum trailing twelve months EBITDA covenant. The Company is in discussion with EWB to waive the current noncompliance with the financial covenant. While the Company expects to negotiate an acceptable resolution, there can be no assurance that it will be able to negotiate a waiver or an amendment, or that such waiver or amendment will be on terms acceptable to the Company. If the Company is unable to obtain a waiver from or enter into an amendment with EWB, it could have a material adverse effect on our financial position and our ability to execute our business plan. Because of the maturity date, the debt under the Credit Agreement is classified as current as of September 30, 2024.

The Credit Agreement contains customary representations and warranties and includes affirmative and negative covenants applicable to the borrowers and their respective subsidiaries. The affirmative covenants include, among others, covenants requiring the Company to maintain its legal existence and governmental compliance, deliver certain financial reports and maintain insurance coverage. The negative covenants include, among others, restrictions on indebtedness, liens, investments, mergers, dispositions, prepayment of other indebtedness and dividends and other distributions.

The Credit Agreement also includes customary events of default, including, among other things, non-payment defaults, covenant defaults, inaccuracy of representations and warranties, defaults under any of the loan documents, certain cross-defaults to other indebtedness, certain bankruptcy and insolvency events, invalidity of guarantees or grant of security interest, certain ERISA-related transactions and events, certain orders of forfeiture, change of control, certain undischarged attachments, sequestrations, or similar proceedings, and certain undischarged or non-stayed judgments, in certain cases subject to certain thresholds and grace periods. The occurrence of an event of default could result in the acceleration of the obligations under the Credit Agreement of the Company or other borrowers. During the nine months ended September 30, 2024, the Company did not incur any deferred financing costs associated with the Credit Agreement. As of September 30, 2024, there was \$9.7 million outstanding under the Credit Agreement which was classified as short term due to the Maturity Date being within twelve months of the reporting period.

The collateral securing the obligations under the 2021 Credit Facility and the Credit Agreement is subject to intercreditor agreements between Lafayette Square and EWB.

#### Silicon Valley Bank ("SVB") Financing

On January 9, 2023, the Company entered into the SVB Loan Agreement, by and among SVB, as lender, and DDH LLC, the Company, Huddled Masses, Colossus Media and Orange 142, as borrowers. The SVB Loan Agreement provided for a revolving credit facility (the "SVB Revolving Credit Facility") in the original principal amount of \$5 million, subject to a borrowing base determined based on eligible accounts, and up to an additional \$2.5 million incremental revolving facility subject to the lender's consent, which would increase the aggregate principal amount of the Credit Facility to \$7.5 million. Loans under the SVB Revolving Credit Facility were to mature on September 30, 2024 unless the Credit Facility was otherwise terminated pursuant to the terms of the Loan Agreement.

On March 10, 2023, the California Department of Financial Protection and Innovation closed SVB and appointed the Federal Deposit Insurance Corporation as receiver. As the Company had not yet drawn any amounts under the SVB Revolving Credit Facility, on March 13, 2023, the Company issued a notice of termination of the SVB Loan Agreement. The termination of the SVB Revolving Credit Facility became effective April 20, 2023. Prior to issuing the notice of termination, the Company received consent to terminate the SVB Revolving Credit Facility and a waiver of the terms relating to the SVB Revolving Credit Facility under its Term Loan and Security Agreement, dated as of December 3, 2021, with Lafayette Square Loan Servicing, LLC ("Lafayette Square"). The Company did not hold material cash deposits or securities at Silicon Valley Bank and did not experience any adverse impact to its liquidity or to its current and projected business operations, financial condition or results of operations as a result of the SVB closure. During the nine months ended September 30, 2023, the Company incurred \$0.4 million of deferred financing costs. After the Company issued the notice of termination, total deferred financing costs of \$0.3 million were expensed to loss on early termination of line of credit during the nine months ended September 30, 2023.

#### U.S. Small Business Administration Loans

#### Economic Injury Disaster Loan

In 2020, the Company applied and was approved for a loan pursuant to the Economic Injury Disaster Loan ("EIDL"), administered by the U.S. Small Business Administration ("SBA"). The Company received the loan proceeds of \$0.2 million on June 15, 2020. The loan bears interest at a rate of 3.75% and matures on June 15, 2050. Installment payments, including principal and interest, of less than \$0.1 million began monthly on December 15, 2022. Each payment will first be applied to pay accrued interest, then the remaining balance will be used to reduce principal. The loan is secured by substantially all assets of DDH LLC.

Accrued and unpaid interest expense as of September 30, 2024 and December 31, 2023 was less than \$0.1 million and is included in accrued liabilities on the condensed consolidated balance sheets.

#### Overall

As of September 30, 2024, future minimum payments related to long-term debt are as follows (in thousands):

	¢	
Remaining 2024	\$	1,109
2025		11,160
2026		25,656
2027		3
2028		3
Thereafter		140
Total		38,071
Less current portion		(36,667)
Less deferred financing costs		(1,254)
Long-term debt, net	\$	150



#### Note 4 — Stockholders' Deficit and Stock-Based Compensation

#### Stockholders' Equity - Initial Public Offering

Following the completion of the Organizational Transactions, DDH LLC's limited liability company agreement was amended and restated to, among other things, appoint the Company as the sole managing member of DDH LLC and effectuate a recapitalization of all outstanding preferred units and common units into (i) economic nonvoting units of DDH LLC held by the Company and, through their indirect ownership of DDM, the Company's Chairman and Chief Executive Officer and President, and (ii) noneconomic voting units of DDH LLC, 100% of which are held by the Company. In August 2022 and December 2023, DDM tendered 100,000 and 410,000, respectively, of its limited liability company units to the Company in exchange for newly issued shares of Class A Common Stock of the Company on a one-for-one basis. In connection with these exchanges, an equivalent number of the holder's shares of Class B Common Stock were cancelled. As of September 30, 2024, DDM held 10,868,000 shares of Class B Common Stock.

The Company is authorized to issue 160,000,000 shares of Class A Common Stock, par value \$0.001 per share, 20,000,000 shares of Class B Common Stock, par value \$0.001 per share, and 10,000,000 shares of preferred stock, par value \$0.001 per share.

On February 15, 2022, the Company completed its initial public offering of 2,800,000 units ("Units"), each consisting of (i) one share of its Class A Common Stock and (ii) one warrant entitling the holder to purchase one share of its Class A Common Stock at an exercise price of \$5.50 per share. The warrants became immediately exercisable upon issuance and were exercisable for a period of five years after the issuance date. The shares of Class A Common Stock and warrants were immediately transferable separately upon issuance. As of September 30, 2024, none of these warrants were outstanding. The underwriters in our initial public offering were granted a 45-day option to purchase up to an additional 420,000 shares and/or warrants, or any combination thereof, to cover over-allotments, which they initially exercised, in part, electing to purchase warrants to purchase an additional 420,000 shares of Class A Common Stock. As of September 30, 2024, none of the underwriters of 0, 2024, none of these warrants to purchase an additional 420,000 shares of Class A Common Stock. As of September 30, 2024, none of these warrants were outstanding. In connection with the Company's initial public offering, the Company issued to the underwriters of the offering a unit purchase option to purchase (i) an additional 140,000 Units at a per Unit sold in the initial public offering, and (ii) warrants to purchase 31,000 shares of Class A Common Stock at a per warrant exercise price of \$0.012, which was equal to 120% of the public offering price per warrant sold in the offering. A group of underwriters exercised 70,000 Units and 10,500 warrants in February 2024.

The warrants had a fair value of \$0 that was calculated using the Black-Scholes option-pricing model. Variables used in the Black-Scholes option-pricing model include: (1) discount rate of 1.94% based on the applicable U.S. Treasury bill rate, (2) expected life of 5 years, (3) expected volatility of approximately 66% based on the trading history of similar companies, and (4) zero expected dividends.

On August 29, 2023, the Company filed a Tender Offer Statement on Schedule TO pursuant to which the Company offered to purchase all of its outstanding warrants for \$1.20 per warrant in cash. The Tender Offer expired at one minute after 11:59 PM, Eastern Time on September 28, 2023. The Company accepted all validly tendered warrants for purchase and settlement on October 2, 2023. As a result of the Tender Offer, a total of 2,213,652 warrants were tendered and not validly withdrawn prior to the expiration of the tender offer for a total purchase price of approximately \$2.7 million. On October 23, 2023, the Company distributed a notice of redemption to the registered holders of the remaining outstanding warrants announcing the redemption of those warrants for \$0.35 per warrant. The redemption closed on October 30, 2023, and all remaining 1,004,148 warrants were purchased for an aggregate price of approximately \$0.4 million.

#### Equity Reserve Facility

On October 18, 2024, the Company entered into a Share Purchase Agreement (the "Purchase Agreement") with New Circle Principal Investments LLC, a Delaware limited liability company ("New Circle"), pursuant to which New Circle has committed to purchase, subject to certain limitations, up to \$20 million (the "Total Commitment") of the Company's Class A common stock, par value \$0.001 per share (the "Class A Common Stock"). Under the applicable Nasdaq rules, the Company may not issue to New Circle under the Purchase Agreement more than 19.99% of the shares of all classes of the Company's common stock outstanding immediately prior to the execution of the Purchase Agreement (the "Exchange Cap"), unless (i) the Company of the Stockholder approval to issue shares of its Class A Common Stock in excess of the Exchange Cap in accordance with applicable Nasdaq rules, or (ii) the average purchase price per share paid by New Circle for all shares of the Company's Class A Common Stock, if any, that the Company elects to sell to New Circle under the Purchase Agreement equals or exceeds certain minimums permitted under the rules of the Nasdaq Stock Market. The purchase price of the shares that may be sold to New Circle under the Purchase Agreement will be based on an agreed upon fixed discount to the market price of our Class A Common Stock as computed under the Purchase Agreement.

As consideration for New Circle's irrevocable commitment to purchase shares of the Company's Class A Common Stock upon the terms of and subject to satisfaction of the conditions set forth in the Purchase Agreement, the Company paid New Circle structuring and legal fees of less than \$0.1 million. In addition, the Company will pay a commitment fee of \$150,000 to New Circle, which we may issue in the form of the Company's Class A Common Stock (the "Commitment Fee"), the market value of which shall be determined based on the closing price of the Class A Common Stock on the date the Registration Statement is declared effective by the SEC; provided, however, that the Company may, in its sole discretion, elect to pay any portion of the Commitment Fee in cash, so long as such amount is paid on or prior to the day of filing of the Registration Statement filed in order to register the Company's Class A Common Stock sold under the Purchase Agreement.

The Purchase Agreement will automatically terminate on the earliest of (i) the 36-month anniversary of the of the Purchase Agreement, (ii) the date on which New Circle shall have made payment to the Company for Class A Common Stock equal to the Total Commitment or (iii) the date any statute, rule, regulation, executive order, decree, ruling or injunction that would prohibit any of the transactions contemplated by the Purchase Agreement goes into effect. The Company has the right to terminate the Purchase Agreement at any time, at no cost or penalty, upon five trading days' prior written notice to New Circle so long as (a) there are no outstanding purchase notices under which our Class A Common Stock have yet to be issued and (b) the Company has paid all amounts owed to New Circle pursuant to the Purchase Agreement. The Company and New Circle may also agree to terminate the Purchase Agreement by mutual written consent.

#### Noncontrolling Interest

Direct Digital Holdings, Inc. is the sole managing member of DDH LLC, and consolidates the financial results of DDH LLC. Therefore, Direct Digital Holdings, Inc. reports a noncontrolling interest ("NCI") based on the common units of DDH LLC held by DDM. While Direct Digital Holdings, Inc. retains its controlling interest in DDH LLC, changes in its ownership interest in DDH LLC are accounted for as equity transactions. As such, future redemptions or direct exchanges of LLC Units by DDM will result in a change in ownership and reduce or increase the amount recorded as noncontrolling interest and increase or decrease additional paid-in capital when DDH LLC has positive or negative net assets, respectively.

#### Stock-Based Compensation Plans

In connection with the initial public offering, the Company adopted the 2022 Omnibus Incentive Plan ("2022 Omnibus Plan") to facilitate the grant of equity awards to the Company's employees, consultants and non-employee directors. The Company's board of directors reserved 1,500,000 shares of Class A Common Stock for issuance in equity awards under the 2022 Omnibus Plan. Information on activity for both the stock options and RSUs is detailed below.

During the nine months ended September 30, 2024 and 2023, the Company recognized \$0.8 million and \$0.5 million, respectively, of total stock-based compensation expense in the condensed consolidated statement of operations in compensation, tax and benefits.

#### Stock Options

Options to purchase shares of common stock vest annually on the grant date anniversary over a period of three years and expire 10 years following the date of grant. The following table summarizes the stock option activity under the 2022 Omnibus Plan as of September 30, 2024:

	Stock Options							
	Shares		Weighted Average Exercise Price	Weighted Average Contractual Life (in years)	Aggregate Intrinsic Value (in thousands)			
Outstanding at January 1, 2024	371,116	\$	2.51	8.77	\$ 4,591			
Granted	—	\$	—	— \$	§ —			
Exercised	(12,557)	\$	2.39	— 9	\$ 179			
Forfeited	(17,701)	\$	2.98	— \$	\$ 36			
Outstanding at September 30, 2024	340,858	\$	2.49	7.92	\$ 97			
Vested and exercisable at September 30, 2024	172,543	\$	2.17	7.64	\$ 63			

As of September 30, 2024, unrecognized stock-based compensation of \$0.2 million related to 168,315 of unvested stock options which will be recognized on a straight-line basis over a weighted-average vesting period of 1.20 year.

#### Restricted Stock Units

RSUs generally vest annually on the grant date anniversary over a period of three years. A summary of RSU activity during the nine months ended September 30, 2024 and related information is as follows:

	Rest	ricted Stock Units	
	Number of Shares	Grant D	nted Average Date Fair Value er Share
Unvested- January 1, 2024	542,396	\$	2.87
Granted	99,474	\$	16.90
Vested	(362,799)	\$	6.76
Forfeited	(14,111)	\$	3.01
Unvested- September 30, 2024	264,960	\$	2.79

The majority of vested RSUs were net share settled such that the Company withheld shares with a value equivalent to the employees' obligation for the applicable income and other employment taxes. The total shares withheld were 98,036 and were based on the value of the RSUs on their respective vesting dates as determined by the Company's closing stock price. As of September 30, 2024, there was unrecognized stock-based compensation of \$0.5 million related to unvested RSUs which will be recognized on a straight-line basis over a weighted average period of 1.23 years.

#### Note 5 — Tax Receivable Agreement and Income Taxes

#### Tax Receivable Agreement

The Company's TRA with DDH LLC and DDM (together, the "TRA Holders") provides for payment by the Company to the TRA Holders of 85% of the net cash savings, if any, in U.S. federal, state and local income tax and franchise tax that the Company actually realizes or is deemed to realize in certain circumstances. The Company retains the benefit of the remaining 15% of these net cash savings.

The TRA liability is calculated by determining the tax basis subject to the TRA ("tax basis") and applying a blended tax rate to the basis differences and calculating the resulting impact. The blended tax rate consists of the U.S. federal income tax rate and assumed combined state and local income tax rate driven by the apportionment factors applicable to each state. Any taxable income or loss generated by the Company will be allocated to TRA Holders in accordance with the LLC Agreement and the TRA, and distributions to the owners of LLC Units in an amount sufficient to fund their tax obligations will be made. Pursuant to the Company's election under Section 754 of the Code in 2022, the Company expects to obtain an increase in its share of the tax basis in the net assets of DDH, LLC when LLC interests are redeemed or exchanged by the members of DDH, LLC. In August 2022 and December 2023, members of DDM exchanged 100,000 and 410,000 Class B shares into Class A shares, respectively.

The Company has recorded a liability related to the tax receivable agreement of less than \$0.1 million and \$5.2 million as of September 30, 2024 and December 31, 2023, respectively. The Company has recorded a deferred tax asset primarily from the outside basis difference in the partnership interest of \$0.0 million and \$6.1 million as of September 30, 2024 and December 31, 2023, respectively. The deferred tax asset is net of a valuation allowance of \$7.3 million and \$0.5 million as of September 30, 2024 and December 31, 2023, respectively. The deferred tax asset is net of a valuation allowance of \$7.3 million and \$0.5 million as of September 30, 2024 and December 31, 2023, respectively. TRA payments of \$0 and less than \$0.1 million were made during the nine months ended September 30, 2024 and 2023, respectively. The payments under the TRA will not be conditional on holder of rights under the TRA having a continued ownership interest in either DDH LLC or the Company. The Company may elect to defer payments due under the TRA if the Company does not have available cash to satisfy its payment obligations under the TRA. Any such deferred payments under the TRA generally will accrue interest from the due date for such payment until the payment date. The Company accounts for any amounts payable under the TRA in accordance with ASC Topic 450, *Contingencies*, and recognizes subsequent period changes to the measurement of the liability from the TRA in the statement of operations as a component of income before taxes. For the three and nine months ended September 30, 2024, \$5.2 million was recognized as income under other income (expense) due to the derecognition of the TRA liability, as a valuation allowance was recorded against the deferred taxes associated with the TRA.



The term of the TRA commenced upon completion of the initial public offering and will continue until all tax benefits that are subject to the TRA have been utilized or expired, unless the Company exercises its right to terminate the TRA. If the Company elects to terminate the TRA early (or it is terminated early due to changes in control), the obligations under the TRA would accelerate and the Company would be required to make an immediate payment equal to the present value of the anticipated future payments to be made by the Company under the TRA.

#### Income Taxes

Through the Organizational Transactions completed in February 2022, the Company formed an Up-C structure which allows DDM to continue to realize tax benefits associated with owning interests in an entity that is treated as a partnership for U.S. federal income tax purposes. Under the Up-C structure, the Company is subject to corporation income tax on the variable ownership changes. The ownership was 20.45% as of January 1, 2023 and increased to 23.35% in the fourth quarter of 2023. There was no exchange of shares of Class B common stock for shares of Class A common stock in the nine months ended September 30, 2024.

The Company recorded a tax benefit for federal and state income tax for which the components and the effective income tax rates are as follows (in thousands):

	Three Months Ended September 30,			Nine Mo Septer	nths End nber 30,	
	2024		2023	 2024		2023
Income tax expense	\$ 6,606	\$	166	\$ 6,132	\$	166
Effective income tax rate	2884.7 %		4.7 %	(85.1 %)	)	4.9 %

The effective tax rates were lower than the statutory tax rates for the three and nine months ended September 30, 2023 primarily due to the Company's partnership loss that is not subject to federal and state taxes. The effective tax rates were different from the statutory rates for the three and nine months ended September 30, 2024 primarily due to recording a valuation allowance against deferred taxes.

As of September 30, 2024, the Company had federal net operating loss carryforwards of \$4.4 million that can be carried forward indefinitely.

The Company files income tax returns in the United States federal jurisdiction and various state jurisdictions. In the normal course of business, the Company can be examined by various tax authorities, including the Internal Revenue Service in the United States. There are currently no federal or state audits in process. The Company analyzes its tax filing positions in all of the U.S. federal, state and local tax jurisdictions where it is required to file income tax returns, as well as for all open tax years in these jurisdictions. Federal and various states returns for the years ended December 2022 and 2021 remain open as of September 30, 2024. The Company evaluates tax positions taken or expected to be taken in the course of preparing an entity's tax returns to determine whether it is "more-likely-than-not" that each tax position will be sustained by the applicable tax authority. As of September 30, 2024 and December 31, 2023, the Company had no uncertain tax positions. Accordingly, the Company has not recognized any penalty, interest or tax impact related to uncertain tax positions.

#### Note 6 — Related Party Transactions

#### Related Party Transactions

#### Member Payable

As of September 30, 2024 and December 31, 2023, the Company had a net receivable from members that totaled \$1.7 million, which is included as a related party receivable on the condensed consolidated balance sheets.

#### Up-C Structure

In February 2022, the Company completed an initial public offering of its securities, and through the Organizational Transactions, formed an Up-C structure, which is often used by partnerships and limited liability companies and allows DDM, a Delaware limited liability company indirectly owned by Mark Walker ("Walker") and Keith Smith ("Smith"), to retain its equity ownership in DDH LLC and to continue to realize tax benefits associated with owning interests in an entity that is treated as a partnership, or "pass-through" entity, for U.S. federal income tax purposes. DDM holds economic



nonvoting LLC Units in DDH LLC and holds noneconomic voting equity interests in the form of the Class B Common Stock in Direct Digital Holdings (See Note 4 — Stockholders' Deficit and Stock-Based Compensation). One of the tax benefits to DDM associated with this structure is that future taxable income of DDH LLC that is allocated to DDM will be taxed on a pass-through basis and therefore will not be subject to corporate taxes at the entity level. Additionally, DDM may, from time to time, redeem or exchange its LLC Units for shares of the Company's Class A Common Stock on a one-for-one basis. The Up-C structure also provides DDM with potential liquidity that holders of non-publicly traded limited liability companies are not typically afforded. If the Company ever generates sufficient taxable income to utilize the tax benefits, DDH expects to benefit from the Up-C structure because, in general, the Company expects cash tax savings in amounts equal to 15% of certain tax benefits arising from such redemptions or exchanges of DDM's LLC Units for Class A Common Stock or cash and certain other tax benefits covered by the TRA. As described in Note 5 — Tax Receivable Agreement and Income Taxes, for the three and nine months ended September 30, 2024, \$5.2 million was recorded as income in other income (expense) for such change as the deferred taxes giving rise to the TRA have a valuation allowance recorded to offset the deferred tax assets.

The aggregate balance of tax receivable liabilities as of September 30, 2024 and December 31, 2023, is as follows (in thousands):

	September 30, 2024	December 31, 2023
Liability related to tax receivable agreement		
Short term	\$ 41	\$ 41
Long term	—	5,201
Total liability related to tax receivable agreement	\$ 41	\$ 5,242

#### Note 7 — Segment Information

Revenue by business segment is as follows (in thousands):

	Three Months Ended September 30,			Nine Months Ended September 30,				
		2024		2023		2024		2023
Sell-side advertising	\$	2,202	\$	51,622	\$	33,001	\$	89,006
Buy-side advertising		6,873		7,850		20,204		27,093
Total revenues	\$	9,075	\$	59,472	\$	53,205	\$	116,099

Operating loss by business segment reconciled to loss before income taxes is as follows (in thousands):

	Three Months Ended September 30,			Nine Months Ended September 30,				
		2024		2023		2024		2023
Sell-side advertising	\$	(873)	\$	6,206	\$	536	\$	9,657
Buy-side advertising		1,349		1,551		3,490		6,980
Corporate office expenses		(4,134)		(3,263)		(12,555)		(10,030)
Total operating (loss) income		(3,658)		4,494		(8,529)		6,607
Corporate other expense		3,887		(977)		1,323		(3,229)
Income (loss) before income taxes	\$	229	\$	3,517	\$	(7,206)	\$	3,378



Total assets by business segment are as follows (in thousands):

	September 30, 2024		December 31, 2023
Sell-side advertising	\$ 3,	807 \$	34,354
Buy-side advertising	20,	604	22,539
Corporate office	6,	80	13,779
Total assets	\$ 31,	.91 \$	5 70,672

#### Note 8 — Net (Loss) Income Per Share

The Company has two classes of common stock, Class A and Class B. Shares of the Company's Class B Common Stock do not share in the earnings or losses attributable to Direct Digital Holdings, Inc. and are therefore not participating securities. The Company uses the two-class method to calculate basic and diluted earnings per share as a result of outstanding participating securities in the form of warrants for the three and nine months ended September 30, 2023. The following table sets forth the computation of the Company's basic and diluted net (loss) income per share (in thousands, except per share amounts):

Three Months Ended September 30,					Nine Months Ended September 30,				
	2024		2023		2024		2023		
\$	(2,690)	\$	571	\$	(4,055)	\$	549		
	_		296		—		285		
\$	(2,690)	\$	275	\$	(4,055)	\$	264		
	3,793		2,990		3,667		2,938		
	_		_		_		_		
	_		20		_		24		
	_		34		_		118		
	3,793		3,044		3,667		3,080		
\$	(0.71)	\$	0.09	\$	(1.11)	\$	0.09		
\$	(0.71)	\$	0.09	\$	(1.11)	\$	0.09		
	\$ \$ \$ \$	Septem           2024           \$ (2,690)	September 30           2024         \$         (2,690)         \$           \$         (2,690)         \$         \$           \$         (2,690)         \$         \$           3,793	September 30,           2024         2023           \$         (2,690)         \$         571           —         296         \$         275           \$         (2,690)         \$         275           \$         (2,690)         \$         275           -         -         -         -           -         -         -         -           -         -         -         -           -         -         200         -         -           -         -         200         -         -         -           -         -         200         -         -         -         -           -         -         -         200         -         -         34         -         3,793         3,044         -         -         0.09         -	September 30,           2024         2023           \$         (2,690)         \$         571         \$	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	September 30,         September 30           2024         2023         2024           \$ (2,690)         \$ 571         \$ (4,055)         \$ $-$ 296 $ -$ \$ (2,690)         \$ 275         \$ (4,055)         \$           \$ (2,690)         \$ 275         \$ (4,055)         \$ $-$ 296 $ -$ \$ (2,690)         \$ 275         \$ (4,055)         \$ $  -$		

The following weighted-average outstanding shares of common stock equivalents were excluded from the computation of diluted net (loss) income per share attributable to common stockholders for the periods presented because including them would have been anti-dilutive (in thousands):

	Three Mont Septemb		Nine Mon Septem	
	2024	2023	2024	2023
Class B Common Stock	10,868	11,278	10,868	11,278
Options to purchase common stock	346	343	362	365
Unvested restricted stock units	268	495	402	382
Total excludable from net loss per share attributable to common stockholders - diluted	11,482	12,116	11,632	12,025

#### Note 9 — Commitments and Contingencies

#### Litigation

We may from time to time be subject to various legal or administrative claims and proceedings arising in the ordinary course of business. As of the date hereof, except as set forth below, we are not a party to any material legal or administrative proceedings nor are there any proceedings in which any of our directors, executive officers or affiliates, or

any registered or beneficial stockholder, is an adverse party or has a material interest adverse to our interest. Litigation or any other legal or administrative proceeding, regardless of the outcome, is likely to result in substantial cost and diversion of our resources, including our management's time and attention.

On May 10, 2024, the Company was the subject of a defamatory article / blog post. In connection with this post, one of the Company's sell-side customers paused its connection to the Company while the allegations were investigated. This customer reconnected the Company on May 22, 2024 and sell-side volumes have resumed but not yet at the levels experienced prior to the pause in May 2024. The Company is actively working with its partners to achieve prior volume levels. On May 14, 2024, the Company filed a lawsuit against the author of the defamatory article and is vigorously pursuing its rights. The Company cannot make any predictions about the final outcome of this litigation matter or the timing thereof.

On May 23, 2024, an alleged stockholder, purportedly on behalf of the persons or entities who purchased or acquired publicly traded securities of the Company between April 2023 and March 2024, filed a putative class action against the Company, certain of our officers and directors, and other defendants in the U.S. District Court for the Southern District of Texas, alleging violations of federal securities laws related to alleged false or misleading disclosures made by the Company in its public filings. On July 9, 2024, another alleged stockholder filed a similar securities class action against the Company, certain of our officers and directors, also in the Southern District of Texas. The two actions have been consolidated. Each of these complaints seeks unspecified damages, plus costs, fees, and attorneys' fees. The Company cannot make any predictions about the final outcome of this matter or the timing thereof but believes that plaintiffs' claims lack merit and intends to vigorously defend these lawsuits.

#### **Operating** Leases

During the nine months ended September 30, 2024 and 2023, the Company incurred fixed rent expense associated with operating leases for real estate of \$0.2 million. The Company did not have any finance leases, short-term leases nor variable leases over this time period. During the three and nine months ended September 30, 2024 and 2023, the Company had the following cash and non-cash activities associated with leases (in thousands):

	Three Months Ended September 30,					Nine Months Ended September			
		2024		2023		2024		2023	
Cash paid for amounts included in the measurement of lease liabilities:									
Operating cash outflow for operating leases	\$	71	\$	41	\$	142	\$	121	
Non-cash changes to the operating lease ROU assets and operating lease									
liabilities									
Additions and modifications to ROU asset obtained from new operating liabilities	\$	_	\$	_	\$	200	\$		

The weighted-average remaining lease term and discount rate for the Company's operating leases is 4.8 years and 8.3%, respectively, as of September 30, 2024. The weighted-average remaining lease term and discount rate for the Company's operating leases is 6.1 years and 8.4%, respectively, as of September 30, 2023.

The future payments due under operating leases as of September 30, 2024 are as follows (in thousands):

2024	\$ 64
2025	258
2026	265
2027	269
2028	167
Thereafter	200
Total undiscounted lease payments	1,223
Less effects of discounting	(208)
Less current lease liability	(183)
Total operating lease liability, net of current portion	\$ 832

#### Note 10 — Property, Equipment and Software, net

Property, equipment and software, net consists of the following (in thousands):

	Useful Life (Years)	September 30, 2024	December 31, 2023
Furniture and fixtures	5	\$ 137	\$ 128
Computer equipment	3	20	20
Leasehold improvements	15	42	36
Capitalized software	3	702	702
Property, equipment and software, gross		901	886
Less: accumulated depreciation and amortization		(492)	(287)
Total property, equipment and software, net		\$ 409	\$ 599

The following table summarizes depreciation and amortization expense related to property, equipment and software by line item for the three and nine months ended September 30, 2024 and 2023 (in thousands):

	Three Months Ended September 30,					Nine Months Ended September 30,					
		2024		2023		2024		2023			
Cost of revenue	\$	41	\$	42	\$	125	\$	125			
General and administrative		26		22		80		60			
Total depreciation and amortization	\$	67	\$	64	\$	205	\$	185			

#### Note 11 — Intangible Assets, net

The Company records amortization expense on a straight-line basis over the life of the identifiable intangible assets related to an acquisition in September 2020. For the three months ended September 30, 2024 and 2023, amortization expense of \$0.5 million and for the nine months ended September 30, 2024 and 2023, amortization expense of \$1.5 million, respectively, was recognized. As of September 30, 2024 and December 31, 2023, intangible assets net of accumulated amortization was \$10.2 million and \$11.7 million, respectively.



As of September 30, 2024, intangible assets and the related accumulated amortization, weighted-average remaining life and future amortization expense are as follows:

		September 30, 2024										
	Weighted-Average	Original			Accumulated		Net					
	Remaining Life (Years)		Amount		Amortization		Total					
Customer Lists	6.0	\$	13,028	\$	(5,211)	\$	7,817					
Trademarks and tradenames	6.0		3,501		(1,400)		2,101					
Non-compete agreements	1.0		1,505		(1,204)		301					
Total intangible assets, net		\$	18,034	\$	(7,815)	\$	10,219					

	December 31, 2023										
	Weighted-Average Remaining Life (Years)	Original Amount		Accumulated Amortization		Net Total					
Customer Lists	6.8	\$	13,028	\$		\$	8,794				
Trademarks and tradenames	6.8	Ψ	3,501	Ψ	(1,138)	φ	2,363				
Non-compete agreements	1.8		1,505		(978)		527				
Total intangible assets, net		\$	18,034	\$	(6,350)	\$	11,684				

	Total
2024	\$ 488
2025	1,879
2026	1,653
2027	1,653
2028	1,653
Thereafter	2,893
Total future amortization expense	\$ 10,219

#### ITEM 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

You should read the following discussion together with our unaudited condensed consolidated financial statements and the related notes included elsewhere in this Quarterly Report on Form 10-Q and our audited consolidated financial statements and the related notes included in our Annual Report on Form 10-K for the fiscal year ended December 31, 2023. This discussion contains forward-looking statements based upon current expectations that involve risks and uncertainties. Our actual results may differ materially from those anticipated in these forward-looking statements as a result of various factors, including those set forth under the section titled "*Risk Factors*" in our Annual Report on Form 10-K or in other parts of this Quarterly Report on Form 10-Q. See "*Cautionary Note Regarding Forward-Looking Statements*" below. Our historical results are not necessarily indicative of the results that may be expected for any period in the future.

#### **Cautionary Note Regarding Forward-Looking Statements**

This Quarterly Report on Form 10-Q contains forward-looking statements within the meaning of the federal securities laws and which are subject to certain risks, trends and uncertainties. We use words such as "could," "would," "may," "might," "will," "expect," "likely," "believe," "continue," "anticipate," "estimate," "intend," "plan," "project" and other similar expressions to identify forward-looking statements, but not all forward-looking statements include these words. All of our forward-looking statements involve estimates and uncertainties that could cause actual results to differ materially from those expressed in or implied by the forward-looking statements. Accordingly, any such statements are qualified in their entirety by reference to the information described under the caption "*Risk Factors*" in our Annual Report on Form 10-K for the fiscal year ended December 31, 2023.

The forward-looking statements contained in this Quarterly Report on Form 10-Q are based on assumptions that we have made in light of our industry experience and our perceptions of historical trends, current conditions, expected future developments and other factors we believe are appropriate under the circumstances. As you read and consider this Quarterly Report on Form 10-Q, you should understand that these statements are not guarantees of performance or results. They involve risks, uncertainties (many of which are beyond our control) and assumptions.

Although we believe that these forward-looking statements are based on reasonable assumptions, you should be aware that many factors could affect our actual operating and financial performance and cause our performance to differ materially from the performance expressed in or implied by the forward-looking statements. We believe these factors include, but are not limited to, the following:

- the restrictions and covenants imposed upon us by our credit facilities;
- · the substantial doubt about our ability to continue as a going concern, which may hinder our ability to obtain future financing;
- · our ability to secure additional financing to meet our capital needs;
- ineligibility to file short-form registration statements on Form S-3, which may impair our ability to raise capital;
- failure to satisfy applicable listing standards of the Nasdaq Capital Market resulting in a potential delisting of our common stock;
- costs, risks and uncertainties related to the restatement of certain prior period financial statements;
- any significant fluctuations caused by our high customer concentration;
- risks related to non-payment by our clients;
- · reputational and other harms caused by our failure to detect advertising fraud;
- operational and performance issues with our platform, whether real or perceived, including a failure to respond to technological changes or to upgrade our technology systems;
- restrictions on the use of third-party "cookies," mobile device IDs or other tracking technologies, which could diminish our platform's effectiveness;
- unfavorable publicity and negative public perception about our industry, particularly concerns regarding data privacy and security relating to our industry's technology and practices, and any perceived failure to comply with laws and industry self-regulation;
- our failure to manage our growth effectively;



- the difficulty in identifying and integrating any future acquisitions or strategic investments;
- any changes or developments in legislative, judicial, regulatory or cultural environments related to information collection, use and processing;
- challenges related to our buy-side clients that are destination marketing organizations and that operate as public/private partnerships;
- any strain on our resources or diversion of our management's attention as a result of being a public company;
- the intense competition of the digital advertising industry and our ability to effectively compete against current and future competitors;
- any significant inadvertent disclosure or breach of confidential and/or personal information we hold, or of the security of our or our customers', suppliers' or other partners' computer systems;
- as a holding company, we depend on distributions from Direct Digital Holdings, LLC ("DDH LLC") to pay our taxes, expenses (including payments under the Tax Receivable Agreement) and any amount of any dividends we may pay to the holders of our common stock;
- the fact that DDH LLC is controlled by DDM, whose interest may differ from those of our public stockholders;
- · any failure by us to maintain or implement effective internal controls or to detect fraud; and
- other factors and assumptions discussed under "Risk Factors" and elsewhere in this Quarterly Report on Form 10-Q and in our Annual Report on Form 10-K for the fiscal year ended December 31, 2023.

Should one or more of these risks or uncertainties materialize or should any of these assumptions prove to be incorrect, our actual operating and financial performance may vary in material respects from the performance projected in these forward-looking statements. Further, any forward-looking statement speaks only as of the date on which it is made, and except as required by law, we undertake no obligation to update any forward-looking statement contained in this Quarterly Report on Form 10-Q to reflect events or circumstances after the date on which it is made or to reflect the occurrence of anticipated or unanticipated events or circumstances. New factors that could cause our business not to develop as we expect emerge from time to time, and it is not possible for us to predict all of them. Further, we cannot assess the impact of each currently known or new factor on our results of operations or the extent to which any factor, or combination of factors, may cause actual results to differ materially from those contained in any forward-looking statements.

#### Overview

Direct Digital Holdings, Inc. and its subsidiaries (collectively the "Company," "DDH," "we," "us" and "our"), headquartered in Houston, Texas, is an end-to-end, fullservice programmatic advertising platform primarily focused on providing advertising technology, data-driven campaign optimization and other solutions intended for underserved and less efficient markets on both the sell- and buy-side of the digital advertising ecosystem. Direct Digital Holdings, Inc. is the holding company that, since the completion of our initial public offering on February 15, 2022, owns certain common units, and serves as the manager of DDH LLC, which operates the business formed in 2018 through the acquisition of Colossus Media, LLC ("Colossus Media"), a sell-side marketing platform, and Huddled Masses, LLC ("Huddled Masses<sup>TM</sup>" or "Huddled Masses"), a buy-side marketing platform.

In late September 2020, DDH LLC acquired Orange142, LLC ("Orange 142") to further bolster its overall programmatic buy-side advertising platform and enhance its offerings across multiple industry verticals such as travel, education, healthcare, financial services, consumer products and other sectors, with particular emphasis on small- and mid-sized businesses transitioning into digital with growing digital media budgets.

Direct Digital Holdings, Inc. owns 100% of the voting interest in Direct Digital Holdings, LLC. As of September 30, 2024, DDH owns 25.9% of the economic interest in Direct Digital Holdings, LLC. See further discussion of the Up-C structure in Note 6 of our condensed consolidated financial statements. Direct Digital Holdings, LLC was formed on June

21, 2018 and acquired by DDH on February 15, 2022 in connection with the Organizational Transactions. Direct Digital Holdings, LLC's wholly-owned subsidiaries are as follows:

Subsidiary	Current % Ownership	Business Segment	Date of Formation	Date of Acquisition
Colossus Media, LLC	100%	Sell-side	September 8, 2017	June 21, 2018
Orange142, LLC	100%	Buy-side	March 6, 2013	September 30, 2020
Huddled Masses, LLC	100%	Buy-side	November 13, 2012	June 21, 2018

Colossus Media operates our proprietary sell-side programmatic platform operating under the trademarked banner of Colossus SSP<sup>TM</sup> ("Colossus SSP"). Colossus SSP is a stand-alone sell-side platform ("SSP") intended to deliver targeted advertising to diverse and multicultural audiences as well as to general audiences, including African Americans, Latin Americans, Asian Americans and LGBTQIA+ customers, as well as general audiences. Both buy-side advertising businesses, Orange 142 and Huddled Masses, offer technology-enabled advertising solutions and consulting services to clients through demand side platforms ("DSPs").

Providing both the front-end, buy-side advertising businesses coupled with our proprietary sell-side operations enables us to curate the first through the last mile in the ad tech ecosystem execution process to drive higher results.

Operating segments are components of an enterprise for which separate financial information is available and evaluated regularly by our chief operating decision maker ("CODM") for purpose of allocating resources and assessing performance. Our CODM is our Chairman and Chief Executive Officer. We operate as two reportable segments: sell-side advertising, which includes the results of Colossus Media, and buy-side advertising, which includes the results of Orange 142 and Huddled Masses. All our revenues are attributable to the United States.

#### **Recent Developments**

#### Nasdaq Rule Noncompliance.

On October 18, 2024, we received a deficiency letter (the "Letter") from the Listing Qualifications Department of Nasdaq (the "Staff") notifying the Company that it was not in compliance with the minimum stockholders' equity requirement for continued listing on Nasdaq under Nasdaq Listing Rule 5550(b)(1). Nasdaq Listing Rule 5550(b)(1) requires companies listed on Nasdaq to maintain stockholders' equity of at least \$2.5 million (the "Stockholders' Equity Requirement"). The Company's Quarterly Report on Form 10-Q for the period ended June 30, 2024 reported a stockholders' deficit of \$8.77 million. The Letter further noted that as of the letter date, the Company did not have a market value of listed securities of \$35 million, or net income from continued operations of \$500,000 in the most recently completed fiscal year or in two of the last three most recently completed fiscal years, which are the alternative quantitative standards to the Stockholders' Equity Requirement for continued listing on Nasdaq. In accordance with the Nasdaq Listing Rules, the Company was provided 45 calendar days, or until December 2, 2024, to submit a plan to regain compliance (the "Compliance Plan"). If the Compliance Plan is acceptable to the Staff, the Staff may grant an extension of up to 180 calendar days from the date of the Letter. If the Staff does not accept the Compliance Plan, the Staff will provide written notification to the Company that the Compliance Plan has been rejected. At that time, the Company may appeal the Staff's determination to a Nasdaq Hearings Panel. The Company intends to submit a Compliance Plan or before December 2, 2024. Further, the Company intends to take all reasonable measures available to regain compliance under the Nasdaq Listing Rules and remain listed on Nasdaq, including by capital-raising activities such as through the Purchase Agreement, defined below, for the equity reserve facility described in Note 4 — Stockholders' Deficit and Stock-Based Compensation. However, there can be no assurance tha

#### Equity Reserve Facility.

On October 18, 2024, the Company entered into a Share Purchase Agreement (the "Purchase Agreement") with New Circle Principal Investments LLC, a Delaware limited liability company ("New Circle"), pursuant to which New Circle has committed to purchase, subject to certain limitations, up to \$20 million (the "Total Commitment") of the Company's Class A common stock, par value \$0.001 per share (the "Class A Common Stock"). Under the applicable Nasdaq rules, the Company may not issue to New Circle under the Purchase Agreement more than 19.99% of the shares of all classes of the Company's common stock outstanding immediately prior to the execution of the Purchase Agreement (the "Exchange Cap"), unless (i) the Company obtains stockholder approval to issue shares of its Class A Common Stock in excess of the Exchange Cap in accordance with applicable Nasdaq rules, or (ii) the average purchase price per share paid by New Circle for all shares of the Company's Class A Common Stock, if any, that the Company elects to sell to New Circle under the



Purchase Agreement equals or exceeds certain minimums permitted under the rules of the Nasdaq Stock Market. The purchase price of the shares that may be sold to New Circle under the Purchase Agreement will be based on an agreed upon fixed discount to the market price of our Class A Common Stock as computed under the Purchase Agreement. As consideration for New Circle's irrevocable commitment to purchase shares of the Company's Class A Common Stock upon the terms of and subject to satisfaction of the conditions set forth in the Purchase Agreement, the Company paid New Circle structuring and legal fees of less than \$0.1 million. In addition, the Company will pay a commitment fee of \$150,000 to New Circle, which we may issue in the form of the Company's Class A Common Stock (the "Commitment Fee"), the market value of which shall be determined based on the closing price of the Class A Common Stock on the date the Registration Statement is declared effective by the SEC; provided, however, that the Company may, in its sole discretion, elect to pay any portion of the Commitment Fee in cash, so long as such amount is paid on or prior to the day of filing of the Registration Statement filed in order to register the Company's Class A Common Stock sold under the Purchase Agreement. The Purchase Agreement will automatically terminate on the earliest of (i) the 36-month anniversary of the Purchase Agreement, (ii) the date on which New Circle shall have made payment to the Company for Class A Common Stock equal to the Total Commitment or (iii) the date any statute, rule, regulation, executive order, decree, ruling or injunction that would prohibit any of the transactions contemplated by the Purchase Agreement goes into effect. The Company has the right to terminate the Purchase Agreement at any time, at no cost or penalty, upon five trading days' prior written notice to New Circle so long as (a) there are no outstanding purchase notices under which our Class A Common Stock have yet to be issued and (b) the Company has paid a

#### Relationship with Sell-Side Customer.

On May 10, 2024, the Company was the subject of a defamatory article / blog post which the Company believes was part of a coordinated misinformation campaign. In connection with this post, one of the Company's sell-side customers paused its connection to the Company while the allegations were investigated. This customer reconnected the Company on May 22, 2024 and sell-side volumes have resumed but not yet at the levels experienced prior to the pause in May 2024. The Company is actively working with its partners to achieve prior volume levels. On May 14, 2024, the Company filed a lawsuit against the author of the defamatory article and is vigorously pursuing its rights. The Company cannot make any predictions about the final outcome of this litigation matter or the timing thereof.

#### Buy-side Unification.

On October 31, 2024, the Company announced the unification of its buy-side businesses, Orange 142 and Huddled Masses. The unification did not have a material impact on the Company's business operations nor on the Company's condensed consolidated financial statements.

#### **Key Factors Affecting Our Performance**

We believe our growth and financial performance are dependent on many factors, including those described below.

#### Sell-side advertising business

#### Increasing revenue from customers through increased advertising spend from buyers

Colossus Media operates our proprietary sell-side programmatic platform operating under the trademarked banner of Colossus SSP. Our customers (or buyers) include ad exchanges, DSPs, agencies and individual advertisers. We have broad exposure to the ecosystem of buyers, reaching on average approximately 176,000 advertisers per month in the three months ended September 30, 2024, an increase of 71% over the 103,000 advertisers per month in the three months ended September 30, 2023. As spending on programmatic advertising increasingly becomes a larger share of the overall ad spend, advertisers and agencies are seeking greater control of their digital advertising supply chains. To take advantage of this industry shift, we have entered into Supply Path Optimization agreements directly with customers which address acceptable advertisements and data usage. As part of these agreements, we provide advertisers and agencies with benefits ranging from custom data and workflow integrations, product features, volume-based business terms, and visibility into campaign performance data and methodology. As a result of these direct relationships, our existing advertisers and agencies are incentivized to allocate an increasing percentage of their advertising budgets to our platform.

We also strive to retain existing publishers and add new publishers. Establishing multiple header bidding integrations by leveraging our technology capabilities allows us to maximize our access to publishers' ad formats, devices and various properties that a publisher may own. We may also up-sell additional products including our header bidding management, identity, and audience solutions. We enter into master service agreements with our publishers which, among other terms, set a fixed rate for content to be sold on Colossus SSP. Our strategy on the sell-side advertising business represents growth

potential, and we believe we are well positioned to be able to bring underserved multicultural publishers into the advertising ecosystem, thereby increasing our value proposition across all customers, including large advertisers and agencies.

#### Monetizing ad impressions for publishers and buyers

We curate advertisers and increase access to publishers with valuable ad impressions. We focus on monetizing digital impressions by coordinating daily real-time auctions and bids. Each time the publisher's web page loads, an ad request is sent to multiple ad exchanges and, in some cases, to the demand side platform directly from Colossus SSP. In case of real-time bidding ("RTB") media buys, many DSPs would place bids to the impressions being offered by the publisher during the auction. The advertiser that bids a higher amount compared to other advertisers will win the bid and pay the second highest price for the winning impression to serve the ads. We continuously review our available inventory from existing publishers across every format (mobile, desktop, digital video, OTT, CTV, and rich media). The factors we consider when determining which impressions we process include transparency, viewability, and whether or not the impression is human sourced. By consistently applying these criteria, we believe the ad impressions we process will be valuable and marketable to advertisers.

#### Enhancing ad inventory quality

In the advertising industry, inventory quality is assessed in terms of invalid traffic ("IVT") which can be impacted by fraud such as "fake eyeballs" generated by automated technologies set up to artificially inflate impression counts. Through our platform design and proactive IVT mitigation efforts, we address and minimize IVT on a number of fronts, including sophisticated technology, which detects and avoids IVT on the front end; direct publisher and inventory relationships, for supply path optimization; and ongoing campaign and inventory performance review, to ensure inventory quality and brand protection controls are in place.

#### Growing access to valuable ad impressions

Historically, our growth has been driven by a variety of factors including increased access to mobile web (display and video) and mobile app (display and video) impressions and desktop video impressions. Our performance is affected by our ability to maintain and grow our access to valuable ad impressions from current publishers as well as through new relationships with publishers. For the three months ended September 30, 2024, we processed over 513.3 billion average monthly bid requests, down 51% from 2023, reflecting the pause in service by one of our sell-side customers in May 2024 which has not yet returned to pre-pause levels.

#### Expanding and managing investments

Each impression or transaction occurs in a fraction of a second. Given that most transactions take place in an auction/bidding format, we continue to make investments across the platform to further reduce the processing time. In addition to the robust infrastructure supporting our platform, it is also critical that we align with key industry partners in the digital supply chain. The Colossus SSP is agnostic to any specific demand side platform.

We automate workflow processes whenever feasible to drive predictable and value-added outcomes for our customers and increase productivity of our organization. In the first half of 2023, we transitioned our server platform to HPE Greenlake, which provides increased capacity, faster response time, and expansion capabilities to align with growth in our business.

#### Managing industry dynamics

We operate in the rapidly evolving digital advertising industry. Due to the scale and complexity of the digital advertising ecosystem, direct sales via manual, person-toperson processes are insufficient for delivering a real-time, personalized ad experience, creating the need for programmatic advertising. In turn, advances in programmatic technologies have enabled publishers to auction their ad inventory to more buyers, simultaneously, and in real time through a process referred to as header bidding. Header bidding has also provided advertisers with transparent access to ad impressions. As advertisers keep pace with ongoing changes in the way that consumers view and interact with digital media we anticipate further innovation and expect that header bidding will be extended into new areas such as OTT/CTV. We believe our focus on publishers and buyers has allowed us to understand their needs and our ongoing innovation has enabled us to quickly adapt to changes in the industry, develop new solutions and do so cost effectively. Our performance depends on our ability to keep pace with industry changes such as header bidding and the evolving needs of our publishers and buyers while continuing our cost efficiency.

#### Seasonality

In the advertising industry, companies commonly experience seasonal fluctuations in revenue. For example, in our sell-side advertising segment, many advertisers allocate the largest portion of their budgets to the fourth quarter of the calendar year in order to coincide with increased holiday purchasing. We expect our sell-side revenue to continue to fluctuate based on seasonal factors that affect the advertising industry as a whole.

#### **Buy-side advertising business**

#### New Customer Acquisitions

On the buy-side of our business, our customers consist of purchasers of programmatic advertising inventory (ad space) looking to place their advertisements. We serve the needs of over 230 small and mid-sized clients, consisting of advertising space buyers, including small and mid-sized companies, large advertising holding companies (which may manage several agencies), independent advertising agencies and mid-market advertising service organizations. We serve a variety of customers across multiple industries including travel/tourism (including destination marketing organizations ("DMOs")), education, energy, consumer packaged goods, healthcare, financial services (including cryptocurrency technologies) and other industries.

We are focused on increasing the number of customers that use our buy-side advertising businesses as their advertising partner. Our long-term growth and results of operations will depend on our ability to attract more customers, including DMOs, across multiple geographies.

#### **Expand Sales to Existing Customers**

Our customers understand the independent nature of our platform and relentless focus on driving results based on return on investment ("ROI"). Our value proposition is complete alignment across our entire digital supply platform beginning with the first dollar in and last dollar out. We are technology, DSP and media agnostic, and we believe our clients trust us to provide the best opportunity for success of their brands and businesses. As a result, our clients have been loyal, with approximately 88% client retention amongst the clients that represent approximately 80% of our revenue during the nine months ended September 30, 2024. In addition, we cultivate client relationships through our platform. The managed services delivery model allows us to combine our technology with a highly personalized offering to strategically design and manage advertising campaigns.

#### Shift to Digital Advertising

Media has increasingly become more digital as a result of three key ongoing developments:

- Advances in technology with more sophisticated digital content delivery across multiple platforms;
- · Changes in consumer behavior, including spending longer portions of the day using mobile and other devices; and
- Better audience segmentation with more efficient targeting and measurable results.

The resulting shift has enabled a variety of options for advertisers to efficiently target and measure their advertising campaigns across nearly every media channel and device. These efforts have been led by big-budgeted, large, multi-national corporations incentivized to cast a broad advertising net to support national brands.

#### Increased Adoption of Digital Advertising by Small-and Mid-Sized Companies

Only recently have small and mid-sized businesses begun to leverage the power of digital media in meaningful ways, as emerging technologies have enabled advertising across multiple channels in a highly localized nature. Campaign efficiencies yielding measurable results and higher advertising ROI have prompted these companies to begin utilizing digital advertising on an accelerated pace. We believe this market is rapidly expanding, and that small-to-mid-sized advertisers will continue to increase their digital spend.

#### Seasonality

In the advertising industry, companies commonly experience seasonal fluctuations in revenue. Historically, for our buy-side advertising segment, the second and third quarters of the year reflect our highest levels of advertising activity and



the first quarter reflects the lowest level of such activity. We expect our buy-side revenue to continue to fluctuate based on seasonal factors that affect the advertising industry as a whole.

#### **Components of Our Results of Operations**

#### Revenue

For the sell-side advertising segment, we generate revenue by selling advertising inventory (digital ad units) that we purchase from publishers to advertisers through a process of monetizing ad impressions on our proprietary sell-side programmatic platform operating under the trademarked banner Colossus SSP. For the buy-side advertising segment, we generate revenue from customers that enter into agreements with us to provide managed advertising campaigns, which include digital marketing and media services to purchase digital advertising space, data and other add-on features.

In connection with our analysis of principal vs agent considerations, we have evaluated the specified goods or services and we considered whether we control the goods or services before they are provided to the customer including the three indicators of control. Based upon this analysis and our specific facts and circumstances, we concluded that we are a principal for the goods or services sold through both our sell-side advertising segment and our buy-side segment because we control the specified good or service before it is transferred to the customer and we are the primary obligor in the agreement with the customer. Therefore, we report revenue on a gross basis inclusive of all supplier costs. We pay suppliers for the cost of digital media, advertising inventory, data and any add-on services or features.

Our revenue recognition policies are discussed in more detail under "-Critical Accounting Estimates and Related Policies" set forth in our Annual Report on Form 10-K for the year ended December 31, 2023.

#### Cost of revenues

For cost of revenues for our sell-side advertising segment, we pay publishers a fee, which is typically a percentage of the value of the ad impressions monetized through our platform. Cost of revenues consists primarily of publisher media fees and data center co-location costs. Media fees include the publishing and real time bidding costs to secure advertising space. For the buy-side advertising segment, cost of revenues consists primarily of digital media fees, third-party platform access fees, and other third-party fees associated with providing services to our customers.

#### **Operating expenses**

Operating expenses consist of compensation expenses related to our executive, sales, finance and administrative personnel (including salaries, commissions, stock-based compensation, bonuses, benefits and taxes); general and administrative expenses (including rent expense, professional fees, independent contractor costs, selling and marketing fees, administrative and operating system subscription costs, insurance, amortization expense related to our intangible assets); and other expenses (including transactions that are unusual in nature or which are occurring infrequently).

#### Other income (expense), net

Other income. Other income includes income associated with recovery of receivables and other miscellaneous credit card rebates.

Interest expense. Interest expense is mainly related to our debt as further described below in "-Liquidity and Capital Resources."

Loss on early termination of line of credit. In January 2023, we entered into a Loan and Security Agreement (the "Loan Agreement"), by and among Silicon Valley Bank ("SVB"), which provided for a revolving credit facility (the "Credit Facility"). In March 2023, we issued a notice of termination and recognized a loss on the write-off of the deferred financing fees.

Derecognition of tax receivable agreement liability. The Company derecognized its tax receivable agreement liability in connection with the full valuation allowance recorded on the Company's deferred tax assets.

#### **Results of Operations**

#### Comparison of the Three and Nine Months Ended September 30, 2024 and 2023

The following tables set forth our condensed consolidated results of operations for the periods presented (in thousands). The period-to-period comparison of results is not necessarily indicative of results for future periods.

	Thre	e Months En	ded S	eptember 30,	Change	e	N	ine Months End	led S	eptember 30,	oer 30, Change		
		2024		2023	 Amount	%		2024		2023		Amount	%
Revenues					 								
Sell-side advertising	\$	2,202	\$	51,622	\$ (49,420)	(96)%	\$	33,001	\$	89,006	\$	(56,005)	(63)%
Buy-side advertising		6,873		7,850	 (977)	(12)%		20,204		27,093		(6,889)	(25)%
Total revenues		9,075		59,472	(50,397)	(85)%		53,205		116,099		(62,894)	(54)%
					 _								
Cost of revenues													
Sell-side advertising		2,654		44,606	(41,952)	(94)%		30,670		77,190		(46,520)	(60)%
Buy-side advertising		2,907		3,113	(206)	(7)%		8,091		10,650		(2,559)	(24)%
Total cost of revenues		5,561		47,719	 (42,158)	(88)%		38,761		87,840		(49,079)	(56)%
Gross profit		3,514		11,753	(8,239)	(70)%		14,444		28,259		(13,815)	(49)%
Operating expenses		7,172		7,259	(87)	(1)%		22,973		21,652		1,321	6 %
(Loss) income from operations		(3,658)		4,494	 (8,152)	(181)%		(8,529)		6,607		(15,136)	(229)%
Other income (expense), net		3,887		(977)	4,864	(498)%		1,323		(3,229)		4,552	(141)%
Income (loss) before income taxes		229		3,517	(3,288)	(93)%		(7,206)		3,378		(10,584)	(313)%
Income tax expense		6,606		166	6,440	3880 %		6,132		166		5,966	3594 %
Net (loss) income	\$	(6,377)	\$	3,351	\$ (9,728)	(290)%	\$	(13,338)	\$	3,212	\$	(16,550)	(515)%
Adjusted EBITDA <sup>(1)</sup>	\$	(2,855)	\$	5,371	\$ (8,226)	(153)%	\$	(5,858)	\$	8,978	\$	(14,836)	(165)%

(1) For a definition of Adjusted EBITDA, an explanation of our management's use of this measure, and a reconciliation of Adjusted EBITDA to net income see "- Non-GAAP Financial Measures."

#### Revenues

Our revenues of \$9.1 million for the three months ended September 30, 2024 decreased by \$50.4 million, or 85%, from \$59.5 million for the three months ended September 30, 2023. Sell-side advertising revenue decreased \$49.4 million, or 96%, primarily due to a decrease in impression inventory. This decrease was primarily caused by one of the Company's sell-side customers pausing its connection to the Company during the second quarter while it investigated allegations made against the Company in a defamatory article / blog post which the Company believes was part of a coordinated misinformation campaign. This customer reconnected the Company on May 22, 2024 and sell-side volumes have resumed but not yet at the levels experienced prior to the pause in May 2024, which affected the entirety of the quarter ended September 30, 2024. The Company sold approximately 0.2 billion average monthly impressions over the three months ended September 30, 2024, a decrease of 97% from the prior period. Buy-side revenue decrease from completion of certain one-time campaigns in 2023 partially offset by growth from existing and new customers.

Our revenues of \$53.2 million for the nine months ended September 30, 2024 decreased by \$62.9 million, or 54%, from \$116.1 million for the nine months ended September 30, 2023. Sell-side advertising revenue decreased \$56.0 million, or 63%, primarily due to a decrease in impression inventory, resulting from the customer suspension that occurred during May 2024, which negatively affected revenue in the second and third quarters of 2024. The Company sold approximately 1.4 billion average monthly impressions over the nine months ended September 30, 2024, a decrease of 64% from the prior period. Buy-side revenue decreased \$6.9 million, or 25%, over the nine months ended September 30, 2023 due to a \$4.5



million decrease in spending from our existing customer base and a \$3.6 million decrease from completion of certain one-time campaigns in 2023 partially offset by growth from existing and new customers.

#### Cost of revenues

Consistent with the overall decrease in revenues, cost of revenues of \$5.6 million for the three months ended September 30, 2024 decreased by \$42.2 million, or 88%, from \$47.7 million for the three months ended September 30, 2023. Sell-side advertising cost of revenues decreased \$42.0 million, to \$2.7 million, or 121% of revenue for the three months ended September 30, 2024, compared to \$44.6 million, or 86% of revenue, for the same period in 2023. The decrease in costs was primarily due to the related decrease in revenue, while the increase as a percentage of revenue was due to fixed costs remaining at a level consistent with the prior year. Buy-side advertising cost of revenues decreased \$0.2 million, or 40% of revenue, for the same period in 2023.

Consistent with the overall decrease in revenues, cost of revenues of \$38.8 million for the nine months ended September 30, 2024 decreased by \$49.1 million, or 56% from \$87.8 million for the nine months ended September 30, 2023. Sell-side advertising cost of revenues decreased \$46.5 million, to \$30.7 million, or 93% of revenue for the nine months ended September 30, 2024, compared to \$77.2 million, or 87% of revenue, for the same period in 2023. The decrease in costs was primarily due to the related decrease in revenue, while the 6% increase as a percentage of revenue was due to an increase in fixed costs of approximately \$0.5 million related to an increase in server capacity and approximately \$1.1 million related to new analytic and technology-related costs to support the growth as well as the mix and concentration of publishers and the related costs. Buy-side advertising cost of revenues decreased \$2.6 million, to \$8.1 million, or 40% of revenue for the nine months ended September 30, 2024, compared to \$10.7 million, or 39% of revenue for the nine months ended September 30, 2024.

#### Gross profit

Gross profit was \$3.5 million, or 39% of revenue, for the three months ended September 30, 2024, compared to \$11.8 million, or 20% of revenue, for the same period in 2023, reflecting a decrease of \$8.2 million, or 70%. The change in margin for the three months ended September 30, 2024 is attributable to the mix in revenue between our business segments as our sell-side segment has higher cost of revenues compared to our buy-side segment, as well as the consistent level of fixed costs not covered by current period sell-side revenue.

Sell-side advertising gross profit decreased \$7.5 million for the three months ended September 30, 2024 as compared to the same period in 2023, primarily due to a consistent level of fixed costs not covered by current period sell-side revenue. Sell-side advertising gross margin was (21)% and 14% for the three months ended September 30, 2024 and 2023, respectively. Buy-side advertising gross profit decreased \$0.8 million for the three months ended September 30, 2024, as compared to the same period in the prior year, primarily due to the decrease in revenue. Buy-side advertising gross margin was 58% and 60% for the three months ended September 30, 2024 and 2023, respectively.

Gross profit was \$14.4 million, or 27% of revenue, for the nine months ended September 30, 2024, compared to \$28.3 million, or 24% of revenue, for the same period in 2023, reflecting a decrease of \$13.8 million or 49%. The change in margin for the nine months ended September 30, 2024 is attributable to the mix in revenue between our business segments as our sell-side segment has higher cost of revenues compared to our buy-side segment, as well as the additional sell-side fixed costs related to an increase in server capacity and new analytic and technology-related costs.

Sell-side advertising gross profit decreased \$9.5 million for the nine months ended September 30, 2024 as compared to the same period in 2023, primarily due to the increase in fixed costs of approximately \$1.6 million related to our servers and analytic and technology-related costs and the decrease in revenues. Sell-side advertising gross margin was 7% and 13% for the nine months ended September 30, 2024 and 2023, respectively. Buy-side advertising gross profit decreased \$4.3 million for the nine months ended September 30, 2024 and 2023, respectively. Buy-side advertising gross margin was 60% and 61% for the nine months ended September 30, 2024 and 2023, respectively.

#### Operating expenses

The following table sets forth the components of operating expenses for the periods presented (in thousands):

	Three Months Ended September 30,			Cha	ange	1	Nine Months Ended September 30,				Change		
	2024		2023	Amount	%		2024		2023		Amount	%	
Compensation, tax and benefits	\$ 3,526	\$	4,747	\$ (1,221)	(26)%	\$	12,216	\$	12,934	\$	(718)	(6)%	
General and administrative	3,646		2,512	1,134	45 %		10,757		8,718		2,039	23 %	
Total operating expenses	\$ 7,172	\$	7,259	\$ (87)	(1)%	\$	22,973	\$	21,652	\$	1,321	6 %	

#### Compensation, taxes and benefits

Compensation, taxes and benefits of \$3.5 million, decreased by \$1.2 million, or 26%, for the three months ended September 30, 2024 from \$4.7 million for the same period in 2023. The decrease is due primarily to a decrease in bonus and commissions expense related to the decrease in revenue. Sequentially, compensation, taxes and benefits decreased by \$0.6 million, or 15%, from \$4.2 million for the three months ended June 30, 2024 primarily due to lower payroll costs resulting from headcount reductions made effective July 1, 2024.

Compensation, taxes and benefits of \$12.2 million, decreased by \$0.7 million, or 6%, for the nine months ended September 30, 2024 from \$12.9 million for the same period in 2023. The decrease is primarily due to a decrease in bonus and commission expense related to the decrease in revenue.

On July 1, 2024, we executed an internal reorganization plan that included a staff reduction, a pause on hiring and cost savings measures, which we anticipate will lower certain ongoing expenses, especially as the Company incurs additional one-time expenses to regain compliance with respect to delinquent SEC filings.

#### General and administrative expenses

General and administrative ("G&A") expenses of \$3.6 million for the three months ended September 30, 2024 increased from \$2.5 million for the same period in 2023. G&A expenses as a percentage of revenue was 40% and 4% for the three months ended September 30, 2024 and 2023, respectively. During the three months ended September 30, 2024, we incurred \$1.1 million in costs to regain compliance with respect to delinquent SEC filings as well as \$0.3 million higher legal expenses compared to the prior year. Sequentially, G&A expenses decreased by \$0.2 million, or 5%, from \$3.8 million for the three months ended June 30, 2024 primarily due to lower sales and marketing expenses, travel and franchise tax expense, partially offset by \$0.9 million of higher compliance and legal costs.

General and administrative ("G&A") expenses of \$10.8 million for the nine months ended September 30, 2024 increased from \$8.7 million for the same period in 2023. G&A expenses as a percentage of revenue was 20% and 8% for the nine months ended September 30, 2024 and 2023, respectively. During the nine months ended September 30, 2024, we incurred \$1.3 million in costs to regain compliance with respect to delinquent SEC filings, \$0.4 million higher legal expenses and \$0.4 million higher franchise tax expense.

We expect to continue to invest in and incur additional expenses associated with our operation as a public company, including increased professional fees, investment in automation, and compliance costs associated with developing the requisite infrastructure required for internal controls. However, on July 1, 2024, we executed an internal reorganization plan that included a staff reduction, a pause on hiring and cost savings measures, which we anticipate will lower certain ongoing expenses, especially as the Company ceases incurring additional one-time expenses to regain compliance with respect to delinquent SEC filings, which have now all been filed.

#### Other income (expense), net

The following table sets forth the components of other income (expense), net for the periods presented (in thousands).

	Three Months Ended September 30,				Change		Nine Months Ended September 30,				Change		
	 2024		2023	_	Amount	%		2024		2023	_	Amount	%
Interest expense	\$ (1,413)	\$	(1,060)	\$	(353)	33 %	\$	(4,068)	\$	(3,104)	\$	(964)	31 %
Other income	99		83		16	19 %		190		175		15	9 %
Loss on early termination of line of credit	—		—			nm				(300)		300	nm
Derecognition of tax receivable agreement liability	5,201		—		5,201	nm		5,201		—		5,201	nm
Total other income (expense), net	\$ 3,887	\$	(977)	\$	4,864	nm	\$	1,323	\$	(3,229)	\$	4,552	nm

#### nm – not meaningful

Other income (expense), net primarily includes \$5.2 million relating to the derecognition of tax receivable agreement liability in connection with the full valuation allowance recorded on the Company's deferred tax assets during the three months ended September 30, 2024. Other income (expense), net for the three months ended September 30, 2024 and 2023 also consists of \$1.4 million and \$1.1 million, respectively, of interest expense. Interest expense increased by \$0.4 million due to additional net borrowings of \$12.7 million under the Company's credit facilities, as well as higher interest rates.

Other income (expense), net for the nine months ended September 30, 2024 primarily includes \$5.2 million relating to the derecognition of tax receivable agreement liability in connection with the full valuation allowance recorded on the Company's deferred tax assets. Other income (expense), net also consists of \$4.1 million and \$3.1 million, respectively, of interest expense. Interest expense increased by \$1.0 million due to additional net borrowings of \$12.7 million under the Company's credit facilities, as well as higher interest rates. Lastly, other income (expense), net for the nine months ended September 30, 2023 also includes \$0.3 million related to the loss on early termination of the line of credit with SVB.

#### Liquidity and Capital Resources

#### Going Concern

As discussed in Note 9 to the condensed consolidated financial statements, on May 10, 2024, the Company was the subject of a defamatory article / blog post which the Company believes was part of a coordinated misinformation campaign. In connection with this post, one of the Company's sell-side customers paused its connection to the Company for a couple of weeks in May 2024, which reduced sell-side sales volumes. As of the date of this report, sell-side volumes related to this customer have resumed but not yet at the levels experienced prior to the pause in May 2024 which has created significant disruption in the Company's sell-side business. The Company is actively working with its partners to achieve prior volume levels. However, there can be no assurance that the Company will be able to achieve prior volume levels with its partners or on the timing of achieving such volume levels. Additionally, the Company (1) incurred a net loss of \$13.3 million for the nine months ended September 30, 2024 reflecting the impact of the sell-side disruption described above and a decrease in customer spend on the buy-side, (2) reported an accumulated deficit of \$6.6 million as of September 30, 2024, (4) has borrowed \$9.7 million and \$8.7 million, respectively, as of September 30, 2024 and the date of this report, under the Credit Agreement which matures in July 2025, (5) was notified on April 17, 2024 that the Company's auditor had resigned, (6) was unable to timely file its 2023 annual report and quarterly reports for the first two quarters of 2024 and (7) was notified by Nasdaq on October 18, 2024 that it was not in compliance with Nasdaq's minimum stockholders' equity requirement. The delay in filing the Company's annual and quarterly reports disrupted existing capital-raising efforts and created additional audit, legal and other expenses. These factors raise substantial doubt about the Company's ability to continue as a going concern over the next twelve months.

The Company anticipates sources of liquidity to include cash on hand and cash flow from operations and has taken several actions to address liquidity concerns. These actions include (1) a plan to reduce expenses through a staff reduction, a pause on hiring and cost savings measures that were executed on July 1, 2024, (2) working with lenders to provide temporary various relief from debt covenants (see Note 3 — Long-Term Debt in the condensed consolidated financial statements) while rebuilding sell-side volumes, (3) putting in place a program to raise capital through an equity reserve facility (see Note 4 - Stockholders' Deficit and Stock-Based Compensation in the condensed consolidated financial statements), (4) regaining compliance with respect to delinquent SEC filings on October 15, 2024 which will allow the Company to access the capital markets as well as other financing sources and (5) a plan to achieve compliance with

Nasdaq's minimum stockholders' equity requirement. There can be no assurance that the Company's actions will be successful or that additional financing will be available when needed or on acceptable terms.

#### Sources of Liquidity

The following table summarizes our cash and cash equivalents, working capital, and availability under our Credit Agreement (as defined below) on September 30, 2024 and December 31, 2023 (in thousands):

	September 30, 2024	December 31, 2023
Cash and cash equivalents	\$ 4,087	\$ 5,116
Working capital <sup>(1)</sup>	\$ (33,813)	\$ 3,280
Availability under Credit Agreement	\$ 300	\$ 7,000

(1) Working capital as of September 30, 2024 includes all amounts owed to Lafayette Square, which is shown as current portion of long term debt until a waiver or amendment is finalized with Lafayette Square. See Note 3 — Long-Term Debt in the notes to the condensed consolidated financial statements.

To fund our operations and service our debt thereafter and depending on our growth and results of operations, we may raise additional capital through the issuance of additional equity and/or debt, which could have the effect of diluting our stockholders. Any future equity or debt financings may be on terms which are not favorable to us. As our credit facilities become due, we will need to repay, extend or replace such indebtedness. Our ability to do so will be subject to future economic, financial, business and other factors, many of which are beyond our control.

#### **Credit Facilities**

The terms and conditions of the various credit facilities we entered into are further described in Note 3 — Long-Term Debt in the notes to the condensed consolidated financial statements.

#### **Equity Reserve Facility**

The terms and conditions of the equity reserve facility we entered into with New Circle are further described in Note 4 — Stockholders' Deficit and Stock-Based Compensation in the notes to the condensed consolidated financial statements.

#### **Historical Cash Flows:**

The following table sets forth our cash flows for the nine months ended September 30, 2024 and 2023 (in thousands):

	Nine Months Ended September 30,				
	 2024	2023			
Net cash (used in) provided by operating activities	\$ (7,095)	\$ 4	4,481		
Net cash used in investing activities	(17)		(137)		
Net cash provided by (used in) financing activities	6,083	(2	2,909)		
Net (decrease) increase in cash and cash equivalents	\$ (1,029)	\$ 1	1,435		

Our cash and cash equivalents at September 30, 2024 were held for working capital and general corporate purposes. The decrease in cash and cash equivalents compared with September 30, 2023, primarily resulted from \$7.1 million in cash flows used in operating activities partially offset by \$6.1 million in cash flows provided by financing activities.

# **Operating Activities**

#### For the Nine Months Ended September 30, 2024 and 2023

Cash provided by operating activities has typically been generated from net income and by changes in our operating assets and liabilities, particularly in the areas of accounts receivable and accounts payable and accrued expenses, adjusted for certain non-cash and non-operating expense items such as depreciation, amortization, stock-based compensation and deferred income taxes.

For the nine months ended September 30, 2024, net cash flows used in operating activities were \$7.1 million and consisted of net loss of \$13.3 million, offset by \$4.1 million in adjustments for non-cash and non-operating items and

\$2.1 million of cash inflows from working capital. Adjustments for non-cash and non-operating items mainly consisted of depreciation and amortization expense of \$2.3 million, stock-based compensation expense of \$0.8 million and deferred tax expense of \$6.1 million partially offset by \$5.2 million of derecognition of tax receivable agreement liability in connection with the full valuation allowance recorded on the Company's deferred tax assets. The \$2.1 million increase in cash resulting from changes in working capital primarily consisted of a \$30.9 million decrease in accounts receivable, partially offset by a \$27.5 million decrease in accounts payable and a \$1.5 million decrease in accounts payable and a san a

For the nine months ended September 30, 2023, net cash flows provided by operating activities were \$4.5 million and mainly consisted of net income of \$3.2 million and \$3.2 million in adjustments for noncash and non-operating items, partially offset by \$2.0 million of cash outflows from working capital. Adjustments for non-cash and non-operating items primarily consisted of depreciation and amortization expense of \$2.2 million, stock-based compensation expense of \$0.5 million and loss on early termination of line of credit of \$0.3 million. The \$2.0 million decrease in cash resulting from changes in working capital consisted primarily of a \$28.4 million increase in accounts receivable and a \$0.8 million decrease in accounts payable is mainly due to the seasonal nature of the sell-side segment of the business.

#### **Investing Activities**

#### For the Nine Months Ended September 30, 2024 and 2023

Our investing activities to date have consisted primarily of purchases of software, office furniture and leasehold improvements. For the nine months ended September 30, 2024 and 2023, net cash flows used in investing activities of less than \$0.1 million and \$0.1 million, respectively, were primarily related to office furniture and leasehold improvements.

#### **Financing Activities**

#### For the Nine Months Ended September 30, 2024 and 2023

For the nine months ended September 30, 2024, net cash provided by financing activities was \$6.1 million mainly resulting from \$6.7 million of proceeds from line of credit and \$0.2 million proceeds from warrants exercised partially offset by \$0.6 million for payments on shares withheld for taxes and \$0.4 million paid on term loan.

For the nine months ended September 30, 2023, net cash used in financing activities was \$2.9 million mainly resulting from \$2.0 million for distributions paid to LLC holders, \$0.5 million paid on the term loan and \$0.4 million for deferred financing costs.

#### **Contractual Obligations and Future Cash Requirements**

As of September 30, 2024, our principal contractual obligations expected to give rise to material cash requirements consist of the 2021 Credit Facility, the Credit Agreement and non-cancelable leases for our various facilities. After giving effect to the October 2024 amendments to the Company's debt facilities, we anticipate that the future minimum payments related to our current indebtedness over the next five years will be \$1.0 million in 2024, \$8.7 million in 2025, \$28.2 million in 2026, less than \$0.1 million in 2027, less than \$0.1 million in 2028, and \$0.1 million thereafter, assuming we do not refinance our indebtedness, enter into a new revolving credit facility or make any further draws under the revolving facility. The leases will require minimum payments of \$0.1 million in 2024, \$0.3 million in 2025, \$0.3 million in 2026, \$0.3 million in 2027, \$0.2 million in 2028, and \$0.2 million thereafter. As of September 30, 2024, we had cash and cash equivalents of \$4.1 million.

#### **Non-GAAP Financial Measures**

In addition to our results determined in accordance with U.S. generally accepted accounting principles ("GAAP"), including, in particular operating income, net cash provided by operating activities, and net income, we believe that earnings before interest, taxes, depreciation and amortization, as adjusted for loss on early termination of line of credit, derecognition of tax receivable agreement liability and stock-based compensation ("Adjusted EBITDA"), a non-GAAP measure, is useful in evaluating our operating performance. The most directly comparable GAAP measure to Adjusted EBITDA is net income.



The following table presents a reconciliation of Adjusted EBITDA to net (loss) income for each of the periods presented (in thousands):

Three Months Ended September 30,					Nine Months Ended September 30,			
	2024		2023	2024	1		2023	
\$	(6,377)	\$	3,351	\$	(13,338)	\$	3,212	
	1,413		1,060		4,068		3,104	
	488		488		1,465		1,465	
	149		242		811		546	
	67		64		205		185	
			—		—		300	
	6,606		166		6,132		166	
	(5,201)		_		(5,201)		_	
\$	(2,855)	\$	5,371	\$	(5,858)	\$	8,978	
	\$ <u>\$</u>	Septem 2024 \$ (6,377) 1,413 488 149 67  6,606 (5,201)	September 30,           2024            \$         (6,377)         \$           1,413         488         149           67         -         6,606           (5,201)         (5,201)         -	September 30,           2024         2023           \$         (6,377)         \$         3,351           \$         (6,377)         \$         3,351           1,413         1,060         488         488           149         242         67         64             6,606         166           (5,201)	September 30,           2024         2023         2024         202	September 30,         Septem           2024         2023         2024           \$ (6,377)         \$ 3,351         \$ (13,338)           1,413         1,060         4,068           488         488         1,465           149         242         811           67         64         205                6,606         166         6,132           (5,201)          (5,201)	$\begin{tabular}{ c c c c c c } \hline September 30, & September 30, & \hline & $2024$ & $2024$ & $2024$ & $2024$ & $2024$ & $$2024$ & $$2024$ & $$2024$ & $$2024$ & $$2024$ & $$$11$ & $$$1,413$ & $$1,060$ & $$4,068$ & $$488$ & $$488$ & $$1,465$ & $$$149$ & $$242$ & $$811$ & $$$67$ & $$64$ & $$205$ & $$$$ & $$$ & $$$$ & $$$ & $$$$$ & $$$$ & $$$$ & $$$$$ & $$$$$ & $$$$$ & $$$$$ & $$$$$$ & $$$$$$ & $$$$$$$ & $$$$$$$$ & $$$$$$$$$	

In addition to operating income and net income, we use Adjusted EBITDA as a measure of operational efficiency. We believe that this non-GAAP financial measure is useful to investors for period-to-period comparisons of our business and in understanding and evaluating our operating results for the following reasons:

- Adjusted EBITDA is widely used by investors and securities analysts to measure a company's operating performance without regard to items such as depreciation and
  amortization, interest expense, provision for income taxes, stock-based compensation, derecognition of tax receivable agreement liability, and certain one-time items
  such as acquisition transaction costs, losses from early termination or redemption of credit agreements or preferred units and gains from settlements or loan
  forgiveness that can vary substantially from company to company depending upon their financing, capital structures and the method by which assets were acquired;
- Our management uses Adjusted EBITDA in conjunction with GAAP financial measures for planning purposes, including the preparation of our annual operating budget, as a measure of operating performance and the effectiveness of our business strategies and in communications with our board of directors concerning our financial performance; and
- Adjusted EBITDA provides consistency and comparability with our past financial performance, facilitates period-to-period comparisons of operations, and also
  facilitates comparisons with other peer companies, many of which use similar non-GAAP financial measures to supplement their GAAP results.

Our use of this non-GAAP financial measure has limitations as an analytical tool, and you should not consider it in isolation or as a substitute for analysis of our financial results as reported under GAAP.

#### **Critical Accounting Estimates and Related Policies**

The preparation of financial statements in conformity with U.S. GAAP requires management to make estimates and assumptions that affect the reported amounts of assets, liabilities and disclosure of contingent liabilities at the date of the financial statements and the reported amounts of revenue and expenses during the reporting period. Actual results could differ from these estimates. The Company bases its estimates on past experiences, market conditions, and other assumptions that the Company believes are reasonable under the circumstances, and the Company evaluates these estimates on an ongoing basis.

There have been no material changes to our critical accounting estimates and related policies as compared to the critical accounting estimates and related policies described in our "Management's Discussion and Analysis of Financial Condition and Results of Operations" set forth in our Annual Report on Form 10-K for the year ended December 31, 2023.

#### **Recent Accounting Pronouncements**

See Note 2 to our condensed consolidated financial statements for accounting pronouncements recently adopted and accounting pronouncements not yet adopted.



#### ITEM 3. Quantitative and Qualitative Disclosures About Market Risk

As a "smaller reporting company," we are not required to provide the information required by this Part I, Item 3.

#### **ITEM 4.** Controls and Procedures

#### **Evaluation of Disclosure Controls and Procedures**

The Company maintains disclosure controls and procedures ("Disclosure Controls") as such term is defined in Rules 13a-15(e) and 15d-15(e) under the Exchange Act that are designed to ensure that information required to be disclosed in our Exchange Act reports is recorded, processed, summarized and reported within the time periods specified in the SEC rules and forms, and that such information is accumulated and communicated to management, including the Chief Executive Officer ("CEO") and Chief Financial Officer ("CFO") as appropriate to allow timely decisions regarding required disclosure. The Company conducted an evaluation (the "Evaluation"), under the supervision and with the participation of the CEO and CFO, of the effectiveness of the design and operation of our Disclosure Controls as of September 30, 2024 pursuant to the Rules 13a-5(b) and 15d-15(b) of the Exchange Act. In designing and evaluating the Disclosure Controls, management recognizes that any controls and procedures, no matter how well designed and operated, can provide only reasonable assurance of achieving the desired control objectives and management was required to apply judgement in evaluating its controls and procedures. Based on this Evaluation, due to the material weaknesses described below, the CEO and CFO concluded that the Company's Disclosure Controls were not effective as of September 30, 2024.

#### Material Weaknesses

We identified previously material weaknesses in our controls over the journal entry processes, information technology general controls ("ITGC") and the technical evaluation of accounting matters that existed as of December 31, 2023. A material weakness is a deficiency, or a combination of deficiencies, in internal controls over financial reporting, such that there is a reasonable possibility that a material misstatement of our annual or interim financial statements will not be prevented or detected on a timely basis. The material weaknesses are a result of our processes and related controls not operating effectively related to journal entry processes, ITGC and the technical evaluation of accounting matters. As further detailed in Note 13 – Restatement (Unaudited) of the Company's consolidated financial statements included in our Annual Report on Form 10-K for the fiscal year ended December 31, 2023, the Company identified prior year accounting for and presentation of noncontrolling interests ("NCI"), (2) recognition of accounting for, and presentation of NCI, and (4) timing of the recording of the 2023 redemption of warrants. The Company's management and the audit committee of the Company's Board of Directors concluded that it was appropriate to restate the quarterly unaudited consolidated financial statements for the quarterly periods ended March 31, 2023, June 30, 2023, and September 30, 2023.

Other than the described above, there were no material misstatements as a result of these material weaknesses; however, it could have resulted in a material misstatement to the annual or interim financial statements that would not have been prevented or detected on a timely basis.

#### Management's Plan to Remediate the Previously Reported Material Weaknesses

Management has implemented remediation steps to address the material weaknesses and to improve our internal controls. Specifically, in late 2023, the Company engaged consultants to assist with identifying and testing the design of controls over business processes as well as ITGC. The first phase of the project was completed in the first quarter of 2024. We are now in the process of enhancing the design of certain internal control procedures and implementing new internal controls over (1) the segregation of duties within the journal entry process, (2) the access to program and change management within our information technology environment, and (3) the evaluation of technical accounting matters. These controls are planned to be tested for design and operating effectiveness in future periods. The Company will continue the engagement with outside consultants to review the revised control processes and procedures.

While the Company has implemented remediation steps, the material weaknesses cannot be considered fully remediated until the improved controls have been in place and operate for a sufficient period of time. However, our management, including our Chief Executive Officer and Chief Financial Officer, has concluded that, notwithstanding the identified material weaknesses in our internal controls over financial reporting, the financial statements fairly present, in all material respects, our financial condition, results of operations and cash flows for the periods presented in conformity with U.S. GAAP.



#### Changes in Internal Controls Over Financial Reporting

Other than as discussed above, there were no changes in internal controls over financial reporting during the three months ended September 30, 2024 that have materially affected, or are reasonably likely to materially affect, the Company's internal controls system over financial reporting.

#### PART II. Other Information

#### ITEM 1. Legal Proceedings

We may from time to time be subject to various legal or administrative claims and proceedings arising in the ordinary course of business. As of the date hereof, except as set forth below, we are not a party to any material legal or administrative proceedings nor are there any proceedings in which any of our directors, executive officers or affiliates, or any registered or beneficial stockholder, is an adverse party or has a material interest adverse to our interest. Litigation or any other legal or administrative proceeding, regardless of the outcome, is likely to result in substantial cost and diversion of our resources, including our management's time and attention.

On May 10, 2024, the Company was the subject of a defamatory article / blog post which the Company believes was part of a coordinated misinformation campaign. In connection with this post, one of the Company's sell-side customers paused its connection to the Company while the allegations were investigated. This customer reconnected the Company on May 22, 2024 and sell-side volumes have resumed but not yet at the levels experienced prior to the pause in May 2024. The Company is actively working with its partners to achieve prior volume levels. On May 14, 2024, the Company filed a lawsuit against the author of the defamatory article and is vigorously pursuing its rights. The Company cannot make any predictions about the final outcome of this litigation matter or the timing thereof.

On May 23, 2024, an alleged stockholder, purportedly on behalf of the persons or entities who purchased or acquired publicly traded securities of the Company between April 2023 and March 2024, filed a putative class action against the Company, certain of our officers and directors, and other defendants in the U.S. District Court for the Southern District of Texas, alleging violations of federal securities laws related to alleged false or misleading disclosures made by the Company in its public filings. On July 9, 2024, another alleged stockholder filed a similar securities class action against the Company, certain of our officers and directors, also in the Southern District of Texas. The two actions have been consolidated. Each of these complaints seeks unspecified damages, plus costs, fees, and attorneys' fees. The Company cannot make any predictions about the final outcome of this matter or the timing thereof.

#### ITEM 1A. Risk Factors

As of the date of this Quarterly Report, other than the below, there have not been any material changes to the information related to the ITEM 1A. "Risk Factors" disclosure in the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 2023. Our business involves significant risks. You should carefully consider the risks and uncertainties described in our Annual Report, together with all of the other information in this Quarterly Report on Form 10-Q, as well as our audited consolidated financial statements and related notes as disclosed in our Annual Report. The risks and uncertainties described below and in our Annual Report are not the only ones we face. Additional risks and uncertainties that we are unaware of or that we deem immaterial may also become important factors that adversely affect our business. The realization of any of these risks and uncertainties could have a material adverse effect on our reputation, business, financial condition, results of operations, growth and future prospects as well as our ability to accomplish our strategic objectives. In that event, the market price of our common stock could decline and you could lose part or all of your investment.

# If we fail to satisfy applicable listing standards, including compliance with the rules requiring that our stockholders' equity be at least \$2.5 million, our common stock may be delisted from the Nasdaq Capital Market.

On October 18, 2024, we received the Letter from the Staff of Nasdaq notifying the Company that it was not in compliance with the minimum stockholders' equity requirement for continued listing on The Nasdaq Capital Market under Nasdaq Listing Rule 5550(b)(1). Nasdaq Listing Rule 5550(b)(1) requires companies listed on The Nasdaq Capital Market to maintain stockholders' equity of at least \$2.5 million. The Company's Quarterly Reports on Form 10-Q for the periods ended June 30, 2024 and September 30, 2024 reported stockholders' equity of negative \$8.8 million and negative \$13.3 million, respectively. The Letter further noted that as of the letter date, the Company did not have a market value of listed securities of \$35 million, or net income from continued operations of \$500,000 in the most recently completed fiscal year or in two of the last three most recently completed fiscal years, the alternative quantitative standards for continued listing on The Nasdaq Capital Market.

Neither the notice from Nasdaq nor the Company's non-compliance with the stockholders' equity requirement has an immediate effect on the listing or trading of the Company's securities on Nasdaq, which currently continues to trade on The Nasdaq Capital Market under the symbol "DRCT."

There can be no assurances, however, that we will be successful in regaining compliance with the continued listing requirements and maintaining the listing of our common stock on the Nasdaq Capital Market. Delisting from the Nasdaq could adversely affect our ability to raise additional financing through the public or private sale of equity securities, would significantly affect the ability of investors to trade our securities and would negatively affect the value and liquidity of our common stock. Delisting could also have other negative results, including the potential loss of confidence by employees, the loss of institutional investor interest and fewer business development opportunities. If our common stock is delisted by the Nasdaq, the price of our common stock may decline and our common stock may be eligible to trade on the OTC Markets or other over-the counter quotation system, where an investor may find it more difficult to dispose of their common stock or obtain accurate quotations as to the market value of our common stock. Further, if we are delisted, we would incur additional costs under requirements of state "blue sky" laws in connection with any sales of our securities. These requirements could severely limit the market liquidity of our common stock and the ability of our stockholders to sell our common stock in the secondary market.

#### ITEM 2. Unregistered Sales of Equity Securities and Use of Proceeds

#### Unregistered Sales of Equity Securities

None.

#### Use of Proceeds

None.

#### Purchases of Equity Securities by the Issuer

None.

#### ITEM 3. Defaults Upon Senior Securities

None.

#### ITEM 4. Mine Safety Disclosures

Not applicable.

#### ITEM 5. Other Information

### Rule 10b5-1 Trading Plans

During the nine months ended September 30, 2024, no director or officer of the Company adopted or terminated a "Rule 10b5-1 trading arrangement" or "non-Rule 10b5-1 trading arrangement," as each term is defined in Item 408(a) of Regulation S-K.



# ITEM 6. Exhibits

Exhibit No.	Description	Form	File Number	Date	Exhibit No.	Filed or Furnished herewith
3.1	Amended and Restated Certificate of Incorporation of Direct Digital Holdings, Inc.	8-K	001-41261	February 16, 2022	3.1	
3.2	Amended and Restated Bylaws of Direct Digital Holdings, Inc.	8-K	001-41261	February 16, 2022	3.2	
10.1	Fifth Amendment to Term Loan and Security Agreement, dated as of October 15, 2024, among Direct Digital Holdings, LLC, Lafayette Square Loan Services, LLC ("Lafayette Square"), as administrative agent, and the various lenders party thereto					Х
10.2	<u>Third Amendment to Credit Agreement, dated October 15, 2024,</u> <u>among Direct Digital Holdings, LLC, Huddled Masses LLC,</u> <u>Colossus Media, LLC and Orange 142, LLC, as borrowers, and East</u> <u>West Bank, as lender</u>					Х
10.3	Share Purchase Agreement, dated October 18, 2024, between Direct Digital Holdings, Inc. and New Circle Principal Investments LLC	8-K	001-41261	October 21, 2024	10.1	
10.4	<u>Registration Rights Agreement, dated as of October 18, 2024,</u> <u>between Direct Digital Holdings, Inc. and New Circle Principal</u> Investments LLC	8-K	001-41261	October 21, 2024	10.2	
31.1	Certification of the Chief Executive Officer of Direct Digital Holdings, Inc., pursuant to Rule 13a-14(a) of the Exchange Act, as adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.	0-1	001-41201	00000121,2024	10.2	Х
31.2	<u>Certification of the Chief Financial Officer of Direct Digital</u> <u>Holdings, Inc, pursuant to Rule 13a-14(a) of the Exchange Act, as</u> <u>adopted pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.</u>					Х
32.1*	Certification of the Chief Executive Officer pursuant to Rule 13a- 14(b) of the Exchange Act and 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.					Х
32.2*	Certification of the Chief Financial Officer pursuant to Rule 13a- 14(b) of the Exchange Act and 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.					Х
101.INS*	Inline XBRL Instance Document					Х
101.SCH*	Inline XBRL Taxonomy Extension Schema					Х
101.CAL*	Inline XBRL Taxonomy Extension Calculation Linkbase					Х
101.DEF*	Inline XBRL Taxonomy Extension Definition Linkbase					Х

101.LAB*	Inline XBRL Taxonomy Extension Label Linkbase	Х
101.PRE*	Inline XBRL Extension Presentation Linkbase	Х
104*	Cover Page Interactive Data File (formatted as inline XBRL and contained in Exhibit 101)	Х

\* This exhibit will not be deemed "filed" for purposes of Section 18 of the Exchange Act, or otherwise subject to the liability of that section. Such exhibit will not be deemed to be incorporated by reference into any filing under the Securities Act or the Exchange Act, except to the extent that the Company specifically incorporates it by reference.

# SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934 the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Date: November 13, 2024

DIRECT DIGITAL HOLDINGS, INC.

By:

/s/ Diana P. Diaz

DIANA P. DIAZ Chief Financial Officer (Duly Authorized Signatory, Principal Financial and Accounting Officer)

#### FIFTH AMENDMENT TO TERM LOAN AND SECURITY AGREEMENT

This Fifth Amendment to Term Loan and Security Agreement ("<u>Agreement</u>"), dated as of October 15, 2024 and effective as of June 30, 2024 (the "<u>Effective Date</u>"), is entered into by and between Direct Digital Holdings, LLC, a Texas limited liability company ("<u>Borrower</u>"), Direct Digital Holdings, Inc., a Delaware corporation ("<u>DDH Holdings</u>"), Colossus Media, LLC, a Delaware limited liability company ("<u>Colossus</u>"), Huddled Masses LLC, a Delaware limited liability company ("<u>HM</u>"), Orange142, LLC, a Delaware limited liability company ("<u>Credit Parties</u>"), and each individually a "<u>Guarantor</u>" and together with Borrower, collectively, the "<u>Credit Parties</u>", and each a "<u>Credit Party</u>"), Lafayette Square Loan Servicing, LLC, as agent for the Lenders ("<u>Agent</u>"), and the Lenders party hereto.

#### RECITALS:

WHEREAS, the Borrower, the Guarantors, the Lenders and Agent entered into that certain Term Loan and Security Agreement dated as of December 3, 2021, as amended by the First Amendment to Term Loan and Security Agreement dated as of February 3, 2022, the Second Amendment to Term Loan and Security Agreement dated as of July 28, 2022, the Third Amendment to Term Loan and Security Agreement dated as of January 9, 2023, and the Fourth Amendment to Term Loan and Security Agreement dated as of October 3, 2023 (the "Existing Loan Agreement"; the Existing Loan Agreement as may be further amended, supplemented, or otherwise modified from time to time, including by this Agreement, the "Loan Agreement");

WHEREAS, effective as of June 30, 2024, the Borrower and Agent have agreed to defer the principal payments of the Closing Date Term Loan required under Section 2.1(a)(i) (Closing Date Term Loan) and the Delayed Draw Term Loan required under Section 2.1(b)(iv) (Delayed Draw Term Loans) of the Loan Agreement through and including December 31, 2025, which was memorialized in email correspondence on July 2, 2024;

WHEREAS, the Borrower, each Guarantor, Agent, and the Lenders desire to amend the Existing Loan Agreement as set forth herein;

WHEREAS, Agent and the Lenders are willing to amend the Existing Loan Agreement under the terms and conditions set forth herein.

**NOW, THEREFORE**, in consideration of the premises and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Credit Parties, Agent, and the Lenders party hereto hereby agree as follows:

1. **Same Terms**. The capitalized terms used in this Agreement and not defined herein shall have the same meanings as provided therefor in the Loan Agreement, unless the context hereof otherwise requires or provides.

2. Fees and Expenses. The Credit Parties agree to pay or reimburse Agent for all fees owing to Agent and all fees and expenses (including, without limitation, reasonable attorneys' fees and legal expenses) incurred by Agent in connection with the preparation, negotiation and execution of this Agreement. In consideration of the agreements set forth herein, the Credit Parties agree to pay to Agent the fees set forth in the Fifth Amendment Fee Letter (as defined below).

3. Amendments to Existing Loan Agreement. The Borrower, each Guarantor, Agent, and the Lenders hereby agree that, upon the satisfaction or waiver of the conditions to effectiveness set forth in Section 7, (a) the Loan Agreement shall be amended to delete the stricken text (indicated textually in the same manner as the following example: stricken text and stricken text) and to add the double-underlined text (indicated textually in the same manner as the following example: double-underlined text) as set forth in the Loan Agreement attached as <u>Annex A</u> hereto and (b) Exhibit A, Exhibit C, and Exhibit D to the Loan Agreement shall be amended and restated in their entirety as set forth in Exhibit A, Exhibit C, and Exhibit D, respectively, attached hereto as <u>Annex B</u> hereto.

4. **Ratification**. Except as expressly provided herein, each Credit Party hereby (a) ratifies the Obligations and each of the Loan Agreement and the Other Documents to which it is a party, and agrees and acknowledges that the Loan Agreement and each of the Other Documents to which it is a party shall continue in full force and effect after giving effect to this Agreement; (b) ratifies and confirms that the security instruments executed by each Credit Party, as amended hereby, are not released, diminished, impaired, reduced, or otherwise adversely affected by the Loan Agreement and continue to secure the full payment and performance of the Obligations pursuant to their terms; (c) acknowledges the continuing existence and priority of the Liens granted, conveyed, and assigned to Agent for its benefit and for the ratable benefit of each Lender, under the security instruments; and (d) agrees that the Obligations include, without limitation, the Obligations (after giving effect to this Agreement). Except as expressly provided herein, nothing in this Agreement extinguishes, novates or releases any right, claim, Lien, security interest or entitlement of Agent or the Lenders created by or contained in any of such documents nor is any Credit Party released from any covenant, warranty or obligation created by or contained therein.

5. **Representations and Warranties**. Each Credit Party hereby represents and warrants to Agent that; (a) this Agreement has been duly authorized, executed, and delivered by each Credit Party; (b) no action of, or filing with, any Governmental Body is required to authorize, or is otherwise required in connection with, the execution, delivery, and performance by each Credit Party of this Agreement; (c) the Loan Agreement and the Other Documents, as amended by this Agreement, are valid and binding upon each Credit Party and are enforceable against each such Credit Party, in accordance with their respective terms, except as limited by Debtor Relief Laws; (d) the execution, delivery, and performance by each Credit Party of this Agreement does not require the consent of any other Person, except for any consent that has been duly obtained, and do not and will not constitute a violation of any laws, agreements, or understandings to which each such Credit Party is a party or by which each such Credit Party is bound; (e) after giving effect to this Agreement, all representations and warranties in the Loan Agreement and the Other Documents are true and correct in all material respect as of such earlier date or (ii) the facts on which any of them were based have been changed by transactions contemplated or permitted by the Loan Agreement; and (f) after giving effect to this Agreement, no Default or Event of Default exists.

6. **Release**. In consideration of Agent and Lenders' agreements herein, each Credit Party hereby (a) releases, acquits and forever discharges Agent, each Lender and each of their respective agents, employees, officers, directors, partners, servants, representatives, attorneys, affiliates, successors and assigns (collectively, the "<u>Released Parties</u>") from any and all liabilities, claims, suits, debts, liens, losses, causes of action, demands, rights, damages, costs and expenses of any kind, character or nature whatsoever, known or unknown, fixed or contingent, that such Credit Party may have or claim to have now against any Released Party or which might arise out of or be connected with any act of commission or omission of any Released Party existing or occurring on or prior to the date of this Agreement, including, without limitation, any claims, liabilities or obligations relating to or arising out of or in

connection with the Advances, the Loan Agreement or the Other Documents (including, without limitation, arising out of or in connection with the initiation, negotiation, closing or administration of the transactions contemplated thereby or related thereto), from the beginning of time until the execution and delivery of this release and the effectiveness of this Agreement (the "<u>Released Claims</u>") and (b) agrees forever to refrain from commencing, instituting or prosecuting any lawsuit, action or other proceeding against the Released Parties with respect to any and all Released Claims.

7. **Conditions to Effectiveness**. The transactions contemplated by this Agreement shall be deemed to be effective as of the Effective Date, when the following have been satisfied in a manner satisfactory to Agent:

(a) <u>Agreement</u>. Agent receives a fully executed copy of this Agreement;

(b) <u>Fifth Amendment Fee Letter</u>. Agent receives a fully executed fee letter (the "<u>Fifth Amendment Fee Letter</u>") by Borrower in form and substance satisfactory to Agent;

- (c) <u>2024 ABL Amendment</u>. Agent receives a fully executed copy of the 2024 ABL Amendment (as defined below);
- (d) <u>Fees and Expenses</u>. Agent receives all fees payable to Agent and Lenders on or prior to the Effective Date, including under *Section 2* hereof;

(e) <u>Representations and Warranties</u>. All representations and warranties set forth in this Agreement are true and correct in all material respects as set forth in *Section 5* above; and

(f) <u>Certificates of each Credit Party</u>. Agent receives certificate from the Secretary or Assistant Secretary of each Credit Party, in form and substance reasonably satisfactory to Agent: (i) attesting to the resolutions of such Credit Party's Board of Directors or similar governing body authorizing its execution, delivery, and performance of this Agreement and the transactions contemplated hereby, and such other items and documents as Agent shall reasonably request, (ii) authorizing specific officers of such Credit Party to execute the same and (iii) attesting to the incumbency and signatures of such specific officers of such Credit Party.

8. **Consent to the 2024 ABL Amendment.** Pursuant to Section 6.1 of the Intercreditor Agreement, Agent consents to the entry by the Credit Parties into that certain Third Amendment to Credit Agreement (the "2024 ABL Amendment") dated the date hereof by and among the Credit Parties and East West Bank pursuant to which, among other things, the Credit Parties and East West Bank agree (1) that the Credit Parties shall make mandatory prepayments on the notes under the ABL Credit Agreement equal to (A) \$1,000,000 on or before January 15, 2025 and (B) \$2,000,000 on or before April 15, 2024 pursuant to Section 3.02(b) of the ABL Credit Agreement and (2) to amend certain financial covenants set forth in Article IX of the ABL Credit Agreement.

9. **Counterparts**. For the convenience of the parties, this Agreement may be executed in multiple counterparts, each of which for all purposes shall be deemed to be an original, and all such counterparts shall together constitute but one and the same agreement. Delivery of an executed counterpart of a signature page of this Agreement by telecopy, e-mail, facsimile transmission, electronic mail in "*portable document format*" ("*.pdf*") form or other electronic means intended to preserve the original graphic and pictorial appearance of the item being sent shall be effective as a delivery of a manually executed counterpart of this Agreement.

10. **References to the Loan Agreement**. Upon the effectiveness of this Agreement, (a) each reference in the Loan Agreement to "*this Agreement*", "*hereof*", "*he* 

11. **Effect**. The Agreement is one of the Other Documents. The modifications set forth herein are limited precisely as written and shall not be deemed (a) to be a consent under or a waiver of or an amendment to any other term or condition in the Loan Agreement, or (b) to prejudice any right or rights which Agent or any Lender now has or may have in the future under or in connection with the Loan Agreement, as amended hereby, or any of the other documents referred to herein or therein.

12. **ENTIRE AGREEMENT**. THIS AGREEMENT CONSTITUTES THE ENTIRE AGREEMENT BETWEEN THE PARTIES HERETO WITH RESPECT TO THE SUBJECT HEREOF. FURTHERMORE, IN THIS REGARD, THIS AGREEMENT, THE LOAN AGREEMENT AND THE OTHER DOCUMENTS REPRESENT, COLLECTIVELY, THE FINAL AGREEMENT AMONG THE PARTIES THERETO AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF SUCH PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS AMONG SUCH PARTIES.

13. **Governing Law.** This Agreement, and all matters relating hereto or arising herefrom (whether arising under contract law, tort law or otherwise) shall, in accordance with Section 5-1401 of the General Obligations Law of the State of New York, be governed by and construed in accordance with the laws of the State of New York.

## [REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, this Agreement is deemed executed effective as of the Effective Date.

## **BORROWER:**

# DIRECT DIGITAL HOLDINGS, LLC

By: <u>/s/ Keith W. Smith</u> Name: Keith W. Smith Title: President

# **GUARANTORS:**

# DIRECT DIGITAL HOLDINGS, INC.

By: <u>/s/ Keith W. Smith</u> Name: Keith W. Smith Title: President

## **COLOSSUS MEDIA, LLC**

By:

Name: Keith Smith Title: President

# HUDDLED MASSES LLC

By: <u>/s/ Keith W. Smith</u> Name: Keith W. Smith Title: President

# ORANGE142, LLC

By: <u>/s/ Keith W. Smith</u> Name: Keith W. Smith Title: President

> Signature Page to Fifth Amendment to Term Loan and Security Agreement

# AGENT:

# LAFAYETTE SQUARE LOAN SERVICING, LLC

By: <u>/s/ Damien Dwin</u> Name: Damien Dwin Title: Chief Executive Officer

Signature Page to Fifth Amendment to Term Loan and Security Agreement

# **LENDERS:**

LAFAYETTE SQUARE USA, INC., as a Lender

By: <u>/s/ Damien Dwin</u> Name: Damien Dwin Title: Chief Executive Officer

Signature Page to Fifth Amendment to Term Loan and Security Agreement

# ANNEX A

# [SEE ATTACHED]

Annex A

ANNEX B

[SEE ATTACHED]

Annex B

# TERM LOAN AND SECURITY AGREEMENT

LAFAYETTE SQUARE LOAN SERVICING, LLC (AS AGENT)

THE LENDERS PARTY HERETO (AS LENDERS)

WITH

DIRECT DIGITAL HOLDINGS, LLC (AS A BORROWER),

ANY PERSON JOINED HERETO AS A BORROWER FROM TIME TO TIME

AND

CERTAIN SUBSIDIARIES OF BORROWERS PARTY HERETO (AS GUARANTORS)

December 3, 2021

2 159823.01008/130146990v.3 159823.01008/150720893v.5

# TABLE OF CONTENTS

## Page

- I. DEFINITIONS. 1
  - 1.1 Accounting Terms 1
  - 1.2 Defined Terms 2
  - 1.3 Uniform Commercial Code Terms 4247
  - 1.4 Certain Matters of Construction 4247
  - 1.5 Early Opt-In Election. Agent notified Borrower on June 1, 2023 that pursuant to Section 3.8 hereof, Agent and the Credit Parties have jointly elected to trigger a fallback from USD LIBOR to Term SOFR. Pursuant thereto, effective May 31, 2023 (the "Replacement Date"), Term SOFR replaced LIBOR for all purposes under this Agreement and the Other Documents in respect of such Benchmark setting without any amendment to, or further action or consent of any other party to, the Loan Agreement or any Other Document. On the Replacement Date, the Benchmark Replacement shall mean the sum of (x) Term SOFR and (y) 0.15% (15 basis points) for an Available Tenor of three-months' duration. In connection with the implementation and administration of Term SOFR, Agent reserves the right to make Benchmark Replacement Conforming <u>Changes from time to time in accordance with this Agreement</u>. 43 48

# II. ADVANCES, PAYMENTS. 4349

- 2.1 <u>Term Loans</u> 4349
- 2.2 <u>Repayment of Advances</u> 4551
- 2.3 [<u>Reserved</u>] 46<u>51</u>
- 2.4 <u>Statement of Account</u> 46<u>52</u>
- 2.5 Additional Payments 4652
- 2.6 Manner of Borrowing and Payment 4652
- 2.7 Mandatory Prepayments 4753
- 2.8 Use of Proceeds 4954
- 2.9 Defaulting Lender 4955
- 2.10 Joint and Several Liability, Waivers, etc 5056

# III. INTEREST AND FEES. 5358

- 3.1 <u>Interest</u> 5358
- 3.2 [<u>Reserved</u>] <del>53</del><u>59</u>

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5 i

ii

- 3.3 [<u>Reserved</u>] 5359
- 3.4 <u>Fee Letter</u> 5359
- 3.5 <u>Computation of Interest and Fees</u> 5359
- 3.6 <u>Maximum Charges</u> 5359
- 3.7 Increased Costs 5359
- 3.8 Benchmark Replacement Setting 5460
- 3.9 <u>Capital Adequacy</u> 5861
- 3.10 <u>Taxes</u> <u>5962</u>
- 3.11 <u>Replacement of Lenders</u> 6366
- IV. COLLATERAL: GENERAL TERMS. 6467
  - 4.1 <u>Security Interest in the Collateral</u> 6467
  - 4.2 Creation and Perfection of Security Interest 6467
  - 4.3 [<u>Reserved</u>] 6568
  - 4.4 <u>Preservation of Collateral</u> 6568
  - 4.5 <u>Ownership of Collateral</u> 6569
  - 4.6 Defense of Agent's and Lenders' Interests 6669
  - 4.7 <u>Books and Records</u> 6770
  - 4.8 <u>Financial Disclosure</u> 6770
  - 4.9 <u>Compliance with Laws</u> 6770
  - 4.10 Inspection of Premises 6770
  - 4.11 <u>Insurance</u> 6771
  - 4.12 <u>Failure to Pay Insurance</u> 6872
  - 4.13 Payment of Taxes 6872
  - 4.14 Payment of Leasehold Obligations 6972
  - 4.15 <u>Receivables</u> 6972
  - 4.16 <u>Pledge of Personal Property Assets</u> 7275
  - 4.17 <u>Maintenance of Equipment</u> 7275
  - 4.18 Exculpation of Liability 7276
  - 4.19 Environmental Matters 7376
  - 4.20 Financing Statements 7477
  - 4.21 Key Executive Policy 7477
- V. REPRESENTATIONS AND WARRANTIES. 7477
  - 5.1 Organization; Requisite Power and Authority; Qualification 7478
  - 5.2 Equity Interests and Ownership: 7578
  - 5.3 <u>Due Authorization</u> 7578
  - 5.4 <u>No Conflict</u> 7578
  - 5.5 Governmental Consents 7579
  - 5.6 Binding Obligations 7679
  - 5.7 <u>No Material Adverse Effect; No Default</u> 7679

5.8 Tax Returns 7679 5.9 Financial Statements 7679 5.10 Information Regarding the Credit Parties and their Subsidiaries 7780 5.11 Environmental Matters; O.S.H.A. 7780 5.12 Solvency; No Litigation, Violation, Indebtedness or Default 7881 5.13 Patents, Trademarks, Copyrights and Licenses 7982 5.14 Licenses, Permits and Other Approvals 7982 5.15 Default of Indebtedness 7983 5.16 <u>No Default</u> 7983 No Burdensome Restrictions 8083 5.17 5.18 No Labor Disputes 8083 Margin Regulations 8083 5.19 5.20 Investment Company Act 8084 5.21 Disclosure 8084 5.22 Delivery of Certain Documents 8184 5.23 <u>Swaps</u> <u>8184</u> 5.24 Immaterial Subsidiaries 8184 5.25 [<u>Reserved</u>] 8184 5.26 Business and Property of Credit Parties 8184 Insurance 8285 5.27 5.28 [Reserved] 8285 5.29 Anti-Terrorism Laws; Anti-Corruption Laws-5.30 Trading with the Enemy 8386 5.31 [Reserved] 8386 5.32 [Reserved]: 8386 Eligible Impact Service 83 Worker Solutions 86 5.33 5.34 Data Privacy and Security 8386 5.35 Small Business Administration Documents 8487 5.36 Small Business Concern 8487 5.37 Canadian Subsidiary 8487 5.38 Survival of Representations and Warranties 8487

# VI. AFFIRMATIVE COVENANTS. 8488

- 6.1 <u>Lender Meetings</u> 8488
- 6.2 <u>Conduct of Business and Maintenance of Existence and Assets</u> 8588
- 6.3 <u>Violations</u> 8588
- 6.4 <u>Government Receivables</u> 8588
- 6.5 <u>Financial Covenants</u>: 8588
- 6.6 Execution of Supplemental Instruments 8689
- 6.7 <u>Payment of Indebtedness</u> 8689

iii

- 6.8 <u>Real Property Collateral</u> 8690
- 6.9 Federal Securities Laws 8791
- 6.10 Exercise of Rights 8791
- 6.11 [<u>Reserved</u>]: 8891
- 6.12 Additional Subsidiaries 8891
- 6.13 [<u>Reserved</u>] <del>89</del>92
- 6.14 Sanctions and Anti-Corruption Matters 8992
- 6.15 Impact Subscription to or Impact Services. 89
- 6.16 Post-Closing Deliveries- 8993

## VII. NEGATIVE COVENANTS. 9194

- 7.1 Merger, Consolidation, Acquisition and Sale of Assets 9194
- 7.2 <u>Creation of Liens</u> 9195
- 7.3 <u>No Further Negative Pledges</u> 9397
- 7.4 <u>Investments</u> 9397
- 7.5 <u>Sale and Lease-Backs</u> 9498
- 7.6 [<u>Reserved</u>] 9498
- 7.7 <u>Restricted Payments</u> 9498
- 7.8 <u>Indebtedness</u> 9599
- 7.9 <u>Nature of Business</u> 9699
- 7.10 <u>Transactions with Affiliates</u> 9699
- 7.11 <u>Burdensome Agreements</u> 96100
- 7.12 [<u>Reserved</u>] 97<u>100</u>
- 7.13 Fiscal Year and Accounting Changes 97100
- 7.14 [<u>Reserved</u>] 97100
- 7.15 <u>Amendment of Governing Documents</u> 97100
- 7.16 Compliance with ERISA 97101
- 7.17 Prepayment of Indebtedness 97101
- 7.18 Anti-Terrorism Laws: 97101
- 7.19 <u>Membership/Partnership Interests</u> 98102
- 7.20 <u>Trading with the Enemy Act</u> 98102
- 7.21 <u>Management Agreement; Management Fees</u> 98102
- 7.22 Other Agreements 99102
- 7.23 Canadian Subsidiary 99102
- 7.24 ABL Obligations 99 Reserved 103

# VIII. CONDITIONS PRECEDENT. 99103

- 8.1 Conditions to Initial Advances 99103
- 8.2 <u>Conditions to Delayed Draw Term Loans</u> 102106

# IX. INFORMATION AS TO CREDIT PARTIES. <u>103</u>107

- 9.1 Information Regarding Credit Parties 103107
- 9.2 [<u>Reserved</u>] <del>104<u>107</u></del>
- 9.3 [<u>Reserved</u>] <del>104</del><u>107</u>
- 9.4 <u>Litigation, etc</u> <del>104<u>107</u></del>
- 9.5 <u>Material Occurrences</u> 104107
- 9.6 <u>Management Discussion and Analysis</u> 104108
- 9.7 <u>Annual Financial Statements</u> 104108
- 9.8 <u>Quarterly Financial Statements</u> 105108
- 9.9 <u>Monthly Financial Statements</u> 105109
- 9.10 Acquired Intellectual Property 106109
- 9.11 Additional Information 106109
- 9.12 <u>Projections</u> <del>106</del><u>110</u>
- 9.13 Small Business Administration 106110
- 9.14 Notice of Suits, Adverse Events 106110
- 9.15 ERISA Notices and Requests 107110
- 9.16 [<u>Reserved</u>]: 107111
- 9.17 Board Observation Rights 107111

# X. EVENTS OF DEFAULT. 108111

- 10.1 <u>Nonpayment</u> 108111
- 10.2 Breach of Representation 108111
- 10.3 <u>Noncompliance</u> 108112
- 10.4 <u>Reserved</u>: 108112
- 10.5 Judgments and Attachments 108112
- 10.6 <u>Criminal Proceeding</u> 109112
- 10.7 Bankruptcy 109112
- 10.8 Inability to Pay 109113
- 10.9 <u>Government Settlement</u> 109113
- 10.10 Material Adverse Change 109113
- 10.11 Cross Default to other Indebtedness 110113
- 10.12 Breach of Guarantee 110114
- 10.13 Change of Control 110114
- 10.14 Invalidity 110114
- 10.15 <u>Seizures</u> <u>110114</u>
- 10.16 <u>Operations</u> <u>110114</u>
- 10.17 Pension Benefit Plans 111114
- 10.18 Anti-Terrorism Laws 111114
- 10.19 Invalidity of Intercreditor Agreement or Subordination Provisions 111115

v

XI. LENDERS' RIGHTS AND REMEDIES AFTER DEFAULT, ETC. 111116

- 11.1 Rights and Remedies 111116
- 11.2 Agent's Discretion 114118
- 11.3 <u>Setoff</u> 114<u>118</u>
- 11.4 <u>Rights and Remedies not Exclusive</u> 114119
- 11.5 <u>Allocation of Payments After Event of Default</u> 114119

# XII. WAIVERS AND JUDICIAL PROCEEDINGS. <u>115120</u>

- 12.1 <u>Waiver of Notice</u> <u>115120</u>
- 12.2 <u>Delay</u> <del>115</del>120
- 12.3 <u>Jury Waiver</u> 115120

# XIII. EFFECTIVE DATE AND TERMINATION. <u>116120</u>

- 13.1 <u>Term</u> <u>116120</u>
- 13.2 <u>Termination</u> 116120
- 13.3 <u>Revival and Reinstatement of Obligations</u> 116121

# XIV. REGARDING AGENT. <u>117121</u>

- 14.1 <u>Appointment</u> 117121
- 14.2 <u>Nature of Duties</u> 117122
- 14.3 Lack of Reliance on Agent and Resignation 118122
- 14.4 Certain Rights of Agent 119123
- 14.5 <u>Reliance</u> 119123
- 14.6 Notice of Default 119124
- 14.7 <u>Indemnification</u> <u>119124</u>
- 14.8 Agent in its Individual Capacity 119124
- 14.9 <u>Delivery of Documents</u> 120124
- 14.10 Borrowers' Undertaking to Agent 120124
- 14.11 <u>No Reliance on Agent's Customer Identification Program</u> 120125
- 14.12 Other Agreements 120125
- 14.13 Erroneous Payment 120125

# XV. MISCELLANEOUS. 122126

- 15.1 <u>Governing Law</u> 122126
- 15.2 Entire Understanding 122127
- 15.3 <u>Successors and Assigns; Participations; New Lenders</u> 124128
- 15.4 <u>Application of Payments</u> 126131
- 15.5 <u>Indemnity</u> <u>127131</u>
- 15.6 <u>Notice</u> <u>128132</u>
- 15.7 <u>Survival</u> 129134
- 15.8 <u>Severability</u> 130134

- 15.9 <u>Expenses</u> 130134
- 15.10 Injunctive Relief 130135
- 15.11 <u>Damages</u> 130135
- 15.12 Captions 130135
- 15.13 Counterparts, etc 130135
- 15.14 <u>Construction</u> 131135
- 15.15 Confidentiality; Sharing Information 131135
- 15.16 <u>PUBLICITY</u> <del>132</del><u>137</u>
- 15.17 Certifications From Banks and Participants; USA PATRIOT Act 133137
- 15.18 Borrowing Agency Provisions 133138
- XVI. GUARANTY AND SURETYSHIP AGREEMENT. 134138
  - 16.1 Guaranty and Suretyship Agreement 134138
  - 16.2 Guaranty of Payment and Not Merely Collection 134139
  - 16.3 Guarantor and Suretyship Waivers 134139
  - 16.4 <u>Repayments or Recovery from Agent or Lenders</u> 135140
  - 16.5 Enforceability of Obligations 136140
  - 16.6 Guaranty Payable upon Event of Default; Remedies 136141
  - 16.7 <u>Waiver of Subrogation</u> 137141
  - 16.8 Continuing Guaranty and Suretyship Agreement 137141
  - 16.9 <u>General Limitation on Guarantee Obligations</u> 137142
  - 16.10 <u>Right of Contribution</u> 137142
  - 16.11 Intercreditor Agreement 137142

# TABLE OF CONTENTS (continued)

# List of Exhibits and Schedules

Exhibits

- Exhibit A Eligible Worker Solutions
- Exhibit B Form of Transfer Supplement
- Exhibit C Form of Compliance Certificate
- Exhibit D Worker Solutions Certificate
- Exhibit E Key Performance Indicators
- Exhibit F-1 Form of Closing Date Term Note
- Exhibit F-2 Form of Delayed Draw Term Loan Note
- Exhibit G Form of Joinder
- Exhibit H Form of Financial Condition Certificate

# Schedules

- Schedule 1.2(b)Term Loan Commitment PercentagesSchedule 1.2(c)Commercial Tort ClaimsSchedule 4.5Chief Executive Office; Equipment and Inventory LocationsSchedule 4.15(h)Deposit and Investment AccountsSchedule 5.1Jurisdictions of Organization and Good StandingSchedule 5.2(a)Equity Interests and OwnershipSchedule 5.2(b)SubsidiariesSchedule 5.8(d)Pension Benefit PlanSchedule 5.10Information Regarding the Credit Parties and their SubsidiariesSchedule 5.26Liabilities, Assets and ActivitiesSchedule 5.27InsuranceSchedule 7.2Existing LiensSchedule 7.4Existing InvestmentsSchedule 7.8Existing Indebtedness
- Schedule 7.10 Transactions with Affiliates

viii

# TERM LOAN AND SECURITY AGREEMENT

Term Loan and Security Agreement dated as of December 3, 2021 among DIRECT DIGITAL HOLDINGS, LLC, a Texas limited liability company ("<u>DDH</u>", together with any Person joined as a party to this Agreement as a "Borrower" in accordance with <u>Section 6.12</u> hereof, and all of their respective permitted successors and assigns, the "<u>Borrowers</u>"), the Guarantors party hereto (together with the Borrowers, the "<u>Credit Party</u>"), the financial institutions which are now or which hereafter become a party hereto (collectively, the "<u>Lenders</u>" and each individually, a "<u>Lender</u>") and LAFAYETTE SQUARE LOAN SERVICING, LLC ("<u>Lafayette Square</u>"), as agent for Lenders (Lafayette Square, in such capacity, "<u>Agent</u>").

IN CONSIDERATION of the mutual covenants and undertakings herein contained, the Credit Parties, Lenders and Agent hereby agree as follows:

# I. <u>DEFINITIONS.</u>

I.1 Accounting Terms.

÷

(a) Except as otherwise expressly provided herein, all accounting terms used in this Agreement, the Other Documents or any certificate, report or other document made or delivered pursuant to this Agreement not otherwise defined herein shall have the meanings assigned to them in conformity with GAAP. Financial statements and other information required to be delivered by Borrower to Agent and Lenders pursuant to Sections 9.7, 9.8 and 9.9 shall be prepared in accordance with GAAP as in effect at the time of such preparation. If at any time any change in GAAP or in the consistent application thereof would affect the computation of any financial covenant or requirement set forth in this Agreement or any Other Document, and Borrowing Agent shall object in writing to determining compliance based on such change or Agent shall otherwise require, then Agent and Borrowing Agent shall negotiate in good faith to amend such financial covenant, requirement or applicable defined terms to preserve the original intent thereof in light of such change to GAAP, provided that, until so amended such computations shall continue to be made on a basis consistent with the most recent financial statements delivered pursuant to <u>Sections 9.7, 9.8 and 9.9</u> as to which no such objection has been made and the Credit Parties shall provide additional financial statements or supplements thereto, attachments to Compliance Certificates and/or calculations regarding financial covenants as Agent may reasonably require in order to provide the appropriate financial information required hereunder with respect to the Credit Parties both reflecting any applicable changes in GAAP and as necessary to demonstrate compliance with the financial covenants before giving effect to the applicable changes in GAAP.

(b) Notwithstanding the above, the parties hereto acknowledge and agree that all calculations of the financial covenants in Section 6.5 shall be made on a Pro Forma Basis.

1

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5 (c) Notwithstanding the above, for purposes of determining compliance with any covenant (including the computation of any financial covenant) contained herein, Indebtedness of DDH and its Subsidiaries (or DDH Holdings and its Subsidiaries if applicable) shall be deemed to be carried at 100% of the outstanding principal amount thereof, and the effects of FASB ASC 825 and FASB ASC 470-20 on financial liabilities shall be disregarded.

# (d) [Reserved].

(e) To the extent a Fiscal Year or Fiscal Quarter of DDH (or DDH Holdings if applicable) ends on a date that is not a calendar year or calendar quarter end, references herein to a Fiscal Year or Fiscal Quarter ending on March 31, June 30, September 30 or December 31 shall be deemed to refer to the Fiscal Year or Fiscal Quarter most recently ending on or about such date.

# I.2 Defined Terms

. For purposes of this Agreement the following terms shall have the following meanings:

"ABL Amendment" shall have the meaning set forth in Section 6.16(e) hereof.

"<u>ABL Credit Agreement</u>" shall mean (a) that certain Credit Agreement dated as of <u>September 30, 2020July 7, 2023</u> among DDH, certain Subsidiaries thereof, and ABL Lender, as such agreement has been amended and may hereafter be amended, restated or otherwise modified in accordance with the Intercreditor Agreement, and (b) any Replacement ABL Credit Agreement entered into subsequent to the <u>Fifth Amendment</u> Closing Date.

"<u>ABL Lender</u>" shall mean (a) East West Bank, a California bank, and any successors and assigns to the extent permitted pursuant to the Intercreditor Agreement, or (b) any lender under a Replacement ABL Credit Agreement.

"<u>ABL Loan Documents</u>" shall mean any and all notes, agreements, documents, instruments now or at anytime evidencing, securing, guarantying or otherwise executed and delivered in connection with the ABL Credit Agreement, as the same may be amended, restated, supplemented or modified in accordance with the Intercreditor Agreement, including all "Loan Documents" as defined in the ABL Credit Agreement.

"<u>ABL Obligations</u>" shall mean the liabilities and Indebtedness of the Credit Parties outstanding under the ABL Credit Agreement from time to time, including the "Obligations" under and as defined in the ABL Credit Agreement in effect on the date hereof.

## "Accountants" shall have the meaning set forth in Section 9.7 hereof.

"<u>Acquisition</u>" shall mean, with respect to any Person, the acquisition by such Person, in a single transaction or in a series of related transactions, of (a) all or any substantial portion of the property of another Person, or any division, line of business or other business unit of another

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5 Person or (b) at least a majority of the Voting Stock of another Person, in each case whether or not involving a merger or consolidation with such other Person and whether for cash, property, services, assumption of Indebtedness, securities or otherwise.

"<u>Acquisition Documents</u>" shall mean, with respect to any Acquisition, the purchase agreement and all material other agreements related to such Acquisition, in each case as amended in conformity with the terms of this Agreement.

<u>"Adjusted Term SOFR" shall mean for purposes of any calculation, the rate per annum (based on a year of 360 days and actual days elapsed and rounded upward to the nearest 1/100 of 1%) equal to Term SOFR for such calculation plus the Term SOFR Adjustment; provided, that if Term SOFR is less than one percent (1%) per annum at any time, Adjusted Term SOFR at such time shall be equal to the Floor.</u>

"Advances" shall mean and include each advance of the Term Loans.

"<u>Adverse Proceeding</u>" shall mean any action, suit, proceeding (whether administrative, judicial or otherwise), governmental investigation or arbitration (whether or not purportedly on behalf of any Credit Party or any of its Subsidiaries) at law or in equity, or before or by any Governmental Body, whether pending, threatened in writing against any Credit Party or any of its Subsidiaries or any material property of any Credit Party or any of its Subsidiaries.

"<u>Affiliate</u>" of any Person shall mean any other Person that directly, or indirectly through one or more intermediaries, Controls, is Controlled by, or is under common Control with such the Person specified.

"Agent" shall have the meaning set forth in the preamble to this Agreement and shall include its successors and permitted assigns.

"<u>Agreement</u>" shall mean this Term Loan and Security Agreement, as the same may be amended, restated, supplemented or otherwise modified from time to time.

"ALTA" shall mean American Land Title Association.

"<u>Anti-Corruption Laws</u>" shall mean the Foreign Corrupt Practices Act, 15 U.S.C. §§ 78dd-1, *et seq*, the UK Bribery Act of 2010 and all other laws, rules, and regulations of any jurisdiction applicable to any Credit Party or any of its Affiliates from time to time concerning or relating to bribery or corruption.

"<u>Anti-Terrorism Laws</u>" shall mean any Applicable Laws relating to terrorism, trade sanctions programs and embargoes, import/export licensing, money laundering or bribery and any regulation, order or directive promulgated, issued or enforced pursuant to such Applicable Laws, including Executive Order No. 13224, the USA PATRIOT Act, the Applicable Laws comprising or implementing the Bank Secrecy Act, and the Applicable Laws administered by

[Different first page setting changed from on in original to off in modified.].

159823.01008/150720893v.5

OFAC (as any of the foregoing Applicable Laws may from time to time be amended, modified, supplemented, renewed, extended, or replaced).

"<u>Applicable Impact Discount</u>" shall mean (a) as of the Closing Date and through and including the date immediately prior to the first Applicable Impact Discount Reset Date, a rate per annum equal to 0.00 % and (b) on and after each Applicable Impact Discount Reset Date, (i) a rate per annum as set forth in the table below based upon the Impact Subscription to each of the Eligible Impact Services by Borrowers:

No Eligible Impact Services	<del>0.00%</del>
One or more Eligible Impact Services	<del>0.05%</del>

plus (ii) solely to the extent Borrowers received a B Corp Certification by Standards Analysts at the non-profit B Lab (or any successor certification or administrator identified by Agent in its sole discretion), as listed in <u>https://bcorporation.net/directory</u> (or any successor directory identified by Agent in its sole discretion) or such other comparable certification from a nationally recognized and reputable third party assessment and validation of social and environmental performance identified to and approved by Agent in writing, a rate per annum equal to 0.05% so long as Borrowers maintain such certification.

The component of the Applicable Impact Discount set forth in clause (b)(i) above shall (i) remain in effect for the period of the applicable Impact Subscription to any applicable Eligible Impact Service(s) notwithstanding removal of such provider, program or service from <u>Exhibit A</u> (unless Agent has notified Borrowers in writing that such provider, program or service is removed or to be removed from <u>Exhibit A</u> prior to the commencement of the period of any such Impact Subscription) and (ii) be adjusted, to the extent applicable, (x) as of each Applicable Impact Discount Reset Date based upon and to the extent that Borrowers have certified in the applicable Impact Certificate as to the Impact Subscription to any such applicable Eligible Impact Service or (y) otherwise at such time agreed by Agent and Borrowers as a result of a mutual determination to (1) cease or otherwise terminate any applicable Eligible Impact Service (thereby reducing the Applicable Impact Discount) or (2) Impact Subscribe to an Eligible Impact Service which had not been the subject of an Impact Subscription immediately prior to such determination (thereby increasing the Applicable Impact Discount to the extent applicable).

If Agent determines that Borrowers did not successfully make an Impact Subscription to any such Eligible Impact Service during an earlier period in which the Applicable Impact Discount applied (or were otherwise not certified as a B Corporation under clause (b)(ii) above during any such earlier period), without limiting any other rights or remedies arising as a result of any certification in an Impact Certificate which may prove to have been false or misleading,

4

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5 [Different first page setting changed from on in original to off in modified.].

Borrowers shall automatically and retroactively be obligated to pay to Agent, for the benefit of the applicable Lenders, promptly on demand by Agent, an amount equal to the excess of the amount of interest that should have been paid for such period without the Applicable Impact Discount over the amount of interest actually paid for such period.

"<u>Applicable Impact Discount Reset Date</u>" shall mean the first day of the month following the date on which financial statements, Compliance Certificate and Impact Certificate are required to be delivered pursuant to <u>Section 9.8</u> after the end of each related Fiscal Quarter.

"<u>Applicable Laws</u>" shall mean all applicable laws, including all applicable provisions of constitutions, statutes, rules, ordinances, regulations and orders of all Governmental Bodies and all orders, rulings, writs and decrees of all courts, tribunals and arbitrators.

"Applicable Margin" shall mean (a) as of the Second Amendment Date and through and including the date on which the quarterly financial statements of the Credit Parties on a Consolidated Basis required under Section 9.8 hereof for the fiscal quarter ending June 30, 2022 are delivered, the margin corresponding to Level V below (the date of such delivery, the "Initial Adjustment Date"), and (b) effective on the first day of the month following receipt by Agent of the quarterly financial statements of the Credit Parties on a Consolidated Compliance Certificate required under Section 9.8 hereof for the fiscal quarter ending subsequent to the Initial Adjustment Date (such first day of the applicable month, an "Adjustment Date"), the Applicable Margin for the Term Loans shall be adjusted, if necessary, to the applicable percent per annum set forth in the pricing table below corresponding to the Consolidated Total Net-Leverage Ratio for the trailing four quarter period ending on the last day of the most recently completed fiscal quarter prior to the applicable Adjustment Date:

[Different first page setting changed from on in original to off in modified.].

Level	Consolidated Total <u>Net-Leverage Ratio</u>	Applicable Margin
Ι	Less than or equal to 1.00 to 1.00	7.00%
Π	Greater than 1.00 to 1.00 and less than or equal to 1.50 to 1.00	7.50%
III	Greater than 1.50 to 1.00 and less than or equal to 2.00 to 1.00	8.00%
IV	Greater than 2.00 to 1.00 and less than or equal to 2.50 to 1.00	8.50%
V	Greater than 2.50 to 1.00 and less than or equal to 3.00 to 1.00	9.00%
VI	Greater than 3.00 to 1.00 and less than or equal to 3.50 to 1.00	9.50%
VII	Greater than 3.50 to 1.00	10.00%

If the Credit Parties shall fail to deliver the financial statements, certificates and/or other information required under Sections 9.7 or 9.8 hereof by the dates required pursuant to such sections, each Applicable Margin shall be conclusively presumed to equal the highest Applicable Margin specified in the pricing table set forth above until the date of delivery of such financial statements, certificates and/or other information, at which time the rate will be adjusted based upon the Consolidated Total Net-Leverage Ratio reflected in such statements. Notwithstanding anything to the contrary set forth herein, immediately and automatically upon the occurrence and continuation of any Event of Default, each Applicable Margin shall increase to and equal the highest Applicable Margin specified in the pricing table set forth above and shall continue at such highest Applicable Margin until the date (if any) on which such Event of Default shall be waived in accordance with the provisions of this Agreement, at which time the rate will be adjusted based upon the Consolidated Total Net-Leverage Ratio reflected on the most recently delivered financial statements and Compliance Certificate delivered by the Credit Parties to Agent pursuant to Section 9.7 or 9.8 hereof (as applicable). Any increase in interest rates payable

6

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5 by the Credit Parties under this Agreement and the Other Documents pursuant to the provisions of the foregoing sentence shall be in addition to and independent of any increase in such interest rates resulting from the occurrence of any Event of Default (including, if applicable, any Event of Default arising from a breach of Section 9.7 or 9.8 hereof) and/or the effectiveness of the Default Rate provisions of Section 3.1 hereof.

If, as a result of any restatement of, or other adjustment to, the financial statements of the Credit Parties or for any other reason, Agent determines in its Permitted Discretion that (a) the Consolidated Total Net-Leverage Ratio as previously calculated as of any applicable date for any applicable period was inaccurate, and (b) a proper calculation of the Consolidated Total Net-Leverage Ratio for any such period would have resulted in different pricing for such period, then (i) if the proper calculation of the Consolidated Total Net-Leverage Ratio would have resulted in a higher interest rate for such period, automatically and immediately without the necessity of any demand or notice by Agent or any other affirmative act of any party, the interest accrued on the applicable outstanding Advances for such period under the provisions of this Agreement and the Other Documents shall be deemed to be retroactively increased by, and the Credit Parties shall be obligated to immediately pay to Agent for the ratable benefit of Lenders an amount equal to the excess of the amount of interest that should have been paid for such period over the amount of interest actually paid for such period; and (ii) if the proper calculation of the Consolidated Total Net-Leverage Ratio would have resulted in a lower interest rate for such period, then Agent shall, upon such determination, notify DDH of the amount of interest over-paid and the Borrowers shall be entitled to a credit (which shall be applied to future accrued interest) from the Lenders who were Lenders that held the Loans during the relevant period and continue to hold Loans on the date Agent made such determination in an amount equal to any such excess interest paid to such Lender; provided, that, if as a result of any restatement or other event or other determination by Agent a proper calculation of the Consolidated Total Net-Leverage Ratio would have resulted in a higher interest rate for one or more periods and a lower interest rate for one or more other periods (due to the shifting of income or expenses from one period to another period or any other reason), then the amount payable by the Credit Parties pursuant to clause (i) above shall be based upon the excess, if any, of the amount of interest that should have been paid for all applicable periods over the amounts of interest actually paid for such periods.

"<u>Asset Sale</u>" shall mean a sale, lease, Sale and Leaseback Transaction, assignment, conveyance, exclusive license (as licensor), Securitization Transaction, transfer or other disposition to, or any exchange of property with, any non-Credit Party Person, in one transaction or a series of transactions, of all or any part of any Credit Party or any of its Subsidiaries' businesses, assets or properties of any kind, whether real, personal, or mixed and whether tangible or intangible, whether now owned or hereafter acquired, created, leased or licensed, including the Equity Interests of DDH or any Subsidiary of DDH, provided that "Asset Sales" shall not include any (a) dispositions of surplus, obsolete or worn out property or property in the ordinary course of business; (b) dispositions of inventory sold, and Intellectual Property licensed, in the ordinary course of business; (c) dispositions of accounts or payment intangibles (each as defined in the UCC) resulting from the compromise or settlement thereof in the ordinary course of business for less than the full amount thereof; (d) dispositions of Cash Equivalents in the

[Different first page setting changed from on in original to off in modified.].

ordinary course of business; (e) licenses, sublicenses, leases or subleases granted to any third parties in the ordinary course of business that do not interfere in any material respect with the business of Borrower or any of its Subsidiaries; (f) any Redemption/Exchange Transactions; and (g) any transactions consummated pursuant to the Warrant Tender Offer.

"<u>Attributable Principal Amount</u>" shall mean (a) in the case of Capital Leases, the amount of Capital Lease obligations determined in accordance with GAAP, (b) in the case of Synthetic Leases, an amount determined by capitalization of the remaining lease payments thereunder as if it were a Capital Lease determined in accordance with GAAP, (c) in the case of Securitization Transactions, the outstanding principal amount of such financing, after taking into account reserve amounts and making appropriate adjustments, determined by Agent in its reasonable judgment and (d) in the case of Sale and Leaseback Transactions, the present value (discounted in accordance with GAAP at the debt rate implied in the applicable lease) of the obligations of the lessee for rental payments during the term of such lease.

"Authority" shall have the meaning set forth in Section 4.19(c).

"<u>Authorized Officer</u>" shall mean, as applied to any Person, any individual holding the position of chairman of the board (if an officer), chief executive officer, president or one of its vice presidents (or the equivalent thereof), chief financial officer or treasurer and, solely for purposes of making the certifications required under <u>Sections 8.1(c)</u> and (d), any secretary or assistant secretary.

"Available Tenor" shall mean, as of any date of determination and with respect to the then-current Benchmark, as applicable, (a) if such Benchmark is a term rate, any tenor for such Benchmark (or component thereof) that is or may be used for determining the length of an interest period pursuant to this Agreement or (b) otherwise, any payment period for interest calculated with reference to such Benchmark (or component thereof) that is or may be used for determining any frequency of making payments of interest calculated with reference to such Benchmark, in each case, as of such date and not including, for the avoidance of doubt, any tenor for such Benchmark that is then-removed pursuant to Section 3.8 hereof from the Interest Period options available to the Borrowers.

"Bankruptcy Code" shall mean Title 11 of the United States Code entitled "Bankruptcy," as now and hereafter in effect, or any successor statute.

"Benchmark" shall mean, initially, the Term SOFR Reference Rate; provided that if a Benchmark Transition Event has occurred with respect to the Term SOFR Reference Rate or the then-current Benchmark, then "Benchmark" shall mean the applicable Benchmark Replacement to the extent that such Benchmark Replacement has replaced such prior benchmark rate pursuant to Section 3.8(a) hereof.

"Benchmark Replacement" shall mean with respect to any Benchmark Transition Event, the first alternative set forth in the order below that can be determined by Agent for the applicable Benchmark Replacement Date:

(a) Daily Simple SOFR; or

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

(b) the sum of: (i) the alternate benchmark rate that has been selected by Agent and the Borrowing Agent giving due consideration to (A) any selection or recommendation of a replacement benchmark rate or the mechanism for determining such a rate by the Relevant Governmental Body or (B) any evolving or then-prevailing market convention for determining a benchmark rate as a replacement to the then-current Benchmark for Dollar-denominated syndicated credit facilities and (ii) the related Benchmark Replacement Adjustment.

If the Benchmark Replacement as determined pursuant to clause (a) or (b) above would be less than the Floor, the Benchmark Replacement will be deemed to be the Floor for all the purposes of this Agreement and the Other Documents.

"Benchmark Replacement Adjustment" shall mean, with respect to any replacement of the then-current Benchmark with an Unadjusted Benchmark Replacement, the spread adjustment, or method for calculating or determining such spread adjustment, (which may be a positive or negative value or zero) that has been selected by Agent and the Borrowing Agent giving due consideration to (a) any selection or recommendation of a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Benchmark with the applicable Unadjusted Benchmark Replacement by the Relevant Governmental Body or (b) any evolving or then-prevailing market convention for determining a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Benchmark with the applicable Unadjusted Benchmark Replacement for Dollar-denominated syndicated credit facilities.

"Benchmark Replacement Date" shall mean the earliest to occur of the following events with respect to the then-current Benchmark:

- (a) in the case of clause (a) or (b) of the definition of "Benchmark Transition Event", the later of (i) the date of the public statement or publication of information referenced therein and (ii) the date on which the administrator of such Benchmark (or the published component used in the calculation thereof) permanently or indefinitely ceases to provide all Available Tenors of such Benchmark (or such component thereof); or
- (b) in the case of clause (c) of the definition of "Benchmark Transition Event", the first date on which such Benchmark (or the published component used in the calculation thereof) has been determined and announced by or on behalf of the administrator of such Benchmark (or such component thereof) or the regulatory supervisor for the administrator of such Benchmark (or such component thereof) to be non-representative or non-compliant with or non-aligned with the International Organization of Securities Commissions (IOSCO) Principles for Financial Benchmarks; provided that such non-representativeness, non-compliance or non-alignment will be determined by reference to the most recent statement or publication referenced in such clause (c) and even if any Available Tenor of such Benchmark (or such component thereof), continues to be provided on such date. For the avoidance of doubt, the "Benchmark Replacement Date" will be deemed to have occurred in the case of clause (a) or (b) with respect to any Benchmark upon the occurrence of the applicable event or events set forth therein with



respect to all then-current Available Tenors of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Transition Event" shall mean the occurrence of one or more of the following events with respect to the then-current Benchmark:

- (a) a public statement or publication of information by or on behalf of the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that such administrator has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof), permanently or indefinitely; provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof);
- (b) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof), the Federal Reserve Board, the Federal Reserve Bank of New York, an insolvency official with jurisdiction over the administrator for such Benchmark (or such component), a resolution authority with jurisdiction over the administrator for such Benchmark (or such component) or a court or an entity with similar insolvency or resolution authority over the administrator for such Benchmark (or such component), which states that the administrator of such Benchmark (or such component) has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof) permanently or indefinitely; provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof); or
- (c) a public statement or publication of information by or on behalf of the administrator of such Benchmark (or the published component used in the calculation thereof) or the regulatory supervisor for the administrator of such Benchmark (or such component thereof) announcing that all Available Tenors of such Benchmark (or such component thereof) are not, or as of a specified future date will not be, representative or in compliance with or aligned with the International Organization of Securities Commissions (IOSCO) Principles for Financial Benchmarks.

For the avoidance of doubt, a "Benchmark Transition Event" will be deemed to have occurred with respect to any Benchmark if a public statement or publication of information set forth above has occurred with respect to each then-current Available Tenor of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Unavailability Period" shall mean, the period (if any) (a) beginning at the time that a Benchmark Replacement Date has occurred if, at such time, no Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Other Document in accordance with Section 3.8 hereof and (b) ending at the time that a Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Other Document in accordance with Section 3.8 hereof.

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

"Borrower" and "Borrowers" shall mean DDH and each other Person joined as a party to this Agreement as a "Borrower" in accordance with Sections 6.12 hereof, and all of their respective permitted successors and assigns.

"Borrowers' Account" shall have the meaning set forth in Section 2.4.

"Borrowing Agent" shall mean DDH.

"<u>Business Day</u>" shall mean any day other than Saturday or Sunday, a legal holiday on which commercial banks are authorized or required by law to be closed for business in New York, New York or legal holiday on which Agent is not open for business in its main office in New York.

"<u>Capital Expenditures</u>" shall mean, for any period of determination, expenditures made or liabilities incurred during such period for the acquisition of any fixed assets or improvements, replacements, substitutions or additions thereto which have a useful life of more than one year (including, for the purpose of clarity, the total principal portion of Capitalized Lease Obligations paid in such period), which, in accordance with GAAP, would be classified as capital expenditures, but excluding (a) any such expenditures made with proceeds of any Involuntary Disposition to the extent such expenditures are used to purchase property that is the same as or similar to the property subject to such Involuntary Disposition or (b) Permitted Acquisitions; provided that, for purposes hereof, Capital Expenditures shall be calculated net of the value of any property traded-in or exchanged in connection with any such expenditure.

"<u>Capital Lease</u>" shall mean, with respect to any Person, any lease of (or other agreement conveying the right to use) any property by such Person as a lessee that, in conformity with GAAP (subject to <u>Section 1.1</u>), is or should be accounted for as a capital lease on the balance sheet of such Person.

"Capitalized Lease Obligation" shall mean any obligation under a Capital Lease.

"Cash Equivalents" shall mean, as of any date of determination, any of the following: (a) marketable securities (i) issued or directly and unconditionally guaranteed as to interest and principal by the United States government, or (ii) issued by any agency of the United States the obligations of which are backed by the full faith and credit of the United States, in each case maturing within one (1) year after such date; (b) marketable direct obligations issued by any state of the United States or any political subdivision of any such state or any public instrumentality thereof, in each case maturing within one (1) year after such date and having, at the time of the acquisition thereof, a rating of at least A-1 from S&P or at least P-1 from Moody's; (c) commercial paper maturing no more than one (1) year from the date of creation thereof and having, at the time of the acquisition thereof, a rating of at least A-1 from S&P or at least P-1 from Moody's; (d) certificates of deposit or bankers' acceptances maturing within one (1) year after such date and issued or accepted by any commercial bank that is organized under the laws of the United States or any state thereof or the District of Columbia that (i) is at least "adequately capitalized" (as defined in the regulations of its primary federal banking regulator), and (ii) has

[Different first page setting changed from on in original to off in modified.].

Tier 1 capital (as defined in such regulations) of not less than 100,000,000; and (e) shares of any money market mutual fund that (i) has substantially all of its assets invested continuously in the types of investments referred to in <u>clauses (a)</u> and <u>(b)</u> above, (ii) has net assets of not less than 500,000,000, and (iii) has the highest rating obtainable from either S&P or Moody's.

"<u>CERCLA</u>" shall mean the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended, 42 U.S.C. §§9601 et seq.

## "Change of Control" shall mean:

(a) with respect to DDH Holdings, (x) at any time, any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act), other than Direct Digital Management, LLC or its Control Affiliates shall become, or obtain rights (whether by means of warrants, options or otherwise) to become, the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Exchange Act), directly or indirectly, of forty percent (40%) or more of the ordinary voting power for the election of directors, partners, managers and members, as applicable, of DDH Holdings (determined on a fully diluted basis) other than (i) by the sale of the Borrower's equity securities in a public offering, a private placement or to venture capital or private equity investors so long as the Borrower identifies to Agent the venture capital or private equity investors at least seven (7) Business Days prior to the closing of the transaction and provides to Agent a description of the material terms of the transaction and (ii) by increases or decreases in the beneficial ownership of DDH Holdings securities by Direct Digital Management, LLC or its Control Affiliates; or (y) during any period of 12 consecutive months, a majority of the members of the board of directors of DDH Holdings cease to be composed of individuals (i) who were members of that board or equivalent governing body on the first (1st) day of such period, (ii) whose election or nomination for election to that board or other equivalent governing body was approved by individuals referred to in clause (i) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body was approved by individuals referred to in clauses (i) and (ii) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body;

(b) with respect to the Borrower, (x) at any time, any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act) other than DDH Holdings, Direct Digital Management, LLC or its Control Affiliates shall become, or obtain rights (whether by means of warrants, options or otherwise) to become, the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Exchange Act), directly or indirectly, of forty percent (40%) or more of the ordinary voting power for the election of directors, partners, managers and members, as applicable, of the Borrower (determined on a fully diluted basis); or (y) except as permitted by Section 7.1, at any time, DDH Holdings and Direct Digital Management, LLC or its Control Affiliates shall collectively cease to own and control, of record and beneficially, directly or indirectly, one hundred percent (100%) of each class of outstanding stock, partnership,

[Different first page setting changed from on in original to off in modified.].

membership, or other ownership interest or other equity securities of the Borrower free and clear of all Liens (except Permitted Encumbrances); or

(c) with respect to each Subsidiary of the Borrower, (x) at any time, any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act) other than DDH Holdings, the Borrower and/or Direct Digital Management, LLC or its Control Affiliates shall become, or obtain rights (whether by means of warrants, options or otherwise) to become, the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Exchange Act), directly or indirectly, of forty percent (40%) or more of the ordinary voting power for the election of directors, partners, managers and members, as applicable, of such Subsidiary (determined on a fully diluted basis); or (y) except as permitted by Section 7.1, at any time, DDH Holdings, the Borrower and/or Direct Digital Management, LLC or its Control Affiliates shall together cease to own and control, of record and beneficially, directly or indirectly, one hundred percent (100%) of each class of outstanding stock, partnership, membership, or other ownership interest or other equity securities of such Subsidiary free and clear of all Liens (except Permitted Encumbrances).

"Closing Date" shall mean December 3, 2021.

"Closing Date Term Loan" shall mean the Advances made pursuant to Section 2.1(a) hereof.

"Closing Date Term Loan Amount" shall mean \$22,000,000.

"Closing Date Term Note" shall mean, collectively, the promissory notes described in Section 2.1(a) hereof.

"<u>Closing Date Transactions</u>" shall mean, collectively, (a) the consummation of the transactions contemplated by and this Agreement, (b) the Preferred A Redemption, (c) the repayment of Indebtedness of the Borrowers with the proceeds of the Closing Date Term Loan, in each case, which have been, or are contemplated to be, consummated on or prior to the Closing Date and (d) the payment of fees and expenses in connection therewith.

"Code" shall mean the Internal Revenue Code of 1986, as the same may be amended or supplemented from time to time, and any successor statute of similar import, and the rules and regulations thereunder, as from time to time in effect.

"<u>Collateral</u>" shall mean and include, all right, title and interest of each Credit Party in all of the following property and assets of such Credit Party, in each case, whether now existing or hereafter arising or created and whether now owned or hereafter acquired and wherever located:

- (a) all Receivables;
- (b) all cash and currency;
- (c) all Certificates of Title;

- (d) all Chattel Paper;
- (e) those certain Commercial Tort Claims set forth on <u>Schedule 1.2(c)</u> hereto;

(f) all Intellectual Property (other than any United States intent to use applications prior to the filing of a "Statement of Use" or "Amendment to Allege Use");

- (g) all Deposit Accounts;
- (h) all Documents;
- (i) all Equipment;
- (j) all Fixtures;
- (k) all General Intangibles (including, without limitation, the Key Executive Policy);
- (l) all Goods;
- (m) all Instruments;
- (n) all Inventory;
- (o) all Investment Property;
- (p) all Letter-of-Credit Rights;
- (q) all Payment Intangibles;
- (r) all Subsidiary Stock;
- (s) all Software;
- (t) all Supporting Obligations;
- (u) all books and records pertaining to any of the foregoing; and
- (v) all Accessions and Proceeds of any and all of the foregoing.

Notwithstanding the foregoing, Collateral shall not include, and no Credit Party shall be deemed to have granted a security interest in the following (each of the following, "Excluded Property"): (a) any intent-to-use trademarks at all times prior to the first use thereof, whether by the actual use thereof in commerce, the recording of a statement of use with the United States Patent and Trademark Office or otherwise; (b) any interest of any Credit Party as a lessee or sublessee under a real property lease under a real property lease or an Equipment lease if such Credit Party is prohibited by the terms of such lease from granting a security interest in such

[Different first page setting changed from on in original to off in modified.].

lease or under which such an assignment or Lien would cause a default to occur under such lease (but only to the extent that such prohibition is enforceable under all applicable laws including, without limitation, the UCC); provided, however, that upon termination of such prohibition, such interest shall immediately become Collateral without any action by any Credit Party or Agent; (c) rights held under a license that are not assignable by their terms without the consent of the licensor thereof (but only to the extent such restriction on assignment is enforceable under applicable law); or (d) any interest of any Credit Party as a lessee under an Equipment lease if such Credit PARTY is prohibited by the terms of such lease from granting a security interest in such lease or under which such an assignment or Lien would cause a default to occur under such lease; provided, however, that upon termination of such prohibition, such interest shall immediately become Collateral without any action by any Credit Party or Agent.

"<u>Commitment Percentage</u>" of any Lender shall mean, as applicable, its Term Loan Commitment Percentage and/or its Delayed Draw Term Loan Commitment Percentage, as applicable.

"<u>Common Units Redemption</u>" shall mean the contemplated redemption of all of the common units held in DDH by USDM Holdings, Inc. pursuant to the Redemption Agreement.

"<u>Company Agreement</u>" shall mean the amended and restated limited liability company agreement of DDH entered into in connection with a Qualified IPO.

"<u>Compliance Certificate</u>" shall mean a compliance certificate in substantially the form attached hereto as <u>Exhibit C</u> to be signed by an Authorized Officer of DDH.

"Conforming Changes" shall mean, with respect to either the use or administration of Term SOFR or the use, administration, adoption or implementation of any Benchmark Replacement, any technical, administrative or operational changes (including changes to the definition of "Business Day", the definition of "U.S. Government Securities Business Day", the definition of "Interest Period", "Interest Rate Reset Period" or any similar or analogous definition (or the addition of a concept of "interest period"), timing and frequency of determining rates and making payments of interest, timing of borrowing requests or prepayment, conversion or continuation notices, the applicability and length of lookback periods, the applicability of breakage provisions and other technical, administrative or operational matters) that Agent decides may be appropriate to reflect the adoption and implementation of any such rate or to permit the use and administration thereof by Agent in a manner substantially consistent with market practice (or, if Agent decides that adoption of such market practice is not administratively feasible or if Agent determines that no market practice for the administration of any such rate exists, in such other manner of administration as Agent decides is reasonably necessary in connection with the administration of this Agreement and the Other Documents).

"<u>Consents</u>" shall mean all filings and all licenses, permits, consents, approvals, authorizations, qualifications and orders of Governmental Bodies and other third parties, domestic or foreign, necessary to carry on the Credit Parties' business or necessary (including to avoid a conflict or breach under any agreement, instrument, other document, license, permit or other authorization) for the execution, delivery or performance of this Agreement, the Other Documents, the Acquisition Documents and all related documents and agreements, including any

Consents required under the ABL Loan Documents and all applicable federal, state or other Applicable Law.

"<u>Consolidated Basis</u>" shall mean, (a) prior to a Qualified IPO, with respect to DDH and its Subsidiaries, as the context may require, the consolidation in accordance with GAAP of the accounts and other items of DDH and its Subsidiaries, and (b) subsequent to a Qualified IPO, with respect to DDH Holdings and its Subsidiaries as the context may require, the consolidation in accordance with GAAP of the accounts and other items of DDH Holdings and its Subsidiaries.

"<u>Consolidated Capital Expenditures</u>" shall mean, for any period, for DDH (or DDH Holdings if applicable) and its Subsidiaries on a Consolidated Basis, the sum of all Capital Expenditures of such entity for such period.

"<u>Consolidated Current Assets</u>" shall mean, as of any date of determination, the total assets of DDH (or DDH Holdings if applicable) and its Subsidiaries on a Consolidated Basis that may properly be classified as current assets in accordance with GAAP, excluding cash and Cash Equivalents.

"<u>Consolidated Current Liabilities</u>" shall mean, as of any date of determination, the total liabilities of DDH (or DDH Holdings if applicable) and its Subsidiaries on a Consolidated Basis that may properly be classified as current liabilities in accordance with GAAP and shall be calculated by excluding the current portion of the Obligations.

"Consolidated EBITDA" shall mean, for any period, for DDH Holdings and its Subsidiaries on a Consolidated Basis, an amount equal to (a) Consolidated Net Income for such period <u>plus</u>, (b) to the extent deducted in determining such Consolidated Net Income, the sum, without duplication, of (i) Consolidated Interest Charges during such period, (ii) all federal, state, local and/or foreign income taxes payable by DDH Holdings and its Subsidiaries during such period, (iii) depreciation expenses of DDH Holdings and its Subsidiaries during such period, (v) any non-cash loss or expense resulting from any impairment charge or asset write-off or write-down related to intangible assets, long-lived assets and other assets that occurs during such period, (vi) one-time loss associated with debt refinancing, (vii) without duplication, non-recurring actual, documented legal or consulting expenses or retention bonuses paid in an amount up to  $\frac{5500,000750,000}{550,000}$  during any 12 month period, (vii) any cash payments (including all premiums) made with respect to the Key Executive Policies required pursuant to Section 4.21, and (ivix) reasonable and documented one-time, non-recurring severance and retention expenses in an aggregate amount not to exceed \$750,000 for employees and management of the Credit Parties, in each case to the extent paid on or before June 30, 2024, and (x) with respect to the fiscal year ended on December 31, 2023, accounting, audit, auditor and/or legal costs and expenses associated with any audits, regulatory compliance, investigations, remediation of audit financings, and the implementation of required changes as a result of audit outcomes or the change in auditor due to the resignation and replacement of the Borrowers' Accountants, provided that such add-backs shall not exceed \$1,500,000 with respect to such fiscal year, (xi).

[Different first page setting changed from on in original to off in modified.].

any amounts paid out to employees in stock, including the 2023 bonus accrual which the compensation committee of the board of directors of DDH Holdings approved to pay out in stock in February 2024, (xii) the gross amount of the profit on the disputed short pay which impacted December 2023 income in the amount up to a total of \$1,781,000, and (xiii) any amounts paid to publishers associated with the disputed short pay when recorded as expense in the financial statements up to a total of \$8,800,000 minus (c) any extraordinary, non-recurring and/or non-cash gains or income during such period as reported in the monthly, quarterly, and annual financials of DDH Holdings and its Subsidiaries.

"<u>Consolidated Excess Cash Flow</u>" shall mean, for any period for DDH and its Subsidiaries (or DDH Holdings and its Subsidiaries if applicable) on a Consolidated Basis, an amount equal to the sum, without duplication, of (a) Consolidated EBITDA, <u>minus</u> (b) Unfinanced Capital Expenditures, <u>minus</u> (c) the cash portion of Consolidated Interest Charges, <u>minus</u> (d) Consolidated Taxes paid in cash <u>minus</u> (e) Consolidated Scheduled Funded Debt Payments, <u>minus</u> (f) the Consolidated Working Capital Adjustment, <u>minus</u> (g) the aggregate amount of all voluntary prepayments of any Indebtedness permitted hereunder of DDH and its Subsidiaries (or DDH Holdings and its Subsidiaries if applicable) made during such Fiscal Year (other than in respect of any revolving credit facility to the extent there is not an equivalent permanent reduction in commitments thereunder), <u>minus</u> (h) all cash expenses added back in computing Consolidated EBITDA. For the avoidance of doubt, payments with respect to Preferred Equity and obligations owing pursuant to the Redemption Agreement, in each case, made with proceeds of the Delayed Draw Term Loan advanced on the Second Amendment Date shall not be deducted in calculating Consolidated Excess Cash Flow; provided, that it is understood that the principal with respect to the Delayed Draw Term Loan shall constitute Consolidated Interest Charges.

"<u>Consolidated Fixed Charge Coverage Ratio</u>" shall mean for DDH and its Subsidiaries (or DDH Holdings and its Subsidiaries if applicable) on a Consolidated Basis, as of any date of determination, the ratio of (a) Consolidated EBITDA for the period of the four (4) Fiscal Quarters most recently ended, <u>minus</u> Taxes paid in cash or permitted Tax Distributions during such period, <u>minus</u> without duplication, Unfinanced Capital Expenditures made during such period, <u>minus</u> Restricted Payments (other than the Preferred A Redemption) for such period paid in cash (without limiting any restrictions in respect thereof under <u>Section 7.7</u>) to (b) the sum of (i) Consolidated Fixed Charges for such period.

"<u>Consolidated Fixed Charges</u>" shall mean, for any period, for DDH (or DDH Holdings if applicable) and its Subsidiaries on a Consolidated Basis, an amount equal to the sum of (a) the cash portion of Consolidated Interest Charges for such period <u>plus</u> (b) Consolidated Scheduled Funded Debt Payments for such period, all as determined in accordance with GAAP.

"<u>Consolidated Funded Debt</u>" shall mean Funded Debt of DDH (or DDH Holdings if applicable) and its Subsidiaries on a Consolidated Basis determined in accordance with GAAP excluding any obligations under an operating lease.

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

"<u>Consolidated Interest Charges</u>" shall mean, for any period, the interest expense of DDH (or DDH Holdings if applicable) and its Subsidiaries on a Consolidated Basis for the period in question, determined on a consolidated basis and in accordance with GAAP.

"<u>Consolidated Net Income</u>" shall mean, for any period, for DDH (or DDH Holdings if applicable) and its Subsidiaries on a Consolidated Basis, the net income (or loss) of DDH (or DDH Holdings if applicable) and its Subsidiaries for that period, as determined in accordance with GAAP.

"<u>Consolidated Scheduled Funded Debt Payments</u>" shall mean for any period for DDH (or DDH Holdings if applicable) and its Subsidiaries on a Consolidated Basis, the sum of all scheduled payments of principal on Consolidated Funded Debt, as determined in accordance with GAAP. For purposes of this definition, "scheduled payments of principal" (a) shall be deemed to include, without duplication, scheduled principal payments with respect to the Attributable Principal Amount of Capital Leases (other than in respect of any Securitization Transactions, Sale and Leaseback Transactions and Synthetic Leases) and (b) shall not include any voluntary prepayments or mandatory prepayments required pursuant to the ABL Credit Agreement except to the extent such prepayments are accompanied by a permanent reduction of the commitments thereunder.

"<u>Consolidated Senior Net Leverage Ratio</u>" shall mean, as of any date of determination, the ratio of (a) the result of (i) the Funded Debt of DDH (or DDH Holdings if applicable) and its Subsidiaries on a Consolidated Basis, but excluding any Subordinated Debt and any obligations under an operating lease, minus (ii) the Unrestricted Cash of DDH and the Credit Parties in excess of \$1,000,000; provided that the amount of Unrestricted Cash included in the calculation pursuant to this clause (ii) shall not exceed \$15,000,000 to (b) Consolidated EBITDA for the period of the four (4) Fiscal Quarters most recently ended.

"<u>Consolidated Taxes</u>" shall mean, for any period, for DDH and its Subsidiaries (or DDH Holdings and its Subsidiaries if applicable) on a Consolidated Basis, the aggregate of all taxes, as determined in accordance with GAAP.

"<u>Consolidated Total Net-Leverage Ratio</u>" shall mean, as of any date of determination, the ratio of (a) the result of (i) the Funded Debt of DDH (or DDH Holdings if applicable) and its Subsidiaries on a Consolidated Basis, excluding any obligations under an operating lease, minus (ii) the Unrestricted Cash of DDH and the Credit Parties in excess of \$1,000,000; provided that the amount of Unrestricted Cash included in the ealeulation pursuant to this elause (ii) shall not exceed \$15,000,000 to (b) Consolidated EBITDA for the period of the four (4) Fiscal Quarters most recently ended.

"Consolidated Working Capital" shall mean, as of any date of determination, the excess of Consolidated Current Assets less Consolidated Current Liabilities.

"<u>Consolidated Working Capital Adjustment</u>" shall mean, for any period on a Consolidated Basis, the amount (which may be a negative number) by which Consolidated

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

Working Capital as of the end of such period exceeds (or is less than) Consolidated Working Capital as of the beginning of such period.

"<u>Contractual Obligation</u>" shall mean, as applied to any Person, any provision of any Security issued by that Person or of any indenture, mortgage, deed of trust, contract, undertaking, agreement or other instrument to which that Person is a party or by which it or any of its properties is bound or to which it or any of its properties is subject.

"<u>Control</u>" shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ability to exercise voting power, by contract or otherwise. "<u>Controlling</u>" and "<u>Controlled</u>" have meanings correlative thereto.

"<u>Control Affiliate</u>" shall mean, with respect to any Person, each other Person that owns or controls directly or indirectly the Person, any Person that controls or is controlled by or is under common control with the Person, and each of that Person's senior executive officers, directors, partners and, for any Person that is a limited liability company, that Person's managers and members.

"Control Agreement(s)" shall have the meaning set forth in Section 6.16(c) hereof.

"<u>Covered Entity</u>" shall mean (a) each Credit Party, each Subsidiary of each Credit Party, and all pledgors of Collateral and (b) each Person that, directly or indirectly, is in control of a Person described in clause (a) above. For purposes of this definition, control of a Person shall mean the direct or indirect (x) ownership of, or power to vote, twenty-five percent (25%) or more of the issued and outstanding equity interests having ordinary voting power for the election of directors of such Person or other Persons performing similar functions for such Person, or (y) power to direct or cause the direction of the management and policies of such Person whether by ownership of equity interests, contract or otherwise.

"Credit Parties" shall mean the Borrowers and the Guarantors, and "Credit Party" shall mean any of them.

"<u>Customer</u>" shall mean and include the account debtor with respect to any Receivable of a Person and/or the prospective purchaser of goods, services or both with respect to any contract or contract right, and/or any party who enters into or proposes to enter into any contract or other arrangement with such Person, pursuant to which such Person is to deliver any personal property or perform any services.

"Daily Simple SOFR" shall mean, for any day, SOFR, with the conventions for this rate (which will include a lookback) being established by Agent in accordance with the conventions for this rate selected or recommended by the Relevant Governmental Body for determining "Daily Simple SOFR" for syndicated business loans; provided that if Agent decides that any such convention is not administratively feasible for Agent, then Agent may establish another convention in its reasonable discretion.

[Different first page setting changed from on in original to off in modified.].

"DDH" shall have the meaning ascribed to such term in the preamble to this Agreement.

"DDH Holdings" shall mean Direct Digital Holdings, Inc., a Delaware corporation, which entity has been incorporated to enter into-a Qualified IPO and shall, upon consummation of the Qualified IPO, be the managing member of DDH at all times while the umbrella partnership corporation structure is in effect.

"<u>Debt Transaction</u>" shall mean, with respect to DDH or any of its Subsidiaries, any sale, issuance, placement, assumption or guaranty of Funded Debt, whether or not evidenced by a promissory note or other written evidence of Indebtedness, except for Funded Debt permitted to be incurred pursuant to <u>Section 7.8</u>.

"Debtor Relief Laws" shall mean the Bankruptcy Code, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief laws of the United States or other applicable jurisdictions from time to time in effect.

"Default" shall mean an event, circumstance or condition which, with the giving of notice or passage of time or both, would constitute an Event of Default.

"Default Rate" shall have the meaning set forth in Section 3.1 hereof.

"Defaulting Lender" shall have the meaning set forth in Section 2.9(a) hereof.

"Delayed Draw Term Loan" shall have the meaning set forth in Section 2.1(b)(i).

"<u>Delayed Draw Term Loan Commitment Percentage</u>" of any Lender shall mean the percentage set forth on <u>Schedule 1.2(b)</u> as its "Delayed Draw Term Loan Commitment Percentage," as the same may be adjusted upon any assignment by a Lender pursuant to <u>Section 15.3(c)</u> or <u>15.3(d)</u>.

"Delayed Draw Term Loan Funding Date" shall mean, with respect to the Second Amendment Delayed Draw Term Loan, the Second Amendment Date and with respect to the Fourth Amendment Delayed Draw Term Loan, the Fourth Amendment Date.

"Delayed Draw Term Loan Note" shall mean, collectively, the promissory notes described in Section 2.1(b).

"Depository Accounts" shall have the meaning set forth in Section 4.15(h)(i) hereof.

"<u>Disqualified Assignee</u>" shall mean any Person with respect to which (a) a voluntary or involuntary case (or comparable proceeding) has been commenced under any applicable state or federal bankruptcy, insolvency, receivership, reorganization or other debtor relief laws, (b) a custodian, conservator, receiver or similar official has been appointed for such Person or for any substantial part of such Person's assets (or any motion or request therefor has been filed with any

[Different first page setting changed from on in original to off in modified.].

Governmental Body), (c) such Person has made or agreed to make a general assignment for the benefit of creditors or is otherwise adjudicated as, or determined by any Governmental Body having regulatory authority over such Person or its assets to be, insolvent, bankrupt, or deficient in meeting any capital adequacy or liquidity standard of such Governmental Body, (d) such Person is a Defaulting Lender, or otherwise is unable to fulfill its obligations as a lender, under any credit facility to which it is a party (or has publicly announced that it believes that it will be any of the foregoing), or (e) which is a competitor of any Credit Party or its Subsidiaries.

"Disqualified Equity Interest" means any Equity Interest that, by its terms (or the terms of any security or other Equity Interests into which it is convertible or for which it is exchangeable), or upon the happening of any event or condition (a) matures or is mandatorily redeemable (other than solely for Equity Interests that are not Disqualified Equity Interests), pursuant to a sinking fund obligation or otherwise (except as a result of a change of control or asset sale so long as any rights of the holders thereof upon the occurrence of a change of control or asset sale event shall be subject to the prior repayment in full of the Advances and all other Obligations that are accrued and payable and the termination of this Agreement), (b) is redeemable at the option of the holder thereof, in whole or in part, (c) provides for scheduled payments of dividends in cash, or (d) is or becomes convertible into or exchangeable for debt or any other Equity Interests that would constitute Disqualified Equity Interests, in each case, prior to the date that is ninety one (91) days after the Term; provided that if such Equity Interests are issued pursuant to a plan for the benefit of employees of any Credit Party or any Subsidiary of a Credit Party or by any such plan to such employees, such Equity Interests shall not constitute Disqualified Equity Interests solely because they may be required to be repurchased by any Credit Party or any of its Subsidiaries in order to satisfy applicable statutory or regulatory obligations or as a result of such employee's termination, death or disability.

"Dollar" and the sign "<u>\$</u>" shall mean lawful money of the United States of America.

"Domestic Subsidiary" shall mean each Subsidiary of DDH that is not a Foreign Subsidiary.

"EBITDA Shortfall Amount" shall have the meaning set forth in Article X hereof.

"Eligible Assignee" shall mean (a)(i) a commercial bank, commercial lender, financial institution, or Affiliate of such entity that represents and warrants that it is not a Disqualified Assignee; provided that so long as no Event of Default has occurred and is continuing, such Person shall be approved by Borrowing Agent (which approval of Borrowing Agent shall not be unreasonably withheld, conditioned or delayed), or (ii) any other Person approved by Borrowing Agent (which approval of Borrowing Agent shall not be unreasonably withheld, conditioned or delayed) <u>provided</u> that, in each case, Borrowing Agent will be deemed to have consented to any such assignment in this clause (a) unless it objects thereto by written notice to Agent within ten (10) days after having received notice thereof, (b) any Lender or Affiliate (other than a natural Person) of a Lender or a Related Fund (other than a natural Person) of a Lender and (c) during

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

the continuation of an Event of Default under Sections 10.1, 10.3 (solely as a result of a failure to comply with Section 6.5) or 10.7, any other Person (other than a natural person).

"<u>Eligible Impact Services</u>Worker Solutions" shall mean Impact Services<u>Worker Solutions</u> and/or Impact<u>Worker Solutions</u> Providers, as applicable, acceptable to Agent in its Permitted Discretion as part of its mission and identified on Exhibit A attached hereto and incorporated herein, as such Exhibit may be amended or supplemented in writing from time to time by written notice from Agent to Borrowers or otherwise as requested by Borrowers and approved by Agent in Agent's sole discretion.

"<u>Enterprise Value</u>" shall mean (i) with respect to any Acquisition of Property or any Acquisition of 100% of the Equity Interests of a Person, the aggregate cash and non-cash consideration (including any assumption of Indebtedness, deferred purchase price and any equity consideration) paid for such Property or Equity Interests, and (ii) with respect to any Acquisition of less than 100% of the Equity Interests of a Person, the total enterprise value of the Person acquired, calculated based upon the aggregate cash and non-cash consideration (including any assumption of Indebtedness, deferred purchase price and any equity consideration) paid for the percentage of Equity Interests that were acquired in such Acquisition, as reasonably determined in good faith by Borrower and mutually agreed to by Agent.

"Environmental Claim" shall mean any known investigation, written notice of violation, written claim, action, suit, proceeding, written demand, abatement order or other written order or directive (conditional or otherwise), by any Person arising (a) pursuant to or in connection with any actual or alleged violation of any Environmental Law; (b) in connection with any Hazardous Material or any actual or alleged Hazardous Materials Activity; or (c) in connection with any actual or alleged damage, injury, threat or harm to human health, safety, natural resources or the environment.

"Environmental Complaint" shall have the meaning set forth in Section 4.19(c) hereof.

"Environmental Laws" shall mean any and all current or future federal or state (or any subdivision of either of them), statutes, ordinances, orders, rules, regulations, judgments, Governmental Authorizations, or any other written requirements of Governmental Bodies relating to (a) any Hazardous Materials Activity; (b) the generation, use, storage, transportation or disposal of Hazardous Materials; or (c) protection of human health and the environment from pollution, in each case, in any manner applicable to any Credit Party or any of its Subsidiaries or their respective Real Property.

"<u>Environmental Liability</u>" shall mean any liability, contingent or otherwise (including any liability for damages, costs of environmental remediation, fines, penalties or indemnities), of DDH, any other Credit Party or any of their respective Subsidiaries directly or indirectly resulting from or based upon (a) violation of any Environmental Law, (b) the generation, use, handling, transportation, storage, treatment or disposal of any Hazardous Materials, (c) exposure to any Hazardous Materials, (d) the Release or threatened Release of any Hazardous Materials

[Different first page setting changed from on in original to off in modified.].

into the environment or (e) any contract, agreement or other consensual arrangement pursuant to which DDH or any Subsidiary of DDH assumed liability with respect to any of the foregoing.

"<u>Equipment</u>" shall mean and include, with respect to any Person, all of such Person's goods (other than Inventory) whether now owned or hereafter acquired and wherever located including all equipment, machinery, apparatus, motor vehicles, fittings, furniture, furnishings, fixtures, parts, accessories and all replacements and substitutions therefor or accessions thereto.

"Equity Cure" shall have the meaning set forth in Article X hereof.

"Equity Interests" of any Person shall mean any and all shares, rights to purchase, options, warrants, general, limited or limited liability partnership interests, member interests, participation or other equivalents of or interest in (regardless of how designated) equity of such Person, whether voting or nonvoting, including common stock, preferred stock, convertible securities or any other "equity security" (as such term is defined in Rule 3a11-1 of the General Rules and Regulations promulgated by the SEC under the Exchange Act), including in each case all of the following rights relating to such Equity Interests, whether arising under the Governing Documents of the Person issuing such Equity Interests (the "Equity Issuer") or under the Applicable Laws of such Equity Issuer's jurisdiction of organization relating to the formation, existence and governance of corporations, limited liability companies or partnerships or business trusts or other legal entities, as the case may be: (i) all economic rights (including all rights to receive dividends and distributions) relating to such Equity Interests; (ii) all voting rights and rights to consent to any particular actions by the applicable Equity Issuer; (iii) all management rights with respect to such Equity Issuer; (iv) in the case of any Equity Interests consisting of a general partner interest in a partnership, all powers and rights as a general partner with respect to the management, operations and control of the business and affairs of the applicable Equity Issuer; (v) in the case of any Equity Interests consisting of the membership/limited liability company interests of a managing member in a limited liability company, all powers and rights as a managing member with respect to the management, operations and control of the business and affairs of the applicable Equity Issuer; (vi) all rights to designate or appoint or vote for or remove any officers, directors, managers, general partners or managing members of such Equity Issuer and/or any members of any board of members/managers/partners/directors that may at any time have any rights to manage and direct the business and affairs of the applicable Equity Issuer under its Governing Documents as in effect from time to time or under Applicable Law; (vii) all rights to amend the Governing Documents of such Equity Issuer, (viii) in the case of any Equity Interests in a partnership or limited liability company, the status of the holder of such Equity Interests as a "partner", general or limited, or "member" (as applicable) under the applicable Governing Documents and/or Applicable Law; and (ix) all certificates evidencing such Equity Interests.

"<u>Equity Transaction</u>" shall mean, with respect to DDH or any of its Subsidiaries, any issuance or sale by DDH or any such Subsidiary of shares of its Equity Interests, to include, without limitation, any capital contribution by such Person's existing members, partners or shareholders, as applicable, other than an issuance (a) to DDH Holdings or Permitted Holders,

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

(b) in connection with a conversion of debt securities to equity, (c) in connection with the exercise by a present or former employee, officer or director under a stock incentive plan, stock option plan or other equity-based compensation plan or arrangement, (d) which occurred on or prior to the Closing Date, (e) as consideration for an Acquisition permitted by Section 7.1(a), or (f) the proceeds of which are used to fund the purchase price consideration for a contemporaneous Acquisition permitted by Section 7.1(a) or capital expenditure permitted hereunder. For the avoidance of doubt, the equity line of credit established in connection with the registration statement filed on or before October 31, 2024 by DDH Holdings with the Securities and Exchange Commission shall not constitute Equity Transaction.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended from time to time, any successor statute, and the rules and regulations promulgated thereunder.

"ERISA Affiliate" shall mean, as applied to any Person, (a) any corporation which is a member of a controlled group of corporations within the meaning of Section 414(b) of the Code of which that Person is a member; (b) any trade or business (whether or not incorporated) which is a member of a group of trades or businesses under common control within the meaning of Section 414(c) of the Code of which that Person is a member; and (c) any member of an affiliated service group within the meaning of Section 414(m) or (o) of the Code for purposes of provisions relating to Section 412 of the Code of which that Person, any corporation described in clause (a) above or any trade or business described in clause (b) above is a member.

"ERISA Event" shall mean (a) a "reportable event" within the meaning of Section 4043 of ERISA and the regulations issued thereunder with respect to any Pension Plan (excluding those for which notice to the PBGC has been waived by regulation) that is reasonably likely to result in material liability to any Credit Party; (b) the failure to meet the minimum funding standard of Section 412 of the Code with respect to any Pension Plan (whether or not waived in accordance with Section 412(c) of the Code), the failure to make by its due date any minimum required contribution or any required installment under Section 430(j) of the Code with respect to any Pension Plan or the failure to make by its due date any required contribution to a Multiemployer Plan in either case if such failure is reasonably likely to result in liability to any Credit Party in excess of \$250,000 (net of third party indemnification); (c) the provision by the administrator of any Pension Plan pursuant to Section 4041(a)(2) of ERISA of a notice of intent to terminate such plan in a distress termination described in Section 4041(c) of ERISA; (d) the withdrawal from any Pension Plan with two (2) or more contributing sponsors or the termination of any such Pension Plan, in either case resulting in material liability pursuant to Section 4063 or 4064 of ERISA; (e) the institution by the PBGC of proceedings to terminate any Pension Plan, or the occurrence of any event or condition reasonably likely to constitute grounds under ERISA for the termination of, or the appointment of a trustee to administer, any Pension Plan; (f) the imposition of liability pursuant to Section 4062(e) or 4069 of ERISA or by reason of the application of Section 4212(c) of ERISA, in each case, that is reasonably likely to result in material liability to any Credit Party; (g) the withdrawal of any Credit Party, any of its Subsidiaries or any of their respective ERISA Affiliates in a complete or partial withdrawal (within the meaning of Sections 4203 and 4205 of ERISA) from any Mu

[Different first page setting changed from on in original to off in modified.].

withdrawal is reasonably likely to result in liability in excess of \$250,000 (net of third party indemnification), or the receipt by any Credit Party, any of its Subsidiaries or any of their respective ERISA Affiliates of notice from any Multiemployer Plan that it is in critical status or insolvency pursuant to Section 4241 or 4245 of ERISA, or that it is in "critical" or "endangered" status within the meaning of Section 103(f)(2)(G) of ERISA, or that it intends to terminate or has terminated under Section 4041A or 4042 of ERISA, if such critical status, insolvency or termination is reasonably likely to result in liability to any Credit Party in excess of \$250,000 (net of third party indemnification); (h) the imposition of fines, penalties, taxes or related charges under Chapter 43 of the Code or under Section 409, Section 502(c), (i) or (l), or Section 4071 of ERISA in respect of any Pension Plan if such fines, penalties, taxes or related charges are reasonably likely to result in material liability to any Credit Party; (i) the assertion of a claim (other than routine claims for benefits and funding obligations in the ordinary course) against any Pension Plan other than a Multiemployer Plan or the assets thereof, or against any Person in connection with any Pension Plan such Person sponsors or maintains which is reasonably likely to result in material liability to any Credit Party; (j) receipt from the Internal Revenue Service of a final written determination of the failure of any Pension Plan intended to be qualified under Section 401(a) of the Code; or (k) the imposition of a lien pursuant to Section 430(k) of the Code or pursuant to Section 303(k) or 4068 of ERISA in each case that is reasonably likely to result in material liability to any Credit Party.

"Erroneous Payment" shall have the meaning given to such term in Section 14.13(a) hereof.

"Erroneous Payment Notice" shall have the meaning given to such term in Section 14.13(b) hereof.

"Event of Default" shall have the meaning set forth in Article X hereof.

"Exchange Act" shall mean the Securities Exchange Act of 1934, as amended from time to time, and any successor statute.

"Excluded Deposit Accounts" means any deposit account (including, for the avoidance of doubt, any cash, cash equivalents or other property contained therein): (i) solely to the extent, and for so long as, such deposit account is pledged to secure obligations arising under Section 7.02(0) of the defined term "Permitted Encumbrances", and whether such pledge is by escrow or otherwise, in all cases with a balance no greater than such obligations under Section 7.02(0) of the defined term "Permitted Encumbrances"; (ii) used exclusively for payroll, payroll taxes and other employee wage and benefit payments with a balance no greater than such payroll, payroll taxes and other employee wage and benefit payments with a balance no greater than such payroll, payroll taxes and other employee wage and benefit payments with a balance no greater than such payroll, payroll taxes and other employee wage and benefit payments with a balance no greater than such payroll, payroll taxes and other employee wage and benefit payments with a balance no greater than such payroll, payroll taxes and other employee wage and benefit payments obligations that are to be paid within any two-week period; (iii) constituting a "zero balance" deposit account; or (iv) consisting of a disbursement account established with a payment processor to process vendor payments so long as the average monthly balance in such account (and in all such accounts in the aggregate) does not exceed \$250,000 at any one time.

"<u>Excluded Perfection Action</u>" shall mean any action with respect to assets otherwise constituting Collateral with respect to which Agent and Borrowing Agent shall have determined in their reasonable discretion and agree in writing that the costs or other consequences of such perfection action are excessive in relation to the benefit to the Lenders of the security intended to be afforded thereby.

"Excluded Property" shall have the meaning set forth in the definition of "Collateral" above.

"Excluded Taxes" shall mean any of the following Taxes on or with respect to a Payee, (a) Taxes imposed on or measured by net income (however denominated), franchise Taxes, and branch profits Taxes, in each case, imposed as a result of such recipient being organized under the laws of, or having its principal office or, in the case of any Payee, its applicable lending office located in, the jurisdiction imposing such Tax (or any political subdivision thereof), (b) in the case of a Payee, U.S. federal withholding Taxes imposed on amounts payable to or for the account of such Payee with respect to an applicable interest in an Advance or commitment pursuant to any Applicable Law in effect on the date on which (i) such Lender acquired such interest in the Advance or commitment (other than an assignee pursuant to a request by Borrowing Agent under Section 3.11) or (ii) such Lender changes its lending office, except in each case to the extent that, pursuant to Section 3.10, amounts with respect to such Taxes were payable either to such Lender's assignor immediately before such Lender became a party hereto or to such Lender immediately before it changed its lending office, (c) Taxes attributable to such Lender's failure to comply with Section 3.10(g)(i), or (d) any withholding Taxes imposed under FATCA.

"Executive Order No. 13224" shall mean the Executive Order No. 13224 on Terrorist Financing, effective September 24, 2001, as the same has been, or shall hereafter be, renewed, extended, amended or replaced.

"Extraordinary Receipts" shall mean any cash in excess of \$250,000 received by or paid to or for the account of DDH or any of its Subsidiaries not in the ordinary course of business (net of taxes paid or payable in connection with such receipts, including any applicable permitted Tax Distributions, if applicable), including, without limitation, under any indemnification provisions, all proceeds from releases of any escrowed amounts, any proceeds from insurance policies, all payments received in respect of any indemnification obligation and any applicable purchase price adjustments and in connection with income Tax refunds.

"<u>Family</u>" shall mean, with respect to any Person (a) such Person's spouse, (b) any immediate family member of such Persons, and (c) any other natural person who has been adopted by such Person.

"<u>FATCA</u>" shall mean Sections 1471 through 1474 of the Code, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with), any current or future regulations thereunder or official

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

interpretations thereof and any agreements entered into pursuant to Section 1471(b)(1) of the Code.

"<u>Fee Letter</u>" shall mean, collectively, (a) the fee letter dated the date hereof among the Borrowers and Agent, as the same may be amended, restated, supplemented or otherwise modified from time to time, (b) the fee letter dated as of the Second Amendment Date among the Borrowers and Agent, as the same may be amended, restated, supplemented or otherwise modified from time to time, and (c) the fee letter dated as of the Fourth Amendment Date among the Borrowers and Agent, as the same may be amended, restated, supplemented or otherwise modified from time to time.

"Fifth Amendment Closing Date" shall mean October 15, 2024.

"Financial Covenant Default" shall have the meaning set forth in Article X hereof.

"<u>Financial Officer Certification</u>" shall mean, with respect to the financial statements for which such certification is required, the certification of the chief financial officer of DDH that such financial statements fairly present, in all material respects, the financial condition of DDH and its Subsidiaries as of the dates indicated on a Consolidated Basis and the results of their operations and their cash flows for the periods indicated on a Consolidated Basis, subject to changes resulting from audit and normal year-end adjustments.

"Fiscal Quarter" shall mean a fiscal quarter of any Fiscal Year.

"Fiscal Year" shall mean the fiscal year of DDH and its Subsidiaries, as such end date may be adjusted in accordance with Section 7.13.

"<u>Flood Hazard Property</u>" shall mean any Real Property Collateral located in an area designated by the Federal Emergency Management Agency (and any successor Governmental Body performing a similar function) as having special flood or mud slide hazards.

## "Floor" shall mean one percent (1.00%) per annum.

"Foreign Subsidiary" shall mean any Subsidiary of DDH that is not organized or incorporated in the United States, any State thereof or the District of Columbia.

"Fourth Amendment" shall mean that certain Fourth Amendment to Term Loan and Security Agreement dated as of the Fourth Amendment Date among the Credit Parties, Agent and Lenders.

"Fourth Amendment Date" shall mean October 3, 2023.

"Fourth Amendment Delayed Draw Term Loan" shall have the meaning set forth in Section 2.1(b)(i).

"Fourth Amendment Delayed Draw Term Loan Amount" shall mean \$3,587,274.03.

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5 "<u>Funded Debt</u>" shall mean, as to any Person at a particular time, without duplication, all of the following, whether or not included as indebtedness or liabilities in accordance with GAAP:

(a) all obligations for borrowed money, whether current or long-term (including the Obligations hereunder, the ABL Obligations, any Subordinated Debt (other than the Preferred Equity) and all Capitalized Lease Obligations), all obligations evidenced by bonds, debentures, notes, loan agreements or other similar instruments;

(b) all obligations in respect of the deferred purchase price of property or services (other than trade accounts payable in the ordinary course of business and, in each case, not past due for more than sixty (60) days after the date on which such trade account payable is due), including, without limitation, any earn-out obligations recognized as a liability on the balance sheet of such Person and its Subsidiaries in accordance with GAAP;

(c) all non-contingent obligations under letters of credit (including standby and commercial), bankers' acceptances and similar instruments (including bank guaranties);

(d) the Attributable Principal Amount of Capital Leases, Synthetic Leases, Securitization Transactions and Sale and Leaseback Transactions;

(e) [reserved];

(f) all Funded Debt of others secured by (or for which the holder of such Funded Debt has an existing right, contingent or otherwise, to be secured by) any Lien on, or payable out of the proceeds of production from, property owned or acquired by such Person, whether or not the obligations secured thereby have been assumed;

(g) all Guarantees in respect of Funded Debt of another Person; and

(h) Funded Debt of any partnership or joint venture or other similar entity in which such Person is a general partner or joint venturer, and, as such, has personal liability for such obligations, but only to the extent there is recourse to such Person for payment thereof.

For purposes hereof, the amount of Funded Debt shall (I) be determined (x) based on the outstanding principal amount in the case of borrowed money indebtedness under <u>clause (a)</u> and purchase money indebtedness and the deferred purchase obligations under <u>clause (b)</u>, (y) without duplication for any letter of credit sublimit within a loan facility commitment amount, based on the maximum amount available to be drawn in the case of letter of credit obligations and the other obligations under <u>clause (c)</u>, and (z) based on the amount of Funded Debt that is the subject of the Guarantees and for which there is recourse to such Person in the case of Guarantees under <u>clause (g)</u> and (II) not be reduced by any origination fees related thereto.

"Funding Account" shall have the meaning set forth in Section 4.15(h)(iii).

"Funds Flow Statement" shall mean the funds flow statement delivered to Agent as of the Closing Date with respect to the Closing Date Transactions.

"GAAP" shall mean, subject to the limitations on the application thereof set forth in Section 1.1, generally accepted accounting principles in the United States of America in effect from time to time.

"General Intangibles" shall mean and include, with respect to any Person, all of such Person's general intangibles, whether now owned or hereafter acquired, including all payment intangibles, all choses in action, causes of action, corporate or other business records, inventions, designs, patents, patent applications, equipment formulations, manufacturing procedures, quality control procedures, trademarks, trademark applications, service marks, trade secrets, goodwill, copyrights, design rights, software, computer information, source codes, codes, records and updates, registrations, licenses, franchises, customer lists, tax refunds, tax refund claims, computer programs, all claims under guaranties, security interests or other security held by or granted to such Person to secure payment of any of the Receivables (other than to the extent covered by Receivables), all rights of indemnification and all other intangible property of every kind and nature (other than Receivables).

"<u>Governing Documents</u>" shall mean, for a corporation, its articles of incorporation or certificate of incorporation and its bylaws, for a limited liability company, its articles of organization or certificate of formation and its operating agreement, and for a partnership, its certificate of formation and its partnership agreement.

"Governmental Authorization" shall mean any permit, license, authorization, plan, directive, consent order or consent decree of or from any Governmental Body.

"<u>Governmental Body</u>" shall mean any nation or government, any state or other political subdivision thereof or any entity, authority, agency, division or department exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to a government (including any supra-national bodies such as the European Union or the European Central Bank) and any group or body charged with setting financial accounting or regulatory capital rules or standards (including, without limitation, the Financial Accounting Standards Board, the Bank for International Settlements or the Basel Committee on Banking Supervision or any successor or similar authority to any of the foregoing).

"<u>Guarantee</u>" shall mean, as to any Person, (a) any obligation, contingent or otherwise, of such Person guaranteeing or having the economic effect of guaranteeing any Indebtedness or other obligation payable or performable by another Person (the "primary obligor") in any manner (including as a result of joint and several liability with such primary obligor), whether directly or indirectly, and including any obligation of such Person, direct or indirect, (i) to purchase or pay (or advance or supply funds for the purchase or payment of) such Indebtedness or other obligation, (ii) to purchase or lease property, securities or services for the purpose of assuring the obligee in respect of such Indebtedness or other obligation of the payment or performance of

such Indebtedness or other obligation, (iii) to maintain working capital, equity capital or any other financial statement condition or liquidity or level of income or cash flow of the primary obligor so as to enable the primary obligor to pay such Indebtedness or other obligation, or (iv) entered into for the purpose of assuring in any other manner the obligee in respect of such Indebtedness or other obligation of the payment or performance thereof or to protect such obligee against loss in respect thereof (in whole or in part), or (b) any Lien on any assets of such Person securing any Indebtedness or other obligation of any other Person, whether or not such Indebtedness or other obligation is assumed by such Person (or any right, contingent or otherwise, of any holder of such Indebtedness to obtain any such Lien). The amount of any Guarantee shall be deemed to be an amount equal to the stated or determinable amount of the related primary obligation, or portion thereof, in respect of which such Guarantee is made or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof as determined by the guaranteeing Person in good faith. The term "Guarantee" as a verb has a corresponding meaning.

"Guaranteed Obligations" shall have the meaning set forth in Section 16.1 hereof.

"<u>Guarantor</u>" shall mean (i) Orange142, LLC, (ii) Huddled Masses LLC, (iii) Colossus Media, LLC, (iv) subject to <u>Section 6.16</u> hereof, DDH Holdings, (v) each Person who subsequent to the Closing Date becomes a party to this Agreement as a Guarantor, and (vi) each other Person who may hereafter guarantee payment or performance of the whole or any part of the Obligations and "<u>Guarantors</u>" shall mean collectively all such Persons.

"<u>Guarantor Security Agreement</u>" shall mean, individually and collectively, any Security Agreement, or joinder thereto, executed by any Guarantor in favor of Agent securing the Guarantee of such Guarantor or otherwise securing the Obligations, in form and substance satisfactory to Agent.

"Hazardous Discharge" shall have the meaning set forth in Section 4.19(c) hereof.

"<u>Hazardous Materials</u>" shall mean any hazardous substances defined by CERCLA, including any hazardous waste as defined under 40 C.F.R. Parts 260-270, gasoline or petroleum (including crude oil or any fraction thereof), asbestos or polychlorinated biphenyls.

"<u>Hazardous Materials Activity</u>" shall mean any past, current, proposed or threatened activity, event or occurrence involving any Hazardous Materials, including the use, manufacture, possession, storage, holding, presence, existence, location, Release, threatened Release, discharge, placement, generation, transportation, processing, construction, treatment, abatement, removal, remediation, disposal, disposition or handling of any Hazardous Materials, and any corrective action or response action with respect to any of the foregoing.

"Immaterial Subsidiary" shall mean any direct or indirect Subsidiary of DDH Holdings which (a) is not actively engaged in any business activities, and (b) maintains assets of not more \$250,000 in the aggregate at any time. In no event shall the aggregate amount of Consolidated EBITDA of all Immaterial Subsidiaries for any fiscal quarter and consolidated assets of all

[Different first page setting changed from on in original to off in modified.].

[Different first page setting changed from on in original to off in modified.].

Immaterial Subsidiaries exceed 5.0% of Consolidated EBITDA for such fiscal quarter or the consolidated total assets of DDH Holdings and its Subsidiaries, respectively.

"Impact Certificate" shall mean a certificate in substantially the form attached hereto as Exhibit D to be signed by an Authorized Officer of Borrowers.

"Impact Provider" shall mean any third-party provider or other third party vendor providing Impact Services.

"Impact Services" shall mean programs or services intended to provide to companies and their employees, as applicable, long-term, sustainable impact in financial inclusion and empowerment, housing, jobs, wellness, diversity and inclusion or such other core values acceptable to Agent in its sole discretion.

"<u>Impact Subscription</u>" shall mean the subscription to, participation in or other purchase or acquisition of one or more (as applicable) Impact Services by Borrowers (or any of them, as applicable) for any period identified and offered by an Impact Provider to Borrowers. "<u>Impact Subscribe</u>" and "<u>Impact Subscribe</u>" shall have the corollary meaning thereto.

"Inchoate Obligations" shall mean contingent indemnification or expense reimbursement Obligations other than those related to claims, causes of action, or liabilities that have been asserted or threatened in writing or that otherwise can be reasonably identified by Agent or any Lender based on the then-known facts and circumstances.

"Indebtedness" shall mean, as to any Person, at any time, any and all indebtedness, obligations or liabilities (whether matured or unmatured, liquidated or unliquidated, direct or indirect, absolute or contingent, or joint or several) of such Person for or in respect of all of the following, whether or not included as indebtedness or liabilities in accordance with GAAP (without duplication):

- (a) all Funded Debt;
- (b) net obligations under any Swap Agreement;
- (c) all Guarantees in respect of Indebtedness of another Person; and

(d) all Indebtedness of the types referred to in <u>clauses (a)</u> through (c) above of any partnership or joint venture (other than a joint venture that is itself a corporation or limited liability company) in which any Borrower or a Subsidiary is a general partner or joint venturer, unless such Indebtedness is non-recourse to such Borrower or Subsidiary and excluded as Indebtedness of such Borrower or Subsidiary for purposes of GAAP.

For purposes hereof, the amount of Indebtedness shall be determined based on Swap Termination Value in the case of net obligations under any Swap Agreement under <u>clause (b)</u>.

"Indemnified Taxes" shall mean (a) Taxes other than Excluded Taxes, imposed on or with respect to any payment made by or on account of any obligations of any Credit Party under this Agreement or any Other Document, and (b) to the extent not otherwise described in (a), Other Taxes.

"Intellectual Property" shall mean property constituting under any Applicable Law a patent, patent application, copyright, trademark, service mark, trade name, mask work, trade secret or license or other right to use any of the foregoing.

"Intercreditor Agreement" shall mean that certain Intercreditor Agreement dated as of January 9July 7, 2023 by and between Agent and ABL Lender, as the same may be amended, restated, supplemented or otherwise modified from time to time.

"Interest Period" shall mean the period provided for any SOFR Loan, which shall be one (1) month or three (3) months, as applicable.

"Interest Rate Reset Period" shall mean; (a) with respect to any Term Loan, (a) initially, the period commencing on the date such Term Loan is made and ending on the last day of the fiscal quarter immediately succeeding the date such Term Loan is made the interest period(s) in effect immediately prior to the Fifth Amendment Closing Date, November 29, 2024, and (b) thereafter, (1) with respect to a one (1) month Interest Period, the period commencing on the date immediately after the end of the previous Interest Period and ending on the earlier of (i) the last day of the fiscal quarter then ending and (ii) the day before the last day of the Term, first calendar day of the month during which an Advance or any portion thereof is continued and ending on the last Business Day of such month and (2) with respect to a three (3) month Interest Period, the period commencing on the first calendar day of the month during which an Advance or any portion thereof is continued and ending on the last Business Day of such month and (2) with respect to a three (3) month Interest Period, the period commencing on the first calendar day of the month during which an Advance or any portion thereof is continued and ending on the last Business Day of the third month during such three-month Interest Period; provided, that if any Interest Rate Reset Period would end on a day other than a Business Day, such Interest Rate Reset Period shall be extended to the next succeeding Business Day unless such next succeeding Business Day. Borrowing Agent shall elect the duration of each succeeding Interest Period by giving irrevocable written notice to Agent of such duration not later than 3:00 p.m. on the day which is three (3) Business Days prior to the last day of the then current Interest Period applicable to such Term SOFR Rate Loan. If Agent does not receive timely notice of the Interest Period elected by Borrowing Agent, Borrowing Agent shall be deemed to have elected a one (1) month Interest Period.

"Intermediate Holdco" shall have the meaning set forth in Section 6.12 hereof.

"Inventory" shall mean and include, with respect to any Person, all of such Person's now owned or hereafter acquired inventory, goods, merchandise and other personal property, wherever located, to be furnished under any consignment arrangement, contract of service or held for sale or lease, all raw materials, work in process, finished goods and materials and supplies of any kind, nature or description which are or might be used or consumed in such

[Different first page setting changed from on in original to off in modified.].

Person's business or used in selling or furnishing such goods, merchandise and other personal property, and all documents of title or other documents representing them.

"Investment" shall mean, as to any Person, any direct or indirect acquisition or investment by such Person, whether by means of (a) the purchase or other acquisition of Equity Interests of another Person, (b) a loan, advance or capital contribution to, Guarantee or assumption of debt of, or purchase or other acquisition of any other debt or equity participation or interest in, another Person, including any partnership or joint venture interest in such other Person and any arrangement pursuant to which the investor Guarantees Indebtedness of such other Person, or (c) an Acquisition. For purposes of covenant compliance, the amount of any Investment shall be the amount actually invested, without adjustment for subsequent increases or decreases in the value of such Investment.

"Investment Property" shall mean and include, with respect to any Person, all of such Person's now owned or hereafter acquired securities (whether certificated or uncertificated), securities entitlements, securities accounts, commodities contracts and commodities accounts.

"Involuntary Disposition" shall mean the receipt by DDH or any of its Subsidiaries of any cash insurance proceeds or condemnation awards payable by reason of theft, loss, physical destruction or damage, taking or similar event with respect to any of its Property in excess of \$100,000.

"Key Executive Policy" shall have the meaning set forth in Section 4.21.

"Key Performance Indicators" shall mean the key performance indicators set forth on Exhibit E attached hereto and such other key performance indicators as Borrowers and Agent shall agree from time to time in replacement of (provided that such replacement shall provide a reasonably similar metric) or in addition to any of the foregoing; provided, however, that any foregoing key performance indicators may be removed or modified by Borrowers with the consent of Agent (such approval not to be unreasonably withheld, delayed or conditioned); provided, further, that all foregoing key performance indicators shall be developed and disclosed to Agent within three months of the Closing Date.

"Lafayette Square" shall have the meaning set forth in the preamble to this Agreement and shall extend to all of its successors and permitted assigns.

"Leasehold Interests" shall mean each Credit Party's right, title and interest in and to the leased Real Property.

"Lender" and "Lenders" shall have the meaning ascribed to such term in the preamble to this Agreement and shall include each Person which becomes a permitted transferee, successor or permitted assign of any Lender.

"<u>LIBOR Rate</u>" shall mean, for any Interest Period, the greater of (i) the rate per annum determined by Agent by dividing (a) the Published Rate as of 11:00 a.m. (London time) two (2)

Business Days prior to the first day of the applicable Interest Period by (b) a number equal to 1.00 minus the percentage prescribed by the Federal Reserve for determining the maximum reserve requirements with respect to any euroeurrency funding by banks on the date of determination and (ii) one percent (1.00%).

"Lien" shall mean (a) any mortgage, deed of trust, pledge, hypothecation, collateral assignment, security interest, lien (whether statutory or otherwise), charge, claim or encumbrance, or preference, priority or other security agreement or preferential arrangement held or asserted in respect of any asset of any kind or nature whatsoever including any conditional sale or other title retention agreement, any lease having substantially the same economic effect as any of the foregoing, and the filing of, or agreement to give, any financing statement under the Uniform Commercial Code or comparable law of any jurisdiction and (b) in the case of Securities, any purchase option, call or similar right of a third party with respect to such Securities.

"Lien Waiver Agreement" shall mean an agreement which is executed in favor of Agent, in form and content satisfactory to Agent in its Permitted Discretion, by a Person who owns or occupies premises at which any Collateral may be located from time to time and by which such Person shall waive or subordinate any Lien that such Person may ever have with respect to any of the Collateral and shall authorize Agent from time to time to enter upon the premises to inspect or remove the Collateral from such premises or to use such premises to store or dispose of such Collateral.

"<u>Liquidity</u>" shall mean, as of any date of determination, the sum of (a) Unrestricted Cash plus (b) Revolving Credit Availability (as defined in the ABL Credit Agreement as in effect on the date hereof Fifth Amendment Closing Date).

## "Liquidity Shortfall Amount" shall have the meaning set forth in Article X hereof.

"<u>LLC Division</u>" shall mean, in the event a Credit Party is a limited liability company, (a) the division of any such Borrower or Guarantor into two or more newly formed limited liability companies (whether or not such Borrower or Guarantor is a surviving entity following any such division) pursuant to Section 18-217 of the Delaware Limited Liability Company Act or any similar provision under any similar act governing limited liability companies organized under the laws of any other State or Commonwealth or of the District of Columbia, or (b) the adoption of a plan contemplating, or the filing of any certificate with any applicable Governmental Body that results or may result in, any such division.

"<u>Management Fee Subordination Agreement</u>" shall mean that certain Management Fee Subordination Agreement, dated as of the date hereof, by and among Mark Walker, Keith Smith, the Credit Parties, and Agent, as the same may be amended, restated, amended and restated, supplemented or otherwise modified from time to time, which agreement shall terminate upon the occurrence of a Qualified IPO.

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

"<u>Management Fees</u>" shall mean management fees or advisory fees required to be paid to the Keith Smith or Mark Walker pursuant to those certain Board Services and Consulting Agreements each dated as of September 30, 2020, by and between DDH, on the one hand, and Keith Smith and Mark Walker on the other hand, together with any other similar fees pursuant to any management agreement or management services agreement entered into now or hereafter by and between Keith Smith or Mark Walker and a Credit Party.

"Margin Stock" shall have the meaning as defined in Regulation U of the Board of Governors of the Federal Reserve System as in effect from time to time.

"<u>Material Adverse Effect</u>" shall mean any effect, event, condition, action, omission, change or state of facts that, individually or in the aggregate, has resulted in, or could reasonably be expected to result in, a material adverse effect with respect to (a) the business operations, properties, assets, or financial condition of DDH and its Subsidiaries; (b) the ability of the Credit Parties to fully and timely perform the Obligations; (c) the legality, validity, binding effect, or enforceability against an Credit Party of this Agreement or any Other Document to which it is a party; (d) the priority of Liens in the whole of the Collateral in favor of Agent; or (e) the material rights, remedies and benefits available to, or conferred upon Agent under this Agreement or any Other Document.

"<u>Material Contract</u>" shall mean any contract or other arrangement (other than the Other Documents) to which any Credit Party or any Subsidiary of any Credit Party is a party as to which the breach, nonperformance, cancellation or failure to renew by any party thereto could reasonably be expected to have a Material Adverse Effect.

"Modified Transfer Supplement" shall have the meaning set forth in Section 15.3(d).

"Moody's" shall mean Moody's Investor Services, Inc., together with its successors.

"<u>Mortgage</u>" shall mean, with respect to any Real Property Collateral, any mortgage, deed of trust or deed to secure debt that purports to grant to Agent a Lien on such Real Property or any interest (including with respect to any improvements and fixtures) of any Credit Party in such Real Property.

"<u>Multiemployer Plan</u>" shall mean a "multiemployer plan" as defined in Sections 3(37) of ERISA which is sponsored, maintained or contributed to by, or required to be contributed to by, any Credit Party or any of its ERISA Affiliates or with respect to which any Credit Party or any of its ERISA Affiliates previously sponsored, maintained or contributed to or was required to contribute to, and still has liability.

"<u>Net Cash Proceeds</u>" shall mean the aggregate proceeds paid in cash or Cash Equivalents received by DDH or any of its Subsidiaries in connection with any Asset Sale, Debt Transaction, Involuntary Disposition, Equity Transaction or Extraordinary Receipts, net of (a) direct costs incurred or estimated costs for which reserves are maintained, in connection therewith (including reasonable legal, accounting and investment banking fees and expenses, sales commissions and

underwriting discounts); (b) estimated taxes paid or payable (including sales, use or other transactional taxes and any net marginal increase in income taxes) directly, or indirectly, by DDH or any of its Subsidiaries, including any such estimated taxes paid or payable by any holder of the Equity Interest issued by DDH (including, DDH Holdings), in each case as a result thereof; and (c) the amount required to retire any Indebtedness secured by a Permitted Encumbrance (excluding any Permitted Encumbrance that is pari passu or junior to the Agent's Lien on or security interest in the Collateral) on the related property (in all cases paid to Persons other than Affiliates of a Credit Party). For purposes hereof, "Net Cash Proceeds" includes any cash or Cash Equivalents received upon the disposition of any non-cash consideration received by DDH or any of its Subsidiaries in any Asset Sale, Debt Transaction, Involuntary Disposition, Equity Transaction or Extraordinary Receipts.

"Non-Defaulting Lender" shall have the meaning set forth in Section 2.9(a).

"Notes" shall mean Term Notes.

"Obligations" shall mean and include, in each case to the extent arising under or in connection with this Agreement and/or the Other Documents: any and all loans, advances, debts, liabilities, obligations, covenants and duties owing by the Credit Parties (or any of them) to Lenders or Agent of any kind or nature, present or future (including any interest or other amounts accruing thereon after maturity, or after the filing of any petition in bankruptcy, or the commencement of any insolvency, reorganization or like proceeding relating to any Credit Party, whether or not a claim for post-filing or post-petition interest or other amounts is allowed in such proceeding), whether or not evidenced by any note, guaranty or other instrument, whether arising under any agreement, instrument or document (including this Agreement and the Other Documents) whether or not for the payment of money, whether arising by reason of an extension of credit, opening of a letter of credit, loan, equipment lease or guarantee, under any interest or currency swap, future, option or other similar agreement, or in any other manner, whether arising out of overdrafts or deposit or other accounts or electronic funds transfers (whether through automated clearing houses or otherwise) or out of Agent's or any Lenders non-receipt of or inability to collect funds or otherwise not being made whole in connection with depository transfer check or other similar arrangements, whether direct or indirect (including those acquired by assignment or participation), absolute or contingent, joint or several, due or to become due, now existing or hereafter arising, contractual or tortious, liquidated or unliquidated, regardless of how such indebtedness or liabilities arise or by what agreement or instrument they may be evidenced or whether evidenced by any agreement or instrument. including, but not limited to, any and all of any Credit Parties' Indebtedness and/or liabilities under this Agreement, the Other Documents or under any other agreement between Agent or Lenders and any Credit Party and any amendments, extensions, renewals or increases and all costs and expenses of Agent and any Lender incurred in the documentation, negotiation, modification, enforcement, collection or otherwise in connection with any of the foregoing, including but not limited to reasonable, documented (to the extent readily available) attorneys' fees, allocated costs of internal counsel and out-of-pocket expenses and all obligations of the Credit Parties (or any of them) to Agent or Lenders to perform acts or refrain from taking any action.

[Different first page setting changed from on in original to off in modified.].

"OFAC" shall mean the U.S. Department of the Treasury's Office of Foreign Assets Control.

"Orange142 Canada" shall mean Orange142 Advertising Canada, Inc., a corporation organized under the laws of British Columbia.

"<u>Other Connection Taxes</u>" shall mean, with respect to any Payee, Taxes imposed as a result of a present or former connection between such Payee and the jurisdiction imposing such Tax (other than connections arising from such Payee having executed, delivered, become a party to, performed its obligations under, received payments under, received or perfected a security interest under, engaged in any other transaction pursuant to or enforced this Agreement or any Other Document, or sold or assigned an interest in any Obligation or Other Document).

"Other Documents" shall mean the Notes, the Fee Letter, all related Guarantees, all Security Agreements, all Guarantor Security Agreements, all Pledge Agreements, the Intercreditor Agreement, the Management Fee Subordination Agreement, any subordination agreement relating to Subordinated Debt, if applicable, any Swap Agreement with any Lender or Affiliate thereof, any Lien Waiver Agreements, Mortgages (if any), environmental indemnities with respect to Real Property, and any and all other agreements, instruments and documents, including guaranties, pledges, powers of attorney, consents, interest or currency swap agreements or other similar agreements and all other writings heretofore, now or hereafter executed by any Credit Party and/or delivered to Agent or any Lender in respect of the transactions contemplated by this Agreement; provided that "Other Documents" shall not include any document evidencing any rights of any Lender as a holder of Equity Interests of DDH or DDH Holdings.

"Other Taxes" shall mean all present or future stamp, court or documentary, intangible, recording, filing, other excise or property Taxes or similar Taxes, charges or similar levies that arise from any payment made under, from the execution, delivery, performance, enforcement or registration of, from the receipt or perfection of a security interest under, or otherwise with respect to, this Agreement or any Other Document, except any such Taxes that are Other Connection Taxes imposed with respect to an assignment (other than an assignment made pursuant to Section 3.11).

"Participant" shall have the meaning set forth in Section 15.3(b) hereof.

"Payee" shall mean Agent, any Lender, assignee of any Lender or Participant.

"<u>Payment Conditions</u>" shall mean that prior to and after giving effect to the relevant dividend, distribution, acquisition, investment or applicable transaction as to which the satisfaction of the Payment Conditions is being determined, each of the following conditions is satisfied: (i) no Default or Event of Default is continuing or would occur after giving Pro Forma effect to such action, (ii) DDH and its Subsidiaries shall, on a Pro Forma Basis, have a Consolidated Senior Net Leverage Ratio of not greater than 1.5 to 1.00, and (iii) DDH and its Subsidiaries shall, on a Pro Forma Basis, have Liquidity of not less than \$15,000,000.

[Different first page setting changed from on in original to off in modified.].

"PBGC" shall mean the Pension Benefit Guaranty Corporation established pursuant to Subtitle A of Title IV of ERISA or any successor thereto.

"<u>Pension Plan</u>" shall mean any "employee pension benefit plan" as defined in Section 3(2) of ERISA other than a Multiemployer Plan, which is subject to Section 412 of the Code or Section 302 of ERISA and which is sponsored, maintained or contributed to by, or required to be contributed to by, any Credit Party or any of its ERISA Affiliates or with respect to which any Credit Party or any of its ERISA Affiliates previously sponsored, maintained or contributed to, or was required to contribute to, and still has liability.

"Periodic Term SOFR Determination Day" shall have the meaning set forth in the definition of "Term SOFR".

"<u>Permitted Acquisition</u>" shall mean any Acquisition by a Credit Party that satisfies the following conditions:

(a) the Property acquired (or the Property of the Person acquired) in such Acquisition is a business or is used or useful in a business permitted under <u>Section 7.9</u> and constitutes all or substantially all of the assets (or all or substantially all of the assets constituting a business unit, division, product line, or line of business of any Person);

(f) with respect to the acquisition of Equity Interests, the target shall have a positive EBITDA, calculated in accordance with GAAP for the trailing twelve month period immediately prior to such acquisition,

(g) in the case of an Acquisition of Equity Interests, (i) the board of directors (or other comparable governing body) of such other Person shall have approved the Acquisition, (ii) such Person shall be organized and existing under the laws of any state of the United States or the District of Columbia and (iii) all of the Equity Interests of such Person are being acquired;

(h) the total costs and liabilities (including without limitation, all assumed liabilities, all earn-out payments, deferred payments and the value of any other stock or assets transferred, assigned or encumbered with respect to such acquisitions) each such acquisition consummated during the Term do not exceed, as of the date of determination, 40% of the Credit Parties' Consolidated EBITDA calculated on a Pro Forma Basis;

(i) after giving effect to such Acquisition (including the incurrence of any Indebtedness and Delayed Draw Term Loans incurred in connection therewith), DDH and its Subsidiaries (or DDH Holdings and its Subsidiaries after a Qualified IPO) shall, on a Pro Forma Basis, have a Consolidated Senior Net Leverage Ratio of not greater than 1.5 to 1.00;

(j) (i) no Default or Event of Default shall exist and be continuing immediately before or immediately after giving effect thereto, (ii) the representations and warranties made by each of the Credit Parties in this Agreement and each Other Document shall be true and correct in all material respects as if made on the date of such Acquisition (after

[Different first page setting changed from on in original to off in modified.].

giving effect thereto) except to the extent such representations and warranties expressly relate to an earlier date, in which case they shall be true and correct in all material respects as of such earlier date (or, in each case, in all respects with respect to any representation or warranty which, by its terms is expressly qualified by materiality, Material Adverse Effect or dollar amount thresholds), and (iii) the Borrowers shall have given Agent written notice of such Acquisition with reasonable detail thereof, and all business, financial/accounting and legal due diligence and documentation with respect thereto (including any quality of earnings reports obtained by the Credit Parties with respect thereto), in each case, no later than thirty (30) days prior to the consummation thereof (or such shorter notice as may be acceptable to Agent in its sole discretion); and

(k) the Property acquired in such Acquisition shall constitute Collateral, and the Person acquired in such Acquisition shall be a Credit Party.

"<u>Permitted Discretion</u>" shall mean a determination made in the exercise of reasonable (from the perspective of a secured creditor) business judgment.

"<u>Permitted Dividends</u>" shall mean (a) dividends and/or distributions required to be paid to the holders of the Preferred Equity (as set forth in the Governing Documents of DDH in effect on the date hereof) so long as (i) the Consolidated Fixed Charge Coverage Ratio, calculated on a Pro Forma Basis, would not be less than 1.5 to 1.00 after giving effect thereto, (ii) no Default or Event of Default is continuing or would occur after giving Pro Forma effect to such dividend, and (iii) expressly permitted under the terms of the Preferred Equity Subordination Agreement and (b) after consummation of a Qualified IPO, dividends and distributions to DDH Holdings and any shareholders of DDH Holdings so long as the Payment Conditions are satisfied.

"Permitted Encumbrances" shall mean each of the Liens permitted pursuant to Section 7.2.

"Permitted Holders" shall mean Keith Smith, Mark Walker, Leah Woolford and/or their respective Related Persons.

"<u>Permitted Lines of Business</u>" shall mean any business engaged in by any Credit Party or its Subsidiaries as of the Closing Date and businesses that are substantially similar, related or incidental thereto, and, the offering of products and services that are complementary or ancillary thereto, as may be adjusted from time to time.

"<u>Permitted Management Fee Payments</u>" shall mean at any time prior to the consummation of a Qualified IPO, the payment of regularly scheduled Management Fees in accordance with the Board Services and Consulting Agreements each dated as of September 30, 2020 (as in effect on the date hereof) so long as at the time of and after giving pro forma effect thereto (calculated with respect to the financial covenants set forth in Section 6.5 as of the last day of the fiscal quarter for which Borrowers were required to deliver financial statements pursuant to Section 9.8 hereof as if such payment was made on the last day of such fiscal quarter

[Different first page setting changed from on in original to off in modified.].

and without giving effect to the addback set forth in clause (v) of the definition Consolidated EBITDA) no Default or Event of Default has occurred or would occur.

"Person" shall mean any individual, sole proprietorship, partnership, corporation, business trust, joint stock company, trust, unincorporated organization, association, limited liability company, limited liability partnership, institution, public benefit corporation, joint venture, entity or Governmental Body (whether federal, state, county, city, municipal or otherwise, including any instrumentality, division, agency, body or department thereof).

"Personal Information" shall mean, in addition to any definition for any similar term (e.g., "personal data," "personally identifiable information," or "PII") provided by applicable law or by a Credit Party or Subsidiary in any of its privacy policies, notices or contracts, all information that identifies, relates to, describes, is reasonably capable of being associated with, or could reasonably be linked, directly or indirectly, with a particular individual or household, including (a) name, physical address, telephone number, email address, financial information, financial account number or government-issued identifier (including Social Security number and driver's license number), medical, health or insurance information, gender, date of birth, educational or employment information, religious or political views or affiliations, marital or other status, and any other data used or intended to be used to identify, contact or precisely locate an individual (e.g., geolocation data), (b) any data regarding an individual's activities online or on a mobile device or application, and (c) Internet Protocol addresses and unique device identifiers. Personal Information may relate to any individual, including a current, prospective or former customer, end user or employee of any Person and includes information in any form or media, whether paper, electronic, or otherwise.

"<u>Pledge Agreements</u>" shall mean that certain Pledge Agreement, dated as of the Closing Date, by DDH pursuant to which DDH pledges all of the Equity Interests owned by DDH in its Subsidiaries and any other pledge agreement executed and delivered by any other Person subsequent to the Closing Date to secure the Obligations.

"Preferred A Redemption" shall mean the redemption of the preferred A units held in DDH pursuant to the Redemption Agreement.

"Preferred B Redemption" shall mean the contemplated redemption of the preferred B units held in DDH pursuant to the Redemption Agreement.

"<u>Preferred Equity</u>" shall mean the preferred B units issued to USDM Holdings, Inc., a Texas corporation and other "Preferred Unit Holders" pursuant to the Amended and Restated Limited Liability Company Agreement of DDH dated September 30, 2020.

"<u>Preferred Equity Subordination Agreement</u>" shall mean that certain Preferred Equity Subordination Agreement of even date herewith, by and among DDH, the holder(s) of the Preferred Equity and Agent, as amended, restated, supplemented or otherwise modified from time to time, which agreement shall terminate upon the occurrence of a Qualified IPO and shall expressly permit the transactions contemplated by a Qualified IPO.

[Different first page setting changed from on in original to off in modified.].

"Privacy and Security Laws" shall mean any and all applicable laws, legal requirements, executive orders, regulations and self-regulatory guidelines (including of any applicable foreign jurisdiction) relating to the receipt, collection, compilation, use, storage, processing, sharing, safeguarding, security (technical, physical and administrative), disposal, destruction, disclosure or transfer (including cross-border) of any Personal Information, including the Federal Trade Commission Act, the California Consumer Privacy Act (CCPA), the California Privacy Rights Act (CPRA), Virginia's Consumer Data Protection Act (CDPA), the Colorado Privacy Act (CPA), Massachusetts General Law Chapter 93H and its implementing regulations 201 CMR 17.00, New York's Stop Hacks and Improve Electronic Data Security Act (SHIELD Act), the Payment Card Industry Data Security Standard (PCI-DSS), the CAN-SPAM Act, the Telephone Consumer Protection Act (TCPA), the EU General Data Protection Regulation, Regulation 2016/679/EU (GDPR), and any and all applicable federal, state or global laws relating to the privacy and security of Personal Information, and any laws relating to the use of biometric identifiers.

"Privacy Laws" shall have the meaning ascribed to such term in the definition of HIPAA.

"Pro Forma Balance Sheet" shall have the meaning set forth in Section 5.9(c) hereof.

"Pro Forma Basis" shall mean, for purposes of calculating any financial covenants set forth in Section 6.5 or set forth in any other provision hereof (including for purposes of calculating compliance with clause (d) of the definition of Permitted Acquisition), that any Asset Sale, Involuntary Disposition, Acquisition, incurrence of Indebtedness (including the incurrence of any Delayed Draw Term Loan) or Restricted Payment shall be deemed to have occurred as of the first day of the most recent four (4) Fiscal Quarter period ended prior to the date of such transaction for which Borrowers were required to deliver financial statements pursuant to Sections 9.7 or 9.8. In connection with the foregoing, (a) (i) with respect to any Asset Sale or Involuntary Disposition, income statement and cash flow statement items (whether positive or negative) attributable to the property disposed of shall be excluded to the extent relating to any period occurring prior to the date of such transaction and (ii) with respect to any Acquisition, income statement items attributable to the Person or property that is the subject of such transaction shall be included to the extent relating to any period applicable in such calculations to the extent (A) such items are not otherwise included in such income statement items for DDH (or DDH Holdings, if applicable) and its Subsidiaries in accordance with GAAP or in accordance with any defined terms set forth in Section Error! Reference source not found, 1.2 and (B) such items are supported by financial statements or other information reasonably satisfactory to Agent, (b) any Indebtedness (including any Delayed Draw Term Loan) incurred or assumed by any Credit Party or any Subsidiary of a Credit Party (including the Person or property acquired) in connection with such transaction (i) shall be deemed to have been incurred as of the first day of the applicable period and (ii) if such Indebtedness has a floating or formula rate, shall have an implied rate of interest for the applicable period for purposes of this definition determined by utilizing the rate which is or would be in effect with respect to such Indebtedness as of the relevant date of determination, and (c) any Restricted Payment shall be deemed to have been paid as of the first day of the applicable period.

[Different first page setting changed from on in original to off in modified.].

"Pro Forma Financial Statements" shall have the meaning set forth in Section 5.9(d) hereof.

"Projections" shall have the meaning set forth in Section 5.9(d) hereof.

"Property" shall mean an interest of any kind in any property or asset, whether real, personal or mixed, and whether tangible or intangible.

"Public Company Costs" shall mean actual documented costs relating to compliance with the Sarbanes-Oxley Act of 2002, as amended, and other actual documented expenses arising out of or incidental to DDH Holdings' status as a reporting company, including actual documented costs, fees and expenses (including reasonable legal, accounting and other professional fees) relating to compliance with provisions of the Securities Act and the Exchange Act, registration and reporting obligations, the rules of securities exchange companies with listed equity securities, directors' compensation, fees and expense reimbursement, shareholder meetings and reports to shareholders, indemnification and reimbursement of directors, officers and employees in respect of liabilities relating to their serving in any such capacity, directors' and officers' insurance and other professional fees, and listing fees.

"<u>Published Rate</u>" shall mean the rate of interest published each Business Day in The Wall Street Journal "Money Rates" listing under the eaption "London Interbank Offered Rates" for a three month period (or, if no such rate is published therein for any reason, then the Published Rate shall be the Eurodollar rate for a three month period as published in another publication determined by Agent).

"Purchasing CLO" shall have the meaning set forth in Section 15.3(d) hereof.

"Purchasing Lender" shall have the meaning set forth in Section 15.3(c) hereof.

"<u>Qualified IPO</u>" shall mean the issuance of Equity Interests by DDH Holdings in an underwritten primary public offering which had an effective date of February 10, 2022 and closed on February 15, 2022 pursuant to a registration statement filed with the Securities and Exchange Commission in accordance with the Securities Act and pursuant to which the Reorganization Transactions occurred.

"Quality of Earnings Report" shall mean that certain Quality of Earnings Report prepared by DoerenMayhew dated November 15, 2021 (with quality of earnings diligence analysis through August 31, 2021).

"<u>Real Property</u>" shall mean all of each Credit Party's right, title and interest in and to owned and leased premises now owned or hereafter acquired by any Credit Party.

"Real Property Collateral" shall mean all Real Property (including any Leasehold Interests) subject to the Liens of Agent.

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

"<u>Receivables</u>" shall mean and include, as to any Person, all of such Person's accounts, contract rights, instruments (including those evidencing indebtedness owed to such Person by its Affiliates), documents, chattel paper (including electronic chattel paper), general intangibles relating to accounts, drafts and acceptances, credit card receivables and all other forms of obligations owing to such Person arising out of or in connection with the sale or lease of Inventory or the rendition of services, all supporting obligations, guarantees and other security therefor, whether secured or unsecured, now existing or hereafter created, and whether or not specifically sold or assigned to Agent hereunder.

"<u>Redemption Agreement</u>" shall mean that certain redemption agreement dated as of November 14, 2021, by and between DDH and USDM Holdings, Inc., as amended by that certain Amendment to Redemption Agreement dated as of February 15, 2022, as further amended by that certain Second Amendment to Redemption Agreement dated as of July 28, 2022.

"<u>Redemption/Exchange Transactions</u>" shall mean at any time following the consummation of a Qualified IPO, the transactions pursuant to which DDH redeems Class A Common Units for either (A)(1) a stock exchange payment or (2) a cash exchange payment, in each case in accordance with the Company Agreement or (B) the direct purchase by DDH Holdings of vested Class A Common Units and paired voting stock pursuant any call right, in accordance with the Company Agreement.

"Register" shall have the meaning set forth in Section 15.3(e) hereof.

"<u>Related Fund</u>" shall mean, with respect to any Person that is an investment fund, any other investment fund that is managed or advised by the same investment advisor as such Person or by an Affiliate of such investment advisor.

"<u>Related PartiesPerson</u>" shall mean, with respect to a particular Person: (a) each other member of such Person's Family; (b) any Person that is directly or indirectly Controlled by such Person and/or any one or more members of such Person's Family; (c) any Person with respect to which such Person and/or one or more members of such Person's Family constitute all of the executors or trustees thereof (or in a similar capacity); and (d) any estate planning trust or limited partnership formed or organized for the benefit of such Person or such Person's Family so long as during his/her lifetime, such Person remains in Control of the voting rights with respect to any actions to be taken by such trust or limited partnership.

"<u>Release</u>" shall mean any release, spill, emission, leaking, pumping, pouring, injection, escaping, deposit, disposal, discharge, dispersal, dumping, leaching or migration of any Hazardous Material into the indoor or outdoor environment (including the abandonment or disposal of any barrels, containers or other closed receptacles containing any Hazardous Material), including the movement of any Hazardous Material through the air, soil, surface water or groundwater.

[Different first page setting changed from on in original to off in modified.].

"Relevant Governmental Body" shall mean the Federal Reserve Board or the Federal Reserve Bank of New York, or a committee officially endorsed or convened by the Federal Reserve Board or the Federal Reserve Bank of New York, or any successor thereto.

"<u>Reorganization Transactions</u>" shall mean the transactions contemplated pursuant to which DDH will effect an initial public offering, including, (i) the preferred equity units held in DDH will be redeemed, including pursuant to the Preferred A Redemption and the Preferred B Redemption, (ii) the Common Units Redemption shall be consummated with proceeds from a Qualified IPO, (iii) the Equity Interests held in DDH will be reclassified into two classes of units, (iv) the holders of Class A common units will contribute all of their Class A common units to DDH Holdings in exchange for Class A common stock, (v) immediately after which the holders of Class B voting units will contribute all of their Class B voting units to DDH Holdings in exchange for Class B common stock, (vi) immediately following such exchanges, DDH Holdings will be designated as the managing member of DDH, and (vii) DDH Holdings shall use the net proceeds received from an initial public offering to purchase Class A common units from DDH.

"<u>Replacement ABL Credit Agreement</u>" shall mean any loan or credit agreement entered into subsequent to the date hereof which (a) makes available to Borrower a revolving credit facility in an amount equal to \$7,500,000 (or such other amount reasonably acceptable to Agent and the Required Lenders), (b) has terms which are not in any material respect less favorable to the Borrower than the terms in effect under the ABL Credit Agreement in effect on the <u>Fifth Amendment</u> Closing Date, (c) is subject to an intercreditor agreement granting Agent the same priority collateral and substantially the same rights with respect to the Collateral as those set forth in the Intercreditor Agreement, and (d) is otherwise satisfactory to Agent and the Required Lenders in its reasonable discretion exercised in good faith.

"Replacement Lender" shall have the meaning set forth in Section 3.11.

"Replacement Notice" shall have the meaning set forth in Section 3.11.

"<u>Required Lenders</u>" shall mean Lenders holding more than fifty percent (50%) of the sum of (a) the aggregate unused Term Loan Commitment Percentages and Delayed Draw Term Loan Commitment Percentages, and (b) the aggregate outstanding principal amount of the Closing Date Term Loan and Delayed Draw Term Loans; <u>provided</u>, <u>however</u>, <u>that</u> (x) so long as Agent is also a Lender hereunder, Required Lenders shall include Agent, and (y) any portion of unused Term Loan Commitment Percentages and Delayed Draw Term Loan Commitment Percentages (or portion of outstanding Closing Date Term Loans or Delayed Draw Term Loans) held, or deemed held by, a Defaulting Lender shall be excluded for purposes of making a determination of Required Lenders.

"<u>Restricted Payment</u>" shall mean any dividend or other distribution (whether in cash, securities or other property) with respect to any Equity Interests of DDH Holdings, DDH or any Subsidiary thereof, or any payment (whether in cash, securities or other property and including, any cash payments to any holders of DDH's Preferred Equity), including any sinking fund or

[Different first page setting changed from on in original to off in modified.].

#### [Different first page setting changed from on in original to off in modified.].

similar deposit, on account of the purchase, redemption, retirement, acquisition, cancellation or termination of any such Equity Interests or on account of any return of capital to any such Person's stockholders, partners or members (or the equivalent Person thereof), or any setting apart of funds or property for any of the foregoing.

"S&P" shall mean Standard & Poor's Financial Services LLC, a subsidiary of The McGraw-Hill Companies, Inc., together with its successors.

"<u>Sale and Leaseback Transaction</u>" shall mean, with respect to DDH or any Subsidiary of DDH, any arrangement, directly or indirectly, with any Person (other than a Credit Party) whereby DDH or such Subsidiary shall sell or transfer to a third party any property, real or personal, used or useful in its business, whether now owned or hereafter acquired, and thereafter rent or lease such property or other property that it intends to use for substantially the same purpose or purposes as the property being sold or transferred.

"<u>Sanctioned Country</u>" shall mean (a) a country, region, territory or a government of a country or territory, (b) an agency of the government of a country, region or territory, or (c) an organization directly or indirectly owned or controlled by a country, region, territory or its government, that is subject to Sanctions.

"<u>Sanctioned Person</u>" shall mean (a) a Person named on the list of "Specially Designated Nationals" or any other Sanctions related list of designated Persons maintained by OFAC, the U.S. Department of State, the United Nations Security Council, the European Union or any European Union member state, (b) any Person operating, organized or resident in a Sanctioned Country or (c) any Person owned or controlled by any such Person or Persons described in the foregoing <u>clauses (a)</u> or (b).

"Sanctions" shall mean economic or financial sanctions or trade embargoes imposed, administered or enforced from time to time by (a) the U.S. government, including those administered by OFAC or the U.S. Department of State, (b) the United Nations Security Council, (c) the European Union, (d) any European Union member state, (e) Her Majesty's Treasury of the United Kingdom or (f) any other relevant sanctions authority.

"<u>SBA Forms</u>" shall mean, collectively, (i) Portfolio Financing Report (SBA Form 1031), (ii) Size Status Declaration (SBA Form 480), (iii) Assurance of Compliance for Nondiscrimination (SBA Form 652), and (iv) the SBIC Regulatory Side Letter.

"SBIC" shall mean a small business investment company licensed under the SBIC Act.

"SBIC Act" shall mean the Small Business Investment Act of 1958, as amended.

"SBIC Lender" shall mean a Lender that is an SBIC and is subject to the SBIC Act and the regulations promulgated thereunder by the SBA relating to the small business investment company program.

[Different first page setting changed from on in original to off in modified.].

"SBIC Regulatory Side Letter" shall mean a letter agreement, dated on or after the Closing Date, made by certain of the Credit Parties in favor of the SBIC Lender.

"SEC" shall mean the Securities and Exchange Commission or any successor thereto.

"Second Amendment" shall mean that certain Second Amendment and Joinder to Term Loan and Security Agreement dated as of the Second Amendment Date among the Credit Parties, Agent and Lenders.

"Second Amendment Date" shall mean July 28, 2022.

"Second Amendment Delayed Draw Term Loan" shall have the meaning set forth in Section 2.1(b)(i).

"Securities" shall mean any stock, shares, partnership interests, limited liability company interests, voting trust certificates, certificates of interest or participation in any profit-sharing agreement or arrangement (e.g., stock appreciation rights), options, warrants, bonds, debentures, notes, or other evidences of indebtedness, secured or unsecured, convertible, subordinated or otherwise, or in general any instruments commonly known as "securities" or any certificates of interest, shares or participations in temporary or interim certificates for the purchase or acquisition of, or any right to subscribe to, purchase or acquire, any of the foregoing.

"Securities Act" shall mean the Securities Act of 1933, as amended.

"Securitization Transaction" shall mean any financing or factoring or similar transaction (or series of such transactions) entered by DDH or any of its Subsidiaries pursuant to which DDH or such Subsidiary may sell, convey or otherwise transfer, or grant a security interest in, accounts, payments, receivables, rights to future lease payments or residuals or similar rights to payment (the "Securitization Receivables") to a special purpose subsidiary or affiliate (a "Securitization Subsidiary") or any other Person.

"Security Agreement" shall mean any security agreement or pledge agreement executed and delivered in favor of Agent pursuant to this Agreement or any Other Document, including, without limitation, the Pledge Agreements and any other security agreement or pledge agreement delivered in connection with the Other Documents, including any intellectual property assignments or security agreements required to be delivered by this Agreement or any Other Document, in each case, as amended, amended and restated, modified and supplemented from time to time.

"Security Incident" shall mean any unauthorized access, acquisition, use, disclosure, modification, deletion or destruction of Personal Information or interference with the operation of computer systems, including software, hardware, servers, networks and interfaces used by any Credit Party or Subsidiary, both for internal purposes and for the customers of any Credit Party or Subsidiary.

[Different first page setting changed from on in original to off in modified.].

"Senior Officer" shall mean, with respect to any Person, any individual holding the position of chairman of the board (if an officer), chief executive officer, president, vice president, chief financial officer or treasurer of such Person.

"SOFR" shall mean a rate equal to the secured overnight financing rate as administered by the SOFR Administrator.

"SOFR Administrator" shall mean the Federal Reserve Bank of New York (or a successor administrator of the secured overnight financing rate).

"SOFR Loan" shall mean a Loan that bears interest at a rate based on Adjusted Term SOFR.

"Solvent" or "Solvency" shall mean, with respect to any Person as of a particular date, that on such date (a) such Person is able to pay its debts and other liabilities, contingent obligations and other commitments as they mature in the ordinary course of business, (b) such Person does not intend to, and does not believe that it will, incur debts or liabilities beyond such Person's ability to pay as such debts and liabilities mature in their ordinary course, (c) such Person is not engaged in a business or a transaction, and is not about to engage in a business or a transaction, for which such Person's property would constitute unreasonably small capital after giving due consideration to the prevailing practice in the industry in which such Person is engaged or is to engage, (d) the fair value of the property of such Person is greater than the total amount of liabilities, including, without limitation, contingent liabilities, of such Person and (e) the present fair salable value of the assets of such Person is not less than the amount that will be required to pay the probable liability of such Person on its debts as they become absolute and matured. In computing the amount of contingent liabilities at any time, it is intended that such liabilities will be computed at the amount which, in light of all the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability.

"Subordinated Debt" shall mean any Indebtedness of any Credit Party to any Person (including any shareholder, officer, director, employee, consultant, or Affiliate of any Borrower) which Indebtedness shall be subordinated to any Term Loan on terms and conditions satisfactory to Agent in its Permitted Discretion, including but not limited to the Preferred Equity.

"<u>Subordination Agreement</u>" shall mean, collectively, those certain subordination agreements with respect to Subordinated Debt, including but not limited to the Preferred Equity Subordination Agreement, which shall be executed and delivered in connection therewith and as a condition thereto, each of which must be acceptable to Agent in its sole and absolute discretion.

"<u>Subsidiary</u>" shall mean, with respect to any Person, any corporation, partnership, limited liability company, association, joint venture or other business entity of which more than fifty percent (50%) of the Voting Stock is at the time owned or controlled, directly or indirectly, by that Person, or the accounts of which would be consolidated with those of such Person in its consolidated financial statements in accordance with GAAP, if such statements were prepared as

[Different first page setting changed from on in original to off in modified.].

of such date, or one or more of the other Subsidiaries of that Person or a combination thereof; <u>provided</u>, in determining the percentage of ownership interests of any Person controlled by another Person, no ownership interest in the nature of a "qualifying share" of the former Person shall be deemed to be outstanding. Unless otherwise provided, "Subsidiary" shall refer to a Subsidiary of a Credit Party; provided, that so long as Orange142 Canada does not conduct any business operations, have any material assets (other than holding a bank account with a de minimis account balance), incur any material liabilities or engage in any transactions with the Credit Parties, such entity shall not constitute a Subsidiary.

"Subsidiary Stock" shall mean all of the issued and outstanding Equity Interests of any Subsidiary owned by a Credit Party.

"Swap Agreement" shall mean (a) any and all rate swap transactions, basis swaps, credit derivative transactions, forward rate transactions, commodity swaps, commodity options, forward commodity contracts, equity or equity index swaps or options, bond or bond price or bond index swaps or options or forward bond or forward bond price or forward bond index transactions, interest rate options, forward foreign exchange transactions, currency swap transactions, cross-currency rate swap transactions, currency options, cap transactions, floor transactions, collar transactions, spot contracts, or any other similar transactions or any combination of any of the foregoing (including any options or warrants to enter into any of the foregoing), whether or not any such transaction is governed by, or otherwise subject to, any master agreement or any netting agreement, and (b) any and all transactions or arrangements of any kind, and the related confirmations, which are subject to the terms and conditions of, or governed by, any form of master agreement (or similar documentation) published from time to time by the International Swaps and Derivatives Association, Inc., any International Foreign Exchange Master Agreement, or any other master agreement (any such agreement or documentation, together with any related schedules, a "<u>Master Agreement</u>"), including any such obligations or liabilities under any Master Agreement.

"Swap Termination Value" shall mean, in respect of any one or more Swap Agreements, after taking into account the effect of any legally enforceable netting agreement relating to such Swap Agreements, (a) for any date on or after the date such Swap Agreements have been closed out and termination value(s) determined in accordance therewith, such termination value(s) and (b) for any date prior to the date referenced in <u>clause</u> (<u>a</u>), the amount(s) determined as the mark-to-market value(s) for such Swap Agreements, as determined based upon one or more mid-market or other readily available quotations provided by any recognized dealer in such Swap Agreements (which may include Agent, any Lender or any Affiliate of Agent or any Lender).

"<u>Synthetic Lease</u>" shall mean a lease transaction under which the parties intend that (a) the lease will be treated as an "operating lease" by the lessee pursuant to Statement of Financial Accounting Standards No. 13, as amended and (b) the lessee will be entitled to various tax and other benefits ordinarily available to owners (as opposed to lessees) of like property.

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

"<u>Tax</u>" or "<u>Taxes</u>" (and, with correlative meaning, "<u>Taxable</u>" or "<u>Taxing</u>") shall mean all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholdings), assessments, fees or other charges imposed by any Governmental Body, including any interest, additions to tax or penalties applicable thereto.

"Tax Distribution" shall mean the Taxes assumed to be payable by a direct or indirect shareholder or member of any Credit Party as a result of the allocation of income of such Credit Party to such shareholder or member due to such Credit Party's status for federal, state or local Tax purposes as a partnership, subchapter S corporation or any other entity that is a pass-through entity or disregarded entity for federal, state and local Tax purposes (as applicable) but only for so long as such Credit Party continues to be so treated as a pass-through entity or disregarded entity for federal, state and local tax purposes, as evidenced and substantiated by the tax returns filed by such Credit Party (as applicable), with such Taxes assumed to be payable by such shareholder or member of any Credit Party being calculated for all members or shareholders, as applicable, at the highest combined marginal federal, state and local Tax rate applicable to any Credit Party that is allocated to such member or shareholder of the Credit Party, taking into consideration (A) the character and nature of such income or gain (i.e., whether such income is subject to Tax at capital gains rates, ordinary income rates or any special rates), (B) losses previously allocated to each such member or shareholder, as applicable, by such Credit Party to the extent such losses have not previously been applied to reduce the Tax Distribution hereunder, provided that capital losse carry forwards shall be taken into account only to the extent they are currently usable to offset income or gain allocated by such Credit Party to a member or shareholder, as applicable; and provided, further, that to the extent that any losses allocated by such Credit Party result in a payback by a member to such Credit Party of previous Tax Distributions pursuant to Section 7.7 hereof, then such losses shall not be taken into account for purposes of determining the Tax Distribution hereunder, and (C) the deduction under Section 199A of the Internal Revenue Code in respect

"Tax Receivable Agreement" shall mean that certain tax receivable agreement to be entered into by and among DDH Holdings, DDH, and Direct Digital Management, LLC.

"Term" shall have the meaning set forth in Section 13.1 hereof.

"<u>Term Loan Commitment Percentage</u>" of any Lender shall mean the percentage set forth on <u>Schedule 1.2(b)</u> as its "Term Loan Commitment Percentage," as the same may be adjusted upon any assignment by a Lender pursuant to <u>Section 15.3(c)</u> or <u>15.3(d)</u>.

"<u>Term Loan Rate</u>" shall mean an interest rate per annum equal to the <u>LIBOR Rate plussum of</u> the Applicable Margin <u>plus Adjusted Term</u> <u>SOFR for the then-current Interest Rate Reset Period</u> less any <u>Applicable Impact Discount</u><u>Worker Solutions Performance Incentive</u>.

"Term Loans" shall mean, collectively, the Closing Date Term Loan and the Delayed Draw Term Loan.

[Different first page setting changed from on in original to off in modified.].

"Term Notes" shall mean, collectively, the Closing Date Term Note and the Delayed Draw Term Loan Notes.

"Term SOFR" shall mean, for any calculation with respect to a SOFR Loan for any Interest Rate Reset Period, the Term SOFR Reference Rate for a tenor comparable to the Interest Period on the day (such day, the "Periodic Term SOFR Determination Day") that is two (2) U.S. Government Securities Business Days prior to the first day of such Interest Rate Reset Period, as such rate is published by the Term SOFR Administrator; provided, however that if as of 5:00 p.m. on any Periodic Term SOFR Determination Day the Term SOFR Reference Rate for the applicable tenor has not been published by the Term SOFR Administrator and a Benchmark Replacement Date with respect to the Term SOFR Reference Rate has not occurred, then Term SOFR will be the Term SOFR Administrator on the first preceding U.S. Government Securities Business Day is not more than three (3) U.S. Government Securities Business Days prior to such Periodic Term SOFR Determination Day.

"Term SOFR Adjustment" shall mean 0.10% (10 basis points) for an Interest Period of one-month's duration and 0.15% (15 basis points) for an Interest Period of three-months' duration.

"Term SOFR Administrator" shall mean CME Group Benchmark Administration Limited (CBA) (or a successor administrator of the Term SOFR Reference Rate selected by Agent in its reasonable discretion).

"Term SOFR Reference Rate" shall mean the forward-looking term rate based on SOFR.

"Title Policy" shall have the meaning set forth in Section 6.8(b)(iii) hereof.

"<u>Trading with the Enemy Act</u>" shall mean the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B, Chapter V, as amended) and any enabling legislation or executive order relating thereto.

"<u>Transfer Supplement</u>" shall mean a document in the form of <u>Exhibit B</u> hereto, properly completed and otherwise in form and substance satisfactory to Agent by which the Purchasing Lender purchases and assumes a portion of (i) the obligation of Lenders to make Advances under this Agreement or (ii) the right and option of Lenders to make Delayed Draw Term Loans under this Agreement.

"Transferee" shall have the meaning set forth in Section 15.3(d) hereof.

"Unadjusted Benchmark Replacement" shall mean the applicable Benchmark Replacement excluding the related Benchmark Replacement Adjustment.

"<u>Unfinanced Capital Expenditures</u>" shall mean, for any specified Persons, all Consolidated Capital Expenditures of such Persons other than those made utilizing financing provided by the applicable seller or third party lenders (including (x) by any Lender to the extent

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

that the Indebtedness arising with respect to such financing does not constitute Advances or Obligations, and (y) expenditures made with advance under the ABL Credit Agreement).

"Uniform Commercial Code" or "UCC" shall have the meaning set forth in Section 1.3 hereof.

"United States" and "U.S." shall mean the United States of America.

"<u>Unrestricted Cash</u>" shall mean that portion of the Credit Parties' cash that is (a) free and clear of all Liens, other than Liens in favor of Agent and ABL Lender pursuant to the ABL Credit Agreement, (b) maintained by the Credit Parties in one or more deposit accounts that are located in the United States and are subject to Control Agreements and (c) not listed as "restricted" on the Credit Parties' balance sheet.

"<u>USA PATRIOT Act</u>" shall mean the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, Public Law 107-56, as the same has been, or shall hereafter be, renewed, extended, amended or replaced.

"U.S. Government Securities Business Day" shall mean any day except for (a) a Saturday, (b) a Sunday or (c) a day on which the Securities Industry and Financial Markets Association recommends that the fixed income departments of its members be closed for the entire day for purposes of trading in United States government securities.

"<u>Voting Stock</u>" shall mean, with respect to any Person, Equity Interests issued by such Person the holders of which are ordinarily, in the absence of contingencies, entitled to vote for the election of directors (or persons performing similar functions) of such Person, even though the right so to vote has been suspended by the happening of such a contingency.

"<u>Warrant Tender Offer</u>" shall mean DDH Holdings offer to purchase for cash any and all of its outstanding warrants at a price of \$1.20 per warrant as set forth in the combined Tender Offer Statement and Rule 13e-3 Transaction Statement on Schedule TO, including the Second Amended and Restated Offer to Purchase and Consent Solicitation dated September 21, 2023, originally filed with the Securities and Exchange Commission on August 29, 2023, as amended on September 14, 2023, September 21, 2023 and September 29, 2023.

"Worker Solutions" shall mean programs or services intended to provide to companies and their employees, as applicable, long-term, sustainable impact in financial inclusion and empowerment, housing, jobs, wellness, diversity and inclusion or such other core values acceptable to Agent in its sole discretion.

"Worker Solutions Certificate" shall mean a certificate in substantially the form attached hereto as Exhibit D to be signed by an Authorized Officer of Borrowers.

"Worker Solutions Performance Incentive" shall mean (a) as of the Closing Date and through and including the date immediately prior to the first Worker Solutions Performance

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

Incentive Reset Date, a rate per annum equal to 0.00 % and (b) on and after each Worker Solutions Performance Incentive Reset Date, (i) a rate per annum as set forth in the table below based upon the Worker Solutions Subscription to each of the Eligible Worker Solutions by Borrowers:

No Eligible Worker Solutions	0.00%
One or more Eligible Worker Solutions	<u>0.05%</u>

plus (ii) solely to the extent Borrowers received a B Corp Certification by Standards Analysts at the non-profit B Lab (or any successor certification or administrator identified by Agent in its sole discretion), as listed in https://bcorporation.net/directory (or any successor directory identified by Agent in its sole discretion) or such other comparable certification from a nationally recognized and reputable third party assessment and validation of social and environmental performance identified to and approved by Agent in writing, a rate per annum equal to 0.05% so long as Borrowers maintain such certification.

The component of the Worker Solutions Performance Incentive set forth in clause (b)(i) above shall (i) remain in effect for the period of the applicable Worker Solutions Subscription to any applicable Eligible Worker Solution(s) notwithstanding removal of such provider, program or service from Exhibit A (unless Agent has notified Borrowers in writing that such provider, program or service is removed or to be removed from Exhibit A prior to the commencement of the period of any such Worker Solutions Subscription) and (ii) be adjusted, to the extent applicable, (x) as of each Worker Solutions Performance Incentive Reset Date based upon and to the extent that Borrowers have certified in the applicable Worker Solutions Certificate as to the Worker Solutions Subscription to any such applicable Eligible Worker Solution or (y) otherwise at such time agreed by Agent and Borrowers as a result of a mutual determination to (1) cease or otherwise terminate any applicable Eligible Worker Solution (thereby, reducing the Worker Solutions Performance Incentive) or (2) Worker Solutions Subscripte to a Worker Solution which had not been the subject of a Worker Solutions Subscription immediately prior to such determination (thereby increasing the Worker Solutions Performance Incentive to the extent applicable).

If Agent determines that Borrowers did not successfully make a Worker Solutions Subscription to any such Eligible Worker Solution during an earlier period in which the Worker Solutions Performance Incentive applied (or were otherwise not certified as a B Corporation under clause (b) (ii) above during any such earlier period), without limiting any other rights or remedies arising as a result of any certification in a Worker Solutions Certificate which may prove to have been false or misleading, Borrowers shall automatically and retroactively be obligated to pay to Agent, for the benefit of the applicable Lenders, promptly on demand by Agent, an amount equal to the

[Different first page setting changed from on in original to off in modified.].

excess of the amount of interest that should have been paid for such period without the Worker Solutions Performance Incentive over the amount of interest actually paid for such period.

"Worker Solutions Performance Incentive Reset Date" shall mean the first day of the month following the date on which financial statements, Compliance Certificate and Worker Solutions Certificate are required to be delivered pursuant to Section 9.8 after the end of each related Fiscal Quarter.

# "Worker Solutions Provider" shall mean any third-party provider or other third party vendor providing Worker Solutions.

<u>"Worker Solutions Subscription" shall mean the subscription to, participation in or other purchase or acquisition of one or more (as applicable) Worker Solutions by Borrowers (or any of them, as applicable) for any period identified and offered by a Worker Solutions Provider to Borrowers. "Worker Solutions Subscribe" and "Worker Solutions Subscribed" shall have the corollary meaning thereto.</u>

# I.3 Uniform Commercial Code Terms

. All terms used herein and defined in the Uniform Commercial Code as adopted in the State of New York from time to time (the "<u>Uniform Commercial Code</u>") shall have the meaning given therein unless otherwise defined herein. Without limiting the foregoing, the terms "accessions", "accounts", "certificates of title", "chattel paper", "commercial tort claims", "commodities accounts", "commodities contracts", "documents", "deposit accounts", "electronic chattel paper", "equipment", "fixtures", "general intangibles", "goods", "instruments", "inventory", "investment property", "letter-of-credit rights", "payment intangibles", "proceeds", "securities", "securities accounts", "securities entitlements", "supporting obligations" and "software", as and when used in the description of Collateral (and related underlying definitions) have the meanings given to such terms in Articles 8 or 9 of the Uniform Commercial Code. To the extent the definition of any category or type of collateral is expanded by any amendment, modification or revision to the Uniform Commercial Code, such expanded definition will apply automatically as of the date of such amendment, modification or revision.

## I.4 Certain Matters of Construction.

Ŧ

(a) <u>The terms "herein", "hereof" and "hereunder" and other words of similar import refer to this Agreement as a whole and not to any particular section, paragraph or subdivision. All references herein to Articles, Sections, Exhibits and Schedules shall be construed to refer to Articles and Sections of, and Exhibits and Schedules to, this Agreement. Any pronoun used shall be deemed to cover all genders. Wherever appropriate in the context, terms used herein in the singular also include the plural and vice versa. All references to statutes and related regulations shall include any amendments of same and any successor statutes and regulations. Unless otherwise provided, all references to any instruments or agreements to which</u>

[Different first page setting changed from on in original to off in modified.].

Agent is a party, including references to any of the Other Documents, shall include any and all modifications or amendments thereto and any and all extensions, renewals or restatements thereof. All references herein to the time of day shall mean the time in New York, New York. Unless otherwise provided, all financial calculations shall be determined on a first-in-first-out basis. Whenever the words "including" or "include" shall be used, such words shall be understood to mean "including, without limitation" or "include, without limitation". A Default or Event of Default shall be deemed to exist at all times during the period commencing on the date that such Default or Event of Default occurs to the date on which such Default or Event of Default is waived in writing pursuant to this Agreement or, in the case of a Default, is remedied within any grace period expressly provided for in this Agreement; and an Event of Default shall "continue" or be "continuing" until such Event of Default has been waived in writing by the Required Lenders. Any Lien referred to in this Agreement or any of the Other Documents as having been created in favor of Agent, any agreement entered into by Agent pursuant to this Agreement or any of the Other Documents, any payment made by or to or funds received by Agent pursuant to or as contemplated by this Agreement or any of the Other Documents, or any act taken or omitted to be taken by Agent, shall, unless otherwise expressly provided, be created, entered into, made or received, or taken or omitted, for the benefit or account of Agent and Lenders. Wherever the phrase "to the best of each Credit Party's knowledge" or words of similar import relating to the knowledge or the awareness of a Credit Party are used in this Agreement or Other Documents, such phrase shall mean and refer to (i) the actual knowledge of a Senior Officer of a Credit Party or (ii) the knowledge that a Senior Officer would have obtained if he or she had engaged in good faith and diligent performance of his or her duties, including the making of such reasonably specific inquiries as may be necessary of the employees or agents of a Credit Party and a good faith attempt to ascertain the existence or accuracy of the matter to which such phrase relates. All covenants hereunder shall be given independent effect so that if a particular action or condition is not permitted by any of such covenants, the fact that it would be permitted by an exception to, or otherwise within the limitations of, another covenant shall not avoid the occurrence of a default if such action is taken or condition exists. In addition, all representations and warranties hereunder shall be given independent effect so that if a particular representation or warranty proves to be incorrect or is breached, the fact that another representation or warranty concerning the same or similar subject matter is correct or is not breached will not affect the incorrectness of a breach of a representation or warranty hereunder.

I.5 Early Opt-In Election. Agent notified Borrower on June 1, 2023 that pursuant to Section 3.8 hereof, Agent and the Credit Parties have jointly elected to trigger a fallback from USD LIBOR to Term SOFR. Pursuant thereto, effective May 31, 2023 (the "Replacement Date"), Term SOFR replaced LIBOR for all purposes under this Agreement and the Other Documents in respect of such Benchmark setting without any amendment to, or further action or consent of any other party to, the Loan Agreement or any Other Document. On the Replacement Date, the Benchmark Replacement shall mean the sum of (x) Term SOFR and (y) 0.15% (15 basis points) for an Available Tenor of three-months' duration. In connection with the implementation and administration of Term SOFR, Agent reserves the right to make Benchmark Replacement Conforming Changes from time to time in accordance with this Agreement.

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

L1 Disclaimer. Agent neither warrants nor accepts responsibility for, and shall not have any liability with respect to (a) the continuation of, administration of, submission of, calculation of or any other matter related to the Term SOFR Reference Rate, Adjusted Term SOFR or Term SOFR, or any component definition thereof or rates referred to in the definition thereof, or any alternative, successor or replacement rate thereto (including any Benchmark Replacement), including whether the composition or characteristics of any such alternative, successor or replacement rate (including any Benchmark Replacement) will be similar to, or produce the same value or economic equivalence of, or have the same volume or liquidity as, the Term SOFR Reference Rate, Adjusted Term SOFR, Term SOFR or any other Benchmark prior to its discontinuance or unavailability, or (b) the effect, implementation or composition of any Conforming Changes. Neither Agent nor its affiliates or other related entities may engage in transactions that affect the calculation of the Term SOFR Reference Rate, Term SOFR, Adjusted Term SOFR, any alternative, successor or replacement rate (including any Benchmark Replacement) or any relevant adjustments thereto, in each case, in a manner adverse to the Credit Parties. Agent may select information sources or services in its reasonable discretion to ascertain the Term SOFR Reference Rate, Term SOFR, Adjusted Term SOFR or any other Benchmark, in each case pursuant to the terms of this Agreement, and shall have no liability to any Credit Party, any Lender or any other person or entity for damages of any kind, including direct or indirect, special, punitive, incidental or consequential damages, costs, losses or expenses (whether in tort, contract or otherwise and whether at law or in equity), for any error or calculation of any such rate (or component thereof) provided by any such information source or service.

# II. <u>ADVANCES, PAYMENTS.</u>

- II.1 Term Loans.
- ÷
- (a) <u>Closing Date Term Loan.</u>

(i) Subject to the terms and conditions of this Agreement, each Lender, severally and not jointly, will make a term loan to Borrowers in the amount equal to such Lender's Term Loan Commitment Percentage of the Closing Date Term Loan Amount (collectively, the "<u>Closing Date Term Loan</u>"). Subject to acceleration upon the occurrence and during the continuance of an Event of Default under this Agreement or termination of this Agreement, and subject to the application of any prepayments in accordance with the terms hereof, the Closing Date Term Loan shall be, with respect to principal, due and payable in quarterly installments on the last Business Day of each fiscal quarter: (x) commencing January 1, 2022 (with the first payment due March 31, 2022) through and including December 31, 2023, in an amount equal to \$137,500.00 each, and (y) at all times thereafter, in an amount equal to \$275,000.00 each, with a final installment due at the end of the Term in an amount equal to the entire unpaid principal balance thereof; provided that during the period commencing with the fiscal quarter ended June 30, 2024, and each fiscal quarter thereafter and ending on, but including, the fiscal quarter ending December 31, 2025, there shall be no payments of principal

[Different first page setting changed from on in original to off in modified.].

<u>due and payable under this Section 2.1(a)</u>. The Closing Date Term Loan shall be evidenced by one or more secured promissory notes (collectively, the "<u>Closing Date Term Note</u>") in substantially the form attached hereto as <u>Exhibit F-1</u>.

(ii) Borrowing Agent may, from time to time, prepay the principal amount owing under the Closing Date Term Loan, so long as: (A) Borrowing Agent shall have provided at least three (3) Business Days prior written notice not later than 1:00 p.m. (New York City time) to Agent of such prepayment, specifying the amount of such prepayment (which notice, once given, shall be irrevocable); (B) each such prepayment is in a minimum amount of \$500,000, or an increment of \$250,000 in excess thereof and (C) such prepayment shall be accompanied by the payment of any prepayment fee set forth in the Fee Letter. Such prepayments shall be applied to the principal installments due on the Closing Date Term Loan on a *pro rata* basis; provided that any such prepayments shall be applied on a *pro rata* basis to the Closing Date Term Loan and the then outstanding portion of the Delayed Draw Term Loans.

# (b) <u>Delayed Draw Term Loans.</u>

(i) Pursuant to the Second Amendment, Lenders made a term loan to Borrower on the Second Amendment Date in an aggregate amount equal to \$4,260,000 (the "Second Amendment Delayed Draw Term Loan") in accordance with each Lender's Delayed Draw Term Loan Commitment Percentage, the proceeds of which were applied in accordance with the Second Amendment. On the Fourth Amendment Date and pursuant to the Fourth Amendment, each Lender will make a term loan to Borrower equal to such Lender's Delayed Draw Term Loan Commitment Percentage of \$3,587,274.03 (the "Fourth Amendment Delayed Draw Term Loan", together with the Second Amendment Delayed Draw Term Loan, the "Delayed Draw Term Loans"). Notwithstanding anything to the contrary contained herein, after giving effect to the Fourth Amendment Date, Borrower shall not be permitted to request, and no Lender shall be obligated to make, any additional term loans or Delayed Draw Term Loan.

(ii) [<u>Reserved].</u>

(iii) <u>Subject to the terms and conditions of this Agreement, including satisfaction of the conditions set forth in Section</u> 8.2 hereof, each Lender severally and not jointly, shall make, on the Fourth Amendment Date, the Fourth Amendment Delayed Draw Term Loan in the sum equal to such Lender's Delayed Draw Term Loan Commitment Percentage of the Fourth Amendment Delayed Draw Term Loan Amount.

(iv) Subject to acceleration upon the occurrence and during the continuance of an Event of Default under this Agreement or termination of this Agreement, and subject to the application of any prepayments in accordance with the terms hereof, each Delayed Draw Term Loan shall be, with respect to principal, payable in quarterly installments in an amount equal to (x) commencing January 1, 2022 through and including December 31, 2023, six hundred twenty-five thousandths of one percent (0.625%) of the amounts of the applicable

[Different first page setting changed from on in original to off in modified.].

Delayed Draw Term Loan, and (y) at all times thereafter, one and one quarter of one\_percent (1.25%) of the amounts of the applicable Delayed Draw Term Loan, commencing on the last day of the first full Fiscal Quarter ending after the applicable Delayed Draw Term Loan Funding Date of such Delayed Draw Term Loan, with a final installment due at the end of the Term in an amount equal to the entire unpaid principal balance thereof; provided that during the period commencing with the fiscal quarter ended June 30, 2024, and each fiscal quarter thereafter and ending on, but including, the fiscal quarter ending December 31, 2025, there shall be no payments of principal due and payable under this Section 2.1(b)(iv). Notwithstanding anything to the contrary set forth above, the Second Amendment Delayed Draw Term Loan, the first installment of principal shall be due on December 31, 2023. Each Delayed Draw Term Loan shall be evidenced by one or more secured promissory notes (collectively, the "Delayed Draw Term Loan Note") in substantially the form attached hereto as Exhibit F-2.

(v) Borrowing Agent may, from time to time, prepay the principal amount owing under each Delayed Draw Term Loan, so long as: (A) Borrowing Agent shall have provided at least three (3) Business Days prior written notice not later than 11:00 a.m. (New York City time) to Agent of such prepayment, specifying the amount of such prepayment (which notice, once given, shall be irrevocable); (B) each such prepayment is in a minimum amount of \$500,000, or an increment of \$250,000 in excess thereof and (C) such prepayment shall be accompanied by the payment of any prepayment fee set forth in the Fee Letter. Such prepayments shall be applied to the principal installments due on such Delayed Draw Term Loan on a pro rata basis; provided that any such prepayments shall be applied on a *pro rata* basis to the Closing Date Term Loan and the then outstanding portion of the Delayed Draw Term Loans.

- II.2 <u>Repayment of Advances.</u>
- ÷

(a) <u>The Term Loans shall be due and payable as provided in Section 2.1 and in the Term Notes, subject to mandatory</u> prepayments as herein provided. Notwithstanding the foregoing, all Advances shall be subject to earlier repayment upon (i) acceleration upon the occurrence and during the continuance of an Event of Default under this Agreement or (ii) termination of this Agreement.

(b) Each Borrower recognizes that the amounts evidenced by checks, notes, drafts or any other items of payment relating to and/or proceeds of Collateral may not be collectible by Agent on the date received. In consideration of Agent's agreement to conditionally credit Borrowers' Account as of the next Business Day following Agent's receipt of those items of payment, each Borrower agrees that, in computing the charges under this Agreement, all items of payment shall be deemed applied by Agent on account of the Obligations (i) on the Business Day following Agent's receipt of such payments via wire transfer or electronic depository check or (ii) in the case of payments received by Agent in any other

[Different first page setting changed from on in original to off in modified.].

form, the Business Day such payment constitutes good funds in Agent's account. Agent is not, however, required to credit Borrowers' Account for the amount of any item of payment which is unsatisfactory to Agent in its Permitted Discretion and Agent may charge Borrowers' Account for the amount of any item of payment which is returned to Agent unpaid.

(c) <u>All payments of principal, interest and other amounts payable hereunder, or under any of the Other Documents shall be</u> made to Agent via wire transfer or electronic depository check not later than 1:00 p.m. (New York City time) on the due date therefor in lawful money of the United States of America in federal funds or other funds immediately available to Agent.

(d) <u>Borrowers shall, jointly and severally, pay principal, interest, and all other amounts payable hereunder, or under any related</u> agreement, without any deduction whatsoever, including, but not limited to, any deduction for any setoff or counterclaim.

- (e) <u>Any amount prepaid or repaid in respect of Term Loans may not be reborrowed.</u>
- II.3 [Reserved].
- ÷

# II.4 Statement of Account

. Agent shall maintain, in accordance with its customary procedures, a loan account ("Borrowers' Account") in the name of the Borrowers in which shall be recorded the date and amount of each Advance made by Lenders, the Lender(s) making such Advance and the date and amount of each payment in respect thereof; provided, however, the failure by Agent to record the date and amount of any Advance shall not adversely affect Agent or any Lender. The records of Agent with respect to the loan account shall be prima facie evidence absent manifest error of the amounts of Advances and other charges thereto and of payments applicable thereto.

### II.5 Additional Payments

. Any sums expended by Agent or any Lender due to any Borrower's failure to perform or comply with its obligations under this Agreement or any Other Document including the Borrowers' obligations under <u>Sections 4.2</u>, <u>4.4</u>, <u>4.12</u>, <u>4.13</u> and <u>4.14</u> hereof, may be charged to Borrowers' Account and added to the Obligations.

## II.6 Manner of Borrowing and Payment.

Ŧ

(a) <u>Each payment (including each prepayment) by Borrowers on account of the principal of and interest on the Term Loans</u> shall be applied to the Term Loans *pro rata* 

[Different first page setting changed from on in original to off in modified.].

according to the Term Loan Commitment Percentages of Lenders. Except as expressly provided herein, all payments (including prepayments) to be made by Borrowers on account of principal, interest and fees shall be made without set off or counterclaim and shall be made to Agent on behalf of the Lenders, in each case, via wire transfer or electronic depository check on or prior to 1:00 p.m. (New York City time) in Dollars and in immediately available funds.

(b) Except as otherwise expressly provided in this Agreement with respect to any such specified payment required to be made by Borrowers hereunder or with respect to any payment obtained by a Lender as consideration for the assignment of or sale of a participation in any of its Advances to any assignee or participant as contemplated under Section 15.3 of this Agreement, if any Lender or Participant (a "benefited Lender") shall at any time receive any payment of all or part of its Advances, or interest thereon, or receive any Collateral in respect thereof (whether voluntarily or involuntarily or, if consented to by Agent, by set-off) in a greater proportion than any such payment to and Collateral received by any other Lender, if any, in respect of such other Lender's Advances, or interest thereon, and such greater proportionate payment or receipt of Collateral is not expressly permitted hereunder, such benefited Lender shall purchase for cash from the other Lenders a participation in such portion of each such other Lender's Advances, or shall provide such other Lender with the benefits of any such Collateral, or the proceeds thereof, as shall be necessary to cause such benefited Lender to share the excess payment or benefits of such Collateral or proceeds ratably with each of the other Lenders; provided, however, that if all or any portion of such excess payment or benefits is thereafter recovered from such benefited Lender, such purchase shall be rescinded, and the purchase price and benefits returned, to the extent of such recovery, but without interest. Each Lender so purchasing a portion of another Lender's Advances may exercise all rights of payment (including, if consented to by Agent, rights of set-off) with respect to such portion as fully as if such Lender were the direct holder of such portion.

II.7 Mandatory Prepayments.

(a) Asset Dispositions and Involuntary Dispositions. The Credit Parties shall prepay the Obligations in an amount equal to one hundred percent (100%) of the Net Cash Proceeds received from any Asset Sale (other than Asset Sales permitted pursuant to Section 7.1(b)) or Involuntary Disposition by DDH or any of its Subsidiaries on the Business Day following receipt of Net Cash Proceeds required to be prepaid pursuant to the provisions hereof; provided that such prepayment shall not be required to be made to the extent Borrowing Agent delivers to Agent a certificate stating that the applicable Credit Party intends to reinvest such Net Cash Proceeds in replacement assets (excluding Consolidated Current Assets) that are useful in the business of DDH or any of its Subsidiaries within 180 days after the date of such Asset Sale or Involuntary Disposition and such Net Cash Proceeds are so reinvested during such period, provided, further that in the case of an Involuntary Disposition, if a definitive agreement to so reinvest has been executed within such 180-day period, then such reinvestment shall have been consummated within 180 days after the date such definitive agreement was executed; it being

[Different first page setting changed from on in original to off in modified.].

÷

understood that any such Net Cash Proceeds not so reinvested during such periods, shall be used to prepay the Obligations immediately following the expiration of such respective period. Until the date of payment, such Net Cash Proceeds required to be prepaid shall be held in trust for Agent and, upon such prepayment, shall be applied to the Term Loans in the inverse order of maturities thereof (x) in respect of the Closing Date Term Loan and the Delayed Draw Term Loans, and (y) to all remaining principal installments thereof until paid in full. The foregoing shall not be deemed to be implied consent to any such sale otherwise prohibited by the terms and conditions hereof.

(b) [<u>Reserved</u>].

(c) Equity Proceeds. The Credit Parties shall prepay the Obligations in an amount equal to one hundred percent (100%) of the Net Cash Proceeds from any Equity Transactions (other than a Qualified IPO and Equity Transactions used solely to consummate Permitted Acquisitions and completed substantially contemporaneously therewith), including any Equity Cure, on the Business Day following receipt thereof unless waived by Agent in its sole and absolute discretion. Until the date of payment, such Net Cash Proceeds required to be prepaid shall be held in trust for Agent. The foregoing shall not be deemed to be implied consent to any such issuance otherwise prohibited by the terms and conditions hereof. Such repayments shall be applied to the outstanding principal installments of the Term Loans in the inverse order of maturities thereof (x) in respect of the Closing Date Term Loan and the Delayed Draw Term Loans, and (y) to all remaining principal installments thereof until paid in full.

(d) Indebtedness Proceeds. The Credit Parties shall prepay the Obligations in an amount equal to one hundred percent (100%) of the Net Cash Proceeds from any Debt Transactions on the Business Day following receipt thereof. Until the date of payment, such proceeds shall be held in trust for Agent. The foregoing shall not be deemed to be implied consent to any such incurrence otherwise prohibited by the terms and conditions hereof. Such repayments shall be applied to the outstanding principal installments of the Term Loans in the inverse order of maturities thereof (x) in respect of the Closing Date Term Loan and the Delayed Draw Term Loans, and (y) to all remaining principal installments thereof until paid in full.

(e) Consolidated Excess Cash Flow. The Credit Parties shall prepay the Obligations, within thirty (30) days of delivery of each annual Compliance Certificate delivered under Section 9.7, commencing with the Compliance Certificate delivered for the Fiscal Year ending December 31, 2022, in an amount equal to (i) so long as no Event of Default has occurred and is continuing, 50% of Consolidated Excess Cash Flow and (ii) following the occurrence and during the continuance of an Event of Default, 100% of Consolidated Excess Cash Flow for the immediately preceding Fiscal Year unless waived in writing by Agent. The prepayments specified in this Section 2.7(e) shall be applied to the outstanding principal installments of the Term Loans in the inverse order of maturities thereof (x) in respect of the Closing Date Term Loan and the Delayed Draw Term Loans, and (y) to all remaining principal installments thereof until paid in full. In the event that the Compliance Certificate required by Section 9.7 is not delivered on or before the due date, then a calculation based upon estimated amounts shall be

[Different first page setting changed from on in original to off in modified.].

made by Agent upon which calculation Borrowers shall make the prepayments required by this Section 2.7(e), subject to adjustment when the applicable financial statements and Compliance Certificate are delivered to Agent as required hereby. Such calculations made by Agent and/or acceptance of any related estimated prepayments shall not be deemed a waiver of any rights Agent or Lenders may have, whether as a result of the failure by Borrowers to timely deliver such financial statements or otherwise.

(f) Extraordinary Receipts. When any Credit Party or Subsidiary thereof receives the cash proceeds of any Extraordinary Receipt, unless such Credit Party or Subsidiary reinvests such proceeds in the same manner that it would be entitled to under Section 2.7(a), the Credit Parties shall repay the Term Loans in an amount equal to 100% of the Net Cash Proceeds of such Extraordinary Receipts, such prepayments to be made promptly but in no event more than five (5) Business Days following receipt of such Net Cash Proceeds, and until the date of payment, such proceeds shall be held in trust for Agent. Such repayments shall be applied to the outstanding principal installments of the Term Loans in the inverse order of maturities thereof (x) in respect of the Closing Date Term Loan and the Delayed Draw Term Loans, and (y) to all remaining principal installments thereof until paid in full.

(g) <u>ABL Loan Documents. The terms of this Section 2.7 are subject to the terms and conditions of the Intercreditor Agreement.</u> To the extent that any mandatory prepayment pursuant to the ABL Loan Documents is, contemporaneously with the event giving rise to such prepayment, required and permitted to be paid to the ABL Obligations pursuant to the Intercreditor Agreement, then the amount of such payment required pursuant to the ABL Loan Documents at such time shall reduce the amount of any mandatory prepayment owing pursuant to this Section 2.7 on a dollar for dollar basis or as otherwise required pursuant to the Intercreditor Agreement.

II.8 Use of Proceeds.

(a) Borrowers shall use the proceeds of (i) the Closing Date Term Loans to (A) fund the Preferred A Redemption, (B) refinance certain Indebtedness of the Borrowers existing immediately prior to the Closing Date, (C) pay fees and expenses associated with the Closing Date Transactions and (D) provide for their general corporate needs and working capital requirements, including, funding for creation and retention of direct and indirect jobs and/or opening of new revenue-generating locations, and (ii) the Second Amendment Delayed Draw Term Loans as set forth in the Second Amendment and Fourth Amendment Delayed Draw Term Loan to make payments due in connection with the consummation of the Warrant Tender Offer in accordance with its terms (not to exceed approximately \$3,021,103.55) and fees and expenses incurred in connection therewith.

(b) <u>Without limiting the generality of Section 2.8(a) above, neither any Borrower, any other Credit Party, nor any other Person</u> which may in the future become a Credit

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

Party, intends to use nor shall they use any portion of the proceeds of the Advances, directly or indirectly, for any purpose in violation of the Trading with the Enemy Act or to purchase or carry, or to reduce or refinance any Indebtedness incurred to purchase or carry, any margin stock (as defined in Regulation U of the Board of Governors of the Federal Reserve System) or for any related purpose governed by Regulations T, U or X of the Board of Governors of the Federal Reserve System.

II.9 Defaulting Lender.

÷

(a) <u>Notwithstanding anything to the contrary contained herein, in the event any Lender (i) has refused (which refusal constitutes a breach by such Lender of its obligations under this Agreement) to make available its portion of any Advance or (ii) notifies either Agent or Borrowing Agent that it does not intend to make available its portion of any Advance and such actual refusal would constitute a breach by such Lender of its obligations under this Agreement (each, a "Lender Default"), all rights and obligations hereunder of such Lender (a "Defaulting Lender") as to which a Lender Default is in effect and of the other parties hereto shall be modified to the extent of the express provisions of this Section 2.9 while such Lender Default remains in effect.</u>

(b) Advances shall be incurred *pro rata* from Lenders (the "Non-Defaulting Lenders") which are not Defaulting Lenders based on their respective applicable Commitment Percentages, and no Commitment Percentage of any Lender or any *pro rata* share of any Advances required to be advanced by any Lender shall be increased as a result of such Lender Default. Amounts received in respect of principal of any type of Advances shall be applied to reduce the applicable Advances of each Lender (other than any Defaulting Lender) *pro rata* based on the aggregate of the outstanding Advances of that type of all Lenders at the time of such application; provided, that, Agent shall not be obligated to transfer to a Defaulting Lender any payments received by Agent for the Defaulting Lender's benefit, nor shall a Defaulting Lender be entitled to the sharing of any payments hereunder (including any principal, interest or fees). Amounts payable to a Defaulting Lender shall instead be paid to or retained by Agent and, if so paid or retained, shall be deemed to have been paid by Borrowers for all purposes hereunder. Agent may hold and, in its discretion, re-lend to Borrowers the amount of such payments received or retained by it for the account of such Defaulting Lender.

(c) <u>A Defaulting Lender shall not be entitled to give instructions to Agent or to approve, disapprove, consent to or vote on any</u> matters relating to this Agreement and the Other Documents. All amendments, waivers and other modifications of this Agreement and the Other Documents may be made without regard to a Defaulting Lender and, solely for purposes of any determination of "Required Lenders" in accordance with the definition thereof, a Defaulting Lender shall be deemed not to be a Lender and not to have either Advances outstanding or a <u>Commitment Percentage</u>.

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

(d) Other than as expressly set forth in this Section 2.9, the rights and obligations of a Defaulting Lender (including the obligation to indemnify Agent) and the other parties hereto shall remain unchanged. Nothing in this Section 2.9 shall be deemed to release any Defaulting Lender from its obligations under this Agreement and the Other Documents, shall alter such obligations, shall operate as a waiver of any default by such Defaulting Lender hereunder, or shall prejudice any rights which any Borrower, Agent or any Lender may have against any Defaulting Lender as a result of any default by such Defaulting Lender hereunder.

(e) <u>In the event a Defaulting Lender retroactively cures to the satisfaction of Agent the breach which caused a Lender to become a Defaulting Lender, such Defaulting Lender shall no longer be a Defaulting Lender and shall be treated as a Lender under this Agreement.</u>

II.10 Joint and Several Liability, Waivers, etc

. Each Borrower hereby agrees as follows.

(a) Each Borrower is accepting joint and several liability hereunder and under the Other Documents in consideration of the financial accommodations to be provided by Agent and the Lenders under this Agreement, for the mutual benefit, directly and indirectly, of each Borrower and in consideration of the undertakings of the other Borrowers to accept joint and several liability for the Obligations.

(b) <u>Each Borrower, jointly and severally, hereby irrevocably and unconditionally accepts, not merely as a surety but also as a co-debtor, joint and several liability with the other Borrowers, with respect to the payment and performance of all of the Obligations (including any Obligations arising under this Section 2.10), it being the intention of each Borrower and the parties hereto that all the Obligations shall be the joint and several obligations of each Borrower without preferences or distinction among them.</u>

(c) <u>If and to the extent that any Borrower shall fail to make any payment with respect to any of the Obligations as and when</u> due or to perform any of the Obligations in accordance with the terms thereof, then in each such event the other Borrowers will make such payment with respect to, or perform, such Obligation.

(d) <u>The Obligations of each Borrower under the provisions of this Section 2.10 constitute the absolute and unconditional, full</u> recourse Obligations of each Borrower enforceable against each Borrower to the full extent of its properties and assets, irrespective of the validity, regularity or enforceability of this Agreement or any other circumstances whatsoever.

(e) Except as otherwise expressly provided in this Agreement, to the fullest extent permitted by Applicable Law, each Borrower hereby waives notice of acceptance of its joint and several liability, notice of any Advances issued under or pursuant to this Agreement, notice of the occurrence of any Default, Event of Default, or of any demand for any payment under this Agreement, notice of any action at any time taken or omitted by Agent or Lenders

[Different first page setting changed from on in original to off in modified.].

under or in respect of any of the Obligations, any requirement of diligence or to mitigate damages and, generally, to the extent permitted by Applicable Law, all demands, notices and other formalities of every kind in connection with this Agreement (except as otherwise provided in this Agreement). Each Borrower hereby assents to, and waives, to the fullest extent permitted by Applicable Law, notice of, any extension or postponement of the time for the payment of any of the Obligations, the acceptance of any payment of any of the Obligations, the acceptance of any partial payment thereon, any waiver, consent or other action or acquiescence by Agent or Lenders at any time or times in respect of any default by any Borrower in the performance or satisfaction of any term, covenant, condition or provision of this Agreement, any and all other indulgences whatsoever by Agent or Lenders in respect of any of the Obligations, and the taking, addition, substitution or release, in whole or in part, at any time or times, of any security for any of the Obligations or the addition, substitution or release, in whole or in part, of any Borrower. Without limiting the generality of the foregoing, each Borrower assents to any other action or delay in acting or failure to act on the part of Agent or any Lender with respect to the failure by any Borrower to comply with any of its respective Obligations, including, without limitation, any failure strictly or diligently to assert any right or to pursue any remedy or to comply fully with Applicable Laws or regulations thereunder, which might, but for the provisions of this Section 2.10 afford grounds for terminating, discharging or relieving any Borrower, in whole or in part, from any of its Obligations under this Section 2.10, it being the intention of each Borrower that, so long as any of the Obligations hereunder remain unsatisfied, the Obligations of each Borrower under this Section 2.10 shall not be discharged except by performance and then only to the extent of such performance. The Obligations of each Borrower under this Section 2.10 shall not be diminished or rendered unenforceable by any winding up, reorganization, arrangement, liquidation, reconstruction or similar proceeding with respect to any Borrower or Agent or any Lender,

(f) Each Borrower represents and warrants to Agent and Lenders that such Borrower is currently informed of the financial condition of the other Borrowers and of all other circumstances which a diligent inquiry would reveal and which bear upon the risk of nonpayment of the Obligations. Each Borrower further represents and warrants to Agent and Lenders that such Borrower has read and understands the terms and conditions of this Agreement and the Other Documents. Each Borrower hereby covenants that such Borrower will continue to keep informed of the other Borrowers' financial condition, the financial condition of other guarantors, if any, and of all other circumstances which bear upon the risk of nonpayment or nonperformance of the Obligations.

(g) Each Borrower waives, to the maximum extent permitted by law, all rights and defenses arising out of an election of remedies by Agent or any Lender, even though that election of remedies, such as a nonjudicial foreclosure with respect to security for a guaranteed obligation, has destroyed Agent's or such Lender's rights of subrogation and reimbursement against any Borrower by the operation of Section 580(d) of the California Code of Civil Procedure, any comparable statute, or otherwise.

(h) [<u>Reserved</u>].

[Different first page setting changed from on in original to off in modified.].

(i) The provisions of this Section 2.10 are made for the benefit of Agent, Lenders and their respective successors and assigns, and may be enforced by it or them from time to time against any Borrower as often as occasion therefor may arise and without requirement on the part of Agent, any Lender, any of their respective successors or assigns first to marshal any of its or their claims or to exercise any of its or their rights against any Borrower or to exhaust any remedies available to it or them against any Borrower or to resort to any other source or means of obtaining payment of any of the Obligations hereunder or to elect any other remedy. The provisions of this Section 2.10 shall remain in effect until all of the Obligations shall have been paid in full in accordance with the terms of this Agreement. If at any time, any payment, or any part thereof, made in respect of any of the Obligations, is rescinded or must otherwise be restored or returned by Agent or any Lender upon the insolvency, bankruptcy or reorganization of any Borrower, or otherwise, the provisions of this Section 2.10 will forthwith be reinstated in effect, as though such payment had not been made.

(j) Until the Obligations (other than Inchoate Obligations) have been paid in full in cash in accordance with the terms hereof, all of the commitments of the Lenders hereunder have been terminated and this Agreement has been terminated, each Borrower hereby agrees that it will not enforce any of its rights of contribution or subrogation against any other Borrower with respect to any liability incurred by it hereunder or under any of the Other Documents, any payments made by it to Agent or any Lender with respect to any of the Obligations or any collateral security therefor until such time as all of the Obligations (other than Inchoate Obligations) have been paid in full in cash. Any claim which any Borrower may have against any other Borrower with respect to any payments to Agent or any Lender hereunder or under any Other Documents are hereby expressly made subordinate and junior in right of payment, without limitation as to any increases in the Obligations arising hereunder or thereunder, to the prior payment in full in cash of the Obligations (other than Inchoate Obligations) and, in the event of any insolvency, bankruptcy, receivership, liquidation, reorganization or other similar proceeding under the laws of any jurisdiction relating to any Borrower, its debts or its assets, whether voluntary or involuntary, all such Obligations shall be paid in full in cash before any payment or distribution of any character, whether in cash, securities or other property, shall be made to any other Borrower therefor.

(k) Each Borrower hereby agrees that, after the occurrence and during the continuance of any Event of Default, the payment of any amounts due with respect to the indebtedness or other obligations owing by any Borrower to any other Borrower is hereby subordinated to the prior payment in full in cash of the Obligations (other than Inchoate Obligations) in accordance with the terms of this Agreement. Each Borrower hereby agrees that after the occurrence and during the continuance of any Event of Default, such Borrower will not demand, sue for or otherwise attempt to collect any indebtedness of any other Borrower owing to such Borrower until the Obligations (other than Inchoate Obligations) shall have been paid in full in cash. If, notwithstanding the foregoing sentence, any Borrower shall collect, enforce or receive any amounts in respect of such indebtedness, such amounts shall be collected, enforced and received by such Borrower as trustee for Agent, and such Borrower shall deliver any such

[Different first page setting changed from on in original to off in modified.].

## amounts to Agent for application to the Obligations in accordance with the terms of this Agreement.

#### III. INTEREST AND FEES.

#### III.1 Interest

. Interest on Advances shall be payable in arrears on the last Business Day of each month. Interest charges shall be computed on the actual principal amount of Advances outstanding during the monthly period, at a rate per annum equal to the applicable Term Loan Rate. Whenever, subsequent to the date of this Agreement, the LIBOR RateAdjusted Term SOFR is increased or decreased, the Term Loan Rate shall be similarly changed without notice or demand of any kind by an amount equal to the amount of such change in the LIBOR RateAdjusted Term SOFR during the time such change or changes remain in effect. Upon and after the occurrence of an Event of Default, and during the continuation thereof, at the option of Agent or at the direction of Required Lenders, the Obligations shall bear interest at the Term Loan Rate plus two (2%) percent per annum (the "Default Rate").

III.2 [Reserved].

.

III.3 [Reserved].

÷

III.4 Fee Letter

. Borrowers shall pay the amounts required to be paid in the Fee Letter in the manner and at the times required by the Fee Letter.

### III.5 Computation of Interest and Fees

. Interest hereunder shall be computed on the basis of a 360 day year and for the actual number of days elapsed. If any payment to be made hereunder becomes due and payable on a day other than a Business Day, the due date thereof shall be extended to the next succeeding Business Day and interest thereon shall be payable at the applicable Term Loan Rate during such extension, unless such next succeeding Business Day would fall in the next calendar month, in which case the due date shall be on the immediately preceding Business Day.

## III.6 Maximum Charges

. In no event whatsoever shall interest and other charges charged hereunder exceed the highest rate permissible under law. In the event interest and other charges as computed hereunder would otherwise exceed the highest rate permitted under law, such excess amount shall be first applied to any unpaid principal balance owed by Borrowers, and if the then

[Different first page setting changed from on in original to off in modified.].

159823.01008/130146990v.3 159823.01008/150720893v.5 remaining excess amount is greater than the previously unpaid principal balance, Lenders shall promptly refund such excess amount to Borrowing Agent and the provisions hereof shall be deemed amended to provide for such permissible rate.

## III.7 Increased Costs

. In the event that any Applicable Law or any change therein or in the interpretation or application thereof, or compliance by any Lender (for purposes of this <u>Section 3.7</u>, the term "Lender" shall include Agent or any Lender and any corporation or bank controlling Agent or any Lender) and the office or branch where Agent or any Lender (as so defined) makes or maintains any Advances with any request or directive (whether or not having the force of law) from any central bank or other financial, monetary or other authority, shall:

(a) <u>subject Agent or any Lender to any Tax of any kind whatsoever (other than (A) Indemnified Taxes and (B) Excluded Taxes)</u> on the Advances, this Agreement, any Other Document, any commitments or other obligations or participations therein, or its deposits, reserves, other liabilities or capital attributable thereto;

(b) <u>impose, modify or deem applicable any reserve, special deposit, compulsory loan, insurance charge, assessment or similar</u> requirement against assets held by, or deposits in or for the account of, advances or loans by, or other credit extended by, any office of Agent or any Lender, including pursuant to Regulation D of the Board of Governors of the Federal Reserve System; or

(c) <u>impose on Agent or any Lender or the London-interbank market any other condition, loss or expense (other than Taxes)</u> affecting this Agreement or any Other Document or any Advance made by any Lender or participation therein;

and the result of any of the foregoing is to increase the cost to Agent or any Lender of making, renewing or maintaining its Advances hereunder by an amount that Agent or such Lender deems to be material or to reduce the amount of any payment (whether of principal, interest or otherwise) in respect of any of the Advances by an amount that Agent or such Lender deems to be material, then, in any case Borrowers shall within 30 days of the events or conditions giving rise thereto pay Agent or such Lender, upon its demand, such additional amount as will compensate Agent or such Lender for such additional cost or such reduction, as the case may be, <u>provided</u> that the foregoing shall not apply to increased costs which are reflected in the LIBOR RateAdjusted Term SOFR, as the case may be. Agent or such Lender shall certify the amount of such additional cost or reduced amount to Borrowing Agent (in writing and providing in reasonable detail the reasons for and calculations of such amounts), and such certification shall be conclusive evidence absent manifest error.

Failure or delay on the part of Agent or any Lender to demand compensation pursuant to the foregoing provisions of this <u>Section 3.7</u> shall not constitute a waiver of Agent's or such Lender's right to demand such compensation, <u>provided</u> that no Credit Party shall be required to compensate Agent or any Lender pursuant to the foregoing provisions of this <u>Section 3.7</u> for any

[Different first page setting changed from on in original to off in modified.].

increased costs incurred or reductions suffered more than nine (9) months prior to the date that Agent or such Lender notifies Borrowing Agent of the circumstance giving rise to such increased costs or reductions and of Agent's or such Lender's intention to claim compensation therefor (except that, if the circumstance giving rise to such increased costs or reductions involves a change in law that is retroactive, then the nine-month period referred to above shall be extended to include the period of retroactive effect thereof).

III.8 Benchmark Replacement Setting.

(a) <u>Replacing\_USD\_LIBOR. On March 5, 2021 the Financial Conduct Authority ("FCA"), the regulatory supervisor of USD</u> <u>LIBOR's administrator ("IBA"), announced in a public statement the future cessation or loss of representativeness of overnight/Spot Next, 1month, 3-month, 6-month and 12-month USD LIBOR tenor settings. On the earlier of (i) the date that all Available Tenors of USD LIBOR have either permanently or indefinitely ceased to be provided by IBA or have been announced by the FCA pursuant to public statement or publication of information to be no longer representative and (ii) the Early Opt-in Effective Date, if Benchmark Replacement. Notwithstanding anything to the contrary herein or in any Other Document, if a Benchmark Transition Event and its related Benchmark Replacement Date have occurred prior any setting of the then-current Benchmark is USD LIBOR, the Benchmark, then (i) if a Benchmark Replacement is determined in accordance with clause (a) of the definition of "Benchmark Replacement" for such Benchmark Replacement Date, such Benchmark Replacement will replace such Benchmark for all purposes hereunder and under this Agreement or any Other Document in respect of any setting of such Benchmark on such day and allsetting and subsequent Benchmark settings without any amendment to, or further action or consent of any other party to, this Agreement or any Other Document. If the, and (ii) if a Benchmark Replacement is based on Daily Simple SOFR pursuant to determined in accordance with clause (iib) of the definition of "Benchmark Replacement", all interest payments will be payable on a monthly basis.</u>

(b) for such <u>Replacing\_Future Benchmarks.</u> Upon the occurrence of a Benchmark Transition Event, the Benchmark Replacement <u>Date, such Benchmark Replacement</u> will replace the then-currentsuch Benchmark for all purposes hereunder and under this Agreement or any Other Document in respect of any Benchmark setting at or after 5:00 p.m. on the fifth (5<sup>th</sup>) Business Day after the date notice of such Benchmark Replacement is provided to the Lenders without any amendment to, or further action or consent of any other party to, this Agreement or any Other Document so long as Agent has not received, by such time, written notice of objection to such Benchmark Replacement from Lenders comprising the Required Lenders.

(a) (e)-Benchmark Replacement Conforming Changes. In connection with the implementation and use, administration, adoption or implementation of a Benchmark Replacement, Agent will have the right to make Benchmark Replacement Conforming Changes

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

from time to time and, notwithstanding anything to the contrary herein or in any Other Document, any amendments implementing such <u>Benchmark Replacement</u> Conforming Changes will become effective without any further action or consent of any other party to this Agreement or any Other Document.

(b) (d)-Notices; Standards for Decisions and Determinations. Agent will promptly notify the BorrowersBorrowing Agent and the Lenders of (i) the implementation of any Benchmark Replacement and (ii) the effectiveness of any Benchmark Replacement-Conforming Changes in connection with the use, administration, adoption or implementation of a Benchmark Replacement. Agent will promptly notify Borrowing Agent of the removal or reinstatement of any tenor of a Benchmark pursuant to Section 3.8. Any determination, decision or election that may be made by Agent or, if applicable, any Lender (or group of Lenders) pursuant to this Section 3.8. Any determination take or refrain from taking any action or any selection, will be conclusive and binding absent manifest error and may be made in its or their sole discretion and without consent from any other party heretoto this Agreement or any Other Document, except, in each case, as expressly required pursuant to this Section 3.8.8.

(c) (e)-Unavailability of Tenor of Benchmark. At Notwithstanding anything to the contrary herein or in any Other Document, at any time (including in connection with the implementation of a Benchmark Replacement), (i) if the then-current Benchmark is a term rate (including the Term SOFR or USD LIBOR), then Agent may remove any tenor of such Benchmark that is unavailable or non-representative for Benchmark (including Benchmark Replacement) settings and (ii) Agent may Reference Rate) and either (A) any tenor for such Benchmark is not displayed on a screen or other information service that publishes such rate from time to time as selected by Agent in its reasonable discretion or (B) the administrator of such Benchmark or the regulatory supervisor for the administrator of such Benchmark has provided a public statement or publication of information announcing that any tenor for such Benchmark is not or will not be representative or in compliance with or aligned with the International Organization of Securities Commissions (IOSCO) Principles for Financial Benchmarks, then Agent may modify the definition of "Interest Period" (or any similar or analogous definition) for any Benchmark settings at or after such time to remove such unavailable, nonrepresentative, non-compliant or non-aligned tenor and (ii) if a tenor that was removed pursuant to clause (i) above either (A) is subsequently displayed on a screen or information service for a Benchmark (including a Benchmark Replacement) or (B) is not, or is no longer, subject to an announcement that it is not or will not be representative or in compliance with or aligned with the International Organization of Securities Commissions (IOSCO) Principles for Financial Benchmarks for a Benchmark (including a Benchmark Replacement), then Agent may modify the definition of "Interest Period" and the related provisions hereof (or any similar or analogous definition) for all Benchmark settings at or after such time to reinstate any such previously removed tenor for Benchmark (including Benchmark Replacement) settings.

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

(d) Benchmark Unavailability Period. Upon the Borrowing Agent's receipt of notice of the commencement of a Benchmark Unavailability Period, the Borrowing Agent may revoke any pending request for an Advance of, conversion to or continuation of SOFR Loans to be made, converted or continued during any Benchmark Unavailability Period.

### (f) Definitions Applicable to this Section 3.8.

"<u>Available Tenor</u>" means, as of any date of determination and with respect to the then-current Benchmark, as applicable, (x) if the then-current Benchmark is a term rate, any tenor for such Benchmark that is or may be used for determining the length of the three month period or (y) otherwise, any payment period for interest calculated with reference to such Benchmark, as applicable, pursuant to this Agreement as of such date.

"Benchmark" means, initially, USD LIBOR; provided that if a replacement of the Benchmark has occurred pursuant to this Section 3.8, then "Benchmark" means the applicable Benchmark Replacement to the extent that such Benchmark Replacement has replaced such prior benchmark rate. Any reference to "Benchmark" shall include, as applicable, the published component used in the calculation thereof.

"Benchmark Replacement" means, for any Available Tenor, the alternative set forth below that that has been selected by the Agent and can be determined by the Agent:

(i) the sum of: (x) Term SOFR and (y) 0.10% (10 basis points) for an Available Tenor of one month's duration, 0.15% (15 basis points) for an Available Tenor of three-months' duration, and 0.25% (25 basis points) for an Available Tenor of six-months' duration, duration,

(ii) the sum of: (x) Daily Simple SOFR and (y) the spread adjustment selected or recommended by the Relevant Governmental Body for the replacement of the tenor of USD LIBOR with a SOFR-based rate having approximately the same length as the interest payment period specified in <u>clause (a)</u> of this <u>Section 3.8</u>, or

(iii) the sum of: (x) the alternate benchmark rate that has been selected by the Agent as the replacement for the then-current Benchmark for the applicable corresponding tenor giving due consideration to (i) any selection or recommendation of a replacement benchmark rate or the mechanism for determining such a rate by the Relevant Governmental Body or (ii) any evolving or then-prevailing market convention for determining a benchmark rate as a replacement for the then-current Benchmark for U.S. dollar-denominated syndicated credit facilities at such time and (y) an adjustment (which may be a positive or negative value or zero), in each case, that has been selected by Agent as the replacement for such Available Tenor of such Benchmark giving due consideration to any evolving or then-prevailing market convention, including any applicable recommendations made by the

[Different first page setting changed from on in original to off in modified.].

Relevant Governmental Body, for U.S. dollar-denominated syndicated credit facilities at such time;

provided that, if the Benchmark Replacement as determined pursuant to this definition of "Benchmark Replacement" would be less than the Floor, the Benchmark Replacement will be deemed to be the Floor for the purposes of this Agreement and the Other Documents.

Any Benchmark Replacement shall be applied in a manner consistent with market practice.

"Benchmark Replacement Conforming Changes" means, with respect to any Benchmark Replacement, any technical, administrative or operational changes (including, without limitation, changes to timing and frequency of determining rates and making payments of interest, timing of borrowing requests or prepayment, conversion or continuation notices, the applicability and length of lookback periods, the applicability of breakage provisions, and other technical, administrative or operational matters) that Agent decides may be appropriate to reflect the adoption and implementation of such Benchmark Replacement and to permit the administration thereof by Agent in a manner substantially consistent with market practice (or, if Agent decides that adoption of any portion of such market practice is not administratively feasible or if Agent decides that no market practice for the administration of such Benchmark Replacement exists, in such other manner of administration as Agent decides is reasonably necessary in connection with the administration of this Agreement and the Other Documents).

"Benchmark Transition Event" means, with respect to any then-current Benchmark other than USD LIBOR, the occurrence of a public statement or publication of information by or on behalf of the administrator of the then-current Benchmark, the regulatory supervisor for the administrator of such Benchmark, the Board of Governors of the Federal Reserve System, the Federal Reserve Bank of New York, an insolvency official with jurisdiction over the administrator for such Benchmark, a resolution authority with jurisdiction over the administrator for such Benchmark, a resolution authority over the administrator for such Benchmark, announcing or stating that (a) such administrator has ceased or will cease on a specified date to provide all Available Tenors of such Benchmark, permanently or indefinitely, provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark or (b) all Available Tenors of such Benchmark are or will no longer be representative of the underlying market and economic reality that such Benchmark is intended to measure and that representativeness will not be restored.

"Daily Simple SOFR" means, for any day, SOFR, with the conventions for this rate (which will include a lookback) being established by Agent in a manner consistent

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

[Different first page setting changed from on in original to off in modified.].

with market practice in accordance with the conventions for this rate recommended by the Relevant Governmental Body for determining "Daily Simple SOFR" for syndicated business loans; <u>provided</u>, that if Agent decides that any such convention is not administratively feasible for Agent, then Agent may establish another convention consistent with market practice in its reasonable discretion.

"Early Opt-in Effective Date" means, with respect to any Early Opt-in Election, the sixth (6th) Business Day after the date notice of such Early Opt-in Election is provided to the Lenders, so long as Agent has not received, by 5:00 p.m. (New York City time) on the fifth (5th) Business Day after the date notice of such Early Opt-in Election is provided to the Lenders, written notice of objection to such Early Opt-in Election from Lenders comprising the Required Lenders.

"Early Opt-in Election" means the occurrence of:

(a) a notification by Agent to (or the request by the Borrowers to Agent to notify) each of the other parties hereto that at least five currently outstanding U.S. dollar-denominated syndicated credit facilities at such time contain (as a result of amendment or as originally executed) a SOFR-based rate (including SOFR, a term SOFR or any other rate based upon SOFR) as a benchmark rate (and such syndicated credit facilities are identified in such notice and are publicly available for review), and

(b) the joint election by Agent and the Borrowers to trigger a fallback from USD LIBOR and the provision by Agent of written notice of such election to the Lenders.

"<u>Floor</u>" means the benchmark rate floor, if any, provided in this Agreement initially (as of the execution of this Agreement, the modification, amendment or renewal of this Agreement or otherwise) with respect to USD LIBOR.

"<u>Relevant Governmental Body</u>" means the Board of Governors of the Federal Reserve System or the Federal Reserve Bank of New York, or a committee officially endorsed or convened by the Board of Governors of the Federal Reserve System or the Federal Reserve Bank of New York, or any successor thereto.

"<u>SOFR</u>" means a rate per annum equal to the secured overnight financing rate for such Business Day published by the Federal Reserve Bank of New York (or a successor administrator of the secured overnight financing rate) on the website of the Federal Reserve Bank of New York, eurrently at http://www.newyorkfed.org (or any successor source for the secured overnight financing rate identified as such by the administrator of the secured overnight financing rate from time to time).

[Different first page setting changed from on in original to off in modified.].

"<u>Term SOFR</u>" means, for the applicable corresponding tenor, the forward-looking term rate based on SOFR that has been selected or recommended by the Relevant Governmental Body.

"USD LIBOR" means the London interbank offered rate for U.S. dollars.

III.9 <u>Capital Adequacy.</u>

(a) In the event that Agent or any Lender shall have determined that any Applicable Law or guideline regarding capital adequacy, or any change therein, or any change in the interpretation or administration thereof by any Governmental Body, central bank or comparable agency charged with the interpretation or administration thereof, or compliance by Agent or any Lender (for purposes of this Section 3.9, the term "Lender" shall include Agent or any Lender and any corporation or bank controlling Agent or any Lender) and the office or branch where Agent or any Lender (as so defined) makes or maintains any Advances with any request or directive regarding capital adequacy (whether or not having the force of law) of any such authority, central bank or comparable agency, has or would have the effect of reducing the rate of return on Agent or any Lender's capital as a consequence of its obligations hereunder to a level below that which Agent or such Lender could have achieved but for such adoption, change or compliance (taking into consideration Agent's and each Lender's policies with respect to capital adequacy) by an amount deemed by Agent or any Lender to be material, then, from time to time, Borrowers shall pay upon demand to Agent or such Lender may use any reasonable averaging or attribution methods. The protection of this Section 3.9 shall be available to Agent and each Lender regardless of any possible contention of invalidity or inapplicability with respect to the Applicable Law or condition.

(b) <u>A certificate of Agent or such Lender setting forth such amount or amounts as shall be necessary to compensate Agent or such Lender with respect to Section 3.9(a) hereof when delivered to Borrowing Agent shall be conclusive evidence absent manifest error.</u>

(c) Failure or delay on the part of Agent or any Lender to demand compensation pursuant to the foregoing provisions of this Section 3.9 shall not constitute a waiver of Agent's or such Lender's right to demand such compensation, provided that no Credit Party shall be required to compensate Agent or any Lender pursuant to the foregoing provisions of this Section 3.9 for any reductions suffered more than six months prior to the date that Agent or such Lender notifies Borrowing Agent of the circumstance giving rise to such reductions and of such Lender's intention to claim compensation therefor (except that, if the circumstance giving rise to such increased costs or reductions involves a change in law that is retroactive, then

[Different first page setting changed from on in original to off in modified.].

[Different first page setting changed from on in original to off in modified.].

the six-month period referred to above shall be extended to include the period of retroactive effect thereof).

III.10 Taxes.

(a) [Reserved].

(b) Payment Free of Taxes. Any and all payments by or on account of any Obligations of any Credit Party under this Agreement or any Other Document shall be made without deduction or withholding for any Taxes, except as required by Applicable Law. If any Applicable Law (as determined in the good faith discretion of Borrowing Agent or a Payee) requires the deduction or withholding of any Tax from any such payment by a Payee, then Borrowing Agent or such Payee, as applicable, shall be entitled to make such deduction or withholding and shall timely pay the full amount deducted or withheld to the relevant Governmental Body in accordance with Applicable Law and, if such Tax is an Indemnified Tax, then the sum payable by the applicable Credit Party shall be increased as necessary so that after such deduction or withholding has been made (including such deductions and withholdings applicable to additional sums payable under this Section) the Payee, as the case may be, receives an amount equal to the sum it would have received had no such deduction or withholding been made. For clarity, notwithstanding the submission of documentation by a Payee under Section 3.10(g)(i) claiming a reduced rate of or exemption from U.S. withholding Tax, Agent shall be entitled to withhold United States federal income Taxes at the full 30% withholding rate if in its reasonable judgment it is required to do so under the due diligence requirements imposed upon Agent under § 1.1441-7(b) of the United States Income Tax Regulations or other Applicable Law.

(c) <u>Payment of Other Taxes by the Credit Parties. The Credit Parties shall timely pay to the relevant Governmental Body in</u> accordance with Applicable Law, or at the option of Agent timely reimburse it for the payment of, any Other Taxes.

(d) Indemnification by the Credit Parties. Each Credit Party shall jointly and severally indemnify Agent and each Payee, within ten (10) days after demand therefor, for the full amount of any Indemnified Taxes (including Indemnified Taxes imposed or asserted on or attributable to amounts payable under this Section) payable or paid by Agent or any Payee or required to be withheld or deducted from a payment to such Agent or Payee, as the case may be, and any penalties, interest and out-of-pocket expenses arising therefrom or with respect thereto, whether or not such Indemnified Taxes were correctly or legally imposed or asserted by the relevant Governmental Body. A certificate as to the amount of such payment or liability delivered to Borrowers by any Payee (with a copy to Agent), or by Agent on its own behalf of a Payee shall be conclusive absent demonstrable error.

(e) Indemnification by the Payees. Each Payee shall severally indemnify Agent, within ten (10) days after demand therefor, for (i) any Indemnified Taxes attributable to

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

such Payee (but only to the extent that any Credit Party has not already indemnified Agent for such Indemnified Taxes and without limiting the obligation of the Credit Parties to do so), and (ii) any Excluded Taxes attributable to such Payee, in each case, that are payable or paid by Agent in connection with this Agreement or any Other Document, and any reasonable expenses arising therefrom or with respect thereto, whether or not such Taxes were correctly or legally imposed or asserted by the relevant Governmental Body. In addition, and for clarity, Agent is hereby indemnified severally by the Lenders under § 1.1461-1(e) of the United States Income Tax Regulations against any claims and demands of any Payee for the amount of any Tax it deducts and withholds in accordance with regulations under § 1441 of the Code. A certificate as to the amount of such payment or liability delivered to any Payee by Agent shall be conclusive absent manifest error. Each Payee hereby authorizes Agent to set off and apply any and all amounts at any time owing to such Payee under any Other Document or otherwise payable by Agent to the Payee from any other source against any amount due to Agent under this clause (e).

(f) <u>Evidence of Payments. As soon as practicable after any payment of Taxes by any Credit Party to a Governmental Body</u> pursuant to this Section 3.10, such Credit Party shall deliver to Agent the original or a certified copy of a receipt issued by such Governmental Body evidencing such payment, a copy of the return reporting such payment or other evidence of such payment reasonably satisfactory to Agent.

(g) Status of Payees. (i) Any Payee that is entitled to an exemption from or reduction of withholding Tax with respect to payments hereunder or under any Other Document shall deliver to Borrowers and Agent, at the time or times reasonably requested by Borrowing Agent or Agent, such properly completed and executed documentation prescribed by Applicable Law as will permit such payments to be made without withholding or at a reduced rate of withholding. In addition, any Payee, if reasonably requested by Borrowing Agent or Agent, shall deliver such other documentation prescribed by Applicable Law or reasonably requested by Borrowing Agent or Agent, shall deliver such other documentation prescribed by Applicable Law or reasonably requested by Borrowing Agent or Agent, shall enable Borrowing Agent or Agent to determine whether or not such Payee is subject to backup withholding or information reporting requirements. Notwithstanding anything to the contrary in the preceding two sentences, the completion, execution and submission of such documentation (other than such documentation set forth in Sections 3.10(g)(A), 3.10(g)(B) and Error! Reference source not found. 1.1(g)(ii)(A), 1.1(g)(ii)(B) and 3.10(g)(ii)(D) below) shall not be required if in Payee's reasonable judgment such completion, execution or submission would subject such Payee to any material unreimbursed cost or expense or would materially prejudice the legal or commercial position of such Payee, and Payee hereby acknowledges and agrees that in such circumstance Borrowers and Agent shall withhold against all payments to be made by Borrowers or Agent to such Payee in accordance with Applicable Law.

(ii) Without limiting the generality of the foregoing, in the event that any Borrower is a "United States Person" as defined in Section 7701(a)(30) of the Code (a "U.S. Person"),

(A) any Payee that is a U.S. Person shall deliver to Borrowing Agent and Agent on or prior to the date on which such Payee becomes a Payee under this

[Different first page setting changed from on in original to off in modified.].

Agreement (and from time to time thereafter upon the reasonable request of the Borrower or Agent), two (2) duly completed and executed copies of IRS Form W-9 certifying that such Payee is exempt from U.S. federal backup withholding Tax;

(B) any Payee that is organized under the laws of a jurisdiction other than that in which the Credit Parties are resident for income Tax purposes (a "Foreign Lender") shall, to the extent it is legally entitled to do so, deliver to Borrowing Agent and Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Payee under this Agreement (and from time to time thereafter upon the reasonable request of any Borrower or Agent), whichever of the following is applicable:

a. in the case of a Foreign Lender claiming the benefits of an income tax treaty to which the United States is a party (x) with respect to payments of interest under this Agreement or any Other Document, two (2) duly completed and executed copies of IRS Form W-8BEN or W-8BEN-E (as applicable) establishing an exemption from, or reduction of, U.S. federal withholding Tax pursuant to the "interest" article of such tax treaty and (y) with respect to any other applicable payments under this Agreement or any Other Document, two (2) duly completed and executed copies of IRS Form W-8BEN or W-8BEN or W-8BEN or W-8BEN or W-8BEN or any Other Document, two (2) duly completed and executed copies of IRS Form W-8BEN or W-8BEN-E (as applicable) establishing an exemption from, or reduction of, U.S. federal withholding Tax pursuant to the "business profits" or "other income" article of such tax treaty;

b. two (2) duly completed and executed copies of IRS Form W-8ECI;

c. in the case of a Foreign Lender claiming the benefits of the exemption for portfolio interest under Section 881(c) of the Code, (x) two (2) duly completed and executed copies of a certificate in form and substance satisfactory to Agent to the effect that such Foreign Lender is not a "bank" within the meaning of Section 881(c)(3)(A) of the Code, a "10 percent shareholder" of the Borrowers within the meaning of Section 881(c)(3)(B) of the Code, or a "controlled foreign corporation" described in Section 881(c)(3)(C) of the Code (a "U.S. Tax Compliance Certificate") and (y) two (2) duly completed and executed copies of IRS Form W-8BEN or IRS Form W-8BEN-E, as applicable; or

d. to the extent a Foreign Lender is not the beneficial owner, two (2) duly completed and executed copies of IRS Form W-8IMY, accompanied by two (2) duly completed and executed copies of IRS Form W-8ECI, IRS Form W-8BEN or W-8BEN-E (as applicable), two (2) duly completed and executed copies of a U.S. Tax Compliance Certificate in form and substance satisfactory to Agent, two (2) duly completed and executed copies of IRS Form W-9, and/or other certification documents from each beneficial owner, as applicable; <u>provided</u> that if the Foreign Lender is a partnership and one or more direct or indirect partners of such Foreign Lender are claiming the portfolio interest exemption, such Foreign Lender may provide two (2) duly completed and executed copies of a U.S. Tax Compliance Certificate in form and substance Satisfactory to Agent on behalf of each such direct and indirect partner;

[Different first page setting changed from on in original to off in modified.].

159823.01008/130146990v.3 159823.01008/150720893v.5

(C) any Foreign Lender shall, to the extent it is legally entitled to do so, deliver to Borrowing Agent and Agent (in such number of copies as shall be requested by the recipient) on or prior to the date on which such Foreign Lender becomes a Payee under this Agreement (and from time to time thereafter upon the reasonable request of Borrowing Agent or Agent), executed copies of any other form prescribed by Applicable Law as a basis for claiming exemption from or a reduction in U.S. federal withholding Tax, duly completed, together with such supplementary documentation as may be prescribed by applicable law to permit Borrowing Agent or Agent to determine the withholding or deduction required to be made;

(D) if a payment to a Payee under this Agreement or any Other Document would be subject to U.S. Federal withholding Taxes imposed by FATCA if such Payee were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the Code, as applicable), such Payee shall deliver to Borrowing Agent and Agent at the time or times prescribed by law and at such time or times reasonably requested by Borrowing Agent or Agent such documentation prescribed by Borrowing Agent or Agent as may be necessary for Credit Parties and Agent to comply with their obligations under FATCA and to determine that such Payee has complied with such Payee's obligations under FATCA or to determine the amount to deduct and withhold from such payment. Solely for purposes of this clause (D), "FATCA" shall include any amendments made to FATCA after the date of this Agreement; and

(E) Any Agent that is not a U.S. Person shall, at the time such Agent becomes a party to this Agreement, provide to the Borrowing Agent two duly-signed properly completed copies of (1) IRS Form W-8ECI, or any successor thereto with respect to payments, if any, received by the Agent for its own account, and (2) with respect to payments received on account of any Payee, executed copies of IRS Form W-8IMY (or any successor form) certifying that the Agent is either (a) a "qualified intermediary" and that it assumes primary withholding responsibility under Chapters 3 and 4 of the Code and primary Form 1099 reporting and backup withholding responsibility for payments it receives for the account of others or (b) a "U.S. branch" and that payment it receives for others are not effectively connected with the conduct of a trade or business in the United States and that is using the form as evidence of its agreement to be treated as a U.S. Person with respect to payments associated with the IRS Form W-8IMY.

Each Payee agrees that if any form or certification it previously delivered expires or becomes obsolete or inaccurate in any respect, it shall update such form or certification or promptly notify Borrowing Agent and Agent in writing of its legal inability to do so.

(h) Treatment of Certain Refunds. If any party determines, in its sole discretion exercised in good faith, that it has received a refund of any Taxes as to which it has been indemnified pursuant to this Section 3.10 (including by the payment of additional amounts

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

pursuant to this Section 3.10), it shall pay to the indemnifying party an amount equal to such refund (but only to the extent of indemnity payments made under this Section with respect to the Taxes giving rise to such refund), net of all out-of-pocket expenses (including Taxes) of such indemnified party and without interest (other than any interest paid by the relevant Governmental Body with respect to such refund). Such indemnifying party, upon the request of such indemnified party, shall repay to such indemnified party the amount paid over pursuant to this clause (h) (plus any penalties, interest or other charges imposed by the relevant Governmental Body) in the event that such indemnified party is required to repay such refund to such Governmental Body. Notwithstanding anything to the contrary in this clause (h), in no event will the indemnified party be required to pay any amount to an indemnifying party pursuant to this clause (h) the payment of which would place the indemnified party in a less favorable net after-Tax position than the indemnified party would have been in if the Tax subject to indemnification and giving rise to such refund had not been deducted, withheld or otherwise imposed and the indemnified party to make available its Tax returns (or any other information relating to its Taxes that it deems confidential) to the indemnifying party or any other Person.

(i) <u>Survival. Each party's obligations under this Section 3.10 shall survive the resignation or replacement of Agent or any</u> assignment of rights by, or the replacement of, a Payee, the termination of the commitments and the repayment, satisfaction or discharge of all obligations under this Agreement or any Other Document.

### III.11 Replacement of Lenders

. If any Lender (an "Affected Lender") (a) makes demand upon any Credit Party for (or if any Credit Party is otherwise required to pay) amounts pursuant to <u>Sections 3.7</u>, <u>3.9</u> or <u>3.10</u> or (b) is a Defaulting Lender, then such Lender shall use reasonable efforts to mitigate such losses, including in the case of clause (a), by designating a different lending office or Affiliate of such Lender for the funding or booking the Loans hereunder and Borrowing Agent may, within ninety (90) days of receipt of such demand (or the occurrence of such other event causing any Credit Party to be required to pay such compensation) or Lender Default, as the case may be, by notice (a "<u>Replacement Notice</u>") in writing to Agent and such Affected Lender (i) request the Affected Lender to cooperate with Borrowing Agent in obtaining a replacement Lender who is an Eligible Assignee and is satisfactory to Agent and Borrowing Agent in their sole discretion (the "<u>Replacement Lender</u>"); (ii) request the non-Affected Lender's Advances and Commitment Percentage as provided herein, but none of such Lenders shall be under an obligation to do so; or (iii) designate a Replacement Lender who is an Eligible Assignee and is satisfactory to Agent and Borrowing Agent in their sole discretion. If any satisfactory Replacement Lender shall be obtained, and/or if any one or more of the non-Affected Lender's Advances and Commitment Percentage, then such Affected Lender shall assign, in accordance with <u>Section 15.3</u>, all of its Advances and Commitment Percentage and obligations under this Agreement and the Other Documents to such Replacement Lender or non-Affected

[Different first page setting changed from on in original to off in modified.].

Lenders, as the case may be, in exchange for payment of the principal amount so assigned and all interest and fees accrued on the amount so assigned, plus all other Obligations then due and payable to the Affected Lender; <u>provided</u>, <u>however</u>, that (A) such assignment shall be without recourse, representation or warranty and shall be on terms and conditions reasonably satisfactory to such Affected Lender and such Replacement Lender and/or non-Affected Lenders, as the case may be, and (B) prior to any such assignment, the Borrowers shall have paid to such Affected Lender all amounts properly demanded and unreimbursed under <u>Sections 3.7</u> and <u>3.9</u>. Upon the effective date of such assignment, the Borrowers shall, if requested, issue replacement Notes to such Replacement Lender and/or non-Affected Lenders, as the case may be, and such institution(s) shall become a "Lender" for all purposes under this Agreement and the other Documents.

## IV. COLLATERAL: GENERAL TERMS.

### IV.1 Security Interest in the Collateral

. To secure the prompt payment and performance to Agent and each Lender (and each other holder of any Obligations) of the Obligations, each Credit Party hereby collaterally assigns, pledges and grants to Agent for its benefit and for the ratable benefit of each Lender a continuing security interest in and to and Lien on all of its Collateral, whether now owned or existing or hereafter created, acquired or arising and wherever located. Each Credit Party shall mark its books and records as may be necessary or appropriate to evidence and perfect Agent's security interest and shall cause its financial statements to reflect such security interest. Each Credit Party shall promptly provide Agent with written notice of all commercial tort claims, such notice to contain the case title together with the applicable court and a brief description of the claim(s). Upon delivery of each such notice, such Credit Party shall be deemed to hereby grant to Agent a security interest and lien in and to such commercial tort claims and all proceeds thereof.

### IV.2 Creation and Perfection of Security Interest

(a) This Agreement and the Other Documents are effective to create in favor of Agent, for the ratable benefit of Lenders and the other holders of the Obligations, a legal, valid and enforceable security interest in the Collateral identified therein, except to the extent the enforceability thereof may be limited by applicable Debtor Relief Laws affecting creditors' rights generally and by equitable principles of law (regardless of whether enforcement is sought in equity or at law), and this Agreement and the Other Documents shall create a fully perfected Lien on, and security interest in, all right, title and interest of the obligors thereunder in such Collateral, in each case prior and superior in right to any other Lien (other than Permitted Encumbrances) (i) with respect to any such Collateral that is a "security" (as such term is defined in the Uniform Commercial Code) and is evidenced by a certificate, when such Collateral is delivered to Agent with duly executed stock powers with respect thereto, (ii) with respect to any such Collateral that is a "security" (as such term is defined in the Uniform Commercial Code)

[Different first page setting changed from on in original to off in modified.].

but is not evidenced by a certificate, when UCC financing statements in appropriate form are filed in the appropriate filing offices in the jurisdiction of organization of the pledgor or when "control" (as such term is defined in the Uniform Commercial Code) is established by Agent over such interests in accordance with the provision of Section 8-106 of the Uniform Commercial Code, or any successor provision, (iii) with respect to any such Collateral that is not a "security" (as such term is defined in the Uniform Commercial Code), when UCC financing statements in appropriate form are filed in the appropriate filing offices in the jurisdiction of organization of the pledgor (to the extent such security interest can be perfected by filing under the Uniform Commercial Code) and (iv) with respect to any such Collateral that is Intellectual Property, when intellectual property security agreements are recorded with the United States Patent and Trademark Office and the United States Copyright Office, together with the financing statements set forth in clause (iii).

Each Credit Party shall take all action that may be necessary or that Agent may request in its Permitted Discretion, so as at all times to maintain the validity, perfection, enforceability and priority of Agent's security interest in and Lien on the Collateral or to enable Agent to protect, exercise or enforce its rights hereunder and in the Collateral including, but not limited to, (i) promptly discharging all Liens other than Permitted Encumbrances, (ii) using commercially reasonable efforts to obtain Lien Waiver Agreements to the extent required under Section 4.5(c), (iii) delivering to Agent, endorsed or accompanied by such instruments of assignment as Agent may specify, and stamping or marking, in such manner as Agent may specify, any and all chattel paper, instruments (other than those deposited for collection), letters of credit and advices thereof and documents evidencing or forming a part of the Collateral, in each case with a face value in excess of \$250,000, individually and in the aggregate, (iv) reserved, and (v) executing and delivering financing statements, Control Agreements, instruments of pledge, mortgages, notices and assignments, in each case in form and substance satisfactory to Agent in its Permitted Discretion, relating to the creation, validity, perfection, maintenance or continuation of Agent's security interest and Lien under the Uniform Commercial Code or other Applicable Law. By its signature hereto, each Credit Party hereby authorizes Agent to file against such Credit Party, one or more financing, continuation or amendment statements pursuant to the Uniform Commercial Code in form and substance satisfactory to Agent (which statements may have a description of collateral which is broader than that set forth herein, including without limitation a description of Collateral as "all assets" and/or "all personal property" of any Credit Party). All charges, expenses and fees Agent may incur in doing any of the foregoing, and any local filing, recording, documentary or similar taxes relating thereto, shall be charged to Borrowers' Account as an Advance and added to the Obligations, or, at Agent's option, shall be paid to Agent for its benefit and for the ratable benefit of Lenders immediately upon demand. Notwithstanding the foregoing, no Credit Party shall be required to undertake any Excluded Perfection Action.

IV.3 [<u>Reserved</u>].

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

# IV.4 Preservation of Collateral

. Following the occurrence and during the continuance of an Event of Default, in addition to the rights and remedies set forth in <u>Section 11.1</u> hereof, Agent: (a) may at any time take such steps as Agent deems reasonably necessary to protect Agent's interest in and to preserve the Collateral and (b) shall have, and is hereby granted, a right of ingress and egress to the places where the Collateral is located to take possession of all or any part of the Collateral and to remove all or any part of the Collateral. Each Credit Party shall cooperate all of Agent's reasonable efforts to preserve the Collateral and will take such actions to preserve the Collateral as Agent may reasonably direct. All of Agent's expenses of preserving the Collateral shall be charged to Borrowers' Account as an Advance and added to the Obligations.

# IV.5 Ownership of Collateral.

÷

Each of the Credit Parties and its Subsidiaries has (i) good, sufficient and legal title to (in the case of fee interests in Real (a) Property), (ii) valid leasehold interests in (in the case of leasehold interests in real or personal property), and (iii) good title to (in the case of all other personal property), all of their respective properties and assets reflected in their financial statements and other information referred to in Section 5.9 and in the most recent financial statements delivered pursuant to Sections 9.7, 9.8 and 9.9, in each case except for assets disposed of since the date of such financial statements as permitted under Section 7.1(b). All such properties and assets are free and clear of Liens other than Permitted Encumbrances. With respect to the Collateral, at the time the Collateral becomes subject to Agent's security interest; (i) a Credit Party. shall have rights in or be the sole owner of and fully authorized and able to sell, transfer, pledge and/or grant a first priority (subject only to Permitted Encumbrances) security interest in each and every item of its Collateral to Agent; (ii) each document and agreement executed by each Credit Party or delivered to Agent or any Lender in connection with this Agreement shall be true and correct in all material respects; (iii) all signatures and endorsements of each Credit Party that appear on such documents and agreements shall be genuine and such Credit Party shall have full capacity to execute same; and (iv) each Credit Party's Equipment with a value in excess of \$100,000 shall be located as of the Closing Date (and as updated pursuant to Sections 4.5 and 9.11(b)) as set forth on Schedule 4.5 and shall not be removed from such location(s) without the prior written consent of Agent; except (A) with respect to the sale of Inventory in the ordinary course of business and Equipment to the extent permitted in Section 7.1 hereof; and (B) for items in transit between Credit Parties' locations, items out for repair, items provided to employees in the ordinary course of business and samples for customers.

# (b) [<u>Reserved].</u>

(c) <u>Notwithstanding anything contained herein to the contrary and without limiting anything contained herein, Borrower shall</u> use commercially reasonable efforts to deliver or cause to be delivered to Agent duly executed Lien Waiver Agreements in form and

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

substance reasonably satisfactory to Agent with respect to (w) 1177 West Loop South, Suite 1170, Houston, Texas 77027, (x) 716 Congress Avenue, Suite 100, Austin, Texas 78701, (y) each other Real Property that has Collateral with a book or fair market value (whichever is greater) in excess of \$250,000 stored or located therein and (z) upon the reasonable request of Agent, any other leased locations where corporate records are maintained.

### IV.6 Defense of Agent's and Lenders' Interests

. Until (a) payment and performance in full of all of the Obligations (other than Inchoate Obligations) and (b) termination of all obligations of Agent and the Lenders to make any Advances hereunder and termination of this Agreement, Agent's interests in the Collateral shall continue in full force and effect. During such period no Credit Party shall, without Agent's prior written consent, pledge, sell (except Inventory in the ordinary course of business and other Collateral to the extent not constituting an Asset Sale), assign, transfer, create or suffer to exist a Lien upon or encumber or allow or suffer to be encumbered in any way except for Permitted Encumbrances and except as permitted by Section 7.1, any part of the Collateral. Each Borrower shall use commercially reasonable efforts to defend Agent's interests in any material portion of the Collateral against any and all Persons whatsoever. At any time following demand by Agent for payment of all Obligations after the occurrence and during the continuance of an Event of Default, Agent shall have the right to take possession of the indicia of the Collateral and the Collateral in whatever physical form contained, including: labels, stationery, documents, instruments and advertising materials. If Agent exercises this right to take possession of the Collateral, each Borrower shall, upon demand, assemble it in the manner requested by Agent, to the extent commercially reasonable, and make it available to Agent at a place reasonably convenient to Agent. In addition, with respect to all Collateral, at all times following the occurrence and during the continuance of an Event of Default, Agent and Lenders shall be entitled to all of the rights and remedies set forth herein and further provided by the Uniform Commercial Code or other Applicable Law. Following the occurrence and during the continuance of an Event of Default, Agent may, at its option, instruct all suppliers, carriers, forwarders, warehousers or others receiving or holding cash, checks, documents or instruments in which Agent holds a security interest to deliver same to Agent and/or subject to Agent's order and if they shall come into such Credit Party's possession, they, and each of them, shall be held by such Credit Party in trust as Agent's trustee, and such Credit Party will promptly deliver them to Agent in their original form together with any necessary endorsement.

#### IV.7 Books and Records

. Each Credit Party shall keep proper books of record and account in which full, true and correct entries shall be made of all dealings or transactions in relation to its business and activities to the extent necessary to prepare the financial statements of DDH and its Subsidiaries, on a Consolidated Basis, in conformity with GAAP consistently applied in the opinion of such independent public accountant as shall then be regularly engaged by DDH.

IV.8 Financial Disclosure

[Different first page setting changed from on in original to off in modified.].

. Subject to applicable federal and state securities laws, each Credit Party hereby irrevocably authorizes and directs all accountants and auditors employed by such Credit Party at any time during the Term to deliver to Agent copies of any Credit Party's financial statements, trial balances or other accounting records of any sort in the accountant's or auditor's possession, and to disclose to Agent any information such accountants may have concerning any Credit Party's financial status and business operations; provided however, Agent will attempt to obtain such information or materials directly from Credit Party hereby authorizes all Governmental Bodies to furnish to Agent copies of reports or examinations relating to any Credit Party, whether made by such Credit Party or otherwise; provided however, Agent will attempt to obtain such information or materials directly from Credit Party, whether made by such Credit Party or otherwise; provided however, Agent will attempt to obtain such information or materials directly from Credit Party hereby authorizes all Governmental Bodies to furnish to Agent copies of reports or examinations relating to any Credit Party, whether made by such Credit Party or otherwise; provided however, Agent will attempt to obtain such information or materials directly from Credit Parties prior to obtaining such information or materials from such Governmental Bodies.

#### IV.9 Compliance with Laws

. Each Credit Party shall comply, and shall cause each of its Subsidiaries to comply, with (a) all Anti-Terrorism Laws, and (b) all other Applicable Laws which noncompliance with, with respect to this <u>clause (b)</u> only, could reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

#### IV.10 Inspection of Premises

. Each Credit Party will, and will cause each of its Subsidiaries to, permit Agent and each Lender and their respective representatives and independent contractors to visit and inspect any of its properties, to conduct field audits and examinations, to examine its corporate, financial and operating records, and make copies thereof or abstracts therefrom, and to discuss its affairs, finances and accounts with its directors, officers, and independent public accountants, all at the expense of Borrowers and at such reasonable times during normal business hours and as often as may be reasonably desired, upon reasonable advance notice to Borrowing Agent (and such Persons shall use commercially reasonable efforts to conduct such visits and inspections in a manner as to minimize interference in the applicable Credit Parties normal operations), provided, however, that so long as no Event of Default exists, without limiting the number of any such visits and/or inspections, Borrowers shall not be obligated to pay for more than two (2) such inspections per year.

#### IV.11 Insurance

. Subject to Section 6.16(f), the Credit Parties will maintain or cause to be maintained, with financially sound and licensed insurers, property insurance, public liability insurance, cybersecurity insurance, and third party property damage insurance with respect to liabilities, losses or damage in respect of the assets, properties and businesses of each Credit Party and its Subsidiaries as may customarily be carried or maintained under similar circumstances by Persons of established reputation engaged in similar businesses, in each case, in such amounts, with such deductibles, covering such risks and otherwise on such terms and conditions as shall be

[Different first page setting changed from on in original to off in modified.].

customary for such Persons. Without limiting the generality of the foregoing, each of Borrower and its Subsidiaries will maintain or cause to be maintained (a) flood insurance with respect to each Flood Hazard Property that is a part of the Collateral, if any, and that is located in a community that participates in the National Flood Insurance Program, in each case (to the extent required) in compliance with any applicable regulations of the Board of Governors of the Federal Reserve System, and (b) replacement value casualty insurance on the Collateral under such policies of insurance, with such insurance companies, in such amounts, with such deductibles, and covering such risks as are at all times carried or maintained under similar circumstances by Persons of established reputation engaged in similar businesses. Each such policy of insurance shall (i) with respect to liability insurance and cybersecurity insurance, name Agent, on behalf of the holders of the Obligations, as an additional insured thereunder as its interests may appear, and (ii) in the case of each property insurance policy, contain a loss payable clause or endorsement with respect to the Collateral, reasonably satisfactory in form and substance to Agent, that names Agent, on behalf of the holders of the Obligations, as the lender's loss payee thereunder and provides for at least thirty (30) days' (or ten (10) days' in the case of non-payment of premium) prior written notice to Agent of any modification or cancellation of such policy; and which, for the avoidance of doubt, in the case of being added as an additional insured with respect to such liability insurance, shall not limit the ability of the applicable insurance carrier to satisfy any claims that may be payable to a third party on behalf of any Credit Party. Credit Parties shall furnish Agent with copies of all policies and evidence of the maintenance of such policies by the renewal thereof as soon as practicable prior to the expiration date thereof. If any insurance losses are paid by check, draft or other instrument payable to any Credit Party and Agent jointly, Agent may endorse such Credit Party's name thereon and do such other things as Agent may deem advisable to reduce the same to cash. Following the occurrence and during the continuance of an Event of Default with respect to all claims for insurance proceeds in excess of \$250,000, Agent is hereby authorized to adjust and compromise claims under insurance coverage referred to in <u>clause (b)</u> above. All loss recoveries received by Agent upon any such insurance may be applied to the Obligations, in such order as Agent in its Permitted Discretion shall determine during the continuance of an Event of Default and, at all other times as set forth in Section 2.7(a). Any surplus shall be paid by Agent to Borrowing Agent or applied as may be otherwise required by law.

#### IV.12 Failure to Pay Insurance

. If Credit Parties fail to obtain insurance as hereinabove provided, or to keep the same in force, Agent, if Agent so elects, may obtain such insurance and pay the premium therefor on behalf of Credit Parties, and charge Borrowers' Account therefor as an Advance and such expenses so paid shall be part of the Obligations.

#### IV.13 Payment of Taxes

. Each Credit Party will, and will cause each of its Subsidiaries to, pay (a) all federal, state and other material Taxes imposed upon it or any of its properties or assets or in respect of any of its income, businesses or franchises before any penalty or fine accrues thereon and (b) all

[Different first page setting changed from on in original to off in modified.].

claims (including claims for labor, services, materials and supplies) for sums that have become due and payable and that by law have or may become a Lien upon any of its properties or assets, prior to the time when any penalty or fine shall be incurred with respect thereto; provided, no such Tax or claim need be paid if it is being contested in good faith by appropriate proceedings promptly instituted and diligently conducted, so long as (i) adequate reserve or other appropriate provision, as shall be required in conformity with GAAP shall have been made therefor, and (ii) in the case of a Tax or claim which has or may become a Lien against any of the Collateral, such contest proceedings conclusively operate to stay the sale of any portion of the Collateral to satisfy such Tax or claim. No Credit Party will, nor will any Credit Party permit any of its Subsidiaries to, file or consent to the filing of any consolidated income tax return with any Person (other than DDH or any Subsidiary of DDH). Subject to the limitations of Section 3.10, if any Tax by any Governmental Body is or may be imposed on or as a result of any transaction between any Credit Party and Agent or any Lender which Agent or any Lender may be required to withhold or pay or if any Taxes, assessments, or other charges remain unpaid after the date fixed for their payment, or if any claim shall be made which, in Agent's or any Lender's Permitted Discretion, may possibly create a valid Lien on the Collateral, Agent may without notice to any Credit Party pay the Taxes, assessments or other claims and each Borrower hereby indemnifies and holds Agent and each Lender harmless in respect thereof. Agent will not pay any Taxes, assessments or claims to the extent being contested as set forth in the proviso in the first sentence in this Section 4.13. The amount of any payment by Agent under this Section 4.13 shall be charged to Borrowers' Account and added to the Obligations and, until Borrowers shall furnish Agent with an indemnity therefor (or supply Agent with evidence satisfactory to Agent that due provision for the payment thereof has been made), Agent may hold without interest any balance standing to Borrowers' credit and Agent shall retain its security interest in and Lien on any and all Collateral held by Agent.

## IV.14 Payment of Leasehold Obligations

. Credit Parties shall at all times pay, when and as due, to the extent failure to pay such amount could result in the termination of such lease, its rental obligations under all leases under which it is a tenant, and shall otherwise comply, in all material respects, with all other terms of such leases and keep them in full force and effect (except for termination of a lease at its scheduled termination date or otherwise in accordance with the terms thereof), to the extent termination of such lease would reasonably result in a Material Adverse Effect.

IV.15 Receivables.

Ŧ

(a) <u>Nature of Receivables. Each of the Receivables shall be a bona fide and valid account representing a bona fide indebtedness</u> incurred by the Customer therein named, for a fixed sum as set forth in the invoice relating thereto (provided that immaterial or unintentional invoice errors shall not be deemed to be a breach hereof) with respect to the sale or lease and delivery of goods upon stated terms of a Credit Party, or work, labor or services theretofore

[Different first page setting changed from on in original to off in modified.].

rendered by a Credit Party as of the date each Receivable is created. Same shall be due and owing in accordance with Credit Party's standard terms of sale without known dispute, setoff or counterclaim except to the extent that such disputes, setoffs, or counterclaims have been disclosed to Agent to the extent involving an amount in excess of \$250,000 individually or in the aggregate.

(b) <u>Solvency of Customers. Each Customer (other than Customers having Receivables not in excess of \$250,000 in the aggregate at any time outstanding), to each Credit Party's actual knowledge, as of the date each Receivable is created, is and will be solvent and able to pay all Receivables on which the Customer is obligated in full when due or with respect to such Customers of Credit Parties who are not, to any Credit Party's actual knowledge, solvent, the applicable Borrower has set up on its books and in its financial records bad debt reserves adequate to cover such Receivables.</u>

(c) Location of Credit Party. Credit Parties' chief executive office is set forth on Schedule 4.5; provided, that, Credit Parties may change the location of their chief executive office upon 10 days prior written notice to Agent. Until written notice is given to Agent by Credit Parties of any other office at which any Credit Party keeps its records pertaining to Receivables, all such records shall be kept at such executive office.

(d) <u>Collection of Receivables. Credit Parties shall deposit in a Depository Account or, upon request by Agent, deliver to Agent, in original form and within three (3) Business Days of receipt thereof, all checks, drafts, notes, money orders, acceptances, cash and other evidences of Indebtedness paid in respect of Receivables.</u>

(e) Notification of Assignment of Receivables. At any time during the continuance of an Event of Default, Agent shall have the right to send notice of the assignment of, and Agent's security interest in and Lien on, the Receivables to any and all Customers or any third party holding or otherwise concerned with any of the Collateral. Agent shall use reasonable efforts to provide notice to Borrowing Agent of the sending of any such notice of assignment, but failure to do so shall not result in any liability to Agent. Thereafter, Agent shall have the sole right to collect the Receivables, take possession of the Collateral, or both. Agent's actual collection expenses, including, but not limited to, stationery and postage, telephone and telegraph, secretarial and clerical expenses and the salaries of any collection personnel used for collection, may be charged to Borrowers' Account and added to the Obligations.

(f) Power of Agent to Act on Credit Parties' Behalf. Agent shall have the right to receive, endorse, assign and/or deliver in the name of Agent or any Credit Party any and all checks, drafts and other instruments for the payment of money relating to the Receivables, and each Credit Party hereby waives notice of presentment, protest and non-payment of any instrument so endorsed. Each Credit Party hereby constitutes Agent or Agent's designee as such Credit Party's attorney with power: (i) to endorse such Credit Party's name upon any notes, acceptances, checks, drafts, money orders or other evidences of payment or Collateral; (ii) to sign such Credit Party's name on any invoice or bill of lading relating to any of the Receivables,

[Different first page setting changed from on in original to off in modified.].

drafts against Customers, assignments and verifications of Receivables; (iii) to send verifications of Receivables to any Customer; (iv) to sign such Credit Party's name on all financing statements or any other documents or instruments deemed necessary or appropriate by Agent to preserve, protect, or perfect Agent's interest in the Collateral and to file same; (v) to demand payment of the Receivables; (vi) to enforce payment of the Receivables by legal proceedings or otherwise; (vii) to exercise all of such Credit Party's rights and remedies with respect to the collection of the Receivables and any other Collateral; (viii) to exercise all of such Credit Party's rights and remedies; (ix) to settle, adjust or compromise any legal proceedings brought to collect Receivables; (x) to prepare, file and sign such Credit Party's name on a proof of claim in bankruptcy or similar document against any Customer; (xi) to prepare, file and sign such Credit Party's name on any notice of Lien, assignment or satisfaction of Lien or similar document in connection with the Receivables; and (xii) to do all other acts and things necessary to carry out this Agreement; provided, however, Agent shall only exercise the rights described in clauses (i), (ii), (v), (vi), (vii), (ix) and (xi) upon the occurrence and during the continuance of an Event of Default. All acts of said attorney or designee are hereby ratified and approved, and said attorney or designee shall not be liable for any acts of omission or commission nor for any error of judgment or mistake of fact or of law, unless done maliciously or with gross (not mere) negligence (as determined by a court of competent jurisdiction in a final non-appealable judgment); this power being coupled with an interest is irrevocable while any of the Obligations (other than Inchoate Obligations) remain unpaid. Agent shall have the right at any time following the occurrence and during the continuance of an Event of Default, to change the address for delivery of mail addressed to any Credit Par

(g) No Liability. Neither Agent nor any Lender shall, under any circumstances or in any event whatsoever, have any liability for any error or omission or delay of any kind occurring in the settlement, collection or payment of any of the Receivables or any instrument received in payment thereof, or for any damage resulting therefrom. Following the occurrence and during the continuance of an Event of Default, Agent may, without notice or consent from any Credit Party, sue upon or otherwise collect, extend the time of payment of, compromise or settle for cash, credit or upon any terms any of the Receivables or any other securities, instruments or insurance applicable thereto and/or release any obligor thereof. Following the occurrence and during the continuance of an Event of Default, Agent is authorized and empowered to accept the return of the goods represented by any of the Receivables, all without discharging or in any way affecting any Credit Party's liability hereunder. Agent shall use reasonable efforts to notify the applicable Credit Party of any such acceptance of returned goods, but shall have no liability for any failure to do so.

(h) Cash Management; Payment Mechanics.

(i) <u>The Credit Parties shall direct each of their Customers to remit all payments on Receivables to, and Credit Parties</u> shall deposit all cash, checks, notes instruments and other funds or items of payment received by such Credit Party from time to time to,

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

depository accounts ("Depository Accounts") established at ABL Lender or such other depository approved by Agent in its reasonable discretion for the deposit of such amounts.

(ii) <u>As of the Closing Date, all deposit accounts and investment accounts of DDH and its Subsidiaries are set forth on</u> <u>Schedule 4.15(h). Each Credit Party will notify Agent in writing at least five (5) Business Days prior to the establishment of any deposit accounts</u> <u>or investment accounts not set forth on Schedule 4.15(h). Subject to Section 6.16 hereof, the Credit Parties shall deliver to Agent Control</u> <u>Agreements with respect to all deposit accounts of the Credit Parties (including without limitation, the Depository Accounts, but excluding any</u> <u>Excluded Deposit Accounts).</u>

(iii) On or before the Closing Date, Borrowing Agent shall have established a deposit account at a depository approved by Agent in its reasonable discretion for the purpose of receiving proceeds of the Advances (the "Funding Account").

(i) <u>Adjustments. No Borrower will, without Agent's consent, compromise or adjust any Receivables (or extend the time for</u> payment thereof) or accept any returns of merchandise or grant any additional discounts, allowances or credits thereon except for compromises, adjustments, returns, discounts, credits and allowances in the ordinary course of business of Borrowers.

IV.16 Pledge of Personal Property Assets

(a) Each Credit Party shall cause one hundred percent (100%) of the issued and outstanding Equity Interests of each Subsidiary of DDH to be subject at all times to a first priority lien (subject to any Permitted Encumbrance) in favor of Agent pursuant to the terms and conditions of this Agreement and the Other Documents, together with opinions of counsel and any filings and deliveries or other items reasonably requested by Agent necessary in connection therewith (to the extent not delivered on the Closing Date) to perfect the security interests therein, all in form and substance reasonably satisfactory to Agent.

(b) Each Credit Party shall (i) cause all of its owned personal property (other than Excluded Property) to be subject at all times to first priority (subject to any Permitted Encumbrances), perfected Liens in favor of Agent to secure the Obligations pursuant to the terms and conditions of this Agreement and the Other Documents or, with respect to any such property acquired subsequent to the Closing Date, such other additional security documents as Agent shall reasonably request, subject in any case to Permitted Encumbrances and (ii) deliver such other documentation as Agent may reasonably request in connection with the foregoing, including, without limitation, appropriate Control Agreements, UCC-1 financing statements, certified resolutions and other organizational and authorizing documents of such Person, opinions of counsel to such Person (which shall cover, among other things, the legality, validity, binding effect and enforceability of the documentation referred to above and the perfection of Agent's Liens thereunder) and other items reasonably requested by Agent necessary in connection

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

# therewith to perfect the security interests therein, all in form, content and scope reasonably satisfactory to Agent.

### IV.17 Maintenance of Equipment

. Each Credit Party shall, and shall cause each of its Subsidiaries to, maintain or cause to be maintained in good repair, working order and condition, ordinary wear and tear excepted, all material properties used or useful in the business of any Credit Party and its Subsidiaries and from time to time will make or cause to be made all appropriate repairs, renewals and replacements thereof substantially in accordance with such Credit Party's past business practices.

# IV.18 Exculpation of Liability

. Nothing herein contained shall be construed to constitute Agent, any Lender or any of their respective officers, directors, Affiliates, attorneys, employees or agents as any Borrower's agent for any purpose whatsoever, nor shall Agent, any Lender or any of their respective officers, directors, Affiliates, attorneys, employees or agents be responsible or liable for any shortage, discrepancy, damage, loss or destruction of any part of the Collateral wherever the same may be located and regardless of the cause thereof, except to the extent arising out of the gross negligence or willful misconduct of Agent or any Lender. Neither Agent nor any Lender, whether by anything herein or in any assignment or otherwise, assume any of any Borrower's obligations under any contract or agreement assigned to Agent or such Lender, and neither Agent nor any Lender shall be responsible in any way for the performance by any Borrower of any of the terms and conditions thereof.

### IV.19 Environmental Matters.

(a) <u>Each Credit Party will deliver to Agent with reasonable promptness, such documents and information with respect to any</u> material Environmental Claim as from time to time may be reasonably requested by Agent.

(b) Each Credit Party shall promptly take, and shall cause each of its Subsidiaries promptly to take, any and all actions necessary to (i) cure any violation of applicable Environmental Laws by such Credit Party or its Subsidiaries that would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, and (ii) respond to any Environmental Claim against such Credit Party or any of its Subsidiaries and discharge any obligations it may have to any Person thereunder where failure to do so would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

(c) In the event any Borrower obtains, gives or receives notice of any Release or threat of Release of a reportable quantity of any Hazardous Materials at the Real Property (any such event being hereinafter referred to as a "Hazardous Discharge") or receives any request for information or notification that it is potentially responsible for investigation or cleanup of

[Different first page setting changed from on in original to off in modified.].

environmental conditions at the Real Property or other written notice with regard to any Hazardous Discharge or any other Environmental Claim (any of the foregoing is referred to herein as an "Environmental Complaint") from any Person, including any state agency responsible in whole or in part for environmental matters in the state in which the Real Property is located or the United States Environmental Protection Agency (any such Person hereinafter the "Authority"), then Borrowers shall, within ten (10) Business Days, give written notice of same to Agent detailing facts and circumstances of which any Borrower is aware giving rise to the Hazardous Discharge or Environmental Complaint. Such information is to be provided to allow Agent to protect its security interest in and Lien on the Real Property and the Collateral and is not intended to create nor shall it create any obligation upon Agent or any Lender with respect thereto.

(d) Borrowers shall respond promptly to any Hazardous Discharge or Environmental Complaint, except where failure to do so could not reasonably be expected to result in material liability to any Borrower, and take all commercially reasonable action in order to safeguard the health of any Person and to avoid subjecting the Collateral or Real Property to any Lien (other than Permitted Encumbrances). If Borrowers shall fail to respond promptly to any Hazardous Discharge or Environmental Complaint or Borrowers shall fail to comply with any of the requirements of any Environmental Laws, Agent on behalf of Lenders may, but without the obligation to do so, for the sole purpose of protecting Agent's interest in the Collateral: (i) give such notices or (ii) enter onto the Real Property (or authorize third parties to enter onto the Real Property) and take such actions as Agent (or such third parties as directed by Agent) deem reasonably necessary, to clean up, remove, mitigate or otherwise deal with any such Hazardous Discharge or Environmental Complaint. All reasonable, documented out-of-pocket costs and expenses incurred by Agent and Lenders (or such third parties) in the exercise of any such rights, including any sums paid in connection with any judicial or administrative investigation or proceedings, fines and penalties shall be paid upon demand by Borrowers, and until paid shall be added to and become a part of the Obligations secured by the Liens created by the terms of this Agreement or any other agreement between Agent, any Lender and any Borrower.

(e) [<u>Reserved</u>].

(f) <u>For purposes of Sections 4.19 and 5.7, all references to Real Property shall be deemed to include all of each Borrower's</u> right, title and interest in and to its owned and leased premises.

## IV.20 Financing Statements

. To the knowledge of the Credit Parties, except for the financing statements filed by Agent and the financing statements evidencing Permitted Encumbrances, no financing statement covering any of the Collateral or any proceeds thereof is on file in any public office.

IV.21 Key Executive Policy

[Different first page setting changed from on in original to off in modified.].

. Promptly, but in any event within one hundred twenty (120) days after the Second Amendment Date, as further security for the payment of the Obligations and the satisfaction by Borrower of all covenants and undertakings contained in this Agreement and the Other Documents, the Borrower shall use commercially reasonable efforts to deliver, for the benefit of Agent and Lenders, a collateral assignment of a key executive life insurance policy owned by Borrower insuring the life of Keith Smith and Mark Walker in the amount of not less than \$5,000,000 for each such Person (and an aggregate amount of not less than \$10,000,000) issued by an insurer acceptable to Agent in its sole discretion (such policy, the "Key Executive Policy"), and in connection therewith, within such period, Borrower shall deliver to Agent all forms and agreements required by the insurer issuing the Key Executive Policy in order to have such assignment of the Key Executive Policy in favor of Agent, for itself and for the benefit of Lenders, acknowledged and reflected on the records of such insurer (all of which such forms shall have been executed by Borrower and any other party necessary thereto); provided, that no Default or Event of Default shall occur if the Borrower is unable to obtain coverage in the amounts or on the terms described herein or if premiums on any such policy exceed a commercially reasonable amount. Nothing contained in the foregoing or in any such forms and agreements required by the insurer shall be construed in any way to contradict or limit (but only to expand and extend) the grant of a security interest and Lien by Borrower in all now existing and hereafter arising General Intangibles and insurance policies provided for in <u>Section 4.1</u> above.

## V. <u>REPRESENTATIONS AND WARRANTIES.</u>

Each Credit Party represents and warrants as follows:

## V.1 Organization; Requisite Power and Authority; Qualification

. Each Credit Party and its Subsidiaries (a) is duly organized, validly existing and in good standing under the laws of its jurisdiction of organization as identified in <u>Schedule 5.1</u>, (b) has all requisite power and authority to own and operate its properties, to carry on its business as now conducted and as proposed to be conducted, to enter into this Agreement and the Other Documents to which it is a party and to carry out the transactions contemplated thereby, and (c) is qualified to do business and in good standing in every jurisdiction where necessary to carry out its business and operations, except in jurisdictions where the failure to be so qualified or in good standing has not had, and could not be reasonably expected to have, a Material Adverse Effect.

## V.2 Equity Interests and Ownership.

(a) <u>Schedule 5.2(a) correctly sets forth the ownership interest of each Credit Party and each Credit Party's Subsidiaries, in each case as of the Closing Date. The Equity Interests of each Credit Party and its Subsidiaries have been duly authorized and validly issued and are fully paid and non-assessable (except with respect to limited liability company interests). Except as set forth on Schedule 5.2(a) as of the Closing Date, there is no existing option, warrant, call, right, commitment, buy-sell, voting trust, contribution agreement or other shareholder</u>

[Different first page setting changed from on in original to off in modified.].

agreement or other agreement with respect to any Credit Party or any Subsidiary, and there is no membership interest or other Equity Interests of any Credit Party or any Subsidiary outstanding which upon conversion or exchange would require, the issuance by any Credit Party, or any Subsidiary of any additional membership interests or other Equity Interests of any Credit Party or any Subsidiary or other Securities convertible into, exchangeable for or evidencing the right to subscribe for or purchase, a membership interest or other Equity Interests of any Credit Party or any Subsidiary.

(b) <u>The only Subsidiaries of each Credit Party are listed on Schedule 5.2(b).</u>

# V.3 Due Authorization

. The execution, delivery and performance of this Agreement and the Other Documents have been duly authorized by all necessary action on the part of each Credit Party that is a party hereto or thereto.

## V.4 No Conflict

. The execution, delivery and performance by Credit Parties of this Agreement and the Other Documents to which they are parties and the consummation of the transactions contemplated by this Agreement and the Other Documents do not and will not (a) violate in any material respect any provision of any Applicable Laws relating to any Credit Party, any of the Governing Documents of any Credit Party, or any order, judgment or decree of any court or other agency of government binding on any Credit Party; (b) conflict with, result in a breach of or constitute (with due notice or lapse of time or both) a default under any material Contractual Obligations of any Credit Party (including the ABL Loan Documents); (c) result in or require the creation or imposition of any Lien upon any of the properties or assets of any Credit Party (other than any Liens created under this Agreement or any of the Other Documents in favor of Agent) whether now owned or hereafter acquired; or (d) require any approval of stockholders, members or partners or any approval or consent of any Person under any Governing Document or any other material Contractual Obligation of any Credit Party, in each case except for any such approval or consent that has been duly obtained.

### V.5 Governmental Consents

. The execution, delivery and performance by the Credit Parties of this Agreement and the Other Documents to which they are parties and the consummation of the transactions contemplated by this Agreement and the Other Documents do not and will not require, as a condition to the effectiveness thereof, any registration with, consent or approval of, or notice to, or other action to, with or by, any Governmental Body except for filings and recordings with respect to the Collateral to be made, or otherwise delivered to Agent for filing and/or recordation, as of the Closing Date and other filings, recordings or consents which have been obtained or made, as applicable.

## V.6 Binding Obligations

[Different first page setting changed from on in original to off in modified.].

. Each of this Agreement and each Other Document has been duly executed and delivered by each Credit Party that is a party thereto and is the legally valid and binding obligation of such Credit Party enforceable against such Credit Party in accordance with its respective terms, except as may be limited by Debtor Relief Laws or by equitable principles relating to enforceability.

# V.7 No Material Adverse Effect; No Default

(a) <u>Since December 31, 2020, no event, circumstance or change has occurred that has caused or evidences, either in any case or in the aggregate, a Material Adverse Effect.</u>

### (b) <u>No Default or Event of Default has occurred and is continuing.</u>

## V.8 <u>Tax Returns</u>

. Each Credit Party and its Subsidiaries have filed all federal, state and other material Tax returns and reports required to be filed or have obtained a valid extension of the deadline for any such returns or reports such that there are no delinquencies for filing, and have paid all federal, state and other material Taxes, assessments, fees and other governmental charges levied or imposed upon them or their respective properties, assets, income, businesses and franchises otherwise due and payable, except those being actively contested in good faith and by appropriate proceedings and for which adequate reserves have been provided in accordance with GAAP. There is no proposed Tax assessment against any Credit Party or any of its Subsidiaries that would, if made, have a Material Adverse Effect.

## V.9 Financial Statements.

÷

(a) The balance sheet of DDH and its Subsidiaries prepared on a Consolidated Basis for the most recent Fiscal Year ended and the related statements of income, equity and cash flows prepared on a Consolidated Basis for such Fiscal Year, including the notes thereto (i) were prepared in accordance with GAAP consistently applied throughout the period covered thereby (except as otherwise noticed in the Quality of Earnings Report); (ii) fairly present the financial condition of DDH and its Subsidiaries in all material respects as of the date thereof and their results of operations for the period covered thereby, in each case, on a Consolidated Basis, in accordance with GAAP consistently applied throughout the Quality of Earnings Report); and (iii) show all material indebtedness and other liabilities, direct or contingent, of DDH and its Subsidiaries as of the date thereof, including liabilities for taxes, material commitments and Indebtedness, to the extent required to be included thereon in accordance with GAAP.

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

(b) The audited balance sheet of DDH and its Subsidiaries prepared on a Consolidated Basis as of December 31, 2020, and the related statements of income, equity and cash flows prepared on a Consolidated Basis for such Fiscal Quarter (i) were prepared in accordance with GAAP consistently applied throughout the period covered thereby, except as otherwise expressly noted therein, (ii) fairly present the financial condition of DDH and its Subsidiaries in all material respects as of the date thereof and their results of operations for the period covered thereby, in each case, on a Consolidated Basis, subject, in the case of clauses (i) and (ii), to the absence of footnotes and to normal year-end audit adjustments, except as otherwise expressly noted therein and (iii) show all material indebtedness and other liabilities, direct or contingent, of DDH and its Subsidiaries as of the date of such financial statements, including liabilities for taxes, material commitments and Indebtedness to the extent required to be included thereon in accordance with GAAP.

(c) <u>The pro forma consolidated balance sheet of DDH and its Subsidiaries (the "Pro Forma Balance Sheet") furnished to Agent</u> on the date hereof reflects the consummation of the Closing Date Transactions and fairly presents in all material respects the financial condition of DDH and its Subsidiaries as of October 31, 2021, after giving pro forma effect to the Closing Date Transactions as if the Closing Date Transactions occurred on such date.

(d) The four quarter consolidated cash flow projections of DDH and its Subsidiaries delivered by DDH as of the Closing Date (the "Projections") were prepared by DDH based upon estimates and assumptions that DDH believes to be reasonable and fair on the Closing Date in light of present circumstances and, as of the Closing Date, reflected DDH's good faith estimates of the future financial performance of DDH and its Subsidiaries on a consolidated basis and of the other information projected therein for the periods set forth therein. Notwithstanding the foregoing, it is understood that such Projections were and remain subject to uncertainties and contingencies, many of which are beyond the control of DDH or any of its Subsidiaries, and that no assurance can be given that such Projections will actually be realized. The Projections together with the Pro Forma Balance Sheet, are referred to as the "Pro Forma Financial Statements".

(e) The forecasted balance sheet and statements of income and cash flows of DDH and its Subsidiaries delivered pursuant to Section 9.12 were prepared in good faith on the basis of the assumptions stated therein, which assumptions were fair in light of the conditions existing at the time of delivery of such forecasts, and represented, at the time of delivery, DDH's best estimate of its future financial condition and performance.

# V.10 Information Regarding the Credit Parties and their Subsidiaries

. Set forth on <u>Schedule 5.10</u>, is the jurisdiction of organization, the exact legal name (and for the prior five (5) years or since the date of its formation has been) and the true and correct U.S. taxpayer identification number (or foreign equivalent, if any) of each Credit Party and each of its Subsidiaries as of the Closing Date.

[Different first page setting changed from on in original to off in modified.].

# V.11 Environmental Matters; O.S.H.A.

(a) <u>Each Credit Party is in compliance with, and its facilities, business, assets, property, leaseholds, Real Property and Equipment are in compliance with, the Federal Occupational Safety and Health Act in all material respects and Environmental Laws, and there are no material outstanding citations, notices or orders of non-compliance issued to any Credit Party or relating to its business, assets, property, leaseholds or Equipment under any such laws, rules or regulations.</u>

(b) No Credit Party nor any of its Subsidiaries nor any of their respective current Real Property (solely during and with respect to such Person's ownership thereof) or operations, and to their knowledge, no former Real Property (solely during and with respect to any Credit Party's or its Subsidiary's ownership thereof), are subject to any outstanding order, consent decree or settlement agreement with any Person relating to any Environmental Law, any Environmental Claim, or any Hazardous Materials Activity that, individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect.

(c) <u>No Credit Party nor any of its Subsidiaries has received any letter or request for information under Section 104 of CERCLA</u> or any comparable state law.

(d) <u>There are and, to each Credit Party's and its Subsidiaries' knowledge, have been, no Hazardous Materials Activities which</u> could reasonably be expected to form the basis of an Environmental Claim against such Credit Party or any of its Subsidiaries that, individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect.

(e) No Credit Party nor any of its Subsidiaries has filed any notice under any Environmental Law indicating past or present treatment of Hazardous Materials at any Real Property owned or leased by such Person (solely during and with respect to such Credit Party's or its Subsidiary's ownership thereof), and no Credit Party's nor any of its Subsidiaries' operations involves the generation, transportation, treatment, storage or disposal of hazardous waste, as defined under 40 C.F.R. Parts 260-270 or any equivalent state rule defining hazardous waste, in each case, except in material compliance with Environmental Law. Compliance with all current requirements pursuant to or under Environmental Laws could not be reasonably expected to have, individually or in the aggregate, a Material Adverse Effect.

V.12 Solvency; No Litigation, Violation, Indebtedness or Default

(a) <u>After giving effect to the Closing Date Transactions, DDH and its Subsidiaries taken as a whole on a Consolidated Basis</u> are, and, upon the incurrence of any Advance on any date on which this representation and warranty is made or deemed made, will be, Solvent.

[Different first page setting changed from on in original to off in modified.].

159823.01008/130146990v.3 159823.01008/150720893v.5

(b) There are no Adverse Proceedings that (i) as of the date hereof purport to affect or pertain to this Agreement or any Other Document, or any of the transactions contemplated hereby or (ii) could reasonably be expected to have a Material Adverse Effect. No Credit Party nor any of its Subsidiaries is subject to or in default with respect to any final judgments, writs, injunctions, decrees, rules or regulations of any Governmental Body that, individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect.

(c) Each Credit Party and its Subsidiaries is in compliance with (i) the USA PATRIOT Act and OFAC rules and regulations as provided in Sections 5.29 and 5.30 and (ii) except such non-compliance with such other Applicable Laws that, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect, all other Applicable Laws.

(d) (i) Except as could not reasonably be expected to have a Material Adverse Effect, each of the Credit Parties and their Subsidiaries are in compliance with all applicable provisions and requirements of ERISA and the Code and the regulations and published interpretations thereunder with respect to each Pension Plan, and have performed all their obligations under each Pension Plan in all material respects, (ii) each Pension Plan which is intended to qualify under Section 401(a) of the Code has received a favorable determination letter or is the subject of a favorable opinion letter from the Internal Revenue Service indicating that such Pension Plan is so qualified and, to the best knowledge of the Credit Parties, nothing has occurred subsequent to the issuance of such determination letter which would cause such Pension Plan to lose its qualified status except where such event could not reasonably be expected to result in a Material Adverse Effect, (ii) except as could not reasonably be expected to have a Material Adverse Effect, no liability to the PBGC (other than required premium payments), the Internal Revenue Service, any Pension Plan (other than for routine claims and required funding obligations in the ordinary course) or any trust established under Title IV of ERISA has been incurred by any Credit Party, any of its Subsidiaries or any of their ERISA Affiliates, (iv) except as would not reasonably be expected to result in liability to any Credit Party or any of its Subsidiaries in excess of \$500,000, no ERISA Event has occurred, and (v) except to the extent required under Section 4980B of the Code and Section 601 et seq. of ERISA or similar state laws and except as could not reasonably be expected to have a Material Adverse Effect, no Pension Plan provides health or welfare benefits (through the purchase of insurance or otherwise) for any retired or former employee of any Credit Party or any of its Subsidiaries.

## V.13 Patents, Trademarks, Copyrights and Licenses

. As of the Closing Date, all registered or material patents, patent applications, trademarks, trademark applications, service marks, service mark applications, copyrights, copyright applications, design rights, tradenames and material licenses therefor owned or utilized by any Credit Party are set forth on <u>Schedule 5.13</u>. Each Credit Party and its Subsidiaries owns or is validly licensed to use all Intellectual Property that is necessary for the present conduct of its business, free and clear of Liens (other than Permitted Encumbrances), without conflict with the rights of any other Person unless the failure to own or benefit from such valid license could

[Different first page setting changed from on in original to off in modified.].

not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect. To the knowledge of each Credit Party, no Credit Party nor any of its Subsidiaries is infringing, misappropriating, diluting, or otherwise violating the Intellectual Property rights of any other Person unless such infringement, misappropriation, dilution or violation could not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

### V.14 Licenses, Permits and Other Approvals

. Each Credit Party (a) is in compliance with and (b) has procured and is now in possession of, all material licenses or permits required by any applicable federal, state, provincial or local law, rule or regulation for the operation of its business in each jurisdiction wherein it is now conducting or proposes to conduct business, except where the failure to so comply or procure such licenses or permits could not reasonably be expected to have a Material Adverse Effect.

#### V.15 Default of Indebtedness

. No Credit Party or Subsidiary thereof is in default in the payment of the principal of or interest on any Indebtedness of \$250,000 or more in principal amount (in the aggregate for all Credit Parties and Subsidiaries) and no such Indebtedness has been accelerated.

### V.16 No Default

. No Credit Party nor any of its Subsidiaries is in default in the performance, observance or fulfillment of any of the obligations, covenants or conditions contained in any of its Contractual Obligations (other than Contractual Obligations relating to Indebtedness) where the consequences, direct or indirect, of such default or defaults, if any, could reasonably be expected to have a Material Adverse Effect.

#### V.17 No Burdensome Restrictions

. As of the Closing Date, each Credit Party has heretofore delivered to Agent or provided Agent access to true and complete copies of all Material Contracts to which it is a party or to which it or any of its properties is subject. No Credit Party or Subsidiary thereof has agreed or consented to cause or permit in the future (upon the happening of a contingency or otherwise) any of its property, whether now owned or hereafter acquired, to be subject to a Lien which is not a Permitted Encumbrance.

# V.18 No Labor Disputes

. No Credit Party or any of its Subsidiaries is engaged in any unfair labor practice that could reasonably be expected to have a Material Adverse Effect. There is (a) no unfair labor practice complaint pending against any Credit Party or any of its Subsidiaries, or to the best knowledge of each Credit Party, threatened against any of them before the National Labor Relations Board and no grievance or arbitration proceeding arising out of or under any collective

[Different first page setting changed from on in original to off in modified.].

bargaining agreement that is so pending against any Credit Party or any of its Subsidiaries or to the best knowledge of each Credit Party, threatened against any of them, (b) no strike or work stoppage in existence or to the knowledge of each Credit Party, threatened that involves any Credit Party or any of its Subsidiaries, and (c) to the best knowledge of each Credit Party, no union representation question existing with respect to the employees of any Credit Party or any of its Subsidiaries and, to the best knowledge of each Credit Party, no union organization activity that is taking place, except (with respect to any matter specified in clause (a), (b) or (c) above, either individually or in the aggregate) such as could not reasonably be expected to have a Material Adverse Effect.

#### V.19 Margin Regulations

. Neither any Credit Party nor any of its Subsidiaries is engaged principally, or as one of its important activities, in the business of extending credit for the purpose of purchasing or carrying any Margin Stock. No part of the proceeds of any Term Loan made to such Credit Party will be used (i) to purchase or carry any such Margin Stock or to extend credit to others for the purpose of purchasing or carrying any such Margin Stock or for any purpose that violates, or is inconsistent with, the provisions of Regulation T, U or X of the Board of Governors of the Federal Reserve System as in effect from time to time or (ii) to finance or refinance any (A) commercial paper issued by such Credit Party or (B) any other Indebtedness, except for Indebtedness that such Credit Party incurred for general corporate or working capital purposes.

#### V.20 Investment Company Act

. No Credit Party or any of its Subsidiaries is subject to regulation under the Investment Company Act of 1940. No Credit Party or any of its Subsidiaries is an "investment company" or a company "controlled" by a "registered investment company" or a "principal underwriter" of a "registered investment company" as such terms are defined in the Investment Company Act of 1940.

#### V.21 Disclosure

. No representation or warranty of any Credit Party contained in this Agreement or any Other Document or in any other documents, certificates or written statements furnished to Agent or any Lender by or on Behalf of any Credit Party or any of its Subsidiaries for use in connection with the transactions contemplated hereby (other than projections and pro forma financial information contained in such materials) contains any untrue statement of a material fact or omits to state a material fact (known to any Credit Party, in the case of any document not furnished by any of them) necessary in order to make the statements contained herein or therein not misleading in any material manner in light of the circumstances in which the same were made. Any projections and pro forma financial information contained in such materials are based upon good faith estimates and assumptions believed by the Credit Parties to be reasonable at the time made, it being recognized by Agent and Lenders that such projections as to future events are not to be viewed as facts and that actual results during the period or periods covered by any such

[Different first page setting changed from on in original to off in modified.].

projections may differ from the projected results and that such differences may be material. There are no facts known to any Credit Party (other than matters of a general economic nature) that, individually or in the aggregate, could reasonably be expected to result in a Material Adverse Effect and that have not been disclosed herein or in such other documents, certificates and statements furnished to Agent or any Lender.

## V.22 Delivery of Certain Documents

. Agent has received true, correct and complete copies of (a) the material ABL Loan Documents, and (b) any material documents governing or giving rise to any Subordinated Debt (including all exhibits, schedules and disclosure letters referred to in any of the foregoing or delivered pursuant thereto), if any and, in each case, all amendments thereto, waivers relating thereto and other side letters or agreements affecting the terms thereof. None of such documents and agreements has been amended or supplemented, nor have any of the provisions thereof been waived, except, in each case to the extent material to the interests of the Lenders, pursuant to a written agreement or instrument which has heretofore been delivered to Agent.

### V.23 Swaps

. No Credit Party or Subsidiary thereof is a party to, nor will it be a party to, any swap agreement whereby such Credit Party has agreed or will agree to swap interest rates or currencies unless same provides that damages upon termination following an event of default thereunder are payable on an unlimited "two-way basis" without regard to fault on the part of either party.

#### V.24 Immaterial Subsidiaries

. Each Subsidiary of DDH Holdings that has not become a Borrower or Guarantor pursuant to <u>Section 6.12</u> of this Agreement is an Immaterial Subsidiary.

V.25 [Reserved].

÷

#### V.26 Business and Property of Credit Parties

. (i) No Credit Party nor Subsidiary thereof engages in or proposes to engage in any business other than as described on <u>Schedule 5.26</u>, and other activities reasonably related thereto and extensions thereof; (ii) none of DDH, Intermediate Holdco or DDH Holdings (subsequent to a Qualified IPO) has engaged in, and does not engage in, any business activities, own any assets or have any liabilities other than (a) acting as a holding company and transactions incidental thereto, including maintaining policies of insurance with respect to directors and officers liability and other insurable risks customary for similarly situated companies, (b) receiving and distributing Restricted Payments permitted pursuant to the terms hereof, (c) liabilities arising under this Agreement, any Other Document and the ABL Loan Documents to which it is a party, (d) owning the Equity Interests of its Subsidiary, (e) activities, ownership of immaterial assets

[Different first page setting changed from on in original to off in modified.].

and contractual rights, in each case, incidental to the foregoing and to the maintenance of its corporate existence, (f) activities required to comply with federal and state securities laws, (g) any obligations under any Acquisition Documents in respect of Permitted Acquisitions, (h) liabilities in respect of the Tax Receivable Agreement, and (i) liabilities in respect of Public Company Costs.

V.27 Insurance

. The properties of the Credit Parties and their Subsidiaries are insured with financially sound and licensed insurance companies not Affiliates of such Persons to the extent required under <u>Section 4.11</u>. The insurance coverage of the Credit Parties and their Subsidiaries as in effect on the Closing Date is outlined as to carrier, policy number, expiration date, type, amount and deductibles on <u>Schedule 5.27</u>.

V.28 [Reserved].

Ŧ

V.29 Anti-Terrorism Laws; Anti-Corruption Laws.

(a) Each Credit Party, its Subsidiaries and their respective officers and employees and, to the knowledge of such Credit Party, its directors and agents, are in compliance with applicable Sanctions and are not engaged in any activity that would reasonably be expected to result in any Credit Party being designated as a Sanctioned Person. None of the Credit Parties, their Subsidiaries nor their respective Affiliates is in violation of any of the country or list based economic and trade sanctions administered and enforced by OFAC that are described or referenced at http://www.ustreas.gov/offices/enforcement/ofac/ or as otherwise published from time to time.

(b) None of the Credit Parties or their Subsidiaries or, to the knowledge of each Credit Party or its Subsidiaries, any of their respective directors, officers, employees or Affiliates (i) is a Sanctioned Person, (ii) has any of its assets located in a Sanctioned Country (unless approved by Agent), or (iii) derives any of its operating income from investments in, or transactions with Sanctioned Persons (unless approved by Agent). The proceeds of any Term Loan or other transaction contemplated by this Agreement or any Other Document have not been used (x) in violation of any Sanctions, (y) to fund any operations in, finance any investments or activities in or make any payments to, a Sanctioned Person or a Sanctioned Country or (z) in any other manner that would result in a violation of Sanctions by any Person (including Agent, any Lender and any other Person participation in any Term Loan, whether as an underwriter, advisor, investor or otherwise).

(c) Each of the Credit Parties and each of its Subsidiaries and, to the knowledge of each Credit Party and its Subsidiaries, each of their respective directors, officers, employees and Affiliates, is in compliance with Anti-Corruption Laws. None of the Credit Parties nor their respective Subsidiaries has made a payment, offering, or promise to pay, or

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

authorized the payment of, money or anything of value (i) in order to assist in obtaining or retaining business for or with, or directing business to, any foreign official, foreign political party, party official or candidate for foreign political office, (ii) to a foreign official, foreign political party or party official or any candidate for foreign political office, and (iii) with the intent to induce the recipient to misuse his or her official position to direct business wrongfully to such Credit Party or any of its Subsidiaries or to any other Person, in violation of any Anti-Corruption Law. No part of the proceeds of any Advance or other transaction contemplated by this Agreement or any Other Document will violate Anti-Corruption Laws.

- (d) To the extent applicable, each Credit Party and its Subsidiaries are in compliance with the USA PATRIOT Act.
- V.30 Trading with the Enemy

. No Credit Party nor any of its Subsidiaries is an "enemy" or an "ally of the enemy" within the meaning of Section 2 of the Trading with the Enemy Act of the United States of America (50 U.S.C. App. §§ 1 *et seq.*), as amended. To its knowledge, no Credit Party nor any of its Subsidiaries is in violation of (a) the Trading with the Enemy Act, as amended, (b) any of the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B, Chapter V, as amended) or any enabling legislation or executive order relating thereto or (c) the USA PATRIOT Act. No Credit Party or any of its Subsidiaries (i) is a blocked person described in Section 1 of the Anti-Terrorism Order or (ii) to the best of its knowledge, engages in any dealings or transactions, or is otherwise associated, with any such blocked person.

V.31 [<u>Reserved</u>].

Ŧ

V.32 [<u>Reserved</u>].

5.33 Eligible Impact ServiceEligible Worker Solutions, Each Credit Party agrees that (i) it is not, and shall not be deemed to be, a representation or warranty of Agent or any Lender as to the adequacy of any "Eligible Impact ServiceWorker Solutions", the provider thereof or any impact thereof and neither Agent nor any Lender has made any such representation or warranty as to such program or service, (ii) it has, independently and without reliance upon Agent or any Lender and based on such documents and information as it has deemed appropriate, made its own appraisal of, and investigation into, each of the Eligible Impact ServicesWorker Solutions and made its own decision as to the ImpactWorker Solutions Subscription to such Eligible Impact ServicesWorker Solutions (and continue to make each such decision) and (iii) Agent shall have no responsibility to ascertain, inquire into, monitor or enforce any such program or service, any provider thereof or any impact thereof.

V.33 Data Privacy and Security

[Different first page setting changed from on in original to off in modified.].

. Each Credit Party and its Subsidiaries has at all times complied in all material respects with (i) all of its policies and notices regarding Personal Information, privacy and data security, including all privacy policies and similar disclosures published on its websites or otherwise communicated to third parties ("Privacy Policies"), (ii) all Privacy and Security Laws, and (iii) all contractual commitments that each Credit Party and its Subsidiaries has entered into with respect to Personal Information. Except as, individually or in the aggregate, could not reasonably be expected to result in a Material Adverse Effect, each Credit Party and its Subsidiaries has not been a party to or the subject of any action, claim, complaint or investigation that alleges that the Credit Party or Subsidiary has violated any Privacy and Security Laws, Privacy Policies or contractual commitments with respect to any Personal Information collected or possessed by or on behalf of, or otherwise subject to the possession or control of, the Credit Party or Subsidiary. To the Credit Party's or Subsidiary's knowledge, there are no facts or circumstances which could form the basis for any such claim or violation that, individually or in the aggregate, could reasonably be expected to result in a Material Adverse Effect. Each Credit Party and its Subsidiaries has at all times taken all steps required (by applicable Privacy and Security Laws and Privacy Policies) and reasonably necessary to protect the security, availability, confidentiality and integrity of Personal Information against loss and against a Security Incident, including by implementing and at all times maintaining reasonable safeguards that are at least consistent with best practices in the industry in which the Credit Party and Subsidiary operates and contractually obligating all third parties who process, store, access, use or otherwise handle any Personal Information for or on behalf of the Credit Party or Subsidiary to comply with applicable Privacy and Security Laws and Privacy Policies and take all commercially reasonable steps to protect Personal Information. Each Credit Party and its Subsidiaries has not experienced (nor have any third parties acting on a Credit Party's or Subsidiary's behalf) any actual or suspected Security Incident that could reasonably be expected to result in a Material Adverse Effect. Each Credit Party and its Subsidiaries has not (nor, to such Credit Party's knowledge, have any third parties acting on such Credit Party's or Subsidiary's behalf) notified, or been required to notify, any Person or Governmental Body of any Security Incident that could reasonably be expected to result in a Material Adverse Effect, including any loss or unauthorized access, use or disclosure, of Personal Information.

#### V.34 Small Business Administration Documents

. If applicable, each Credit Party hereby acknowledges that the SBIC Lender is an SBIC and is subject to the SBIC Act and the regulations promulgated thereunder by the SBA relating to the small business investment company program. Although such regulations are not incorporated into this Agreement, each Credit Party acknowledges that it has been provided with an opportunity to review such regulations. The information set forth in the SBA Forms (if any) regarding each Credit Party shall be accurate and complete in all material respects.

#### V.35 Small Business Concern

. The Credit Parties, together with their "affiliates" (as that term is defined in Title 13, Code of Federal Regulations, § 121.103), do not presently engage in, and do not intend to engage

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

in, any activities, nor does any Credit Party intend to use, directly or indirectly, the proceeds of any Advance for any purpose violative of the SBIC Regulatory Side Letter (if any).

# V.36 Canadian Subsidiary

. Orange142 Canada does not (i) conduct any business operations or own, hold or lease any assets other than holding a bank account with a de minimis account balance or (ii) have any material liabilities.

#### V.37 Survival of Representations and Warranties

. All representations and warranties of the Credit Parties contained in this Agreement and the Other Documents shall be true at the time of the Credit Parties' execution of this Agreement and the Other Documents, and shall survive the execution, delivery and acceptance thereof by the parties thereto and the closing of the transactions described therein or related thereto.

### VI. AFFIRMATIVE COVENANTS.

Each Credit Party shall, and shall cause each of its Subsidiaries to, until payment in full of the Obligations (other than Inchoate Obligations) and termination of all obligations of Agent, the Lenders to make Advances hereunder and termination of this Agreement:

# VI.1 Lender Meetings

. Upon reasonable advance notice and at the request of Agent, participate in a meeting with Agent and Lenders one time each Fiscal Year (or such additional meetings as Borrower may agree to in its sole discretion) to be held at Borrowers' corporate offices (or at such other location as may be agreed to by Borrowers and Agent) at such time as may be agreed to by Borrowers, Agent and Lenders; provided, that prior to receiving any material, non-public information at any such meetings, Agent and Lenders shall execute a non-disclosure agreement with respect to such material, non-public information that is in form and substance satisfactory to the Credit Parties in their reasonable discretion, containing, among other things, standstill provisions preventing transactions in the Credit Parties' stock.

### VI.2 Conduct of Business and Maintenance of Existence and Assets

. At all times preserve and keep in full force and effect its existence and good standing, hold itself out to the public as a legal entity separate and distinct from all other Persons, and preserve and keep all assets, rights, franchises, licenses and permits material to its business, except (i) to the extent permitted by <u>Section 7.1</u>, (ii) to the extent not constituting an Asset Sale hereunder, or (iii) to the extent the failure to maintain such assets, rights, franchises, licenses, or permits could not reasonably be expected to result in a Material Adverse Effect.

### VI.3 <u>Violations</u>

[Different first page setting changed from on in original to off in modified.].

. Promptly notify Agent in writing after learning of any violation of any law, statute, regulation or ordinance of any Governmental Body, or of any agency thereof, applicable to any Credit Party which could reasonably be expected to have a Material Adverse Effect.

# VI.4 Government Receivables

. Upon Agent's reasonable request, take all steps necessary to protect Agent's interest in the Collateral under the Federal Assignment of Claims Act, the Uniform Commercial Code and all other applicable state or local statutes or ordinances and deliver to Agent (or ABL Lender) appropriately endorsed, any instrument or chattel paper connected with any Receivable arising out of contracts between any Credit Party and the United States, any state or any department, agency or instrumentality of any of them, except that no Credit Party shall be required to take such actions with respect to Receivables not exceeding \$250,000 in the aggregate for all Credit Parties unless otherwise requested in writing by Agent during the continuance of an Event of Default.

# VI.5 Financial Covenants.

(a) <u>Consolidated Total Net-Leverage Ratio.</u> Cause to be maintained, when measured as of each date set forth below, a <u>Consolidated Total Net-Leverage Ratio of not more than the amount set forth opposite thereto:</u>

Measurement Dates	<u>Maximum Consolidated Total <del>Net</del>-Leverage</u>
December 31, 2021 and the last day of each Fiscal	3.50 to 1.00
Quarter through and including December 31, 2023	
<u>March 31, 2024</u>	<u>3.25 to 1.00</u>
March 31June 30, 2024 and the last day of each Fiscal	<del>3.25 to 1.00<u>n/a</u></del>
Quarter through and including March 31 June 30, 2025	
June 30, 2025 and September 30, 2025	<del>3.00<u>4.25</u> to 1.00</del>
<u>December 31, 2025</u>	<u>4.00 to 1.00</u>
December 31, 2025 and March 31, 2026	<del>2.75</del> <u>3.75</u> to 1.00
<u>June 30, 2026</u>	<u>3.50 to 1.00</u>
JuneSeptember 30, 2026 and the last day of each fiscal	<del>2.50</del> 3.25 to 1.00
quarter thereafter	

(b) <u>Consolidated Fixed Charge Coverage Ratio</u>. Cause to be maintained a Consolidated Fixed Charge Coverage Ratio as of the end of each Fiscal Quarter, (a) beginning

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5 with the Fiscal Quarter ending December 31, 2021, of not less than 1.51.25 to 1.00 through and including March 31, 2024 and (b) beginning with the Fiscal Quarter ending June 30, 2025 and thereafter, of not less than 1.50 to 1.00.

(e) <u>Minimum Liquidity. Cause to be maintained Liquidity, at all times, (a) beginning with the Fiscal Quarters ending September</u> 30, 2024 and December 31, 2024, of not less than One Million Five Hundred Thousand Dollars (\$1,500,000) and, (b) beginning with the Fiscal Quarter ending March 31, 2025, of not less than Two Million Dollars (\$2,000,000).

(f) Minimum Consolidated EBITDA. Cause to be maintained Consolidated EBITDA, as at the last day of the each of the following four (4) Fiscal Quarters periods, of not less than (a) \$5,000,000 for the most recently completed four (4) Fiscal Quarters ending September 30, 2024, (b) \$3,500,000 for the most recently completed four (4) Fiscal Quarters ending December 31, 2024, (c) \$5,500,000 for the most recently completed four (4) Fiscal Quarters ending March 31, 2025, and (d) \$7,500,000 for the most recently completed four (4) Fiscal Quarters ending June 30, 2025.

# VI.6 Execution of Supplemental Instruments

. Execute and deliver to Agent from time to time, upon demand, such supplemental agreements, statements, assignments and transfers, or instructions or documents relating to the Collateral, and such other instruments as Agent may reasonably request in its Permitted Discretion, in order that the full intent of this Agreement may be carried into effect.

# VI.7 Payment of Indebtedness

. Pay, discharge or otherwise satisfy at or before maturity (subject, where applicable, to specified grace periods and, in the case of the trade payables, to normal payment practices) all its obligations and liabilities of whatever nature, except when the failure to do so could not reasonably be expected to have a Material Adverse Effect or when the amount or validity thereof is currently being properly contested in good faith by appropriate proceedings promptly instituted and diligently conducted and in respect of which reserves have been established as required in conformity with GAAP, subject at all times to any applicable subordination arrangement in favor of Lenders.

# VI.8 Real Property Collateral.

Ŧ

(a) In the event that any Credit Party owns or acquires Real Property (other than any Real Property constituting Excluded Property), then such Credit Party, no later than ninety (90) days (or such longer period as may be agreed in writing by Agent in its sole discretion) after acquiring such Real Property shall take all such actions and execute and deliver, or cause to be executed and delivered, all such Mortgages, documents, instruments, agreements, opinions and certificates similar to those described in clause (b) immediately below that Agent

[Different first page setting changed from on in original to off in modified.].

shall reasonably request to create in favor of Agent a valid and, subject to any filing and/or recording referred to herein, enforceable Lien on, and security interest in such Real Property. Agent may, in its reasonable judgment, grant extensions of time for compliance or exceptions with respect to the provisions of this Section 6.8 by any Credit Party. In addition to the foregoing, the applicable Credit Party shall, at the reasonable request of Agent, deliver, from time to time, to Agent such appraisals as are required by law or regulation of Real Property with respect to which Agent has been granted a Lien.

(b) <u>In order to create in favor of Agent, a valid and, subject to any filing and/or recording referred to herein, enforceable Lien</u> on, and security interest in, any Real Property Collateral that is prior and superior in right to any other Lien (other than Permitted Encumbrances), Agent shall have received from Borrowers with respect to such Real Property Collateral:

(i) <u>a fully executed and notarized Mortgage, in proper form for recording in all appropriate places in all applicable</u> jurisdictions, encumbering such Real Property Collateral;

(ii) <u>an opinion of counsel (which counsel shall be reasonably satisfactory to Agent) in each state in which such Real</u> <u>Property is located with respect to the enforceability of the form of Mortgage to be recorded in such state and such other matters as Agent may</u> <u>reasonably request, in each case in form and substance reasonably satisfactory to Agent;</u>

(iii) (A) ALTA mortgagee title insurance policies or unconditional commitments therefor issued by one or more title companies reasonably satisfactory to Agent (each, a "Title Policy") with respect to such Real Property, in amounts not less than the fair market value of such Real Property, together with a title report issued by a title company with respect thereto and copies of all recorded documents listed as exceptions to title or otherwise referred to therein, each in form and substance reasonably satisfactory to Agent and (B) evidence reasonably satisfactory to Agent that Borrower has paid to the title company or to the appropriate Governmental Bodies all expenses and premiums of the title company and all other sums required in connection with the issuance of each Title Policy and all applicable recording and stamp taxes (including mortgage recording and intangible taxes) payable in connection with recording the Mortgage for such Real Property Collateral in the appropriate real estate records;

(iv) <u>a recently issued flood zone determination certificate;</u>

(v) evidence of flood insurance with respect to each Flood Hazard Property that is located in a community that participates in the National Flood Insurance Program, in each case in compliance with any applicable regulations of the Board of Governors of the Federal Reserve System, in form and substance reasonably satisfactory to Agent;

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

(vi) if an exception to the Title Policy with respect to any Real Property subject to a Mortgage would arise without such ALTA surveys, ALTA surveys of such Real Property; and

(vii) reports and other reasonable information, in form, scope and substance reasonably satisfactory to Agent, regarding environmental matters relating to such Real Property Collateral.

# VI.9 Federal Securities Laws

. All documents, financial statements, notices and other information required to be delivered pursuant to this Agreement may be delivered to Agent electronically and, if so delivered to Agent, shall be deemed to have been delivered on the date (i) on which such materials are publicly available as posted on the Electronic Data Gathering, Analysis and Retrieval system (EDGAR); provided Borrower will, within three (3) Business Days, provide notice to Agent of such material being available on EDGAR; or (ii) on which such documents are posted on a Credit Party's behalf on an Internet or intranet website, if any, to which the Lenders and the Agent have access and approve in writing (whether a commercial or third-party website or whether made available by the Agent.

# VI.10 Exercise of Rights

. Enforce all of its rights under any Acquisition Documents including, but not limited to, all indemnification rights and pursue its available remedies in respect thereof in the exercise of its reasonable business judgment with diligence and in good faith; provided, however, that, unless an Event of Default has occurred and is continuing, if in any particular instance the probable amount of any indemnity payment which would be payable on account of the pursuit of any rights or remedies under any Acquisition Documents would be less than \$250,000 and the applicable Credit Party in the exercise of its reasonable business judgment determines that in such instance the pursuit of such rights and remedies is not in the best interests of such Credit Party, then the failure of such Credit Party in such instance to pursue any such rights and remedies shall not be a violation of this <u>Section 6.10</u>.

#### VI.11 [Reserved].

#### VI.12 Additional Subsidiaries

. Prior to the acquisition or formation of any Subsidiary of DDH after the Closing Date, (i) DDH shall (a) form a wholly owned Subsidiary holding company (such wholly owned Subsidiary, an "Intermediate Holdco"), (b) transfer, sell and assign all Equity Interests owned by DDH in its existing Subsidiaries to Intermediate Holdco, (c) cause Intermediate Holdco to execute and deliver (I) a joinder agreement to this Agreement fully executed by Intermediate Holdco, the Credit Parties and the Agent pursuant to which Intermediate Holdco shall become a Guarantor hereunder and become jointly and severally liable for the Obligations of the Guarantors hereunder and grant a lien and security interest in its property (to the extent such

[Different first page setting changed from on in original to off in modified.].

property would be included in the definition of Collateral), together with amended and restated and schedules to this Agreement, (II) a fully executed Pledge Agreement by Intermediate Holdco pursuant to which Intermediate Holdco shall pledge all of the Equity Interests owned by Intermediate Holdco, (III) such other documents (including without limitation, Control Agreements) as Agent deems necessary to grant to Agent a security interest in any property of Intermediate Holdco (to the extent such property would be included in the definition of Collateral), and (IV) any other documents Agent may reasonably require in connection with the forgoing, including without limitation, legal opinions, certificates, and any documentation and other information requested in connection with applicable "know your customer" and anti-money laundering rules and regulations, including the USA Patriot Act, and (d) DDH shall deliver a fully executed Pledge Agreement pursuant to which DDH shall pledge all of the Equity Interests of Intermediate Holdco, and (ii) upon satisfaction of the conditions set forth in clause (i) above, within thirty (30) days after the acquisition or formation of any Subsidiary of DDH or Intermediate Holdco, the Credit Parties shall:

(a) notify Agent thereof in writing, together with the (i) jurisdiction of formation, (ii) number of shares of each class of Equity Interests outstanding, (iii) number and percentage of outstanding shares of each class owned (directly or indirectly) by any Credit Party or any Subsidiary and (iv) number and effect, if exercised, of all outstanding options, warrants, rights of conversion or purchase and all other similar rights with respect thereto;

(b) cause such Person (other than an Immaterial Subsidiary) to at the election of Agent, either (x) become a Borrower hereunder and under the Other Documents by executing and delivering to Agent such assumption agreements, joinder documentation (including in the form of Exhibit G hereto) or such other documents as Agent shall deem appropriate for such purpose, or (y) execute and deliver a Guarantee, Guarantor Security Agreement and joinders to such Other Documents as Agent may request; and

(c) <u>deliver to Agent (i) documents of the types referred to in Sections 8.1(b), (c), (d), (e), (f) and (u), and favorable opinions of counsel to such Person (other than an Immaterial Subsidiary) (which shall cover, among other things, the legality, validity, binding effect and enforceability of the documentation referred to in the immediately foregoing clause (b)), all in form, content and scope satisfactory to Agent and (ii) all documents (including Pledge Agreements with respect to the Equity Interests of such Subsidiary), including original certificates evidencing the Equity Interests of such Subsidiary and transfer powers with respect thereto executed in blank, required by Agent in its Permitted Discretion to perfect its Lien on the Equity Interests of such Subsidiary (other than an Immaterial Subsidiary).</u>

VI.13 [Reserved].

Ŧ

VI.14 Sanctions and Anti-Corruption Matters

[Different first page setting changed from on in original to off in modified.].

. Each Credit Party will, and will cause each of its Subsidiaries to, conduct its businesses in compliance with Sanctions and the United States Foreign Corrupt Practices Act of 1977, the UK Bribery Act 2010 and other similar anti-corruption legislation in other jurisdictions and maintain policies and procedures designed to promote and achieve compliance with such laws.

VI.15 ImpactWorker Solutions Subscription to or Impact Services Worker Solutions .

(a) Use commercially reasonable and good faith efforts, subject to applicable Federal and state securities laws, to (i) in connection with any ImpactWorker Solutions Subscription to or Impact ServicesWorker Solutions otherwise obtained by, and made generally available to the employees of, each such Credit Party and Subsidiary, educate such employees of each Credit Party and each Subsidiary of a Credit Party as to the offerings in each such Impact ServicesWorker Solutions and the impact and benefits thereof, and (ii) maintain corporate citizenship and utilize best standards of social responsibility, including, without limitation, with respect to human capital management (it being agreed and acknowledged that the job quality standards promulgated by the Good Jobs Institute (https://goodjobsinstitute.org) and the good faith utilization of the "Good Jobs Scoreboard" promulgated thereby shall be deemed to be maintaining corporate citizenship and utilizing best standards of social responsibility). Without limitation, guidance and counsel concerning the management, operations or business objectives and policies of) the Credit Parties and their Subsidiaries with respect to such efforts to the extent reasonably requested by each such Credit Party and Subsidiary and (y) the Credit Parties shall identify and certify as to any and all efforts to educate employees under clause (i) of this Section 6.15(a), including, without limitation, as to any workshops, intranet postings, literature distribution, employee e-mails or otherwise, in each ImpactWorker Solutions Certificate delivered pursuant to Section 9.8.

(b) <u>Deliver to any ImpactWorker Solutions Provider all reasonable documentation such ImpactWorker Solutions Provider may</u> require in order to provide Agent with information from or regarding the Credit Parties.

## VI.16 Post-Closing Deliveries.

(a) Promptly, but in any event, not later than the Second Amendment Date, deliver or cause to be delivered to Agent (i) a joinder agreement to this Agreement fully executed by DDH Holdings, the Credit Parties and the Agent pursuant to which DDH Holdings shall become a Guarantor hereunder and become jointly and severally liable for the Obligations of the Guarantors hereunder and grant a lien and security interest in its property (to the extent such property would be included in the definition of Collateral), together with amended and restated and schedules to this Agreement, (ii) a fully executed Pledge Agreement by DDH Holdings and any Permitted Holders (other than Leah Woolford or any Person owned by Leah Woolford) owning Equity Interests of DDH, pursuant to which DDH Holdings and such Permitted Holders shall pledge all of the Equity Interests owned by such Person in DDH, (iii) such other documents (including without limitation, Control Agreements) as Agent deems necessary to grant to Agent a

[Different first page setting changed from on in original to off in modified.].

security interest in any property of DDH Holdings (to the extent such property would be included in the definition of Collateral), and (iv) any other documents Agent may reasonably require in connection with the forgoing, including without limitation, legal opinions, certificates, and any documentation and other information requested in connection with applicable "know your customer" and anti-money laundering rules and regulations, including the USA Patriot Act.

(b) Deliver to Agent the Key Executive Policy as required under and in accordance with Section 4.21.

(c) Promptly, but in any event, on or before February 28, 2022 (or such later date consented to by Agent, such consent not to be unreasonably withheld, conditioned or delayed), deliver or cause to be delivered to Agent fully executed deposit account control agreements ("Control Agreements") in form and substance reasonably satisfactory to Agent, among Agent, ABL Lender, the financial institutions or other Person at which all applicable deposit accounts are maintained and the Credit Party maintaining such account(s), with respect to all deposit accounts of Credit Parties (other than those utilized solely for making payroll or employee benefit related payments), including, without limitation, the Depository Accounts and the Funding Accounts, which agreements shall be effective for Agent to obtain "control" (as defined in the Uniform Commercial Code) of all such accounts and all monies on deposit in, or credited to, such deposit accounts and all certificates and instruments, if any, representing or evidencing such deposit accounts.

(d) Deliver to Agent the Landlord Waiver Agreements as required under and in accordance with Section 4.5(c).

(e) Within fourteen days after the Closing Date (or such later date agreed to by Agent and the Required Lenders in their reasonable discretion), Borrowers shall deliver to Agent an amendment to the existing ABL Credit Agreement (an "ABL Amendment") in effect on the date hereof, in form and substance satisfactory to Agent and the Required Lenders, pursuant to which ABL Lender agrees to increase the commitments under the existing ABL Credit Agreement to \$5,000,000 and agrees to such other amendments which are satisfactory to Agent and the Required Lenders in their reasonable discretion; provided however that if such ABL Amendment is not executed and delivered on or prior to the due date set forth above, it shall not be an Event of Default hereunder if Borrowers terminate the existing ABL Credit Agreement and repay all obligations owing thereunder no later than the fourteenth day after the Closing Date (or such later date agreed to by Agent and the Required Lenders in their reasonable discretion).

(f) <u>Within ten (10) days of the Closing Date, Borrowers shall deliver evidence that their insurance policies are in compliance</u> with Section 4.11 hereof (or otherwise acceptable to Agent and Required Lenders in their sole discretion) and deliver certificates of insurance evidencing such compliance and the endorsements required to be delivered pursuant to Section 4.11 hereof.

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

#### VII. NEGATIVE COVENANTS.

The Credit Parties shall not, and shall not permit their respective Subsidiaries, until satisfaction in full of the Obligations (other than Inchoate Obligations), termination of all obligations of Agent and the Lenders to make any Advances hereunder and termination of this Agreement, to:

### VII.1 Merger, Consolidation, Acquisition and Sale of Assets

. Enter into any Acquisition or transaction of merger or consolidation, or liquidate, wind-up or dissolve itself (or suffer any liquidation or dissolution), or consummate an LLC Division, or make any Asset Sale (including by way of an LLC Division), or acquire by purchase or otherwise (other than purchases or other acquisitions of inventory and materials and the acquisition of equipment and Capital Expenditures in the ordinary course of business) the business, property or fixed assets of, or Equity Interests or other evidence of beneficial ownership of, any Person or any division or line of business or other business unit of any Person, except: (a) any Subsidiary of DDH may be merged with or into DDH or any Subsidiary of DDH, or be liquidated, wound up or dissolved, or all or any part of its business, property or assets may be conveyed, sold, leased, transferred or otherwise disposed of, in one transaction or a series of transactions, to DDH or any other Subsidiary of DDH; provided, in the case of such a merger, if a Borrower is party to the merger, a Borrower shall be the continuing or surviving Person; (b) Asset Sales, the proceeds of which when aggregated with the proceeds of all other Asset Sales made within the same Fiscal Year, do not exceed \$100,000; provided the consideration received for such assets shall be in an amount at least equal to the fair market value thereof (determined in good faith by the board of directors of the applicable Credit Party (or similar governing body)); (c) Investments made in accordance with <u>Section 7.4</u>; (d) sales or dispositions or redemptions of Equity Interests in connection with the Redemption/Exchange Transactions.

#### VII.2 Creation of Liens

. Directly or indirectly, create, incur, assume or permit to exist any Lien on or with respect to any property or asset of any kind (including any document or instrument in respect of goods or accounts receivable) of any Credit Party or any of its Subsidiaries, whether now owned or hereafter acquired, created or licensed or any income, profits or royalties therefrom, or file or permit the filing of, or permit to remain in effect, any financing statement or other similar notice of any Lien with respect to any such property, asset, income, profits or royalties under the Uniform Commercial Code of any State or under any similar recording or notice statute or under any Applicable Laws related to intellectual property, except:

(a) Liens in favor of Agent granted pursuant to this Agreement or any Other Document;

[Different first page setting changed from on in original to off in modified.].

# (b) Liens in favor of ABL Lender pursuant to the ABL Loan Documents to the extent subject to the Intercreditor Agreement;

(c) Liens for Taxes not yet due and payable or for Taxes if obligations with respect to such Taxes are being contested in good faith by appropriate proceedings promptly instituted and diligently conducted;

(d) <u>statutory Liens of landlords, banks, carriers, warehousemen, mechanics, repairmen, workmen and materialmen, and other</u> <u>Liens imposed by law (other than any such Lien imposed pursuant to Section 430(k) of the Code or Section 303(k) or 4068 of ERISA that would</u> <u>constitute an Event of Default under Section 10.17), in each case incurred in the ordinary course of business (i) for amounts not yet overdue, or (ii)</u> for amounts that are overdue and that are being contested in good faith by appropriate proceedings, so long as such reserves or other appropriate provisions, if any, as shall be required by GAAP shall have been made for any such contested amounts;

(e) Liens incurred in the ordinary course of business in connection with workers' compensation, unemployment insurance and other types of social security, or to secure the performance of tenders, statutory obligations, surety and appeal bonds, bids, leases, government contracts, trade contracts, performance and return-of-money bonds and other similar obligations (exclusive of obligations for the payment of borrowed money or other Indebtedness), so long as no foreclosure, sale or similar proceedings have been commenced with respect to any portion of the Collateral on account thereof;

(f) <u>easements, rights-of-way, restrictions, encroachments, and other minor defects or irregularities in title, in each case which</u> <u>do not and will not interfere in any material respect with the ordinary conduct of the business of any Credit Party or any of its Subsidiaries,</u> <u>including, without limitation, all encumbrances shown on any policy of title insurance in favor of Agent with respect to any Real Property</u> <u>Collateral;</u>

(g) any interest or title of a lessor or sublessor under any lease of real estate permitted hereunder;

(h) Liens solely on any cash earnest money deposits made by any Credit Party or any of its Subsidiaries in connection with any letter of intent, or purchase agreement permitted hereunder;

(i) purported Liens evidenced by the filing of precautionary UCC financing statements relating solely to operating leases of personal property entered into in the ordinary course of business;

(j) Liens in favor of customs and revenue authorities arising as a matter of law to secure payment of customs duties in connection with the importation of goods;

[Different first page setting changed from on in original to off in modified.].

(k) any zoning or similar law or right reserved to or vested in any governmental office or agency to control or regulate the use of any real property;

(l) <u>licenses of patents, trademarks and other intellectual property rights granted by any Credit Party or any of its Subsidiaries in</u> the ordinary course of business and not interfering in any respect with the ordinary conduct of the business of such Credit Party or such Subsidiary;

(m) Liens existing as of the Closing Date and described in Schedule 7.2;

(n) <u>Liens securing purchase money Indebtedness and Capitalized Lease Obligations to the extent permitted pursuant to Section</u> 7.8(g); provided, any such Lien shall encumber only the asset acquired with the proceeds of such purchase money Indebtedness or the assets subject to the applicable Capital Lease, respectively;

(o) Liens consisting of judgment or judicial attachment liens relating to judgments which do not constitute an Event of Default

# hereunder;

(p) licenses (including licenses of Intellectual Property), sublicenses, leases or subleases granted to third parties in the ordinary course of business;

- (q) Liens in favor of collecting banks under Section 4-210 of the Uniform Commercial Code;
- (r) Liens (including the right of set-off) in favor of a bank or other depository institution arising as a matter of law encumbering deposits;

(s) Liens arising out of conditional sale, title retention, consignment or similar arrangements for the sale of goods in the ordinary course of business; and

(t) Liens not otherwise permitted hereunder securing Indebtedness or other obligations not in excess of \$250,000 in the aggregate at any one time outstanding.

# VII.3 No Further Negative Pledges

. Enter into any Contractual Obligation (other than this Agreement and the Other Documents and the ABL Loan Documents) that limits the ability of any Credit Party or any such Subsidiary to create, incur, assume or suffer to exist Liens on property of such Person; <u>provided</u>, <u>however</u>, that this <u>Section 7.3</u> shall not prohibit (i) any negative pledge incurred or provided in favor of any holder of Indebtedness permitted under <u>Section 7.8(g)</u>, solely to the extent any such negative pledge relates to the property financed by or subject to Permitted Encumbrances securing such Indebtedness, (ii) any Permitted Encumbrance or any document or instrument governing any Permitted Encumbrance; <u>provided</u> that any such restriction contained therein relates only to the asset or assets subject to such Permitted Encumbrance, (iii) customary restrictions and conditions contained in any agreement relating to the disposition of any property

[Different first page setting changed from on in original to off in modified.].

or assets permitted under <u>Section 7.1</u> pending the consummation of such disposition, and (iv) customary provisions restricting assignments, subletting or other transfers contained in leases, licenses, joint venture agreements and similar agreements entered into in the ordinary course of business.

### VII.4 Investments

. Directly or indirectly, make or own any Investment in any Person, including any joint venture, or create or form any Subsidiary, except: (a) Investments in cash and Cash Equivalents and deposit accounts or securities accounts in connection therewith; (b) (i) equity Investments owned as of the Closing Date in any Subsidiary and (ii) (A) Investments made after the Closing Date by the Credit Parties in any Credit Party and (B) Investments by a Credit Party in Subsidiaries that are not a Credit Party for the ordinary and necessary operating expenses of such Subsidiaries in an amount not to exceed \$200,000 in the aggregate in any fiscal year and (C) Investments by Subsidiaries that are not a Credit Party in any other Subsidiary for the ordinary and necessary current operating expenses of such Credit Party in an amount not to exceed \$200,000 in the aggregate in any fiscal year or in any Credit Party; (c) intercompany loans to the extent permitted under Section 7.8(d) and guarantees to the extent permitted under Section 7.8(e); (d) Investments existing on the Closing Date and described on Schedule 7.4; (e) Investments constituting Swap Agreements permitted by Section 7.8(h); (f) Permitted Acquisitions; (g) Investments constituting accounts receivable, trade debt and deposits for the purchase of goods, in each case, made in the ordinary course of business; (h) repurchases of stock from current or former employees, directors, or consultants of the Credit Parties under the terms of applicable repurchase agreements: (i) prior to the consummation of a Qualified IPO, at the original issuance price of such securities in an aggregate amount not to exceed \$250,000 in any fiscal year; provided that no Event of Default has occurred, is continuing or could exist after giving effect to the repurchases and (ii) following a Qualified IPO, subject to satisfaction of the Payment Conditions; (i) Investments consisting of travel advances in the ordinary course of business; (j) Investments consisting of loans not involving the net transfer on a substantially contemporaneous basis of cash proceeds to employees, officers or directors relating to the purchase of capital stock or other Equity Interests of the Credit Parties pursuant to employee stock purchase plans or other similar agreements approved by such Credit Party's Board of Directors (or, if not a corporation, its equivalent authorizing body); (k) Investments (including debt obligations) received in connection with the bankruptcy or reorganization of customers or suppliers and in settlement of delinquent obligations of, and other disputes with, customers or suppliers arising in the ordinary course of Borrowers' business; (1) Investments consisting of notes receivable of, or prepaid royalties and other credit extensions to, customers and suppliers who are not Affiliates, in the ordinary course of business; provided that this subparagraph (1) shall not apply to Investments of any Borrower in any Subsidiary; (m) formation of, and Investments in, newly-formed Subsidiaries of DDH or any Subsidiary of DDH so long as DDH and such Subsidiary comply with Section 6.12 hereof; (n) Investments made to consummate the Reorganization Transactions in connection with a Qualified IPO; (o) Investments required to finance the buyout of the SSP licensing agreement by a Credit Party in an amount not to exceed \$750,000 in the aggregate; provided that no Event of Default has occurred and is continuing or

[Different first page setting changed from on in original to off in modified.].

would occur after giving effect to such Investments; and (p) other Investments not listed above and not otherwise prohibited by this Agreement (i) prior to the consummation of a Qualified IPO, in an aggregate amount not to exceed \$250,000 in the aggregate in any fiscal year and (ii) following a Qualified IPO, in an aggregate amount not to exceed \$500,000 in the aggregate in any fiscal year; provided that to the extent such Investments shall be subject to satisfaction of the Payment Conditions.

#### VII.5 Sale and Lease-Backs

. Directly or indirectly, become or remain liable as lessee or as a guarantor or other surety with respect to any lease of any property (whether real, personal or mixed), whether now owned or hereafter acquired, which the Credit Party or any Subsidiary (a) has sold or transferred or is to sell or to transfer to any other Person (other than a Credit Party), or (b) intends to use for substantially the same purpose as any other property which has been or is to be sold or transferred by a Credit Party to any Person (other than a Credit Party) in connection with such lease.

#### VII.6 [Reserved].

Ŧ

# VII.7 Restricted Payments

. Declare or make, directly or indirectly, any Restricted Payment, or incur any obligation (contingent or otherwise) to do so, except: (a) each Subsidiary of a Borrower may make Restricted Payments to a Borrower, (b) the Preferred A Redemption on the Closing Date, (c) Permitted Dividends, (d) redemptions of Equity Interests pursuant to employee, director or consultant repurchase plans or other similar agreements in an aggregate amount not to exceed \$250,000 in any fiscal year; provided, however, in each case (other than any such repurchase or redemption in the ordinary course of business in connection with an employee incentive plan) the repurchase or redemption price does not exceed the original consideration paid for such stock or Equity Interest, (e) the conversion of any of its convertible securities into other securities of such Credit Party pursuant to the terms of such convertible securities, (f) the payment of cash in lieu of fractional shares upon the conversion of any such convertible securities, not to exceed \$500,000 in the aggregate during the Term, (g) Permitted Management Fee Payments or (h) any Credit Party and any Subsidiary may make Tax Distributions to its members or shareholders when the applicable Tax giving rise to such Tax Distribution is due, including in respect of estimated Tax payments, so long as: (i) no Event of Default or Default shall be continuing at the time of and after giving pro forma effect to such Tax Distribution; and (ii) the purpose and calculation of such Tax Distribution shall be set forth in writing and submitted to Agent at least ten (10) days prior to such Tax Distribution, which shall be conclusive absent manifest error, and such Tax Distribution shall in fact be used for such purpose, unless an alternative use is approved by the Agent, (i) subsequent to a Qualified IPO and to the extent made with cash proceeds of a Qualified IPO, cash payments to any holders of DDH's Preferred Equity, (j) cash payments to

[Different first page setting changed from on in original to off in modified.].

the holders of DDH's Preferred Equity and cash payments with respect to obligations owing under the Redemption Agreement subsequent to a Qualified IPO to the extent made with proceeds of the Second Amendment Delayed Draw Term Loan, (k) the transactions contemplated pursuant to a Qualified IPO, (l) with respect to the payment of Public Company Costs; (m) in order to effect the Redemption/Exchange Transactions; and (n) in order to effect the Warrant Tender Offer.

### VII.8 Indebtedness

. Directly or indirectly, create, incur, assume or guaranty, or otherwise become or remain directly or indirectly liable with respect to any Indebtedness other than (a) the Obligations; (b) the ABL Obligations in an amount not to exceed the maximum permitted amount set forth in the Intercreditor Agreement; (c) the Subordinated Debt; provided that such Indebtedness shall at all times be subject to a Subordination Agreement and have an earliest maturity date or call or redemption date, as applicable, no earlier than at least six (6) months after the last day of the Term; (d) Indebtedness of the Credit Parties to any other Credit Party; (e) Guarantees with respect to Indebtedness permitted under this Section 7.8; (f) Indebtedness existing on the Closing Date and described in Schedule 7.8; (g) (x) Capitalized Lease Obligations and (y) purchase money Indebtedness; provided, in the case of clause (x), that any such Indebtedness shall be secured only by the asset subject to the applicable Capital Lease, and, in the case of <u>clause (y)</u>, that any such Indebtedness shall be secured only by the asset acquired in connection with the incurrence of such Indebtedness; provided further that the sum of the aggregate principal amount of any Indebtedness under this clause (g) shall not exceed at any time \$250,000; (h) Indebtedness in respect of any Swap Agreement that is entered into in the ordinary course of business to hedge or mitigate risks to which any Credit Party or any of its Subsidiaries is exposed in the conduct of its business or the management of its liabilities (it being acknowledged by the Credit Parties that a Swap Agreement entered into for speculative purposes or of a speculative nature is not a Swap Agreement entered into in the ordinary course of business to hedge or mitigate risks); (i) Indebtedness in connection with the financing of insurance premiums in the ordinary course of business; (j) Indebtedness representing deferred compensation to officers, directors, employees of DDH and its Subsidiaries; (k) to the extent constituting Indebtedness, accrued Taxes not yet due and payable or that are being contested in good faith by appropriate proceedings promptly instituted and diligently conducted; and (1) other unsecured Indebtedness of the Credit Parties not otherwise permitted hereunder in an aggregate amount not to exceed at any time \$250,000.

# VII.9 Nature of Business

. From and after the Closing Date, engage in any business other than the Permitted Lines of Business. Notwithstanding anything herein to the contrary, none of DDH, Intermediate Holdco or, subsequent to a Qualified IPO, DDH Holdings shall (i) incur any liabilities (other than liabilities arising (a) under this Agreement and the Other Documents and the ABL Loan Documents to which it is a party, (b) tax liabilities arising in the ordinary course of business, (c) obligations under any Acquisition documents in respect of a Permitted Acquisition, under any Acquisition Document with respect to a Permitted Acquisition to which it is a party, (d)

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

liabilities in connection with the Tax Receivables Agreement, (e) liabilities for Public Company Costs, and (f) liabilities in connection with the Redemption/Exchange Transactions), (ii) own or acquire any assets (other than the Equity Interests of its Subsidiaries, immaterial assets and contractual rights) or (iii) engage itself in any operations or business, except for those expressly set forth in <u>Section 5.26</u>.

# VII.10 Transactions with Affiliates

. Directly or indirectly, enter into or permit to exist any transaction (including the purchase, sale, lease or exchange of any property or the rendering of any service) with any officer, director or Affiliate of DDH or any of its Subsidiaries on terms that are less favorable to DDH or such Subsidiary, as the case may be, than those that might be obtained at the time from a Person who is not an officer, director or Affiliate of DDH or any of its Subsidiaries; provided, the foregoing restriction shall not apply to (a) any transaction between or among the Credit Parties; (b) normal and reasonable compensation and reimbursement of expenses of officers and directors in the ordinary course of business; (c) transactions otherwise permitted under this Agreement; (d) Permitted Management Fee Payments; (e) transactions pursuant to the Tax Receivable Agreement; (f) transactions pursuant to the Redemption/Exchange Transactions; (g) the transactions described in <u>Schedule 7.10</u>; and (h) transaction pursuant to the Warrant Tender Offer.

#### VII.11 Burdensome Agreements

. Enter into, or permit to exist, any Contractual Obligation that encumbers or restricts the ability of any such Person to (i) pay dividends or make any other distributions to a Borrower or other Credit Party on its Equity Interests or with respect to any other interest or participation in, or measured by, its profits, (ii) pay any Indebtedness or other obligation owed to a Borrower or any other Credit Party, (iii) make loans or advances to a Borrower or any other Credit Party, (iv) sell, lease or transfer any of its property to any Borrower or any other Credit Party, (v) pledge its property pursuant to this Agreement or any Other Document or any renewals, refinancings, exchanges, refundings or extension thereof, or (vi) act as a Credit Party pursuant to this Agreement and the Other Documents or any renewals, refinancings, exchanges, refundings or extension thereof, except, in each case, for (w) the ABL Loan Documents, this Agreement and the Other Documents, (x) any document or instrument governing Indebtedness incurred pursuant to <u>Section 7.8(g)</u>; provided that any such restriction contained therein relates only to the asset or assets constructed or acquired in connection therewith, (y) any Permitted Encumbrance or any document or instrument governing any Permitted Encumbrance, provided that any such restriction contained therein relates only to the asset or assets subject to such Permitted Encumbrance or (z) customary restrictions and conditions contained in any agreement relating to the sale of any property permitted under <u>Section 7.1(b)</u> pending the consummation of such sale.

VII.12 [Reserved].

-

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

# VII.13 Fiscal Year and Accounting Changes

. (a) Change its Fiscal Year-end employed by the Credit Parties as of the Closing Date other than in the case of any Subsidiary, to cause such Subsidiary's Fiscal Year to end in the same date as the Fiscal Year of DDH or (b) make any significant change in accounting treatment and reporting practices except as required or permitted by GAAP.

VII.14 [Reserved].

÷

## VII.15 Amendment of Governing Documents

. Amend or permit any amendments to its Governing Documents (x) to amend or change its legal name without providing not less than twenty (20) days prior written notice to Agent, or (y) if such amendment could reasonably be expected to be materially adverse to Agent or Lenders, it being understood and agreed that (i) the Reorganization Transactions contemplated by a Qualified IPO shall not be deemed to be materially adverse to the Agent or Lenders, and (ii) any amendment to a Credit Party's Governing Documents that would affect the a pledge of the Equity Interests of such Credit Party (or the rights granted to Agent and Lenders in connection with such pledge) shall be deemed materially adverse.

### VII.16 Compliance with ERISA

. (i) (x) Maintain or (y) become obligated to contribute to, any Pension Plan, other than those Pension Plans disclosed on <u>Schedule 5.8(d)</u> as of the Closing Date or any other Pension Plan for which Agent has provided its prior written consent, (ii) engage in any non-exempt "prohibited transaction", as that term is defined in Section 406 of ERISA and Section 4975 of the Code, with respect to any "plan" as defined in Section 3(3) of ERISA or Section 4975 of the Code which could reasonably be expected to have a Material Adverse Effect, (iii) fail to satisfy, or permit any ERISA Affiliate to fail to satisfy, the minimum funding standards of the pension funding rules with respect to any Pension Plan, which could reasonably be expected to result in a Material Adverse Effect, (iv) terminate, or permit any ERISA Affiliate to terminate, any Pension Plan where such event could result in any material liability of any Credit Party or the imposition of a lien on the property of any Credit Party pursuant to Section 4068 of ERISA, (v) assume, or permit any ERISA Affiliate to incur, any withdrawal liability to any Multiemployer Plan for which Agent has not provided its prior written consent, (vi) incur, or permit any ERISA Affiliate to incur, any withdrawal liability to any Multiemployer Plan in excess of \$500,000 or (vii) fail to comply with the requirements of ERISA or the Code or other Applicable Laws in respect of any "plan" as defined in Section 3(3) of ERISA or Section 4975 of the Code which could reasonably be expected to have a Material Adverse Effect.

VII.17 Prepayment of Indebtedness

[Different first page setting changed from on in original to off in modified.].

. Make any payment, redemption, defeasance or acquisition for value of, or refund, refinance or exchange of, any Subordinated Debt in a manner prohibited by the applicable Subordination Agreement; provided that for the avoidance of doubt, any payment of Subordinated Debt that would also constitute a Restricted Payment is also subject to <u>Section 7.7</u> hereof.

# VII.18 Anti-Terrorism Laws.

(a) Use the proceeds of any Advance except as permitted by Section 2.8. No Credit Party shall use, and each Credit Party shall ensure that its Subsidiaries and its or their respective directors, officers, employees and agents shall not use, the proceeds of any Advance (i) to refinance any commercial paper, (ii) in any manner that causes or may reasonably be expected to cause any Advance or the application of such proceeds to violate any applicable Sanctions, Regulation T, Regulation U or Regulation X of the Board of Governors of the Federal Reserve System as in effect from time to time or any other regulation thereof or to violate the Exchange Act, (iii) in furtherance of an offer, payment, promise to pay, or authorization of the payment or giving of money, or anything else of value, to any Person in violation of any Anti-Corruption Laws, or (iv) for the purpose of funding, financing or facilitating any activities, business or transaction of or with any Sanctioned Person, or in any Sanctioned Country; and

(b) No Covered Entity, either in its own right or through any third party, will (i) have any of its assets in a Sanctioned Country or in the possession, custody or control of a Sanctioned Person in violation of any Anti-Terrorism Law; (ii) do business in or with, or derive any of its income from investments in or transactions with, any Sanctioned Country or Sanctioned Person in violation of any Anti-Terrorism Law; (iii) engage in any dealings or transactions prohibited by any Anti-Terrorism Law or (iv) use the Advances to fund any operations in, finance any investments or activities in, or, make any payments to, a Sanctioned Country or Sanctioned Person in violation of any Anti-Terrorism Law.

The Credit Parties shall deliver to Agent and/or Lenders, as applicable, any certification or other evidence requested from time to time by Agent and/or any Lender, as applicable in their sole discretion, confirming the Credit Parties' compliance with this <u>Section 7.18</u>.

# VII.19 Membership/Partnership Interests

. Unless consented to by Agent (and all actions, filings and deliveries necessary and desirable in the Permitted Discretion of Agent to perfect Agent's first priority security interest (subject to any Permitted Encumbrances) have been completed to Agent's reasonable satisfaction) and except with respect to Subsidiaries existing on the Closing Date with respect to which Agent has received all original certificates evidencing the Equity Interests of such Subsidiaries, elect to treat or permit any of its Domestic Subsidiaries to (a) treat its limited liability company Equity Interests or partnership interests, as the case may be, as securities as contemplated by the definition of "security" in Section 8-102(15) and by Section 8-103 of Article 8 of Uniform Commercial Code or (b) certificate its limited liability company Equity

[Different first page setting changed from on in original to off in modified.].

Interests or partnership interests, as the case may be. To the extent that any of the operating agreements, limited partnership agreements or other agreements governing any Equity Interests of any Credit Party's Subsidiaries, or any law applicable thereto, contains any restrictions on the transfer of such Equity Interests, each Credit Party who owns such Equity Interests and each Credit Party who has issued such Equity Interests hereby (x) consents to the transfer of such Equity Interests by Agent to any Person in connection with any exercise of Agent's remedies pursuant to the terms of this Agreement or any Other Document and (y) consents to any such transferee being admitted as a member or limited partner of such limited liability company or limited partnership (as applicable) with all of the rights of the Credit Party originally owning such Equity Interests.

### VII.20 Trading with the Enemy Act

. Engage in any business or activity in violation of the Trading with the Enemy Act.

#### VII.21 Management Agreement; Management Fees

. At any time prior to a Qualified IPO, except in connection with the consummation of a Qualified IPO, (i) enter into any other management agreement, consulting agreement, or similar arrangement (other than the Management Agreement) without the prior written consent of Agent; or (ii) pay any Management Fees other than Permitted Management Fee Payments.

#### VII.22 Other Agreements

. Enter into any amendment, waiver or modification of (a) any documents governing or giving rise to any Subordinated Debt (except to the extent permitted in the subordination agreement with respect thereto) or, in each case, any related agreements, in each case that could reasonably be expected to be material and adverse to the interests of Agent or Lenders in respect of this Agreement, the Other Documents or the transactions contemplated hereby or thereby, (b) the ABL Loan Documents to the extent prohibited by the Intercreditor Agreement, or (c) any Material Contract or Acquisition Documents in a manner which could reasonably be expected to have any material adverse effect on the interests of Agent or Lenders in respect of this Agreement, the Other Documents or the transactions contemplated hereby and thereby, unless required by law.

#### VII.23 Canadian Subsidiary

. Permit Orange142 Canada to (1) conduct any business operations or own, hold or lease any assets other than holding a bank account with a de minimis account balance or (ii) have any material liabilities.

# VII.24 ABL Obligations [Reserved.]

- Until an ABL Amendment or an ABL Replacement is in effect, request any loans or advances under the ABL Credit Agreement or permit the outstanding principal balance under the ABL Credit

[Different first page setting changed from on in original to off in modified.].

Agreement to exceed \$0 (other than ordinary course fees due thereunder which may be charged as loans under such facility to the extent any such fees are paid within one (1) Business Day).

# VIII. CONDITIONS PRECEDENT.

# VIII.1 Conditions to Initial Advances

. The agreement of Lenders to make the initial Advances requested to be made on the Closing Date is subject to the satisfaction, or waiver by Agent and the Lenders, immediately prior to or concurrently with the making of such Advances, of the following conditions precedent:

(a) <u>Notes. Agent shall have received the Notes duly executed and delivered by an authorized officer of Borrower, other than the</u> <u>Delayed Draw Term Note</u>;

(b) Filings, Registrations and Recordings. Each document (including any Uniform Commercial Code financing statement, including fixture filings) required by this Agreement, any related agreement or under law or reasonably requested by Agent to be filed, registered or recorded in order to create, in favor of Agent, a perfected security interest in or lien upon the Collateral shall have been properly filed, registered or recorded in each jurisdiction in which the filing, registration or recordation thereof is so required or requested by Agent, and Agent shall have received evidence satisfactory to it (i) of each such filing, registration or recordation and satisfactory evidence of the payment of any necessary fee, tax or expense relating thereto and (ii) that no Liens exist with respect to the Collateral other than Permitted Encumbrances;

(c) <u>Company Proceedings of Credit Parties. Agent shall have received a certificate from the Secretary or Assistant Secretary of each Credit Party, in form and substance satisfactory to Agent: (i) attesting to the resolutions of such Credit Party's Board of Directors or similar governing body authorizing its execution, delivery, and performance of this Agreement and the Other Documents to which it is a party, (ii) authorizing specific officers of such Credit Party to execute the same and (iii) attesting to the incumbency and signatures of such specific officers of such Credit Party;</u>

(d) Certificates. Agent shall have received, with respect to each Credit Party, a copy of the Certificate of Incorporation or Certificate of Formation, as applicable, of such Credit Party, and all amendments thereto, certified by the Secretary of State or other appropriate official of its jurisdiction of incorporation or formation together with copies of the bylaws or operating agreement of such Credit Party and all agreements of such Credit Party's Equity Interest holders certified as accurate and complete by the Secretary or Assistant Secretary of such Credit Party;

(e) <u>Good Standing Certificates. Agent shall have received good standing certificates for each Credit Party, dated a recent date, issued by the Secretary of State or other appropriate official of such Credit Party's jurisdiction of incorporation and each jurisdiction</u>

[Different first page setting changed from on in original to off in modified.].

where the conduct of such Credit Party's business activities or the ownership of its properties necessitates qualification;

(f) Legal Opinion. Agent shall have received the executed legal opinions of (i) counsel to the Borrowers in form and substance reasonably satisfactory to Agent, which shall cover such matters incident to the transactions contemplated by this Agreement, and certain material Other Documents and related agreements as Agent may reasonably require;

(g) No Litigation. (i) No litigation, investigation or proceeding before or by any arbitrator or Governmental Body shall be continuing or threatened against any Credit Party or against the officers or directors of any Credit Party (A) in connection with this Agreement, the Other Documents or any of the transactions contemplated thereby and which, in the reasonable opinion of Agent, is deemed material or (B) which could, as determined by Agent in its Permitted Discretion, reasonably be expected to have a Material Adverse Effect; and (ii) no injunction, writ, restraining order or other order of any nature which could, as determined by Agent in its Permitted Discretion of the Closing Date Transactions shall have been issued by any Governmental Body;

(h) <u>Financial Condition Certificates. Agent shall have received an executed Financial Condition Certificate in the form of</u> Exhibit H and related financial statements of DDH and its Subsidiaries evidencing that the Consolidated Total <u>Net</u>-Leverage Ratio of DDH and its Subsidiaries, calculated on a Pro Forma Basis after giving effect to the Closing Date Transactions is not greater than 2.70 to 1.00;

(i) <u>Fees and Expenses. Agent shall have received all fees payable to Agent and Lenders on or prior to the Closing Date hereunder, including under the Fee Letter and pursuant to Article III, and all costs and expenses required to be paid on the Closing Date pursuant to Section 15.9;</u>

(j) <u>Minimum Cash. Borrower shall, calculated on a pro forma basis after giving effect to the making of the Closing Date Term</u> Loan and the transactions contemplated herein, have unrestricted cash on the balance sheet of not less than \$4,000,000;

(k) <u>Structure. The ownership, capital, corporate, tax, organizational and legal structure of DDH and its Subsidiaries (after giving effect to the Closing Date Transactions) shall be satisfactory to Agent in its reasonable discretion;</u>

(1) <u>Insurance. Agent shall have received in form and substance satisfactory to Agent, certified copies of the Credit Parties'</u> casualty insurance policies, together with lender loss payable endorsements naming Agent as lender loss payee, and certified copies of the Credit Parties' liability insurance and cybersecurity insurance policies, together with endorsements naming Agent as a co-insured;

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

(m) <u>Payment Instructions. Agent shall have received written instructions from Borrowing Agent directing the application of proceeds of the initial Advances made pursuant to this Agreement, including the Funds Flow Statement;</u>

(n) <u>Certain Deposit Accounts. Subject to Section 6.17(b), Agent shall have received, in form and substance satisfactory to Agent, duly executed agreements establishing the Funding Account and the Depository Accounts, together with Control Agreements acceptable to Agent;</u>

(o) <u>Consents. Agent shall have received evidence of receipt of any necessary governmental Consents for the Closing Date</u> <u>Transactions and any other Consents where the failure to obtain such consents could reasonably be expected to have a Material Adverse Effect, as</u> <u>reasonably required by Agent;</u>

(p) Other Documents. Agent shall have received (i) the executed Fee Letter, (ii) the executed Pledge Agreement, (iii) the Intercreditor Agreement, (iv) the Preferred Equity Subordination Agreement, and (v) the other executed Other Documents;

(q) Closing Certificate. Agent shall have received a closing certificate signed by the a Senior Officer of each Credit Party dated as of the Closing Date (which certifications may be included in the Financial Condition Certificate), stating that (i) all representations and warranties set forth in this Agreement and each of the Other Documents are true and correct in all material respects (without duplication of any materiality qualifier contained therein) on and as of the Closing Date as if made on the Closing Date (except to the extent that such representation or warranty expressly relates to an earlier date (in which event such representations and warranties are true and correct in all material respects (without duplication of any materiality qualifier contained therein) as of such earlier date)), (ii) each Credit Party is in compliance in all material respects with all of the terms and provisions set forth in this Agreement and the Other Documents required to be complied with or performed by such Credit Party on or before the Closing Date, (iii) on such date no Event of Default or Default has occurred and is continuing, and (iv) the other conditions set forth in Section 8.1 have been met as of the Closing Date;

(r) <u>Compliance with Laws. Agent shall be reasonably satisfied that the Credit Parties are in compliance in all material respects</u> with all pertinent federal, state, local or territorial regulations, including those with respect to the Federal Occupational Safety and Health Act, the Environmental Protection Act, ERISA and the Trading with the Enemy Act;

- (s) [<u>Reserved</u>].
- (t) [<u>Reserved</u>].

(u) <u>Confirmation of Lien Releases. Agent shall have received a fully executed letter agreement from Silverpeak Credit Partners,</u> <u>LP reasonably satisfactory to it confirming that all Liens upon any of the assets of the Credit Parties or any of their Subsidiaries in favor of Silverpeak Credit Partners, LP have been released prior to or as of the Closing Date;</u>

[Different first page setting changed from on in original to off in modified.].

(v) <u>Closing Date Transactions. (i) Substantially concurrently with the funding of the Closing Date Term Loans, Borrower shall have completed the Closing Date Transactions in accordance with the terms of this Agreement and the Other Documents and (ii) Agent shall have received evidence that all necessary consents, permits and approvals (governmental or otherwise) required for the execution, delivery and performance by each Credit Party of the Other Documents have been duly obtained and are in full force and effect;</u>

(w) <u>ABL Obligations. The outstanding principal balance of the ABL Obligations, calculated on a pro forma basis after giving effect to the making of the Closing Date Term Loan and application of the proceeds thereof, shall be \$0.</u>

(x) <u>Background Checks. Agent and each Lender shall have received, in form and substance acceptable to Agent, all</u> documentation and other information requested in connection with applicable "know your customer" and anti-money laundering rules and regulations, including the USA Patriot Act;

(y) Quality of Earnings. Agent shall have received and approved the Quality of Earnings Report, in form and substance reasonably acceptable to Agent, evidencing trailing twelve month adjusted EBITDA (as defined therein) of DDH and its Subsidiaries of not less than \$6,920,000 for the trailing twelve month period ended October 31, 2021; and

(z) <u>Other. All corporate and other proceedings, and all documents, instruments and other legal matters in connection with the</u> <u>Closing Date Transactions shall be in form and substance satisfactory to Agent (and its counsel) in Agent's Permitted Discretion.</u>

VIII.2 Conditions to Delayed Draw Term Loans

. The making of the Fourth Amendment Delayed Draw Term Loan is subject to the satisfaction, or waiver by Agent, immediately prior to or concurrently with the making of such Fourth Amendment Delayed Draw Term Loan, of the following conditions precedent:

(a) <u>No Event of Default or Default shall have occurred and be continuing on such date, or would exist after giving effect to the Advances requested to be made, on such date;</u>

(b) <u>After giving effect to the making of any Delayed Draw Term Loan, DDH Holdings shall, on a Pro Forma Basis, have a</u> <u>Consolidated Total Net-Leverage Ratio of not greater than 2.70 to 1.00;</u>

(c) Each of the representations and warranties made by the Credit Parties in or pursuant to this Agreement, the Other Documents and any related agreements to which it is a party, and each of the representations and warranties contained in any certificate, document or financial or other statement furnished at any time under or in connection with this Agreement, the Other Documents or any related agreement shall be true and correct in all material respects (or in all respects with respect to any representation or warranty which, by its terms is limited as to materiality) on and as of such date as if made on and as of such date, except to the extent such

[Different first page setting changed from on in original to off in modified.].

representation or warranty was expressly made as of an earlier date, in which case, such representation and warranty was true and correct in all material respects (or in all respects with respect to any representation or warranty which, by its terms is limited as to materiality) on and as of such earlier date;

(d) <u>Agent shall have received a Delayed Draw Term Loan Note duly executed and delivered by an authorized officer of each</u> Borrower;

(e) <u>Since the Closing Date, no Material Adverse Effect shall have occurred;</u>

(f) <u>Agent and Lenders funding such Fourth Amendment Delayed Draw Term Loan shall have received all fees payable to</u> <u>Agent and Lenders on or prior to the applicable Delayed Draw Term Loan Funding Date hereunder, including under the Fee Letter, and all costs</u> <u>and expenses required to be paid pursuant to Section 15.9; and</u>

(g) Agent shall have received a closing certificate signed by the a Senior Officer of each Credit Party dated as of the applicable Delayed Draw Term Loan Funding Date, stating that (i) all representations and warranties set forth in this Agreement and the Other Documents are true and correct in all material respects (without duplication of any materiality qualifier contained therein) on and as of such date, except to the extent such representation or warranty was expressly made as of an earlier date, in which case, such representation and warranty was true and correct in all material respects on and as of such earlier date, (ii) each Credit Party is on such date in compliance in all material respects (without duplication of any materiality qualifier contained therein) with all the terms and provisions set forth in this Agreement and the Other Documents, (iii) after giving effect to such Fourth Amendment Delayed Draw Term Loan (and any transaction consummated in connection therewith) on a Pro Forma Basis (but without "netting" the cash proceeds of such Delayed Draw Term Loans funded on such date), the Credit Parties shall be in compliance with the financial covenants set forth in Section 6.5, and (iv) on such date no Default or Event of Default has occurred or is continuing.

# IX. INFORMATION AS TO CREDIT PARTIES.

The Credit Parties shall, until satisfaction in full of the Obligations (other than Inchoate Obligations), the termination of all obligations of Agent and the Lenders to make any Advances hereunder and termination of this Agreement:

# IX.1 Information Regarding Credit Parties

. Promptly upon learning thereof, report to Agent (a) all matters materially and adversely affecting the value, enforceability or collectability of the Collateral, taken as a whole, and (b) any investigation, hearing, proceeding or other inquest by any Governmental Body into any Credit Party or any Affiliate of any Credit Party with respect to Anti-Terrorism Laws; provided, that, to the extent such investigation, hearing, proceeding or other inquest constitutes, as determined by Borrower, material non-public information, no Credit Party shall be required to report any such information to Agent until such time as Agent delivers to the Credit Parties a non-disclosure

[Different first page setting changed from on in original to off in modified.].

agreement that is in form and substance satisfactory to the Credit Parties in their reasonable discretion; provided further that the Credit Parties shall provide Agent prompt notice that such investigation, hearing, proceeding or inquest exists.

IX.2 [<u>Reserved</u>].

IX.3 [Reserved].

÷

# IX.4 Litigation, etc

. Notify Agent promptly in writing of any claim, litigation, suit or administrative proceeding affecting a Credit Party, whether or not the claim is covered by insurance, and of any litigation, suit or administrative proceeding, which in any such case affects a material portion of Collateral, which involves claims for money damages in excess of \$250,000 or which could reasonably be expected to have a Material Adverse Effect.

# IX.5 Material Occurrences

. Deliver, or cause to be delivered, to Agent, promptly upon any Authorized Officer of any Credit Party obtaining knowledge (a) of any condition or event that constitutes a Default or an Event of Default or that notice has been given to any Credit Party with respect there, (b) that any Person has given notice to any Credit Party or any of its Subsidiaries or taken any other action with respect to any event or condition set forth in <u>Section 10.11</u>, (c) of the occurrence of any material violation of, or material non-compliance with, any material requirement of Applicable Laws by any Credit Party, or any of its Subsidiaries, including a description of such violation or non-compliance, or (d) of the occurrence of any Material Adverse Effect (or any other development in the business or affairs of any Credit Party which could reasonably be expected to have a Material Adverse Effect), a certificate of its Authorized Officers specifying the nature and period of existence of such condition, event or change, or specifying the notice given and action taken by any such Person and the nature of such claimed Event of Default, Default, event or condition or change (including, without limitation, with particularity any and all provisions of this Agreement and any Other Document that have been breached, as applicable), and what action the Credit Parties have taken, are taking and propose to take with respect thereto.

# IX.6 Management Discussion and Analysis

. Deliver, or cause to be delivered, to Agent within forty-five (45) days following the end of each Fiscal Quarter of DDH, written discussion and analysis of the prior Fiscal Quarter prepared by the management of DDH in connection with financial statements delivered pursuant to Section 9.8.

[Different first page setting changed from on in original to off in modified.].

#### IX.7 Annual Financial Statements

. Deliver, or cause to be delivered, to Agent within ninety (90) days after the end of each Fiscal Year of DDH, the balance sheets of DDH and its Subsidiaries (or DDH Holdings and its Subsidiaries after a Qualified IPO), prepared on a Consolidated Basis, as of the end of such Fiscal Year and the related statements of income, equity and cash flows of DDH and its Subsidiaries (or DDH Holdings and its Subsidiaries after a Qualified IPO), prepared on a Consolidated Basis, for such Fiscal Year, setting forth in each case after the first anniversary of the Closing Date in comparative form the corresponding figures for the previous Fiscal Year, in reasonable detail and consistent in all material respects with the manner of presentation as of the Closing Date, together with a Financial Officer Certification with respect thereto; and with respect to such financial statements beginning with Borrower's Fiscal Year ending on or about December 31, 2021, be audited by an independent certified public accountants of recognized national or regional standing selected by DDH and reasonably acceptable to Agent (the "Accountants"), which report shall be unqualified as to going concern and scope of audit (other as to qualifications pertaining to periods not audited by said auditor); and, in each case, shall state that such financial statements fairly present, in all material respects, the financial position, on a Consolidated Basis, of DDH and its Subsidiaries (or DDH Holdings and its Subsidiaries after a Qualified IPO) as of the dates indicated and the results of their operations and their cash flows for the periods indicated in conformity with GAAP applied on a basis consistent with prior years (except as otherwise disclosed in such financial statements) and that the examination by such accountants in connection with such financial statements has been made in accordance with generally accepted auditing standards.

### IX.8 Quarterly Financial Statements

. Deliver, or cause to be delivered, to Agent within forty-five (45) days after the end of each Fiscal Quarter of each Fiscal Year, the balance sheets of DDH and its Subsidiaries (or DDH Holdings and its Subsidiaries after a Qualified IPO), prepared on a Consolidated Basis, as of the end of such Fiscal Quarter and the related statements of income, equity and cash flows of DDH and its Subsidiaries (or DDH Holdings and its Subsidiaries after a Qualified IPO), prepared on a Consolidated Basis, for such Fiscal Quarter and for the prior three Fiscal Quarters during the trailing twelve-month calendar period, setting forth in each case after the first anniversary of the Closing Date in comparative form the corresponding figures for the corresponding periods of the previous Fiscal Year, all in reasonable detail and consistent in all material respects with the manner of presentation as of the Closing Date and complete and correct in all material respects, subject to normal and recurring year-end adjustments that individually and in the aggregate are not material to the Credit Parties' business operations, together with a Financial Officer Certification with respect thereto. The reports delivered pursuant to this <u>Section 9.8</u> shall be accompanied by a Compliance Certificate, an Impacta Worker Solutions Certificate, and a report setting forth the Key Performance Indicators for such period; provided, that to the extent that any Key Performance Indicators constitute material non-public information, the Credit Parties shall be under no obligation to deliver such information to Agent or any Lender until Agent and Lenders deliver to the Credit Parties a non-disclosure agreement that is in form and substance

[Different first page setting changed from on in original to off in modified.].

satisfactory to the Credit Parties in their reasonable discretion, containing, among other things, standstill provisions preventing transactions in the Credit Parties' stock.

# IX.9 Monthly Financial Statements

. Deliver, or cause to be delivered, to Agent within thirty (30) days after the end of each calendar month of each Fiscal Year, the balance sheets of DDH and its Subsidiaries (or DDH Holdings and its Subsidiaries after a Qualified IPO), prepared on a Consolidated Basis, as of the end of such calendar month and the related statements of income, equity and cash flows of DDH and its Subsidiaries (or DDH Holdings and its Subsidiaries after a Qualified IPO), prepared on a Consolidated Basis, for such calendar month and for the period from the beginning of the then current Fiscal Year to the end of such calendar month, setting forth in each case (in the case of any such reporting after the first anniversary of the Closing Date) in comparative form the corresponding figures for the corresponding periods of the previous Fiscal Year, all in reasonable detail and consistent in all material respects with the manner of presentation as of the Closing Date and complete and correct in all material respects, subject to normal and recurring year-end adjustments that individually and in the aggregate are not material to the Credit Parties' business operations and the absence of footnotes, together with a Financial Officer Certification with respect thereto. Notwithstanding the foregoing, to the extent any of the foregoing constitutes material non-public information, no Credit Party shall be required to deliver any such financial statements until such time as Agent delivers to the Credit Parties a non-disclosure agreement that is in form and substance satisfactory to the Credit Parties in their reasonable discretion containing, among other things, standstill provisions preventing transactions in the Credit Parties' stock.

#### IX.10 Acquired Intellectual Property

. At any time upon request by Agent, furnish to Agent a report listing all registered copyrights, patents and trademarks (or applications therefor) acquired by any Credit Party during the period specified by Agent.

### IX.11 Additional Information

. Deliver, or cause to be delivered, to Agent (a) such additional information as Agent shall reasonably request in order to enable Agent to determine whether the terms, covenants, provisions and conditions of this Agreement and the Notes have been complied with by the Credit Parties including, without the necessity of any request by Agent, prior written notice of any change (i) in any Credit Party's legal name, (ii) in any Credit Party's Type of legal entity, (iii) in any Credit Party's Federal Taxpayer Identification Number or (iv) in any Credit Party's of any new office or place of business or any Credit Party's closing of any existing office or place of business (which notice, if any, shall be deemed to amend Schedule 4.5), (c) promptly (but in no event later than five days after entry into the same) copies of all amendments, waivers and supplements to any of the Governing

[Different first page setting changed from on in original to off in modified.].

Document or ABL Loan Documents, and (d) such information regarding the operation, business affairs and financial conditions of DDH and its Subsidiaries which Agent may reasonably request.

# IX.12 Projections

. Deliver, or cause to be delivered, to Agent within thirty (30) days after the beginning of each Fiscal Year of DDH, forecasts prepared by management of DDH, on a Consolidated Basis (or for DDH Holdings on a Consolidated Basis subsequent to a Qualified IPO), in form reasonably satisfactory to Agent, of balance sheets and statements of income or operations and cash flows of DDH and its Subsidiaries (or DDH Holdings and its Subsidiaries after a Qualified IPO) on a quarterly basis for the immediately following Fiscal Year (including the Fiscal Year(s) in which the end of the Term occurs)<del>;</del>; provided, that to the extent the foregoing constitutes material non-public information, no Credit Party shall be required to deliver any such financial statements until such time as Agent delivers to the Credit Parties a non-disclosure agreement that is in form and substance satisfactory to the Credit Parties in their reasonable discretion containing, among other things, standstill provisions preventing transactions in the Credit Parties' stock.

# IX.13 Small Business Administration

. So long as any Lender is an SBIC Lender, the Credit Parties shall, and shall cause their Subsidiaries to, within ten (10) days of receipt of request from any SBIC Lender, deliver to such SBIC Lender any information requested by the SBA.

#### IX.14 Notice of Suits, Adverse Events

. Furnish Agent with prompt written notice of (i) any lapse or other termination of any Consent issued to any Credit Party by any Governmental Body or any other Person that is material to the operation of the Credit Parties' business, (ii) any refusal by any Governmental Body or any other Person to renew or extend any such Consent; and (iii) copies of any periodic or special reports filed by any Credit Party with any Governmental Body or Person, if such reports indicate any material change in the business, operations, affairs or condition of any Credit Party, or if copies thereof are requested by Lender, and (iv) copies of any material notices and other communications from any Governmental Body or Person which specifically relate to any Credit Party to the extent any such notices or communications relate to a Material Adverse Effect; provided, that, at any time prior to a Qualified IPO while the Credit Parties are contemplating a Qualified IPO, the foregoing shall not apply to notices and communications with any Governmental Body relating solely to procedural or filing requirements for such Qualified IPO.

# IX.15 ERISA Notices and Requests

. Deliver, or cause to be delivered, to Agent, (i) promptly upon becoming aware of the occurrence of or forthcoming occurrence of any ERISA Event, a written notice specifying the

[Different first page setting changed from on in original to off in modified.].

nature thereof, what action any Credit Party, any of its Subsidiaries or any of their respective ERISA Affiliates has taken, is taking or proposes to take with respect thereto and, when known, any action taken or threatened by the Internal Revenue Service, the Department of Labor, the PBGC, or any other party with respect thereto; (ii) (1) promptly upon reasonable request of Agent, copies of each Schedule B (Actuarial Information) to the annual report (Form 5500 Series) filed by any Credit Party, any of its Subsidiaries or any of their respective ERISA Affiliates with respect to each Pension Plan; and (2) promptly after their receipt, copies of all notices received by any Credit Party, any of its Subsidiaries or any of their respective ERISA Affiliates from a Multiemployer Plan sponsor concerning an ERISA Event; and (iii) such additional information relating to the ERISA Event as Agent may reasonably request. Each notice pursuant to this <u>Section 9.15</u> shall be accompanied by a statement of an Authorized Officer of DDH setting forth details of the occurrence referred to therein and stating what action DDH and/or the other applicable Credit Party has taken and proposes to take with respect thereto.

### IX.16 [Reserved].

#### IX.17 Board Observation Rights

. DH shall notify Agent in writing of the date and time for each meeting of the board of managers or directors, as applicable, which such meetings shall be no less often than quarterly, of each Borrower in accordance with the Governing Documents of such Borrower, and shall deliver to Agent all reports and other materials that are provided to the board of directors; provided that to the extent such reports or materials constitute, as determined by Borrower, material non-public information, no Credit Party shall be required to deliver any such reports or materials until such time as Agent delivers to the Credit Parties a non-disclosure agreement that is in form and substance satisfactory to the Credit Parties in their reasonable discretion containing, among other things, standstill provisions preventing transactions in the Credit Parties' stock. Subject to the execution by Agent of a board observer agreement in form and substance reasonably acceptable to the Borrowers, Agent shall have the right to have up to two (2) designated representatives, at the Borrowers' expense (including, reimbursement of his or her reasonable out-of-pocket expenses incurred in connection with such designation), and subject to reasonable and customary confidentiality obligations, attend such meetings of the board of directors as an observer. Notwithstanding the foregoing, any such observer may be excluded from any such meeting, and such observer and Agent shall not be required to be delivered or receive any such information and materials (or the relevant portions thereof), if DDH or the applicable Borrower reasonably determines that such exclusion or reservation of delivery is necessary, based on advice of counsel, to avoid an actual conflict of interest, including with respect to any indebtedness of DDH or any of its Subsidiaries held by the applicable Agent or any other Lender, or to protect attorney-client privilege.

[Different first page setting changed from on in original to off in modified.].

# X. EVENTS OF DEFAULT.

The occurrence of any one or more of the following events shall constitute an "Event of Default":

# X.1 <u>Nonpayment</u>

. Failure by any Credit Party to pay (i) any principal Obligations when due; or (ii) within three (3) Business Days of when due, any interest on the Obligations or any fee or any other amount due hereunder or in any Other Document, in each case, whether at maturity, by reason of acceleration pursuant to the terms of this Agreement, by notice of intention to prepay, or by required prepayment;

### X.2 Breach of Representation

. Any representation, warranty, certification or other statement made or deemed made by any Credit Party in this Agreement, any Other Document or in any statement or certificate at any time given by any Credit Party or any of its Subsidiaries in writing pursuant hereto or thereto or in connection herewith or therewith shall be false in any material respect (or in any respect with respect to any representation or warranty which, by its terms is limited as to materiality) as of the date made or deemed made;

# X.3 <u>Noncompliance</u>

. Except as otherwise provided for in <u>Section 10.1</u>, failure or neglect of any Credit Party, or any other applicable Person, to perform, keep or observe any term, provision, condition, covenant contained in:

(a) <u>Sections 2.8, 4.9, 4.10, 4.11, 4.19(c), 6.2, 6.5, 6.12(b), 6.14, 6.15, 6.16 (other than 6.16(b)), Article VII, 9.5, 9.6, 9.7, 9.8, 9.9, 9.12 and 9.17, hereof;</u>

(c) <u>this Agreement or any of the Other Documents, other than any such term referred to in any other clause of this Section 10.3</u> and except as provided in Section 4.21, and such default shall not have been remedied or waived within thirty (30) days after the earlier of (i) an Authorized Officer of any Credit Party becoming aware of such default, or (ii) receipt by the Credit Parties of written notice from Agent or any Lender of such default;

X.4 Reserved.

# X.5 Judgments and Attachments

. (i) Any one or more non-appealable money judgments, writs or warrants of attachment or similar process involving an aggregate amount at any time in excess of \$250,000 (to the extent

[Different first page setting changed from on in original to off in modified.].

159823.01008/130146990v.3 159823.01008/150720893v.5

<sup>(</sup>b) Sections 6.3, 6.6, 9.1, 9.11, 9.14, and 9.15 hereof which is not cured within ten (10) from the occurrence of such failure or neglect; or

not adequately covered by insurance as to which a solvent and unaffiliated insurance company has acknowledged coverage) shall be entered or filed against any Credit Party or any of its Subsidiaries or any of their respective assets and shall remain undischarged, unvacated, unbonded or unstayed for a period of sixty (60) days or enforcement proceedings shall have been commenced by a creditor upon such judgment, writ, warrant or process; or (ii) any non-appealable non-monetary judgment or order shall be rendered against any Credit Party or any of its Subsidiaries that could reasonably be expected to have a Material Adverse Effect, and shall remain undischarged, unvacated, unbonded or unstayed for a period of sixty (60) days or enforcement proceedings shall have been commenced by a creditor upon such judgment, writ, warrant or process;

# X.6 Criminal Proceeding

. The institution in any court of a criminal proceeding against any Credit Party or officer or director of any Credit Party which would have a Material Adverse Effect, or the indictment of any Credit Party for any misdemeanors resulting from fraudulent conduct or any felony, and in any such case the proceeding is not stayed or dismissed, or the indictment is not withdrawn, within thirty (30) days of such institution or indictment;

# X.7 Bankruptcy

. (a) (i) A court of competent jurisdiction shall enter a decree or order for relief in respect of any Credit Party or any Subsidiary of a Credit Party in an involuntary case under the Bankruptcy Code or Debtor Relief Laws now or hereafter in effect, which decree or order is not stayed; or any other similar relief shall be granted under any applicable federal or state law; or (ii) (w) an involuntary case shall be commenced against any Credit Party or any Subsidiary of a Credit Party under the Bankruptcy Code or other Debtor Relief Laws now or hereafter in effect: or (x) a decree or order of a court having jurisdiction in the premises for the appointment of a receiver, liquidator, sequestrator, trustee, custodian or other officer having similar powers over any Credit Party or any Subsidiary of a Credit Party, or over all or a substantial part of its property, shall have been entered; or (y) there shall have occurred the involuntary appointment of an interim receiver, trustee or other custodian of any Credit Party or any Subsidiary of a Credit Party for all or a substantial part of its property; or (z) a warrant of attachment, execution or similar process shall have been issued against any substantial part of the property of any Credit Party or any Subsidiary of a Credit Party, and any such event described in this clause (ii) shall continue for ninety (90) days without having been dismissed, bonded or discharged; or (b) (i) any Credit Party or any Subsidiary of a Credit Party shall have an order for relief entered with respect to it or shall commence a voluntary case under the Bankruptcy Code or other Debtor Relief Laws now or hereafter in effect, or shall consent to the entry of an order for relief in an involuntary case, or to the conversion of an involuntary case to a voluntary case, under any such law, or shall consent to the appointment of or taking possession by a receiver, trustee or other custodian for all or a substantial part of its property; (ii) or any Credit Party or any Subsidiary of a Credit Party shall make any assignment for the benefit of creditors; or (iii) or the board of directors (or similar governing body) of any Credit Party or any Subsidiary of a Credit Party or any committee

[Different first page setting changed from on in original to off in modified.].

thereof shall adopt any resolution or otherwise authorize any action to approve any of the actions referred to in this <u>Section 10.7</u> or in <u>Section 10.8</u>;

# X.8 Inability to Pay

. (i) Any Credit Party or any Subsidiary of a Credit Party shall be unable, or shall fail generally, or shall admit in writing its inability, to pay its debts as such debts become due; or (ii) any order, judgment or decree shall be entered against any Credit Party or any Subsidiary of a Credit Party decreeing the dissolution or split up of such Credit Party or such Subsidiary and such order shall remain undischarged or unstayed for a period in excess of thirty (30) days;

# X.9 Government Settlement

. Any Credit Party enters into a settlement agreement with any Governmental Body that (a) results in aggregate liability as to any single or related services of transactions, incidents or conditions of \$250,000 or more, or (ii) would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect;

### X.10 Material Adverse Change

. The occurrence of any event or development which could reasonably be expected to have a Material Adverse Effect;

#### X.11 Cross Default to other Indebtedness

. (i) Failure of any Credit Party or any of its Subsidiaries to pay when due any principal of or interest on or any other amount payable in respect of one or more items of Indebtedness (other than Indebtedness referred to in <u>Section 7.8(a)</u>) in an aggregate principal amount of \$250,000 or more (including, without limitation, an Event of Default under the ABL Loan Documents), in each case beyond the grace or cure period, if any, provided therefor; or (ii) breach or default by any Credit Party with respect to any other term of (1) one or more items of Indebtedness in the aggregate principal amounts referred to in <u>clause (i)</u> above, or (2) any loan agreement, mortgage, indenture or other agreement relating to such item(s) of Indebtedness, in each case beyond the grace or cure period, if any, provided therefor, if the effect of such breach or default is to cause, or to permit the holder or holders of that Indebtedness (or a trustee on behalf of such holder or holders), to cause, that Indebtedness to become or be declared due and payable (or subject to a compulsory repurchase or redeemable) prior to its stated maturity or the stated maturity of any underlying obligation, as the case may be (regardless of whether the holder of such Indebtedness shall actually accelerate, terminate or otherwise exercise any rights and remedies with respect to such Indebtedness);

# X.12 Breach of Guarantee

. Termination or breach of any Guarantee or Guarantor Security Agreement or similar agreement executed and delivered to Agent in connection with the Obligations of Borrowers, or

[Different first page setting changed from on in original to off in modified.].

if any Guarantor attempts, in writing or by legal action, to terminate, challenges the validity of, or its liability under, any such Guarantee or Guarantor Security Agreement or similar agreement;

# X.13 Change of Control

. Any Change of Control shall occur;

# X.14 Invalidity

At any time after the execution and delivery thereof, (i) this Agreement or any Other Document ceases to be in full force and effect (other than by reason of a release of Collateral in accordance with the terms hereof or thereof or the satisfaction in full of the Obligations (other than Inchoate Obligations) in accordance with the terms hereof) or shall be declared null and void, or Agent shall not have or shall cease to have a valid and perfected Lien in any Collateral with the priority required by this Agreement and the Other Documents, or (ii) any Credit Party shall contest the validity or enforceability of this Agreement or any Other Document in writing or deny in writing that it has any further liability, including with respect to future advances by Agent or any Lender under this Agreement or under any Other Document to which it is a party;

# X.15 Seizures

. Any material portion of the Collateral shall be seized, subject to garnishment or taken by a Governmental Body, or any Credit Party, or the title and rights of any Credit Party, shall have become the subject matter of claim, litigation, suit, garnishment or other proceeding which might, as determined by Agent in its Permitted Discretion, upon final determination, result in impairment or loss of the security provided by this Agreement or the Other Documents;

# X.16 Operations

. Other than those business interruptions which are customary in Credit Party's industry, the operations of any Credit Party's manufacturing facility are interrupted at any time for more than fifteen (15) consecutive days, unless such Credit Party shall (i) be entitled to receive for such period of interruption, proceeds of business interruption or similar insurance in an amount sufficient to assure that its per diem cash needs during such period is at least equal to its average per diem cash needs for the consecutive three month period immediately preceding the initial date of interruption and (ii) receive such proceeds in the amount described in <u>clause (i)</u> not later than thirty (30) days following the initial date of any such interruption; <u>provided</u>, <u>however</u>, that notwithstanding the provisions of <u>clauses (i)</u> and (<u>ii</u>) of this Section, an Event of Default shall be deemed to have occurred if such Credit Party shall be receiving such proceeds for a period of thirty (30) consecutive days;

#### X.17 Pension Benefit Plans

. There shall occur one or more ERISA Events which individually or in the aggregate results in liability of any Credit Party, any of its Subsidiaries or any of their respective ERISA

[Different first page setting changed from on in original to off in modified.].

Affiliates in excess of \$250,000 during the term hereof and which is not paid by the applicable due date;

# X.18 Anti-Terrorism Laws

. If (a) any representation or warranty contained in (i) <u>Section Error! Reference source not found.5.29</u> hereof or (ii) any corresponding section of any Guarantee is or becomes false or misleading at any time, (b) any Borrower shall fail to comply with its obligations under <u>Section 7.18</u> hereof, or (c) any Guarantor shall fail to comply with its obligations under any section of any Guarantee containing provisions comparable to those set forth in <u>Sections Error! Reference source not found.5.29</u> and <u>7.18</u> hereof;

# X.19 Invalidity of Intercreditor Agreement or Subordination Provisions

. The Intercreditor Agreement or any subordination agreement with respect to any Subordinated Debt shall cease to be in full force and effect, or any Person party to the Intercreditor Agreement or any such subordination agreement, other than Agent or any Lender party to this Agreement, shall breach the provisions thereof or shall contest in any manner the validity, binding nature or enforceability of any such provision or a proceeding shall be commenced by any such Person or any Governmental Body having jurisdiction over such Person, seeking to establish the invalidity or unenforceability thereof.

Notwithstanding the forgoing, and subject to the terms of this paragraph and the immediately following paragraph, any holder of the Equity Interest issued by DDH Holdings may, but shall not be obligated to, cure any Event of Default (a "Financial Covenant Default") arising from a violation of Section 6.5 hereof by making a direct or indirect capital contribution to DDH Holdings or DDH, with DDH Holdings, as applicable, immediately contributing such capital contribution to a Borrower, in the form of cash equity. (x) in an aggregate amount equal to (but not to exceed) the greater of (x) the amount (the "EBITDA Shortfall Amount") that, when added to Consolidated EBITDA on a dollar-for-dollar basis (with respect to Sections 6.5(a), (b), and (d)) and (y) the amount (the "Liquidity Shortfall Amount") that, when applied to reduce the outstanding balance of Revolving Credit Advances (as defined in the ABL Credit Agreement as in effect on the Fifth Amendment Closing Date) as of the last day of the four (4) Fiscal Quarters period on which the applicable covenant under Section 6.5(c) was tested, would have caused Credit Parties to be in full compliance with Sections 6.5(a), (b), (c), and (d), as applicable, for such testing period (each, an "Equity Cure"); provided, that (a) concurrently with the making of such Equity Cure, the Borrowers receive all of the cash proceeds thereof and concurrently therewith uses such proceeds to make a mandatory prepayment in accordance with Section 2.7(c) hereof, (b) Borrowing Agent shall deliver to Agent written notice of its intent to exercise such Equity Cure no later than the earlier of (i) the date of delivery of the Compliance Certificate describing the Financial Covenant Default and (ii) the date on which such Compliance Certificate was required to have been delivered to Agent, (c) such Equity Cure must be effected no later than the earlier of (i) ten (10) days after the date on which such

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

[Different first page setting changed from on in original to off in modified.].

<u>Compliance Certificate was required to have been delivered to Agent), (d) no such rights to effectuate an Equity Cure may be exercised (i) more than two (2) times in any calendar year, (ii) in consecutive Fiscal Quarters, or (iii) more than four (4) times during the Term of this Agreement, and (e) no Equity Cure may be in an amount greater than necessary to cure the Financial Covenant Defaults in accordance with the terms hereof.</u>

Upon the receipt by DDH of each such Equity Cure and the making of the mandatory prepayment set forth in Section 2.7(c), each such financial covenant that is the subject of a Financial Covenant Default shall be recalculated giving effect to the following pro forma adjustments: Consolidated EBITDA shall be increased by an amount equal to the EBITDA Shortfall Amount and/or the outstanding balance of Revolving Credit Advances (as defined in the ABL Credit Agreement as in effect on the Fifth Amendment Closing Date) as of the last day of the four (4). Fiscal Quarters period on which the applicable covenant was tested is reduced by an amount equal to the Liquidity Shortfall Amount, as applicable, solely for the purpose of determining the existence of any Event of Default arising from the failure to comply with Section 6.5(a), (b), (c), and/or (d), as applicable, with respect to the relevant testing period and all future testing periods that include the last Fiscal Quarter of the testing period in respect of which such Equity Cure was made; provided, that (x) no increase in Consolidated EBITDA or Liquidity on account of the exercise of the Cure Right pursuant to this paragraph and the immediately preceding paragraph above shall be applicable for any other purpose under this Agreement or any Other Documents, including but not limited to determining any applicable margin or fee or the availability or amount of any covenant basket, carve-out or compliance on a Pro Forma Basis with Section 6.5(a), (b), (c), or (d). If, after giving effect to the foregoing recalculations, Credit Parties shall then be in compliance with the requirements of Section 6.5 hereof. Credit Parties shall be deemed to have satisfied the requirements of Section 6.5 hereof with the same effect as though there had been no failure to comply therewith, and the Financial Covenant Default that had occurred shall be deemed not to have occurred for purposes of this Agreement and the Other Documents.

# XI. LENDERS' RIGHTS AND REMEDIES AFTER DEFAULT, ETC.

# XI.1 Rights and Remedies.

Ŧ

(a) Upon the occurrence of (i) an Event of Default pursuant to Section 10.7 or 10.8, all Obligations shall be immediately due and payable and this Agreement and the obligation of Lenders to make Advances shall be deemed terminated; and (ii) any of the other Events of Default and at any time during the continuance thereafter, at the option of Agent or at the direction of Required Lenders all Obligations shall be immediately due and payable and Agent or Required Lenders shall have the right to terminate this Agreement and to terminate the obligation of Lenders to make Advances. Upon the occurrence of any Event of Default, Agent shall have the right to exercise any and all rights and remedies provided for herein, under the Other Documents, under the Uniform Commercial Code and at law or equity generally, including

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

the right to foreclose the security interests granted herein and to realize upon any Collateral by any available judicial procedure and/or to take possession of and sell any or all of the Collateral with or without judicial process. Agent may enter any of any Credit Party's premises or other premises without legal process and without incurring liability to any Credit Party therefor, and Agent may thereupon, or at any time thereafter, in its discretion without notice or demand, take the Collateral and remove the same to such place as Agent may deem advisable and Agent may require the Credit Parties to make the Collateral available to Agent at a convenient place. With or without having the Collateral at the time or place of sale, Agent may sell the Collateral, or any part thereof, at public or private sale, at any time or place, in one or more sales, at such price or prices, and upon such terms, either for cash, credit or future delivery, as Agent may elect. Except as to that part of the Collateral which is perishable or threatens to decline speedily in value or is of a type customarily sold on a recognized market, Agent shall give Borrowing Agent reasonable notification of such sale or sales, it being agreed that in all events written notice mailed to Borrowing Agent at least ten (10) days prior to such sale or sales is reasonable notification. At any public sale Agent or any Lender may bid (including credit bid) for and become the purchaser, and Agent, any Lender or any other purchaser at any such sale thereafter shall hold the Collateral sold absolutely free from any claim or right of whatsoever kind, including any equity of redemption and all such claims, rights and equities are hereby expressly waived and released by each Credit Party. In connection with the exercise of the foregoing remedies, including the sale of Inventory, Agent is granted a perpetual nonrevocable, royalty free, nonexclusive license and Agent is granted permission to use all of each Credit Party's (y) trademarks, trade styles, trade names, patents, patent applications, copyrights, service marks, licenses, franchises and other proprietary rights which are used or useful in connection with Inventory for the purpose of marketing, advertising for sale and selling or otherwise disposing of such Inventory and (z) Equipment for the purpose of completing the manufacture of unfinished goods. The cash proceeds realized from the sale of any Collateral shall be applied to the Obligations in the order set forth in Section 11.5 hereof. Noncash proceeds will only be applied to the Obligations as they are converted into cash. If any deficiency shall arise, the Credit Parties shall remain liable to Agent and Lenders therefor.

(b) To the extent that Applicable Law imposes duties on Agent to exercise remedies in a commercially reasonable manner, each Credit Party acknowledges and agrees that it is not commercially unreasonable for Agent (i) to fail to incur expenses reasonably deemed significant by Agent to prepare Collateral for disposition or otherwise to complete raw material or work in process into finished goods or other finished products for disposition, (ii) to fail to obtain third party consents for access to Collateral to be disposed of, or to obtain or, if not required by other law, to fail to obtain governmental or third party consents for the collection or disposition of Collateral to be collected or disposed of, (iii) to fail to exercise collection remedies against Customers or other Persons obligated on Collateral or to remove Liens on or any adverse claims against Collateral, (iv) to exercise collection remedies against Customers and other Persons obligated on Collateral directly or through the use of collection agencies and other collection specialists, (v) to advertise dispositions of Collateral through publications or media of general circulation, whether or not the Collateral is of a specialized nature, (vi) to contact other Persons, whether or not in the same business as the Credit Parties (or any of them), for

[Different first page setting changed from on in original to off in modified.].

159823.01008/130146990v.3 159823.01008/150720893v.5

expressions of interest in acquiring all or any portion of such Collateral, (vii) to hire one or more professional auctioneers to assist in the disposition of Collateral, whether or not the Collateral is of a specialized nature, (viii) to dispose of Collateral by utilizing internet sites that provide for the auction of assets of the types included in the Collateral or that have the reasonable capacity of doing so, or that match buyers and sellers of assets, (ix) to dispose of assets in wholesale rather than retail markets, (x) to disclaim disposition warranties, such as title, possession or quiet enjoyment, (xi) to purchase insurance or credit enhancements to insure Agent against risks of loss, collection or disposition of Collateral or to provide to Agent a guaranteed return from the collection or disposition of Collateral, or (xii) to the extent deemed appropriate by Agent, to obtain the services of other brokers, investment bankers, consultants and other professionals to assist Agent in the collection or disposition of any of the Collateral. Each Credit Party acknowledges that the purpose of this Section 11.1(b) is to provide non-exhaustive indications of what actions or omissions by Agent shall not be deemed commercially unreasonable solely on account of not being indicated in this Section 11.1(b). Without limitation upon the foregoing, nothing contained in this Section 11.1(b) shall be construed to grant any rights to any Credit Party or to impose any duties on Agent that would not have been granted or imposed by this Agreement or by Applicable Law in the absence of this Section 11.1(b).

# (c) <u>Without limiting any other provisions hereof:</u>

(i) (i) At any bona fide public sale, and to the extent permitted by Applicable Law, at any private sale, Agent shall be free to purchase all or any part of the Investment Property. Any such sale may be on cash or credit. Agent shall be authorized at any such sale (if it deems it advisable to do so) to restrict the prospective bidders or purchasers to persons who will represent and agree that they are purchasing the Investment Property for their own account in compliance with Regulation D of the Securities Act or any other applicable exemption available under the Securities Act. Agent will not be obligated to make any sale if it determines not to do so, regardless of the fact that notice of the sale may have been given. Agent may adjourn any sale and sell at the time and place to which the sale is adjourned. If the Investment Property is customarily sold on a recognized market or threatens to decline speedily in value, Agent may sell such Investment Property at any time without giving prior notice to any Credit Party or other Person.

(ii) (iii)-Each Credit Party recognizes that Agent may be unable to effect or cause to be effected a public sale of the Investment Property by reason of certain prohibitions of the Securities Act, so that Agent may be compelled to resort to one or more private sales to a restricted group of purchasers who will be obligated to agree, among other things, to acquire the Investment Property for their own account, for investment and without a view to the distribution or resale thereof. Each Credit Party understands that private sales so made may be at prices and on other terms less favorable to the seller than if the Investment Property were sold at public sales, and agrees that Agent has no obligation to delay or agree to delay the sale of any of the Investment Property for the period of time necessary to permit the

[Different first page setting changed from on in original to off in modified.].

issuer of the securities which are part of the Investment Property (even if the issuer would agree), to register such securities for sale under the Securities Act. Each Credit Party agrees that private sales made under the foregoing circumstances shall be deemed to have been made in a commercially reasonable manner.

(iii) (iii) The net cash proceeds arising from the disposition of the Investment Property after deducting expenses incurred by Agent will be applied to the Obligations pursuant to Section 11.5 hereof. If any excess remains after the discharge of all of the Obligations, the same will be paid to the applicable Credit Party or to any other Person that may be legally entitled thereto.

(iv) (iv) At any time after the occurrence and during the continuance of an Event of Default (A) Agent may transfer any or all of the Investment Property into its name or that of its nominee and, after providing the Borrowers with one (1) Business Days prior written notice, may exercise all voting rights with respect to the Investment Property, but no such transfer shall constitute a taking of such Investment Property in satisfaction of any or all of the Obligations, and (B) Agent shall be entitled to receive, for application to the Obligations, all cash or stock dividends and distributions, interest and premiums declared or paid on the Investment Property.

# XI.2 Agent's Discretion

. Agent shall have the right in its sole discretion to determine which rights, Liens, security interests or remedies Agent may at any time pursue, relinquish, subordinate, or modify, which procedures, timing and methodologies to employ, and what any other action to take with respect to any or all of the Collateral and in what order, thereto and such determination will not in any way modify or affect any of Agent's or Lenders' rights hereunder as against the Credit Parties or each other.

# XI.3 Setoff

. Subject to <u>Section 14.12</u>, in addition to any other rights which Agent or any Lender may have under Applicable Law, upon the occurrence of an Event of Default hereunder, Agent and such Lender shall have a right, immediately and without notice of any kind, to apply any Credit Party's property held by Agent and such Lender or any of their Affiliates to reduce the Obligations and to exercise any and all rights of setoff which may be available to Agent and such Lender with respect to any deposits held by Agent or such Lender.

# XI.4 <u>Rights and Remedies not Exclusive</u>

. The enumeration of the foregoing rights and remedies is not intended to be exhaustive and the exercise of any rights or remedy shall not preclude the exercise of any other right or remedies provided for herein or otherwise provided by law, all of which shall be cumulative and not alternative.

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

# XI.5 Allocation of Payments After Event of Default

. Notwithstanding any other provisions of this Agreement to the contrary, after the occurrence and during the continuance of an Event of Default, all amounts collected or received by Agent on account of the Obligations or any other amounts outstanding under any of the Other Documents or in respect of the Collateral may, at Agent's discretion, and shall, at the direction of the Required Lenders, be paid over or delivered as follows:

FIRST, to the payment of all reasonable costs and expenses (including reasonable attorneys' fees and allocated costs of internal counsel) of Agent in connection with enforcing its rights and the rights of the Lenders under this Agreement and the Other Documents and any protective advances made by Agent with respect to the Collateral under or pursuant to the terms of this Agreement;

SECOND, to payment of any fees owed to Agent;

THIRD, to the payment of all reasonable out-of-pocket costs and expenses (including reasonable documented (to the extent readily available) attorneys' fees and allocated costs of internal counsel) of each of the Lenders to the extent owing to such Lender pursuant to the terms of this Agreement;

FOURTH, to the payment of all of the Obligations consisting of accrued fees and interest;

FIFTH, to the payment of the outstanding principal amounts of Term Loans, on a pro rata basis;

SIXTH, to all other Obligations and other obligations which shall have become due and payable under the Other Documents or otherwise and not repaid pursuant to clauses "FIRST" through "FIFTH" above; and

SEVENTH, to the payment of the surplus, if any, to whoever may be lawfully entitled to receive such surplus.

In carrying out the foregoing, (i) amounts received shall be applied in the numerical order provided until exhausted prior to application to the next succeeding category; and (ii) each of the Lenders shall receive (so long as it is not a Defaulting Lender) an amount equal to its *pro rata* share (based on the proportion that the then outstanding Advances held by such Lender bears to the aggregate then outstanding Advances) of amounts available to be applied pursuant to clauses "FOURTH", "FIFTH," "SIXTH", and "SEVENTH" above.

XII. WAIVERS AND JUDICIAL PROCEEDINGS.

XII.1 Waiver of Notice

[Different first page setting changed from on in original to off in modified.].

. Each Credit Party hereby waives notice of non-payment of any of the Receivables, demand, presentment, protest and notice thereof with respect to any and all instruments, notice of acceptance hereof, notice of loans or advances made, credit extended, Collateral received or delivered, or any other action taken in reliance hereon, and all other demands and notices of any description, except such as are expressly provided for herein.

#### XII.2 Delay

. No delay or omission on Agent's or any Lender's part in exercising any right, remedy or option shall operate as a waiver of such or any other right, remedy or option or of any Default or Event of Default.

# XII.3 Jury Waiver

. TO THE MAXIMUM EXTENT PERMITTED BY APPLICABLE LAW, EACH PARTY TO THIS AGREEMENT HEREBY EXPRESSLY WAIVES ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, COUNTERCLAIM, DEMAND, ACTION OR CAUSE OF ACTION (a) ARISING UNDER THIS AGREEMENT, ANY OTHER DOCUMENT OR ANY OTHER INSTRUMENT, DOCUMENT OR AGREEMENT EXECUTED OR DELIVERED IN CONNECTION HEREWITH, OR (b) IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTIES HERETO OR ANY OF THEM WITH RESPECT TO THIS AGREEMENT, ANY OTHER DOCUMENT OR ANY OTHER INSTRUMENT, DOCUMENT OR AGREEMENT EXECUTED OR DELIVERED IN CONNECTION HEREWITH, OR THE TRANSACTIONS RELATED HERETO OR THERETO IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER SOUNDING IN CONTRACT OR TORT OR OTHERWISE AND EACH PARTY HEREBY CONSENTS THAT ANY SUCH CLAIM, COUNTERCLAIM, DEMAND, ACTION OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY, AND THAT ANY PARTY TO THIS AGREEMENT MAY FILE AN ORIGINAL COUNTERPART OR A COPY OF THIS SECTION WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENTS OF THE PARTIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.

#### XIII. EFFECTIVE DATE AND TERMINATION.

#### XIII.1 Term

. This Agreement, which shall inure to the benefit of and shall be binding upon the respective successors and permitted assigns of the Credit Parties, Agent and each Lender, shall become effective on the date hereof and shall continue in full force and effect until December 3, 2026 (the "<u>Term</u>"), unless sooner terminated as herein provided. Borrowers may terminate this Agreement at any time upon fifteen (15) Business Days' prior written notice upon payment in full of the Obligations (other than Inchoate Obligations).

XIII.2 Termination

[Different first page setting changed from on in original to off in modified.].

. The termination of this Agreement shall not affect any Credit Party's, Agent's or any Lender's rights, or any of the Obligations having their inception prior to the effective date of such termination or any Inchoate Obligations, and the provisions hereof shall continue to be fully operative until (a) all transactions entered into, rights or interests created or Obligations (other than Inchoate Obligations) have been fully paid in immediately available funds and (b) each of the Credit Parties have released Agent and Lenders from and against any and all claims of any nature whatsoever that any Credit Party may have against Agent and Lenders (or any of them). The security interests, Liens and rights granted to Agent and Lenders hereunder and the financing statements filed in connection herewith shall continue in full force and effect, notwithstanding the termination of this Agreement or the fact that Borrowers' Account may from time to time be temporarily in a zero or credit position, until all of the Obligations (other than Inchoate Obligations) have been paid and performed in full in immediately available funds after the termination of all obligations of Agent and Lenders to make any Advances hereunder or Borrowers have furnished Agent and Lenders with an indemnification satisfactory to Agent and Lenders with respect thereto. Accordingly, each Credit Party waives any rights which it may have under the Uniform Commercial Code to demand the filing of termination statements with respect to the Collateral unless and until all of the Obligations (other than Inchoate Obligations) have been paid and performed in full in immediately available funds after the termination of all obligations of Agent and Lenders to make any Advances hereunder, and Agent shall not be required to send such termination statements to any Credit Party, or to file them with any filing office, until such time. When all of the Obligations (other than Inchoate Obligations) have been paid and performed in full in immediately available funds after the termination of all obligations of Agent and Lenders to make any Advances hereunder, Agent will, at Borrowers' sole expense, execute and deliver any termination statements, lien releases, mortgage releases, re-assignments of trademarks, discharges of security interests, and other similar discharge or release documents (and, if applicable, in recordable form) as are reasonably necessary to release, as of record, Agent's Liens and all notices of security interests and liens previously filed by Agent with respect to the Obligations. Notwithstanding anything to the contrary set forth herein, all representations, warranties, waivers and indemnities of the Credit Parties contained herein or in any Other Document shall survive the payment in full of the Obligations and termination of this Agreement and/or such Other Documents.

### XIII.3 Revival and Reinstatement of Obligations

. If the incurrence or payment of the Obligations by any Credit Party or the transfer to Agent or any Lender of any property should for any reason subsequently be declared to be void or voidable under any state or federal law relating to creditors' rights, including provisions of the Bankruptcy Code or any other Debtor Relief Laws relating to fraudulent conveyances, preferences, or other voidable or recoverable payments of money or transfers of property (each, a "<u>Voidable Transfer</u>"), and if Agent or any Lender is required to repay or restore, in whole or in part, any such Voidable Transfer, or elects to do so upon the reasonable advice of its counsel, then, as to any such Voidable Transfer, or the amount thereof that Agent or such Lender is required or elects to repay or restore, and as to all reasonable costs, expenses, and attorneys' fees of Agent and such Lender related thereto, the liability of each Credit Party automatically shall be

[Different first page setting changed from on in original to off in modified.].

revived, reinstated, and restored and shall exist as though such Voidable Transfer had never been made and any Liens held by Agent previously released or terminated with respect to any Collateral shall be reinstated as of the date on which Agent or Lenders (as applicable) repays or restores such Voidable Transfer.

# XIV. <u>REGARDING AGENT.</u>

### XIV.1 Appointment

. Each Lender hereby designates Lafayette Square to act as Agent for such Lender under this Agreement and the Other Documents. Each Lender hereby irrevocably authorizes Agent to take such action on its behalf under the provisions of this Agreement and the Other Documents and to exercise such powers and to perform such duties hereunder and thereunder as are specifically delegated to or required of Agent by the terms hereof and thereof and such other powers as are reasonably incidental thereto and Agent shall hold all Collateral, payments of principal and interest, fees (except the fees set forth in the Fee Letter), charges and collections received pursuant to this Agreement, for the ratable benefit of Lenders. Agent may perform any of its duties hereunder by or through its agents or employees. Notwithstanding anything to the contrary contained in this Agreement, Agent shall act or refrain from acting or exercise any discretion (and shall be fully protected in so acting or refraining from acting) only with the consent of the Required Lenders; provided, however, that Agent shall not be required to take any action which exposes Agent to liability or which is contrary to this Agreement or the Other Documents or Applicable Law unless Agent is furnished with an indemnification reasonably satisfactory to Agent with respect thereto.

# XIV.2 Nature of Duties

. Agent shall have no duties or responsibilities except those expressly set forth in this Agreement and the Other Documents. Neither Agent nor any of its officers, directors, employees or agents shall be (i) liable for any action taken or omitted by them as such hereunder or in connection herewith, unless caused by their gross (not mere) negligence or willful misconduct (as determined by a court of competent jurisdiction in a final non-appealable judgment), or (ii) responsible in any manner for any recitals, statements, representations or warranties made by any Credit Party or any of their respective officers contained in this Agreement, or in any of the Other Documents or in any certificate, report, statement or other document referred to or provided for in, or received by Agent under or in connection with, this Agreement, or any of the Other Documents or for the value, validity, effectiveness, genuineness, due execution, enforceability or sufficiency of this Agreement, or any of the Other Documents or for any failure of any Credit Party to perform its obligations hereunder or under any Other Document. Agent shall not be under any obligation to any Lender to ascertain or to inquire as to the observance or performance of any of the Credit Parties. The duties of Agent as respects the Advances to Borrowers shall be mechanical and administrative in nature; Agent shall not have by reason of this Agreement a fiduciary

[Different first page setting changed from on in original to off in modified.].

relationship in respect of any Lender; and nothing in this Agreement, expressed or implied, is intended to or shall be so construed as to impose upon Agent any obligations in respect of this Agreement or the transactions described herein except as expressly set forth herein.

# XIV.3 Lack of Reliance on Agent and Resignation

. Independently and without reliance upon Agent or any other Lender, each Lender has made and shall continue to make (a) its own independent investigation of the financial condition and affairs of each Credit Party in connection with the making and the continuance of the Advances hereunder and the taking or not taking of any action in connection herewith, and (b) its own appraisal of the creditworthiness of each Credit Party. Agent shall have no duty or responsibility, either initially or on a continuing basis, to provide any Lender with any credit or other information with respect thereto, whether coming into its possession before making of the Advances or at any time or times thereafter except as shall be provided by the Credit Parties pursuant to the terms hereof. Agent shall not be responsible to any Lender for any recitals, statements, information, representations or warranties herein or in any agreement, document, certificate or a statement delivered in connection with or for the execution, effectiveness, genuineness, validity, enforceability, collectability or sufficiency of this Agreement or any Other Document, or of the financial condition of any Credit Party, or be required to make any inquiry concerning either the performance or observance of any of the terms, provisions or conditions of this Agreement, the Note, the Other Documents or the financial condition of any Credit Party, or the existence of any Event of Default or any Default.

Required Lenders may remove Agent on sixty (60) days' written notice to each of Agent, Lenders and Borrowing Agent and Agent may resign on sixty (60) days' written notice to each of Lenders and Borrowing Agent, and upon such removal or resignation, the Required Lenders will promptly designate a successor Agent reasonably satisfactory to Borrowing Agent (provided that no such approval by Borrowing Agent shall be required (i) in any case where the successor Agent is one of the Lenders or (ii) after the occurrence and during the continuance of any Event of Default). Any such successor Agent shall succeed to the rights, powers and duties of Agent, and shall in particular succeed to all of Agent's right, title and interest in and to all of the Liens in the Collateral securing the Obligations created hereunder or any Other Document, and the term "Agent" shall mean such successor agent effective upon its appointment, and the former Agent's rights, powers and duties as Agent shall be terminated, without any other or further act or deed on the part of such former Agent. However, notwithstanding the foregoing, if at the time of the effectiveness of the new Agent's appointment, any further actions need to be taken in order to provide for the legally binding and valid transfer of any Liens in the Collateral from former Agent to new Agent and/or for the perfection of any Liens in the Collateral as held by new Agent or it is otherwise not then possible for new Agent to become the holder of a fully valid, enforceable and perfected Lien as to any of the Collateral, former Agent shall continue to hold such Liens solely as agent for perfection of such Liens on Behalf of new Agent until such time as new Agent can obtain a fully valid, enforceable and perfected Lien on all Collateral, provided that Agent shall not be required to or have any liability or responsibility to take any further actions after such date as such agent for perfection to continue the perfection of any such Liens

[Different first page setting changed from on in original to off in modified.].

(other than to forego from taking any affirmative action to release any such Liens). After Agent's resignation as Agent, the provisions of this <u>Article XIV</u>, and any indemnification rights under this Agreement, including without limitation, rights arising under <u>Section 15.5</u> hereof, shall inure to its benefit as to any actions taken or omitted to be taken by it while it was Agent under this Agreement (and in the event resigning Agent continues to hold any Liens pursuant to the provisions of the immediately preceding sentence, the provisions of this <u>Article XIV</u> and any indemnification rights under this Agreement, including without limitation, rights arising under <u>Section 15.5</u> hereof, shall inure to its benefit as to any actions taken or omitted to be taken by it in connection with such Liens).

# XIV.4 Certain Rights of Agent

. If Agent shall request instructions from Lenders with respect to any act or action (including failure to act) in connection with this Agreement or any Other Document, Agent shall be entitled to refrain from such act or taking such action unless and until Agent shall have received instructions from the Required Lenders; and Agent shall not incur liability to any Person by reason of so refraining. Without limiting the foregoing, Lenders shall not have any right of action whatsoever against Agent as a result of its acting or refraining from acting hereunder in accordance with the instructions of the Required Lenders.

#### XIV.5 Reliance

. Agent shall be entitled to rely, and shall be fully protected in relying, upon any note, writing, resolution, notice, statement, certificate, email, telecopier or facsimile message, order or other document or telephone message believed by it to be genuine and correct and to have been signed, sent or made by the proper Person, and, with respect to all legal matters pertaining to this Agreement and the Other Documents and its duties hereunder, upon advice of counsel selected by it. Agent may employ agents and attorneys-in-fact and shall not be liable for the default or misconduct of any such agents or attorneys-in-fact selected by Agent with reasonable care.

#### XIV.6 Notice of Default

. Agent shall not be deemed to have knowledge or notice of the occurrence of any Default or Event of Default hereunder or under the Other Documents, unless Agent has received notice from a Lender or Borrowing Agent referring to this Agreement or the Other Documents, describing such Default or Event of Default and stating that such notice is a "notice of default". In the event that Agent receives such a notice, Agent shall give notice thereof to Lenders. Agent shall take such action with respect to such Default or Event of Default as shall be reasonably directed by the Required Lenders; <u>provided</u>, <u>that</u>, unless and until Agent shall have received such directions, Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Default or Event of Default as it shall deem advisable in the best interests of Lenders.

# XIV.7 Indemnification

[Different first page setting changed from on in original to off in modified.].

. To the extent Agent is not reimbursed and indemnified by Borrowers, each Lender will reimburse and indemnify Agent in proportion to its respective portion of the Advances (or, if no Advances are outstanding, according to its applicable Commitment Percentage), from and against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind or nature whatsoever which may be imposed on, incurred by or asserted against Agent in performing its duties hereunder, or in any way relating to or arising out of this Agreement or any Other Document; provided that, Lenders shall not be liable for any portion of such liabilities, obligations, losses, damages, penalties, actions, judgments resulting from Agent's gross (not mere) negligence or willful misconduct (as determined by a court of competent jurisdiction in a final non-appealable judgment).

# XIV.8 Agent in its Individual Capacity

. With respect to the obligation of Agent to lend under this Agreement, the Advances made by it shall have the same rights and powers hereunder as any other Lender and as if it were not performing the duties as Agent specified herein; and the term "Lender" or any similar term shall, unless the context clearly otherwise indicates, include Agent in its individual capacity as a Lender. Agent may engage in business with any Credit Party as if it were not performing the duties specified herein, and may accept fees and other consideration from any Credit Party for services in connection with this Agreement or otherwise without having to account for the same to Lenders.

# XIV.9 Delivery of Documents

. To the extent Agent receives financial statements or notifications required under <u>Sections 9.7</u>, <u>9.8</u>, <u>9.9</u> and <u>9.12</u> pursuant to the terms of this Agreement which the applicable Credit Party is not obligated to deliver to each Lender, Agent will promptly furnish such documents and information to Lenders.

# XIV.10 Borrowers' Undertaking to Agent

. Without prejudice to its obligations to Lenders under the other provisions of this Agreement, Borrowers hereby undertake with Agent to pay to Agent from time to time on demand all amounts from time to time due and payable by it for the account of Agent or Lenders or any of them pursuant to this Agreement to the extent not already paid. Any payment made pursuant to any such demand shall *pro tanto* satisfy the relevant Borrower's obligations to make payments for the account of Lenders or the relevant one or more of them pursuant to this Agreement.

# XIV.11 No Reliance on Agent's Customer Identification Program

. Each Lender acknowledges and agrees that neither such Lender, nor any of its Affiliates, participants or assignees, may rely on Agent to carry out such Lender's, Affiliate's, participant's or assignee's customer identification program, or other obligations required or

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

imposed under or pursuant to the USA PATRIOT Act or the regulations thereunder, including the regulations contained in 31 CFR 103.121 (as hereafter amended, modified, supplemented or replaced, the "<u>CIP Regulations</u>"), or any other Anti-Terrorism Law, including any programs involving any of the following items relating to or in connection with any Credit Party, their Affiliates or agents, this Agreement, the Other Documents or the transactions hereunder or contemplated hereby: (1) any identity verification procedures, (2) any record-keeping, (3) comparisons with government lists, (4) customer notices or (5) other procedures required under the CIP Regulations or such other laws.

# XIV.12 Other Agreements

. Each of the Lenders agrees that it shall not, without the prior written consent of Agent, and that it shall, to the extent it is lawfully entitled to do so, upon the request of Agent, set off against the Obligations, any amounts owing by such Lender to any Credit Party or any deposit accounts of any Credit Party now or hereafter maintained with such Lender. Anything in this Agreement to the contrary notwithstanding, each of the Lenders further agrees that it shall not, unless specifically requested to do so by Agent, take any action to protect or enforce its rights arising out of this Agreement or the Other Documents, it being the intent of Lenders that any such action to protect or enforce rights under this Agreement and the Other Documents shall be taken in concert and at the direction or with the consent of Agent or Required Lenders.

XIV.13 Erroneous Payment.

÷

(a) Each Lender hereby agrees that (i) if Agent (x) notifies a Lender or any Person who has received funds on behalf of a Lender (any such Lender or other recipient, a "Payment Recipient") that Agent has determined in its sole discretion (whether or not after receipt of any notice under immediately succeeding clause (b)) that any funds (as set forth in such notice from Agent) received by such Payment Recipient from Agent or any of its Affiliates were erroneously or mistakenly transmitted to, or otherwise erroneously or mistakenly received by, such Payment Recipient (whether or not known to such Lender or other such Payment Recipient on its behalf) (any such funds, whether transmitted or received as a payment, prepayment or repayment of principal, interest, fees or otherwise, individually and collectively, an "Erroneous Payment") and (y) demands in writing the return of such Erroneous Payment (or a portion thereof), such Payment Recipient shall promptly, but in no event later than one (1) Business Day thereafter (or such later date as Agent may, in its sole discretion, specify in writing), return to Agent the amount of any such Erroneous Payment (or portion thereof) as to which such a demand was made, in same day funds (in the currency so received), together with interest thereon (except to the extent waived in writing by Agent) in respect of each day from and including the date such Erroneous Payment (or portion thereof) was received by such Lender to the date such amount is repaid to Agent in same day funds at the greater of the overnight bank funding rate and a rate determined by Agent in accordance with banking industry rules on interbank compensation from time to time in effect, and (ii) such Lender shall not assert any right

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

or claim to the Erroneous Payment, and hereby waives any claim, counterclaim, defense or right of set-off or recoupment with respect to any demand, claim or counterclaim by Agent for the return of any Erroneous Payments received, including without limitation waiver of any defense based on "discharge for value" or any similar doctrine. A notice of Agent to any Lender under this subsection (a) shall be conclusive, absent manifest error.

(b) Without limiting subsection (a) above, each Lender or other Payment Recipient hereby further agrees that if it receives an Erroneous Payment from Agent (or any of its Affiliates) (i) that is in an amount different than (other than a de minimis difference), or on a different date from, that specified in this Agreement or in a notice of payment sent by Agent (or any of its Affiliates) with respect to such Erroneous Payment (an "Erroneous Payment Notice"), or (ii) that was not preceded or accompanied by an Erroneous Payment Notice, it shall be on notice that, in each such case, an error has been made with respect to such Erroneous Payment. Each Lender or other Payment Recipient further agrees that, in each such case, or if it otherwise becomes aware an Erroneous Payment (or portion thereof) may have been sent in error, such Lender or other such Payment Recipient shall promptly notify Agent of such occurrence and, upon demand from Agent, it shall promptly, but in no event later than one (1) Business Day thereafter, return to Agent the amount of any such Erroneous Payment (or portion thereof) that was received by such Lender or other such Payment Recipient to the date such amount is repaid to Agent in same day funds at the greater of the overnight bank funding rate and a rate determined by Agent in accordance with banking industry rules on interbank compensation from time to time in effect.

(c) Each Credit Party hereby agrees that (i) in the event an Erroneous Payment (or portion thereof) is not recovered from any. Lender or other such Payment Recipient that has received such Erroneous Payment (or portion thereof) for any reason, Agent shall be subrogated to all the rights of such Lender with respect to such amount and (ii) an Erroneous Payment shall not pay, prepay, repay, discharge or otherwise satisfy any Obligations owed any Credit Party hereunder or under any of the Other Documents.

(d) Each party's obligations under this Section 14.13 shall survive the resignation or replacement of Agent or any transfer of rights or obligations by, or the replacement of, a Lender, the termination of this Agreement, the termination of the commitments of the Lenders hereunder and the indefeasible payment in full in cash of the Obligations.

# XV. <u>MISCELLANEOUS.</u>

# XV.1 Governing Law

. This Agreement and each Other Document (unless and except to the extent expressly provided otherwise in any such Other Document), and all matters relating hereto or thereto or arising herefrom or therefrom (whether arising under contract law, tort law or otherwise) shall, in accordance with Section 5-1401 of the General Obligations Law of the State of New York, be governed by and construed in accordance with the laws of the State of New York. Any judicial

[Different first page setting changed from on in original to off in modified.].

proceeding brought by or against any Credit Party with respect to any of the Obligations, this Agreement, the Other Documents or any related agreement may be brought in any court of competent jurisdiction in the State of New York, United States of America, and, by execution and delivery of this Agreement, each Credit Party accepts for itself and in connection with its properties, generally and unconditionally, the non-exclusive jurisdiction of the aforesaid courts, and irrevocably agrees to be bound by any judgment rendered thereby in connection with this Agreement. Each Credit Party hereby waives personal service of any and all process upon it and consents that all such service of process may be made by registered mail (return receipt requested) directed to such Credit Party at its address set forth in Section 15.6 and service so made shall be deemed completed five (5) days after the same shall have been so deposited in the mails of the United States of America, or, at Agent's option, by service upon Borrowing Agent which each Borrower irrevocably appoints as such Borrower's agent for the purposes of accepting service within the State of New York. Nothing herein shall affect the right to serve process in any manner permitted by law or shall limit the right of Agent or any Lender to bring proceedings against any Credit Party in the courts of any other jurisdiction. Each Credit Party waives any objection to jurisdiction and venue of any action instituted hereunder and shall not assert any defense based on lack of jurisdiction or venue or based upon forum non conveniens. Each Credit Party waives the right to remove any judicial proceeding brought against such Credit Party in any state court to any federal court. Any judicial proceeding by any Credit Party against Agent or any Lender involving, directly or indirectly, any matter or claim in any way arising out of, related to or connected with this Agreement or any related agreement, shall be brought only in a federal or state court located in the Borough of Manhat

XV.2 Entire Understanding.

Ŧ

(a) This Agreement and the Other Documents contain the entire understanding between the Credit Parties, Agent and each Lender and supersedes all prior agreements and understandings, if any, relating to the subject matter hereof. Any promises, representations, warranties or guarantees not herein contained and hereinafter made shall have no force and effect unless in writing, signed by each Borrower's, Agent's and the applicable Lender's respective officers. Neither this Agreement nor any portion or provisions hereof may be changed, modified, amended, waived, supplemented, discharged, cancelled or terminated orally or by any course of dealing, or in any manner other than by an agreement in writing, signed by the party to be charged. Each Credit Party acknowledges that it has been advised by counsel in connection with the execution of this Agreement and Other Documents and is not relying upon oral representations or statements inconsistent with the terms and provisions of this Agreement.

(b) The Required Lenders, Agent with the consent in writing of the Required Lenders, and the Credit Parties signatory hereto may, subject to the provisions of this Section 15.2(b), from time to time enter into written supplemental agreements to this Agreement

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

or the Other Documents executed by such Credit Parties, for the purpose of adding or deleting any provisions or otherwise changing, varying or waiving in any manner the rights of Lenders, Agent or such Credit Parties thereunder or the conditions, provisions or terms thereof or waiving any Event of Default thereunder, but only to the extent specified in such written agreements; provided, however, that no such supplemental agreement shall without the consent of all Lenders:

(i) increase the Commitment Percentages or the maximum dollar commitment of any Lender or the amount of any Delayed Draw Term Loan;

(ii) whether or not any Advances are outstanding, extend the Term or the time for payment of principal or interest of any Advance (excluding the due date of any mandatory prepayment of an Advance), or any fee payable to any Lender, or reduce the principal amount of or the rate of interest borne by any Advances or reduce any fee payable to any Lender, without the consent of each Lender directly affected thereby (except that Required Lenders may elect to waive or rescind any imposition of the Default Rate under Section 3.1 hereof (unless imposed by Agent));

(iii) <u>alter the definition of the term Required Lenders or alter, amend or modify this Section 15.2(b);</u>

(iv) release any Collateral during any calendar year (other than in accordance with the provisions of this Agreement) having an aggregate value in excess of \$1,000,000;

- (v) alter, amend or modify the provisions of Section 11.5 hereof;
- (vi) change the rights and duties of Agent; or
- (vii) release any Guarantor.

(c) Any such supplemental agreement shall apply equally to each Lender and shall be binding upon the Credit Parties, Lenders and Agent and all future holders of the Obligations. In the case of any waiver, the Credit Parties, Agent and Lenders shall be restored to their former positions and rights, and any Event of Default waived shall be deemed to be cured and not continuing, but no waiver of a specific Event of Default shall extend to any subsequent Event of Default (whether or not the subsequent Event of Default is the same as the Event of Default which was waived), or impair any right consequent thereon.

(d) In the event that Agent or Borrowers request the consent of a Lender (other than Lafayette Square or any of its Affiliates). pursuant to this Section 15.2 and such consent is denied, then Agent or Borrowers may, at their respective option and upon five (5) Business Days prior written notice to Agent (if given by Borrowers) and such Lender (a "Non-Consenting Lender"), require such Non-Consenting Lender to assign its interest in the Advances to Lafayette Square, or if Lafayette Square shall decline to purchase such interest, to another

[Different first page setting changed from on in original to off in modified.].

Lender or to any other Person eligible to become a Purchasing Lender in accordance with the terms of Section 15.3(c), for a price equal to the then outstanding principal amount thereof plus accrued and unpaid interest and fees due such Non-Consenting Lender, which interest and fees shall be paid when collected from Borrowers. In the event any Non-Consenting Lender is required to assign its interests hereunder, such Non-Consenting Lender, which interest, pursuant to a Transfer Supplement, on the date specified in the notice given to such Non-Consenting Lender, which shall be no later than fifteen (15) Business Days after the date of such notice.

(e) <u>Agent is hereby authorized by each Borrower and the Lenders, from time to time in Agent's sole discretion, (i) after the</u> occurrence and during the continuation of a Default or an Event of Default, or (ii) at any time that any of the other applicable conditions precedent set forth in Section 8.2 hereof have not been satisfied, to make advances to Borrowers on behalf of the Lenders which Agent, in its reasonable business judgment, deems necessary or desirable (x) to preserve or protect the Collateral, or any portion thereof, (y) to enhance the likelihood of, or maximize the amount of, repayment of the Advances and other Obligations, or (z) to pay any other amount chargeable to Borrowers pursuant to the terms of this Agreement.

XV.3 Successors and Assigns; Participations; New Lenders

(a) <u>This Agreement shall be binding upon and inure to the benefit of the Credit Parties signatory hereto, Agent, each Lender, all future holders of the Obligations and their respective successors and permitted assigns, except that no Credit Party may assign or transfer any of its rights or obligations under this Agreement without the prior written consent of Agent and each Lender.</u>

(b) Each Credit Party acknowledges that in the regular course of commercial banking business one or more Lenders may at any time and from time to time sell participating interests in the Advances to other financial institutions (each such transferee or purchaser of a participating interest, a "Participant"). Upon the occurrence and during the continuance of an Event of Default, each Participant may, with the prior consent of Agent, exercise all rights of payment (including rights of set-off) with respect to the portion of such Advances held by it or other Obligations payable hereunder as fully as if such Participant were the direct holder thereof and the Obligations shall be reduced by the amount of any such setoff, provided that Borrowers shall not be required to pay to any Participant more than the amount which it would have been required to pay to Lender which granted an interest in its Advances or other Obligations payable hereunder to such Participant had such Lender retained such interest in the Advances hereunder or other Obligations payable hereunder and in no event shall Borrowers be required to pay any such amount arising from the same circumstances and with respect to the same Advances or other Obligations payable hereunder to both such Lender and such Participant. Each Borrower hereby grants to any Participant a continuing security interest in any deposits, moneys or other property actually or constructively held by such Participant as security for the Participant's

[Different first page setting changed from on in original to off in modified.].

interest in the Advances. Each Lender that sells a participation shall, acting solely for this purpose as an agent of the Borrowers, maintain a register on which it enters the name and address of each Participant and the principal amounts (and stated interest) of each Participant's interest in the Advances and Obligations (the "Participant Register"); provided that no Lender shall have any obligation to disclose all or any portion of the Participant Register (including the identity of any Participant or any information relating to a Participant's interest in any commitments, loans, letters of credit or its other obligations under this Agreement or any Other Document) to any Person except to the extent that such disclosure is necessary to establish that such commitment, loan, letter of credit or other obligation is in registered form under Section 5f.103-1(c) of the United States Treasury Regulations. The entries in the Participant Register shall be conclusive absent manifest error, and such Lender shall treat each Person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary. For the avoidance of doubt, Agent (in its capacity as Agent) shall have no responsibility for maintaining a Participant Register.

Any Lender, with the consent of Agent which shall not be unreasonably withheld or delayed, may sell, assign or transfer all (c) or any part of its rights and obligations under or relating to the Term Loans under this Agreement and the Other Documents to one or more Eligible Assignees and one or more Eligible Assignees may commit to make Advances hereunder (each a "Purchasing Lender"), in minimum amounts of not less than \$1,000,000 in the case of any transfer to any Purchasing Lender which is not a Lender or an Affiliate of Lender prior to such transfer, pursuant to a Transfer Supplement, executed by a Purchasing Lender, the transferor Lender, and Agent and delivered to Agent for recording. Upon such execution, delivery, acceptance and recording, from and after the transfer effective date determined pursuant to such Transfer Supplement, (i) Purchasing Lender thereunder shall be a party hereto and, to the extent provided in such Transfer Supplement, have the rights and obligations of a Lender thereunder with the Commitment Percentages as set forth therein, and (ii) the transferor Lender thereunder shall, to the extent provided in such Transfer Supplement, be released from its obligations under this Agreement, the Transfer Supplement creating a novation for that purpose. Such Transfer Supplement shall be deemed to amend this Agreement to the extent, and only to the extent, necessary to reflect the addition of such Purchasing Lender and the resulting adjustment of the Commitment Percentages arising from the purchase by such Purchasing Lender of all or a portion of the rights and obligations of such transferor Lender under this Agreement and the Other Documents. Each Borrower hereby consents to the addition of such Purchasing Lender and the resulting adjustment of the Commitment Percentages arising from the purchase by such Purchasing Lender of all or a portion of the rights and obligations of such transferor Lender under this Agreement and the Other Documents. Borrowers shall execute and deliver such further documents and do such further acts and things in order to effectuate the foregoing. Notwithstanding the foregoing or anything to the contrary set forth herein, no assignment or transfer shall be made to a Defaulting Lender unless an Event of Default shall have occurred and be continuing.

(d) <u>Any Lender, with the consent of Agent which shall not be unreasonably withheld or delayed, may directly or indirectly sell,</u> assign or transfer all or any portion of its

[Different first page setting changed from on in original to off in modified.].

rights and obligations under or relating to the Term Loans under this Agreement and the Other Documents to an entity, whether a corporation, partnership, trust, limited liability company or other entity that (i) is engaged in making, purchasing, holding or otherwise investing in bank loans and similar extensions of credit in the ordinary course of its business and (ii) is administered, serviced or managed by the assigning Lender or an Affiliate of such Lender (a "Purchasing CLO" and together with each Participant and Purchasing Lender, each a "Transferee" and collectively the "Transferees"), pursuant to a Transfer Supplement modified as appropriate to reflect the interest being assigned ("Modified Transfer Supplement"), executed by any intermediate purchaser, the Purchasing CLO, the transferor Lender, and Agent as appropriate and delivered to Agent for recording. Upon such execution and delivery, from and after the transfer effective date determined pursuant to such Modified Transfer Supplement, (i) Purchasing CLO thereunder shall be a party hereto and, to the extent provided in such Modified Transfer Supplement, have the rights and obligations of a Lender thereunder and (ii) the transferor Lender thereunder shall, to the extent provided in such Modified Transfer Supplement, be released from its obligations under this Agreement, the Modified Transfer Supplement creating a novation for that purpose. Such Modified Transfer Supplement shall be deemed to amend this Agreement to the extent, and only to the extent, necessary to reflect the addition of such Purchasing CLO. Each Borrower hereby consents to the addition of such Purchasing CLO. Borrowers shall execute and deliver such further documents and do such further acts and things in order to effectuate the foregoing. Notwithstanding the foregoing or anything to the contrary set forth herein, no assignment or transfer shall be made to a Defaulting Lender unless an Event of Default shall have occurred and be continuing.

(e) Each Borrower hereby acknowledges and makes the Obligations registered Obligations for United States withholding tax purposes. Agent shall maintain at its address a copy of each Transfer Supplement and Modified Transfer Supplement delivered to it and a register (the "Register") for the recordation of the names and addresses of each Lender and the outstanding principal, accrued and unpaid interest and other fees due hereunder and Agent shall record the assignment of the Advances in the Register. Any assignment of the Advances, whether or not evidenced by a note, shall be effective as to Borrowers only upon appropriate entries with respect thereto being made in the Register. The entries in the Register shall be prima facie evidence, in the absence of manifest error, and Borrowers, Agent and Lenders may treat each Person whose name is recorded in the Register as the owner of the Advances recorded therein for the purposes of this Agreement notwithstanding notice to the contrary. The Lenders, by their acceptance of this Agreement or any Transfer Supplement, agree to be bound by the provisions of this paragraph and to indemnify and hold harmless Agent against any loss or liability arising from the assignment by a holder of the Advances in violation of this subsection (e). The Register shall be available for inspection by Borrowers or any Lender at any reasonable time and from time to time upon reasonable prior notice. Agent shall receive a fee in the amount of \$3,500 payable by the applicable Purchasing Lender and/or Purchasing CLO upon the effective date of each transfer or assignment (other than to an intermediate purchaser) to such Purchasing Lender and/or Purchasing CLO.

[Different first page setting changed from on in original to off in modified.].

(f) Each Credit Party authorizes each Lender to disclose to any Transferee and any prospective Transferee any and all financial information in such Lender's possession concerning any Credit Party which has been delivered to such Lender by or on behalf of any Credit Party pursuant to this Agreement or in connection with such Lender's credit evaluation of any Credit Party.

(g) <u>Any Lender may at any time pledge or assign a security interest in all or any portion of its rights under this Agreement to secure obligations of such Lender, including any pledge or assignment to secure obligations to a Federal Reserve Bank; provided that no such pledge or assignment shall release such Lender from any of its obligations hereunder or substitute any such pledgee or assignee for such Lender as a party hereto.</u>

# XV.4 Application of Payments

. Agent shall have the continuing and exclusive right, in its Permitted Discretion, to apply or reverse and re-apply any payment and any and all proceeds of Collateral to any portion of the Obligations. To the extent that any Credit Party makes a payment or Agent or any Lender receives any payment or proceeds of the Collateral for any Credit Party's benefit, which are subsequently invalidated, declared to be fraudulent or preferential, set aside or required to be repaid to a trustee, debtor in possession, receiver, custodian or any other party under any bankruptcy law, common law or equitable cause, then, to such extent, the Obligations or part thereof intended to be satisfied shall be revived and continue as if such payment or proceeds had not been received by Agent or such Lender.

# XV.5 Indemnity

. Each Credit Party shall defend, protect, indemnify, pay and save harmless Agent, each Lender and each of their respective officers, directors, Affiliates, attorneys, employees and agents (each an "<u>Indemnified Party</u>") for and from and against any and all claims, demands, liabilities, obligations, losses, damages, penalties, fines, actions, judgments, suits, costs, charges, expenses and disbursements of any kind or nature whatsoever (including reasonable fees and disbursements of counsel (including allocated costs of internal counsel)) (collectively, "<u>Claims</u>") which may be imposed on, incurred by, or asserted against any Indemnified Party arising out of or in any way relating to or as a consequence, direct or indirect, of: (a) this Agreement, the Other Documents, the Advances and other Obligations, the Eligible <u>Impact ServicesWorker Solutions</u>, and/or the transactions contemplated hereby including the Closing Date Transactions, (b) any action or failure to act or action taken only after delay or the satisfaction of any conditions by any Indemnified Party in connection with and/or relating to the negotiation, execution, delivery or administration of the Agreement and the Other Documents, the credit facilities established hereunder and thereunder and/or the transactions contemplated hereby including the Closing Date Transactions or warranties made in this Agreement and the Other Documents, (c) any Credit Party's failure to observe, perform or discharge any of its covenants, obligations, agreements or duties under or breach of any of the representations or warranties made in this Agreement and the Other Documents, (d) the enforcement of any of the rights and remedies of Agent or any Lender under the Agreement and the Other Documents, (e)

[Different first page setting changed from on in original to off in modified.].

any threatened or actual imposition of fines or penalties, or disgorgement of benefits, for violation of any Anti-Terrorism Law by any Credit Party, any Affiliate or Subsidiary of any of the Credit Parties, and (f) any claim, litigation, proceeding or investigation instituted or conducted by any Governmental Body or instrumentality, any Borrower, any Affiliate or Subsidiary of any Borrowers, or any Guarantor, or any other Person with respect to any aspect of, or any transaction contemplated by, or referred to in, or any matter related to, this Agreement or the Other Documents, whether or not Agent or any Lender is a party thereto. Without limiting the generality of any of the foregoing, each Credit Party shall defend, protect, indemnify, pay and save harmless each Indemnified Party from any Claims which may be imposed on, incurred by, or asserted against any Indemnified Party under any Environmental Laws with respect to or in connection with any Real Property owned or leased by any Credit Party, any Environmental Liability, the presence of any Hazardous Materials affecting any Real Property owned or leased by any Credit Party (whether or not the same originates or emerges from such Real Property or any contiguous real estate), including any Claims consisting of or relating to the imposition or assertion of any Lien on any Real Property owned or leased by any Credit Party under any Environmental Laws and any loss of value of such Real Property as a result of the foregoing except to the extent such loss, liability, damage and expense is attributable to any Hazardous Materials Activity resulting from actions on the part of Agent or any Lender. The Credit Parties' obligations under this Section 15.5 shall arise upon the discovery of the presence of any Hazardous Materials at any Real Property owned or leased by any Credit Party, whether or not any federal, state, or local environmental agency has taken or threatened any action in connection with the presence of any Hazardous Materials, in each such case except to the extent that any of the foregoing arises out of the gross negligence or willful misconduct of the Indemnified Party (as determined by a court of competent jurisdiction in a final and non-appealable judgment). Without limiting the generality of the foregoing, this indemnity shall extend to any liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses and disbursements of any kind or nature whatsoever (including fees and disbursements of counsel) asserted against or incurred by any of the Indemnified Parties by any Person under any Environmental Laws or similar laws by reason of any Credit Party's or any other Person's failure to comply with laws applicable to solid or hazardous waste materials, including Hazardous Materials. This Section 15.5 shall not apply with respect to Taxes other than Taxes that represent losses, claims, damages, etc. arising from any non-Tax claim.

### XV.6 Notice

. Any notice or request hereunder may be given to any Credit Party or to Agent or any Lender at their respective addresses set forth below or at such other address as may hereafter be specified in a notice designated as a notice of change of address under this Section. Any notice, request, demand, direction or other communication (for purposes of this <u>Section 15.6</u> only, a "<u>Notice</u>") to be given to or made upon any party hereto under any provision of this Agreement shall be given or made by telephone or in writing (which includes by means of electronic transmission (*i.e.*, e-mail) or facsimile transmission or by setting forth such Notice on a website (an "<u>Internet Posting</u>") if Notice of such Internet Posting (including the information necessary to access such site) has previously been delivered to the applicable parties hereto by another means

[Different first page setting changed from on in original to off in modified.].

set forth in this <u>Section 15.6</u>) in accordance with this <u>Section 15.6</u>. Any such Notice must be delivered to the applicable parties hereto at the addresses and numbers set forth under their respective names on <u>Section 15.6</u> hereof or in accordance with any subsequent unrevoked Notice from any such party that is given in accordance with this <u>Section 15.6</u>. Any Notice shall be effective:

(a) <u>In the case of hand-delivery, when delivered;</u>

(b) If given by mail, five days after such Notice is deposited with the United States Postal Service, with first-class postage prepaid, return receipt requested;

(c) In the case of a telephonic Notice, when a party is contacted by telephone, if delivery of such telephonic Notice is confirmed no later than the next Business Day by hand delivery, a facsimile or electronic transmission, an Internet Posting or an overnight courier delivery of a confirmatory Notice (received at or before noon on such next Business Day);

(d) In the case of a facsimile transmission, when sent to the applicable party's facsimile machine's telephone number, if the party sending such Notice receives confirmation of the delivery thereof from its own facsimile machine;

(e) In the case of electronic transmission, when actually received;

(f) In the case of an Internet Posting, upon delivery of a Notice of such posting (including the information necessary to access such site) by another means set forth in this Section 15.6; and

(g) If given by any other means (including by overnight courier), when actually received.

Any Lender giving a Notice to any Credit Party shall concurrently send a copy thereof to Agent, and Agent shall promptly notify the other Lenders of its receipt of such Notice.

(A) If to Agent at:

Lafayette Square Loan Servicing, LLC PO Box 25250

PMB 13941

Miami, Florida 33102-5250 Attention: Susan Golden Email: legal@lafayettesquare.com; lsloanops@lafayettesquare.com

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

with a copy to (which shall not constitute notice) to:

Blank Rome LLP One Logan Square Philadelphia, Pennsylvania 19103 Attention: Shlomo B. Troodler, Esq. Telephone: (215) 569-5338 Facsimile: (215) 832-5338 E-mail: troodler@blankrome.com

(B) If to a Lender other than Agent, as specified on the signature pages hereof;

(C) If to any Credit Party:

Direct Digital Holdings, LLC 12331177 West Loop South, Suite 11701310 Huston, TX 7704277027 Attention: Mark Walker E-mail: \*\*\*

with a copy (which shall not constitute notice) to:

McGuireWoods LLP 2000 McKinney Avenue Suite 1400 Dallas, TX 75201 Attention: Phyllis Y. Young Telephone: (214) 932-6420 E-mail: pyoung@mcguirewoods.com

#### XV.7 Survival

. The obligations of the Credit Parties under <u>Sections 3.7</u>, <u>3.9</u>, <u>3.10</u>, <u>4.19(e)</u>, <u>15.5</u> and <u>15.9</u> and the obligations of Lenders under <u>Section 14.7</u>, shall survive termination of this Agreement and the Other Documents and payment in full of the Obligations.

## XV.8 Severability

. If any part of this Agreement is contrary to, prohibited by, or deemed invalid under Applicable Laws, such provision shall be inapplicable and deemed omitted to the extent so contrary, prohibited or invalid, but the remainder hereof shall not be invalidated thereby and shall be given effect so far as possible.

[Different first page setting changed from on in original to off in modified.].

#### XV.9 Expenses

. The Credit Parties shall pay all reasonable costs and expenses including reasonable attorneys' fees (including the allocated costs of in house counsel) and disbursements incurred by Agent on its Behalf or on Behalf of Lenders: (a) in all efforts made to enforce payment of any Obligation or effect collection of any Collateral, (b) in connection with the entering into, preparation, negotiation, execution, delivery, modification, amendment, administration (including costs and expense with respect to electronic distribution services for distribution of documents and information to the Lenders) and enforcement of this Agreement, the Other Documents or any consents or waivers hereunder or thereunder and all related agreements, documents and instruments (whether or not the transactions contemplated hereby or thereby shall be consummated), (c) in instituting, maintaining, preserving, enforcing and foreclosing on Agent's security interest in or Lien on any of the Collateral, or maintaining, preserving or enforcing any of Agent's or any Lender's rights hereunder, under the Other Documents and under all related agreements, documents, whether through judicial proceedings or otherwise, (d) in defending or prosecuting any actions or proceedings arising out of or relating to Agent's or any Lender's transactions with the Credit Parties or Seller or (e) in connection with any advice given to Agent or any Lender with respect to its rights and obligations under this Agreement, the Other Documents and all related agreements, documents, notwithstanding the foregoing, the Borrowers shall not be required to reimburse the costs of (x) more than one (1) field examinations in any consecutive period of 365 days unless an Event of Default shall have occurred and be continuing and (y) more than one (1) appraisal of machinery and equipment and of Inventory in any consecutive period of 365 days unless an Event of Default shall have occurred and be continuing.

### XV.10 Injunctive Relief

. Each Credit Party recognizes that, in the event such Credit Party fails to perform, observe or discharge any of its obligations or liabilities under this Agreement, or threatens to fail to perform, observe or discharge such obligations or liabilities, any remedy at law may prove to be inadequate relief to Lenders; therefore, Agent, if Agent so requests, shall be entitled to temporary and permanent injunctive relief in any such case without the necessity of proving that actual damages are not an adequate remedy.

### XV.11 Damages

. Neither Agent nor any Lender, nor any agent or attorney for any of them, shall be liable to any Credit Party (or any Affiliate of any such Person) for indirect, punitive, exemplary or consequential damages arising from any breach of contract, tort or other wrong relating to the establishment, administration or collection of the Obligations or as a result of any transaction contemplated under this Agreement or any Other Document.

XV.12 Captions

[Different first page setting changed from on in original to off in modified.].

. The captions at various places in this Agreement are intended for convenience only and do not constitute and shall not be interpreted as part of this Agreement.

## XV.13 Counterparts, etc

. This Agreement may be executed in any number of and by different parties hereto on separate counterparts, all of which, when so executed, shall be deemed an original, but all such counterparts shall constitute one and the same agreement. Any signature delivered by a party by facsimile or other similar method of electronic transmission (including email transmission of a PDF image) shall be deemed to be an original signature hereto.

# XV.14 Construction

. The parties acknowledge that each party and its counsel have reviewed this Agreement and that the normal rule of construction to the effect that any ambiguities are to be resolved against the drafting party shall not be employed in the interpretation of this Agreement or any amendments, schedules or exhibits thereto.

# XV.15 Confidentiality; Sharing Information

. Agent, each Lender and each Transferee shall hold all non-public information obtained by Agent, such Lender or such Transferee (and designated as non-public information by Borrower in writing) pursuant to the requirements of this Agreement in accordance with Agent's, such Lender's and such Transferee's customary procedures for handling confidential information of this nature; provided, however, Agent, each Lender and each Transferee may disclose such confidential information (a)(i) to its examiners, Affiliates, outside auditors, counsel and other professional advisors, (ii) to Agent, any Lender or to any prospective Transferees, (iii) to any investor or prospective investor in Agent, any Lender, or any of their respective affiliates, in each case, subject to a non disclosure agreement containing provisions substantially the same as those of this Section 15.15 and among other things, standstill provision preventing transactions in the Credit Parties' stock and in compliance with applicable Federal and state securities laws. (b) as required by any Governmental Body or representative thereof or pursuant to legal process or in public filings to the extent customary, (c) subject to an agreement (i) in connection with the exercise of any secured creditor remedy under this Agreement or under any of the Other Documents, and (ii) subject to a non disclosure agreement containing provisions substantially the same as those of this Section 15.15 and in compliance with applicable Federal and state securities laws, in connection with a bona fide initial public offering of Equity Interests of Agent, any Lender or any Transferee, to the extent disclosure is required or advisable (in the determination of Agent, such Lender or such Transferee, as applicable, and their respective counsel) under the Securities Act of 1933, as amended; provided, further, that (I) unless specifically prohibited by Applicable Law, Agent, each Lender and each Transferee shall use its reasonable best efforts prior to disclosure thereof, to notify Borrowing Agent of the applicable request for disclosure of such non-public information (A) by a Governmental Body or representative thereof (other than any such request in connection with an examination of the

[Different first page setting changed from on in original to off in modified.].

financial condition of a Lender or a Transferee by such Governmental Body) or (B) pursuant to legal process; provided, that notwithstanding anything to the contrary above in this sub-clause (I), the foregoing notice obligation shall not apply with respect to disclosures of non-public information that are requested or required pursuant to routine regulatory examinations or inquiries that do not reference any Credit Party, this Agreement or any Other Document, to which Agent, Lender or a Transferee is subject in the ordinary course of business, and (II) in no event shall Agent, any Lender or any Transferee be obligated to return any materials furnished by any Credit Party other than those documents and instruments in possession of Agent or any Lender in order to perfect its Lien on the Collateral once the Obligations have been paid in full (other than Inchoate Obligations) and the termination of all obligations of Agent and the Lenders to make any Advances hereunder. Each Credit Party acknowledges that from time to time financial advisory, investment banking and other services may be offered or provided to such Credit Party or one or more of its Affiliates (in connection with this Agreement or otherwise) by any Lender or by one or more Subsidiaries or Affiliates of such Lender and each Credit Party hereby authorizes each Lender to share any information delivered to such Lender by any Credit Party and/or their respective Subsidiaries pursuant to this Agreement or any Other Document, or in connection with the decision of such Lender to enter into this Agreement, to any such Subsidiary or Affiliate of such Lender, it being understood that any such Subsidiary or Affiliate of any Lender receiving such information shall be bound by the provisions of this <u>Section 15.15</u> as if it were a Lender hereunder. Such authorization shall survive the repayment of the other Obligations and the termination of this <u>Agreement</u>.

AGENT AND EACH LENDER ACKNOWLEDGE THAT INFORMATION RECEIVED FROM ANY CREDIT PARTY RELATING TO SUCH CREDIT PARTY OR ANY SUBSIDIARY OR THEIR BUSINESSES OR THE COLLATERAL FURNISHED TO SUCH PERSON PURSUANT TO THIS AGREEMENT MAY INCLUDE INFORMATION IDENTIFIED BY THE CREDIT PARTIES AS MATERIAL NON-PUBLIC INFORMATION CONCERNING SUCH CREDIT PARTY, ITS SUBSIDIARIES AND ITS OTHER AFFILIATES AND THEIR RESPECTIVE SECURITIES, AND CONFIRMS THAT IT HAS DEVELOPED COMPLIANCE PROCEDURES REGARDING THE USE OF SUCH MATERIAL NON-PUBLIC INFORMATION AND THAT IT WILL HANDLE SUCH MATERIAL NON-PUBLIC INFORMATION IN ACCORDANCE WITH THOSE PROCEDURES AND APPLICABLE LAW, INCLUDING FEDERAL AND STATE SECURITIES LAWS.

ALL INFORMATION, INCLUDING REQUESTS FOR WAIVERS AND AMENDMENTS, FURNISHED BY ANY CREDIT PARTY OR THE AGENT PURSUANT TO, OR IN THE COURSE OF ADMINISTERING, THIS AGREEMENT WILL BE SYNDICATE-LEVEL INFORMATION, WHICH MAY CONTAIN INFORMATION IDENTIFIED BY THE CREDIT PARTIES AS MATERIAL NON-PUBLIC INFORMATION CONCERNING THE CREDIT PARTIES, THEIR SUBSIDIARIES AND THEIR OTHER AFFILIATES AND THEIR RESPECTIVE SECURITIES. ACCORDINGLY, EACH OF AGENT AND EACH LENDER REPRESENTS TO THE CREDIT PARTIES THAT SUCH PERSON HAS IDENTIFIED A CREDIT CONTACT (WHICH SUCH PERSON MAY CHANGE FROM TIME TO TIME

[Different first page setting changed from on in original to off in modified.].

160

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

UPON NOTICE FROM SUCH PERSON TO THE CREDIT PARTIES) WHO MAY RECEIVE INFORMATION THAT MAY CONTAIN SUCH MATERIAL NON-PUBLIC INFORMATION AND SHARE SUCH INFORMATION IN ACCORDANCE WITH ITS COMPLIANCE PROCEDURES AND APPLICABLE LAW, INCLUDING FEDERAL AND STATE SECURITIES LAWS.

## XV.16 PUBLICITY

. EACH CREDIT PARTY AND EACH LENDER HEREBY AUTHORIZES AGENT, AT ITS OWN EXPENSE, AND WITH THE PRIOR WRITTEN CONSENT OF DDH (SUCH CONSENT NOT TO BE UNREASONABLY WITHHELD), TO MAKE APPROPRIATE ANNOUNCEMENTS OF THE FINANCIAL ARRANGEMENT ENTERED INTO AMONG THE CREDIT PARTIES, AGENT AND LENDERS AND RELATED SERVICES, AND HEREBY GRANTS PERMISSION FOR AGENT TO USE THE LOGO OF ANY CREDIT PARTY AND LENDER, INCLUDING IN MARKETING MATERIALS AND OTHER ANNOUNCEMENTS WHICH ARE COMMONLY KNOWN AS TOMBSTONES, IN SUCH PUBLICATIONS, media, or forum AND TO SUCH SELECTED PARTIES AS AGENT SHALL IN ITS SOLE AND ABSOLUTE DISCRETION DEEM APPROPRIATE; PROVIDED, THAT NO ANNOUNCEMENTS PURSUANT TO THIS SECTIONSECTION 15.16 SHALL BE MADE PRIOR TO THE CONSUMMATION OF A QUALIFIED IPO OR WHILE ANY CREDIT PARTY IS CONTEMPLATING A QUALIFIED IPO AND ALL SUCH ANNOUNCEMENTS PURSUANT TO THIS SECTIONSECTION 15.16 SHALL BE IN CONFORMANCE WITH APPLICABLE FEDERAL AND STATE SECURITIES LAWS.

XV.17 Certifications From Banks and Participants; USA PATRIOT Act

(a) Each Lender or assignee or participant of a Lender that is not incorporated under the Applicable Laws of the United States of America or a state thereof (and is not excepted from the certification requirement contained in Section 313 of the USA PATRIOT Act and the applicable regulations because it is both (i) an affiliate of a depository institution or foreign bank that maintains a physical presence in the United States or foreign country, and (ii) subject to supervision by a banking authority regulating such affiliated depository institution or foreign bank) shall deliver to Agent the certification, or, if applicable, recertification, certifying that such Lender is not a "shell" and certifying to other matters as required by Section 313 of the USA PATRIOT Act and the applicable regulations: (1) within 10 days after the Closing Date, and (2) as such other times as are required under the USA PATRIOT Act.

(b) <u>The USA PATRIOT Act requires all financial institutions to obtain, verify and record certain information that identifies</u> individuals or business entities which open an "account" with such financial institution. Consequently, any Lender may from time to time request, and each Credit Party shall provide to such Lender, such Credit Party's name, address,

[Different first page setting changed from on in original to off in modified.].

tax identification number and/or such other identifying information as shall be necessary for such Lender to comply with the USA PATRIOT Act and any other Anti-Terrorism Law.

XV.18 Borrowing Agency Provisions.

÷

(a) Each Borrower hereby irrevocably designates Borrowing Agent to be its attorney and agent and in such capacity to borrow, sign and endorse notes, and execute and deliver all instruments, documents, writings and further assurances now or hereafter required hereunder, on Behalf of such Borrower, and hereby authorizes Agent to pay over or credit all loan proceeds hereunder in accordance with the request of Borrowing Agent.

(b) The handling of this credit facility as a co-borrowing facility with a borrowing agent in the manner set forth in this Agreement is solely as an accommodation to Borrowers and at their request. Neither Agent nor any Lender shall incur liability to any Borrower as a result thereof. To induce Agent and Lenders to do so and in consideration thereof, each Borrower hereby indemnifies Agent and each Lender and holds Agent and each Lender harmless from and against any and all liabilities, expenses, losses, damages and claims of damage or injury asserted against Agent or any Lender by any Borrower arising from or incurred by reason of the handling of the financing arrangements of Borrowers as provided herein, reliance by Agent or any Lender on any request or instruction from Borrowing Agent or any other action taken by Agent or any Lender with respect to this Section 15.18 except due to willful misconduct or gross (not mere) negligence by the indemnified party (as determined by a court of competent jurisdiction in a final and non-appealable judgment).

(c) All Obligations shall be joint and several, and each Borrower shall make payment upon the maturity of the Obligations by acceleration or otherwise, and such obligation and liability on the part of each Borrower shall in no way be affected by any extensions, renewals and forbearance granted to Agent or any Lender to any Borrower, failure of Agent or any Lender to give Borrowing Agent or any Borrower notice of borrowing or any other notice, any failure of Agent or any Lender to pursue or preserve its rights against any Borrower, the release by Agent or any Lender of any Collateral now or thereafter acquired from any Borrower, and such agreement by each Borrower to pay upon any notice issued pursuant thereto is unconditional and unaffected by prior recourse by Agent or any Lender to the other Borrowers or any Collateral for any Borrower's Obligations or the lack thereof. Each Borrower waives all suretyship defenses.

# XVI. GUARANTY AND SURETYSHIP AGREEMENT.

# XVI.1 Guaranty and Suretyship Agreement

. Each Guarantor hereby guarantees, and becomes surety for the prompt payment and performance when due (whether at stated maturity, by required prepayment, declaration, demand, by acceleration or otherwise) of all of the Obligations owing by the Credit Parties to

[Different first page setting changed from on in original to off in modified.].

Agent and Lenders (the "<u>Guaranteed Obligations</u>"). The obligations and liabilities of the Guarantors under this <u>Article XVI</u> are joint and several, and each Guarantor hereby acknowledges and accepts such joint and several liability and further acknowledges and agrees that the joint and several liabilities of Guarantors under the provisions of this <u>Article XVI</u> shall be primary and direct liabilities and not secondary liabilities.

# XVI.2 Guaranty of Payment and Not Merely Collection

. The provisions of this <u>Article XVI</u> constitute a guaranty of payment and not of collection and neither Agent nor any Lender shall be required, as a condition of any Guarantor's liability hereunder, to make any demand upon or to pursue any of their rights against any Credit Parties and/or any of the Collateral, or to pursue any rights which may be available to Agent and Lenders with respect to any other person who may be liable for the payment of the Guaranteed Obligations and/or any other collateral or security available to Agent and Lenders therefor.

# XVI.3 Guarantor and Suretyship Waivers.

÷

(a) The provisions of this Article XVI constitute an absolute, unconditional, irrevocable and continuing guaranty and will remain in full force and effect until all of the Guaranteed Obligations have been paid in full. The provisions of this Article XVI will remain in full force and effect even if there are no Guaranteed Obligations outstanding at a particular time or from time to time. The provisions of this Article XVI will remain in full force and effect (i) by any surrender, exchange, acceptance, compromise or release by Agent and Lenders of any other party, or any other guaranty or any Collateral or other collateral or security held by it for any of the Guaranteed Obligations, (ii) by any failure of Agent and Lenders to take any steps to perfect or maintain their Liens or security interest in or to preserve their rights in or to any Collateral or any other security or other collateral for the Guaranteed Obligations or any guaranty, or (iii) by any irregularity, unenforceability or invalidity of the Guaranteed Obligations or any part thereof or any security therefor or other guaranty thereof, and the provisions of this Article XVI will not be affected by any other facts, events, occurrences or circumstances (except payment in full of the Guaranteed Obligations) that might otherwise give rise to any "guarantor" or "suretyship" defenses to which any Guarantor might otherwise be entitled, all of which such "guarantor" or "suretyship" defenses of each Guarantor hereunder shall not be affected, modified or impaired by any counterclaim, set-off, deduction or defense of any kind, including any such counterclaim, set-off, deduction or surety may have against Agent and Lenders, except payment in full of the Guaranteed Obligations.

(b) <u>Notice of acceptance of the agreement to guaranty provided for under the provisions of this Article XVI, notice of extensions of credit to Borrowers from time to time, notice of default, diligence, presentment, notice of dishonor, protest, demand for payment, and</u>

[Different first page setting changed from on in original to off in modified.].

any defense based upon Agent's or any Lender's failure to comply with the notice requirements of §§ 9-611, 9-612 and 9-613 of the Uniform Commercial Code are hereby waived to the fullest extent permitted by law. Each Guarantor hereby waives all defenses based on suretyship or impairment of collateral to the fullest extent permitted by law.

(c) Agent may at any time and from time to time, in accordance with the terms of this Agreement, without impairing or releasing, discharging or modifying any Guarantor's liabilities hereunder and (for purposes of this Article XVI only) without notice to or the consent of any Guarantor: (i) change the manner, place, time or terms of payment or performance of or interest rates or other fees on, or other terms relating to (including the maturity thereof), any of the Guaranteed Obligations; (ii) renew, extend, substitute, modify, amend or alter or refinance, or grant consents or waivers relating to any of the terms and provisions of this Agreement or any of the Other Documents or of the Guaranteed Obligations, or of any other guaranties, or any security for the Obligations or guaranties, (iii) increase (without limit of any kind) or decrease the Guaranteed Obligations (including all loans and extensions of credit thereunder) or modify the terms on which loans and extensions of credit may be made to Borrower (including without limitation by making available to Borrowers under this Agreement and/or any Other Document and as part of the Guaranteed Obligations any new loans, advances or other extensions of credit of any kind, including any such new loans, advances or extension of credit of a new or different type or nature as compared to the loans, advances and extensions of credit available to Borrowers under this Agreement as of the date hereof); (iv) apply any and all payments by whomever paid or however realized including any proceeds of the Collateral or any other collateral or security, to any Guaranteed Obligations in such order, manner and amount as Agent may determine in its sole discretion, but in accordance with the terms of this Agreement; (v) settle, compromise or deal with any other Person, including any Borrower or any other guarantor, with respect to the Guaranteed Obligations in such manner as Agent deems appropriate in its sole discretion; (vi) substitute, exchange, subordinate, sell, compromise or release any security or guaranty for the Guaranteed Obligations; or (vii) take such actions and exercise such remedies hereunder as provided herein.

## XVI.4 Repayments or Recovery from Agent or Lenders

. If any demand or claim is made at any time upon Agent or any Lender for the repayment or recovery of any amount received by it in payment or on account of the Guaranteed Obligations (including any such demand or claim made in respect of or arising out of any laws relating to fraudulent transfers, fraudulent conveyances or preferences) and if Agent or any Lender repays all or any part of such amount by reason of any judgment, decree or order of any court or administrative body in respect of such demand or claim, or by reason of any settlement or compromise of any such demand or claim, the joint and several liability of Guarantors with respect to such portion of the Guaranteed Obligations previously satisfied by the payment of the amount so repaid or recovered shall be reinstated and revived and Guarantors will be and remain jointly and severally liable hereunder for the amount so repaid or recovered to the same extent as if such amount had never been received originally by Agent and/or such Lender, as the case may be. The provisions of this <u>Section 16.4</u> shall survive any release and/or termination of this

[Different first page setting changed from on in original to off in modified.].

Agreement (and/or of any Guarantor's liability under this <u>Article XVI</u>) and will be and remain effective notwithstanding any contrary action which may have been taken by Guarantor in reliance upon such payment, and any such contrary action so taken will be without prejudice to Agent's or any Lender's rights hereunder and any such release and/or termination will be deemed to have been conditioned upon such payment having become final and irrevocable.

## XVI.5 Enforceability of Obligations

. No modification, limitation or discharge of the Guaranteed Obligations arising out of or by virtue of any bankruptcy, reorganization or similar proceeding for relief of debtors under federal or state law with respect to any Borrower or any other guarantor or surety for the Guaranteed Obligations (each, an "<u>Other Obligor</u>") will affect, modify, limit or discharge, other than any Other Obligor, any Guarantors' liability in any manner whatsoever and the provisions of this <u>Article XVI</u> will remain and continue in full force and effect and will be enforceable against each Guarantor to the same extent and with the same force and effect as if any such proceeding had not been instituted.

# XVI.6 Guaranty Payable upon Event of Default; Remedies

(a) Upon the occurrence of any Event of Default under this Agreement, (i) all Guaranteed Obligations may, at the option of Agent, and without demand, notice or legal process of any kind, be declared, and immediately shall become, due and payable, (ii) Agent in its discretion may exercise with respect to any Collateral of any Guarantor or any other collateral or security for the Guaranteed Obligations any one or more of the rights and remedies provided a secured party under the Uniform Commercial Code or any other applicable law or at equity (all of which such rights and remedies are hereby deemed incorporated herein and confirmed and ratified by Guarantors as if expressly set forth and granted and agreed to by Guarantors herein); and/or (iii) Agent in its discretion may exercise from time to time any other rights and remedies available to it or any other Lender at law, in equity or otherwise, provided, however, that upon the occurrence of an Event of Default described in Section 10.7 or Section 10.8 hereof, all of the Guaranteed Obligations shall immediately and automatically become due and payable, without notice of any kind.

(b) The Guarantors jointly and severally agree that, as between the Guarantors and the Agent and Lenders, the obligations of the Credit Parties under this Agreement and the Other Documents may be declared to be forthwith due and payable as provided in Section 11.1 (and shall be deemed to have become automatically due and payable in the circumstances provided in Section 11.1) for purposes of this Article XVI (specifically including Section 16.1 hereof), notwithstanding any stay, injunction or other prohibition preventing such declaration (or such obligations from becoming automatically due and payable) as against the Credit Parties and that, in the event of such declaration (or such obligations being deemed to have become automatically due and payable), such obligations (whether or not due and payable by the Credit

### [Different first page setting changed from on in original to off in modified.].

# Parties) shall forthwith become due and payable by the Guarantors for purposes of this Article XVI (specifically including Section 16.1 hereof).

### XVI.7 Waiver of Subrogation

. Until the Guaranteed Obligations are paid in full, each Guarantor waives in favor of Agent and Lenders any and all rights which such Guarantor may have to (a) assert any claim against any Borrower or any other Guarantor based on subrogation, restitution, reimbursement or contribution rights with respect to payments made under the provisions of this <u>Article XVI</u>, and (b) any realization on any property of any Borrower or any other Guarantor, including participation in any marshalling of any Borrower's or any other Guarantor's assets.

# XVI.8 Continuing Guaranty and Suretyship Agreement

. The provisions of this <u>Article XVI</u> shall constitute a continuing guaranty and suretyship obligation of each Guarantor with respect to all Guaranteed Obligations from time to time outstanding, arising or incurred, and shall continue in effect, and Agent and Lenders may continue to act in reliance hereon, until all of the Guaranteed Obligations have been paid in full, and until such time, no Guarantor shall have any right to terminate or revoke the provisions of this <u>Article XVI</u> nor any of the guarantee and surety agreements and other covenants and undertakings provided for herein.

# XVI.9 General Limitation on Guarantee Obligations

. If, in the course of any legal action or proceeding under any applicable law, including any insolvency, bankruptcy, receivership, liquidation, reorganization or other similar proceeding under the laws of any jurisdiction with respect to any Guarantor(s), the obligations of any Guarantor under the provisions of this <u>Article XVI</u> would otherwise be held or determined to be void, voidable, invalid or unenforceable, or subordinated to the claims of any other creditors, on account of the amount of its liability under the provisions of this <u>Article XVI</u>, then, notwithstanding any other provision to the contrary, the amount of such liabilities of such Guarantor under the provisions of this <u>Article XVI</u> shall, without any further action by such Guarantor, Agent or any Lender or any other person, be automatically limited and reduced to the highest amount (after giving effect to the right of contribution established in <u>Section 16.10</u> hereof) that is valid and enforceable and not subordinated to the claims of other creditors as determined in such action or proceeding.

# XVI.10 <u>Right of Contribution</u>

. Each Guarantor hereby agrees that to the extent that any Guarantor shall have paid more than its proportionate share of any payment made hereunder, such Guarantor shall be entitled to seek and receive contribution from and against any other Guarantor hereunder which has not paid its proportionate share of such payment. Each Guarantor's right of contribution shall be subject to the terms and conditions of <u>Section 16.7</u> hereof. The provisions of this <u>Section 16.10</u> shall in no respect limit the obligations and liabilities of any Guarantor to the

[Different first page setting changed from on in original to off in modified.].

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5

Agent or any Lender, and each Guarantor shall remain liable to Agent and each Lender for the full amount guaranteed by such Guarantor hereunder.

# XVI.11 Intercreditor Agreement

. Notwithstanding anything to the contrary contained in this Agreement, the Liens, security interests and rights granted pursuant to this Agreement shall be subject to the terms, provisions and conditions of (and the exercise of any right or remedy by the administrative agent hereunder or thereunder shall be subject to the terms and conditions of), the Intercreditor Agreement. In the event of any conflict between this Agreement and the Intercreditor Agreement, the Intercreditor Agreement shall control, and no right, power, or remedy granted to the Agent hereunder shall be exercised by the Agent, nor shall any direction be given by the Agent in contravention of, the Intercreditor Agreement. With respect to any requirements herein for any Credit Party to deliver originals of certificated Securities, Instruments or similar documents constituting pledged Collateral, such requirements shall be deemed satisfied to the extent the requirements to deliver the same to the ABL Lender in accordance with the Intercreditor Agreement and the ABL Loan Documents are in effect and are satisfied by such Credit Party. To the extent that any covenants, representations or warranties set forth in this Agreement are untrue or incorrect solely as a result of the delivery to, or grant of possession or control to, the ABL Lender in accordance with this <u>Section 16.11</u>, such representation or warranty shall not be deemed to be untrue or incorrect for purposes of this Agreement.

[signature pages follow]

[Different first page setting changed from on in original to off in modified.].

167

<del>159823.01008/130146990v.3</del> 159823.01008/150720893v.5 Each of the parties has signed this Agreement as of the day and year first above written.

<del>DIRECT DIGITAL HOLDINGS, LLO</del>	3
<del></del>	
Name:	
Title:	

159823.01008/130146990v.3

[Signature Page to Term Loan and Security Agreement]

<del>GUARANTORS</del>	ORANGE142 LLC
	<del>By:</del>
	Name:
	<del>Title:</del>
	HUDDLED MASSES LLC
	<del>By:</del>
	Name:
	<del>Title:</del>
	COLOSSUS MEDIA, LLC
	<del>By:</del>
	Name:
	<del>Title:</del>
	UNIVERSAL STANDARDS FOR DIGITAL MARKETING, LLC
	<del>By:</del>
	Name:
	<del>Title:</del>

[Signature Page to Term Loan and Security Agreement]

159823.01008/130146990v.3

AGENT:	LAFAYETTE SQUARE LOAN SERVICING, LLC
	<del>By:</del>
	Name:
	Title:

[Signature Page to Term Loan and Security Agreement]

159823.01008/130146990v.3

# LENDERS:

Addresses for Notices:	MHĐ	CAP FUNDING XI TRUST, as a Lender
MidCap Funding XI Trust	<del>By:</del>	Apollo Capital Management, L.P., its investment manager
MidCap Financial Trust c/o MidCap Financial Services, LLC, as Servicer	<del>By:</del>	Apollo Capital Management GP, LLC, its general partner
7255 Woodmont Ave., Suite 300	<del>By:</del>	-
Bethesda, MD 20814		Name: Maurice Amsellem
Attention: Account Manager for DDH		Title: Authorized Signatory
Telephone: (301) 941-1450		
Email: nohus@mideapfinancial.com		
	MID	CAP FINANCIAL TRUST, as a Lender
	<del>By:</del>	Apollo Capital Management, L.P., its investment manager
	<del>By:</del>	Apollo Capital Management GP, LLC, its general partner
	<del>By:</del>	_
		Name: Maurice Amsellem
		Title: Authorized Signatory

159823.01008/130146990v.3

[Signature Page to Term Loan and Security Agreement]

# **List of Exhibits and Schedules**

Exhibits			
Exhibit A Eligible Impact Services			
Exhibit B Form of Transfer Supplement			
Exhibit C Form of Compliance Certificate			
Exhibit D Impact Certificate			
Exhibit E Key Performance Indicators			
Exhibit F-1 Form of Closing Date Term Note			
Exhibit F-2 Form of Delayed Draw Term Loan Note			
Exhibit G Form of Joinder			
Exhibit H Form of Financial Condition Certificate			
<u>Schedules</u>			
Schedule 1.2(b) Term Loan Commitment Percentages			
Schedule 1.2(c) Commercial Tort Claims			
Schedule 4.5 Chief Executive Office; Equipment and Inventory Locations			
Schedule 4.15(h) Deposit and Investment Accounts			
Schedule 5.1 Jurisdictions of Organization and Good Standing			
Schedule 5.2(a) Equity Interests and Ownership			
Schedule 5.2(b) Subsidiaries			
Schedule 5.8(d) Pension Benefit Plan			
Schedule 5.10 Information Regarding the Credit Parties and their Subsidiaries			
Schedule 5.13 Intellectual Property			
Schedule 5.26 Liabilities, Assets and Activities			
Schedule 5.27 Insurance			
Schedule 7.2 Existing Liens			
Schedule 7.4 Existing Investments			
Schedule 7.8 Existing Indebtedness			
Schedule 7.10 Transactions with Affiliates			

 $\frac{159823.01008/130146990v.3}{1000}$ 

# Schedule 1.2(b)

# Term Loan Commitment Percentage and Delayed Draw Term Loan Commitment Percentage

Lender	<u>Term Loan Commitment</u> Percentage	<u> <del>Delayed Draw Term Loan</del></u> Commitment Percentage	
LAFAYETTE SQUARE USA, INC.	<del>100%</del>	<del>100%</del>	
TOTAL	<del>100%</del>	<del>100%</del>	

[Different first page link-to-previous setting changed from on in original to off in modified.].

159823.01008/130146990v.3

Summary report:				
Litera Compare for Word 11.9.1.1 Document comparison done on 10/15/2024 4:12:32 PM				
Style name: Default Style				
Intelligent Table Comparison: Active				
Original DMS: iw://dmsus.mwllp.dom/ACTIVE/191329029/1				
Modified DMS: iw://dmsus.mwllp.dom/ACTIVE/191329029/12				
Changes:				
Add	575			
<del>Delete</del>	733			
Move From	207			
<u>Move To</u>	207			
Table Insert	4			
Table Delete	5			
Table moves to	0			
Table moves from	0			
Embedded Graphics (Visio, ChemDraw, Images etc.)	0			
Embedded Excel	0			
Format changes	0			
Total Changes:	1731			

#### THIRD AMENDMENT TO CREDIT AGREEMENT

This Third Amendment to Credit Agreement ("*Agreement*"), dated as of October 15, 2024 and effective as of June 30, 2024 (the "*Effective Date*"), is entered into by and among Direct Digital Holdings, Inc., a Delaware corporation ("*DDH Holdings*"), Direct Digital Holdings, LLC, a Texas limited liability company ("*Direct Digital*"), Colossus Media, LLC, a Delaware limited liability company ("*Colossus*"), Huddled Masses LLC, a Delaware limited liability company ("*HM*"), Orange142, LLC, a Delaware limited liability company ("*Borrowers*" and each individually a "*Borrower*"), and East West Bank, a California state bank ("*Lender*").

#### RECITALS:

WHEREAS, Borrowers and Lender entered into that certain Credit Agreement dated as of July 7, 2023, as amended by that certain First Amendment to Credit Agreement dated as of August 17, 2023, and as further amended by that certain Second Amendment to Credit Agreement dated as of November 27, 2023 (the "*Existing Credit Agreement*"; the Existing Credit Agreement as may be further amended, supplemented or otherwise modified from time to time, including by this Agreement, the "*Credit Agreement*");

WHEREAS, the Borrowers and Lender are willing to make certain modifications to the Credit Agreement, subject to the terms and conditions set forth below;

**NOW, THEREFORE**, in consideration of the premises and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Borrowers and Lender hereby agree as follows:

1. **Same Terms**. The terms used in this Agreement shall have the same meanings as provided therefor in the Credit Agreement, unless the context hereof otherwise requires or provides.

### 2. Amendments to Existing Credit Agreement.

(a) The Borrowers and the Lender hereby agree that, upon the satisfaction or waiver of the conditions precedent set forth in Section 6, the Credit Agreement shall be amended to delete the stricken text (indicated textually in the same manner as the following example: stricken text and stricken text) and to add the double-underlined text (indicated textually in the same manner as the following example: double-underlined text and double-underlined text) as set forth in the Credit Agreement attached as <u>Annex A</u> hereto.

(b) Exhibit B to the Credit Agreement, Compliance Certificate, is replaced in its entirety by the Exhibit B attached hereto as <u>Annex B</u> to this Agreement.

3. **Ratification**. Except as expressly provided herein, each Borrower hereby (a) ratifies the Obligations and each of the Loan Documents to which it is a party, and agrees and acknowledges that the Credit Agreement and each of the other Loan Documents to which it is a party shall continue in full force and effect after giving effect to this Agreement; (b) ratifies and confirms that the security instruments executed by each Borrower, as amended hereby, are not released, diminished, impaired, reduced, or otherwise adversely affected by the Credit Agreement and continue to secure the full payment and performance of the Obligations pursuant to their terms; (c) acknowledges the continuing existence and priority of the Liens granted, conveyed, and assigned to Lender, under the security instruments; and (d)

agrees that the Obligations include, without limitation, the Obligations (after giving effect to this Agreement). Except as expressly provided herein, nothing in this Agreement extinguishes, novates or releases any right, claim, Lien, security interest or entitlement of Lender created by or contained in any of such documents nor is any Borrower released from any covenant, warranty or obligation created by or contained therein.

4. **Representations and Warranties**. Each Borrower hereby represents and warrants to Lender that (a) after giving effect to this Agreement, all representations and warranties in the Loan Documents are true and correct in all material respects except (i) to the extent that such representations and warranties specifically refer to an earlier date, in which case they shall be true and correct in all material respects as of such earlier date or (ii) the facts on which any of them were based have been changed by transactions contemplated or permitted by the Credit Agreement; and (b) after giving effect to this Agreement, no Event of Default exists.

5. **Other Agreements**. Each Borrower (a) agrees to perform such acts and duly authorize, execute, acknowledge, deliver, file, and record such additional pledges, and other agreements, documents, instruments, and certificates as Lender may reasonably deem necessary or appropriate in order to preserve and protect the Collateral granted by such Borrower pursuant to the security instruments executed by such Borrower; and (b) represents and warrants to Lender that such liability and obligation may reasonably be expected to directly or indirectly benefit each Borrower.

6. **Conditions to Effectiveness**. The transactions contemplated by this Agreement shall be deemed to be effective as of the Effective Date, when the following have been satisfied in a manner satisfactory to Lender:

(a) all representations and warranties set forth in this Agreement are true and correct in all material respects as set forth in *Section 4* above;

(b) Lender receives a fully executed copy of this Agreement;

(c) Lender receives payment of the reasonable and documented out-of-pocket fees and expenses of Lender's counsel incurred in connection with this Agreement in immediately available funds to the extent invoiced on or prior to the date hereof;

(d) the representations and warranties set forth in the Loan Documents (after giving effect to this Agreement) are true and correct in all material respects (without duplication of any materiality standards set forth therein), except to the extent that such representations and warranties specifically refer to an earlier date, in which case they shall be true and correct in all material respects as of such earlier date or the facts on which any of them were based have been changed by transactions contemplated or permitted by the Credit Agreement;

(e) Lender receives a Certificate certified by a Responsible Officer from Borrowers, in form and substance reasonably acceptable to Lender, attaching resolutions from the appropriate governing body of Borrowers which authorize the execution, delivery and performance of this Agreement and the transactions contemplated hereby, and such other items and documents as Lender shall reasonably request;

(f) Lender receives a copy of an amendment to the Term Loan Agreement reasonably acceptable to Lender pursuant to which, among other things, Term Loan Lender consents to the form and substance of this Agreement in accordance with Section 6.1 of the Intercreditor Agreement;

(g) Lender receives Eligible Account Reports for July 2024 to August 2024, along with corresponding past-due accounts receivables and accounts payable aging reports.

- (h) after giving effect to this Agreement, no Default or Event of Default exists under the Credit Agreement; and
- (i) Lender receives, in immediately available funds, a principal payment on the Notes equal to \$1,000,000.

7. **Counterparts**. For the convenience of the parties, this Agreement may be executed in multiple counterparts, each of which for all purposes shall be deemed to be an original, and all such counterparts shall together constitute but one and the same agreement. Delivery of an executed counterpart of a signature page of this Agreement by telecopy, e-mail, facsimile transmission, electronic mail in "*portable document format*" ("*.pdf*") form or other electronic means intended to preserve the original graphic and pictorial appearance of the item being sent shall be effective as a delivery of a manually executed counterpart of this Agreement.

8. **References to the Credit Agreement**. Upon the effectiveness of this Agreement, (a) each reference in the Credit Agreement to "*this Agreement*", "*hereof*", "*hereof*", "*herein*", or words of like import shall mean and be a reference to the Credit Agreement, as amended by this Agreement, and (b) each reference to the Credit Agreement in any other document, instrument or agreement executed and/or delivered in connection with the Credit Agreement shall mean and be a reference to the Credit Agreement, as amended by this Agreement shall mean and be a reference to the Credit Agreement, as amended by this Agreement shall mean and be a reference to the Credit Agreement, as amended by this Agreement.

9. **Effect**. This Agreement is one of the Loan Documents. The modifications set forth herein are limited precisely as written and shall not be deemed (a) to be a consent under or a waiver of or an amendment to any other term or condition in the Credit Agreement, or (b) to prejudice any right or rights which the Lender now has or may have in the future under or in connection with the Credit Agreement, as amended hereby, or any of the other documents referred to herein or therein.

10. **RELEASE**. EACH OBLIGATED PARTY HEREBY ACKNOWLEDGES THAT IT HAS NO DEFENSE, COUNTERCLAIM, OFFSET, CROSS COMPLAINT, CLAIM OR DEMAND OF ANY KIND OR NATURE WHATSOEVER THAT CAN BE ASSERTED TO REDUCE OR ELIMINATE ALL OR ANY PART OF ITS LIABILITY TO REPAY ANY OF THE INDEBTEDNESS EVIDENCED BY THE CREDIT AGREEMENT AND ANY OTHER LOAN DOCUMENT OR TO SEEK AFFIRMATIVE RELIEF OR DAMAGES OF ANY KIND OR NATURE FROM LENDER. EACH OBLIGATED PARTY HEREBY VOLUNTARILY AND KNOWINGLY RELEASES AND FOREVER DISCHARGES LENDER AND ITS PREDECESSORS, AGENTS, EMPLOYEES, SUCCESSORS AND ASSIGNS, FROM ALL POSSIBLE CLAIMS, DEMANDS, ACTIONS, CAUSES OF ACTION, DAMAGES, COSTS, EXPENSES, AND LIABILITIES WHATSOEVER, KNOWN OR UNKNOWN, ANTICIPATED OR UNANTICIPATED, SUSPECTED OR UNSUSPECTED, FIXED, CONTINGENT, OR CONDITIONAL, AT LAW OR IN EQUITY (INCLUDING WITHOUT LIMITATION, CLAIMS OF FRAUD, DURESS, MISTAKE, TORTIOUS INTERFERENCE, USURY, BREACH OF FIDUCIARY DUTY, BREACH OF DUTY OF FAIR DEALING, BREACH OF CONFIDENCE, BREACH OF FUNDING COMMITMENT, UNDUE INFLUENCE, NEGLIGENCE OR FRAUD IN RATES AND METHODS USED TO COMPUTE INTEREST, DECEPTIVE TRADE PRACTICE OR THE RACKETEER INFLUENCED AND CORRUPT ORGANIZATIONS ACT), ORIGINATING IN WHOLE OR IN PART ON OR BEFORE THE DATE THIS AGREEMENT IS EXECUTED, WHICH EACH OBLIGATED PARTY MAY NOW OR HEREAFTER HAVE AGAINST LENDER AND ITS

PREDECESSORS, AGENTS, EMPLOYEES, SUCCESSORS AND ASSIGNS, IF ANY, AND IRRESPECTIVE OF WHETHER ANY SUCH CLAIMS ARISE OUT OF CONTRACT, TORT, VIOLATION OF LAW OR REGULATIONS, OR OTHERWISE, AND ARISING FROM ANY CREDIT ACCOMMODATIONS FROM, OR ANY CREDIT EXTENSION MADE UNDER, THE LOAN DOCUMENTS, INCLUDING, WITHOUT LIMITATION, ANY CONTRACTING FOR, CHARGING, TAKING, RESERVING, COLLECTING OR RECEIVING INTEREST IN EXCESS OF THE HIGHEST LAWFUL RATE APPLICABLE, THE EXERCISE OF ANY RIGHTS AND REMEDIES UNDER THE LOAN DOCUMENTS, AND NEGOTIATION FOR AND EXECUTION OF THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT.

11. **ENTIRE AGREEMENT; GOVERNING LAW**. THIS AGREEMENT CONSTITUTES THE ENTIRE AGREEMENT BETWEEN THE PARTIES HERETO WITH RESPECT TO THE SUBJECT HEREOF. FURTHERMORE, IN THIS REGARD, THIS AGREEMENT AND THE OTHER WRITTEN LOAN DOCUMENTS REPRESENT, COLLECTIVELY, THE FINAL AGREEMENT AMONG THE PARTIES THERETO AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF SUCH PARTIES.

### THERE ARE NO UNWRITTEN ORAL AGREEMENTS AMONG SUCH PARTIES.

THIS AGREEMENT AND ANY CONTROVERSY, DISPUTE, CLAIM OR CAUSE OF ACTION ARISING OUT OF OR RELATING TO THIS AGREEMENT, ANY BREACH HEREOF, OR THE TRANSACTIONS CONTEMPLATED HEREBY, SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF TEXAS; *PROVIDED THAT* LENDER SHALL RETAIN ALL RIGHTS UNDER FEDERAL LAW.

### [REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, this Agreement is deemed executed effective as of the date first above written.

BORROWERS:

# DIRECT DIGITAL HOLDINGS, INC.

By: <u>/s/ Keith W. Smith</u> Name: Keith W. Smith Title: President

## DIRECT DIGITAL HOLDINGS, LLC

By: <u>/s/ Keith W. Smith</u> Name: Keith W. Smith Title: President

# **COLOSSUS MEDIA, LLC**

By: <u>/s/ Keith W. Smith</u> Name: Keith W. Smith Title: President

# HUDDLED MASSES LLC

By: <u>/s/ Keith W. Smith</u> Name: Keith W. Smith Title: President

### ORANGE142, LLC

By: <u>/s/ Keith W. Smith</u> Name: Keith W. Smith Title: President

> Signature Page to Third Amendment to Credit Agreement

# EAST WEST BANK,

a California state bank

By: <u>/s/ Hamilton LaRoe</u> Name: Hamilton LaRoe

Title: Senior Vice President

Signature Page to Third Amendment to Credit Agreement

### **CREDIT AGREEMENT**

by and among

# DIRECT DIGITAL HOLDINGS, INC.

# DIRECT DIGITAL HOLDINGS, LLC

# COLOSSUS MEDIA, LLC

# HUDDLED MASSES LLC

# ORANGE142, LLC

and

EAST WEST BANK

Dated as of July 7, 2023

4864-5069-8851 v.9

#### TABLE OF CONTENTS

Page ARTICLE I. DEFINITIONS 1 1.01 Definitions 1 1.02 Accounting Matters 1819 1.03 Other Definitional Provisions 1819 ARTICLE II. ADVANCES 1820 2.01 Advances 1820 2.02 General Provisions Regarding Interest, Etc 1921 2.03 Unused Facility Fee 2122 2.04 Use of Proceeds 2122 2.05 Late Charges 2122 2.06 Funding Loss 2122 2.07 Uncommitted Increase in Commitment 2123 2.08 Letters of Credit 23 ARTICLE III. PAYMENTS 2224 3.01 Method of Payment 2224 3.02 Prepayments 2224 ARTICLE IV. SECURITY 2325 4.01 Collateral 2325 4.02 Setoff 2325 ARTICLE V. CONDITIONS PRECEDENT 2325 5.01 Initial Extension of Credit 2325 5.02 All Extensions of Credit 2527 ARTICLE VI. REPRESENTATIONS AND WARRANTIES 2527 6.01 Corporate Existence 2527 6.02 Financial Statements, Etc 2628 6.03 Action; No Breach 2628 6.04 Operation of Business 2628 6.05 Litigation and Judgments 2628 6.06 Rights in Properties; Liens 2628 6.07 Enforceability 2729 6.08 Approvals 2729 Taxes 2729 6.09 6.10 Use of Proceeds; Margin Securities 2729 6.11 ERISA <del>27</del>29 6.12 Disclosure 2729 6.13 Subsidiaries, Ventures, Etc 2830 6.14 Agreements 2830 6.15 Compliance with Laws 2830 6.16 Regulated Entities 2830 6.17 Environmental Matters 2830 6.18 Intellectual Property 2931 6.19 Foreign Assets Control Regulations and Anti-Money Laundering 2931

4<del>864-5069-8851 v.9</del> i

6.20 Patriot Act 3032

6.21 Solvency <u>3032</u> 6.22 Anti-Corruption Laws 3032 6.23 Beneficial Ownership Regulation 3032 ARTICLE VII. AFFIRMATIVE COVENANTS 3032 7.01 Reporting Requirements <u>3032</u> 7.02 Maintenance of Existence; Conduct of Business 3234 7.03 Maintenance of Properties 3234 7.04 Taxes and Claims 3234 7.05 Insurance <u>3234</u> 7.06 Inspection Rights 3335 7.07 Keeping Books and Records 3335 7.08 Compliance with Laws 3335 7.09 Compliance with Agreements 3335 7.10 Further Assurances 3335 7.11 ERISA 3435 7.12 Depository Relationship 3435 7.13 Subsidiaries 3436 7.14 Keepwell <u>3436</u> 7.15 Deposit Account Control Agreements 3436 ARTICLE VIII. NEGATIVE COVENANTS 3536 8.01 Debt <u>3536</u> 8.02 Limitation on Liens 3537 8.03 Mergers, Etc 3537 8.04 Restricted Payments 3537 8.05 Loans and Investments 3638 8.06 Limitation on Issuance of Equity 3638 8.07 Transactions with Affiliates 3638 8.08 Disposition of Assets 3639 8.09 Sale and Leaseback 3639 8.10 Nature of Business 3639 8.11 Environmental Protection 3639

- 8.12 Accounting <u>3739</u>
- 8.13 No Negative Pledge 3739
- 8.14 Subsidiaries 3739
- 8.15 Hedge Agreements 3739
- 8.16 OFAC 3739
- 8.17 Payments under Term Loan Agreement 3740

# ARTICLE IX. FINANCIAL COVENANTS 3740

- 9.01 Fixed Charge Coverage Ratio 3740
- 9.02 Total Funded Debt to EBITDA Ratio 3740
- 9.03 Liquid Assets: Borrowers 3840
- 9.04 Minimum EBITDA 40

# ARTICLE X. DEFAULT 3841

- 10.01 Events of Default 3841
- 10.02 Remedies Upon Default 4043
- 10.03 Performance by Lender 4043
- <u>10.04 Equity 43</u>

4864-5069-8851 v.9 ii

# <u>10.05</u> <u>Cash Collateral</u> <u>43</u>

ARTICLE XI. MISCELLANEOUS 4043 11.01 Expenses 4043 11.02 INDEMNIFICATION 4144 11.03 Limitation of Liability 4245 11.04 No Duty 4246 11.05 Lender Not Fiduciary 4246 11.06 Equitable Relief  $\frac{4246}{42}$ No Waiver; Cumulative Remedies 4346 11.07 11.08 Successors and Assigns 4346 11.09 Survival 4346 11.10 ENTIRE AGREEMENT; AMENDMENT 4346 11.11 Notices 4346 11.12 Governing Law; Venue; Service of Process 4347 11.13 Counterparts 4447 11.14 Severability 4447 11.15 Headings 4448 11.16 Participations, Etc 4448 11.17 Construction <u>4548</u> 11.18 Independence of Covenants 4548 11.19 WAIVER OF JURY TRIAL 4548 11.20 Additional Interest Provision 4548 11.21 Ceiling Election 4649 11.22 USA Patriot Act Notice 4649 11.23 Confidentiality: 4650

11.24 Intercreditor Legend. 4750

# SCHEDULES

6.13 Subsidiaries, Ventures, Etc.

- 6.18 Intellectual Property
- 8.01 Existing Debt
- 8.02 Existing Liens
- 8.05 Existing Investments

## EXHIBITS

A. A. Eligible Account Report 1.01

- B. B. Compliance Certificate 1.01
- C. Revolving Credit Note 2.01

#### **CREDIT AGREEMENT**

THIS CREDIT AGREEMENT (this "*Agreement*"), dated as of July 7, 2023 (the "*Closing Date*") is by and among Direct Digital Holdings, Inc., a Delaware corporation ("*DDH Holdings*"), Direct Digital Holdings, LLC, a Texas limited liability company ("*Direct Digital*"), Colossus Media, LLC, a Delaware limited liability company ("*Colossus*"), Huddled Masses LLC, a Delaware limited liability company ("*Colossus*"), Huddled Masses LLC, a Delaware limited liability company ("*HM*"), Orange142, LLC, a Delaware limited liability company ("*Colossus*"), Huddled Masses LLC, a Delaware limited liability company ("*Colossus*"), Huddled Masses LLC, a Delaware limited liability company ("*HM*"), Orange142, LLC, a Delaware limited liability company ("*Colossus*"), and East West Bank, a California state bank ("*Lender*").

## **RECITALS:**

Borrowers have requested that Lender extend credit to Borrowers as described in this Agreement. Lender is willing to make such credit available to Borrowers upon and subject to the provisions, terms and conditions hereinafter set forth.

NOW THEREFORE, in consideration of the premises and the mutual covenants herein contained, the parties hereto agree as follows:

### Article I. DEFINITIONS

I.1 **Definitions**. As used in this Agreement, all exhibits, appendices and schedules hereto and in any note, certificate, report or other Loan Documents made or delivered pursuant to this Agreement, the following terms will have the meanings given such terms in this *Article I* or in the provision, section or recital referred to below:

"Acquisition" means the acquisition by any Person of (a) a majority of the Equity Interests of another Person, (b) all or substantially all of the assets of another Person or (c) all or substantially all of a business unit or line of business of another Person, in each case (i) whether or not involving a merger or consolidation with such other Person, and (ii) whether in one transaction or a series of related transactions.

"Advance" means an advance by Lender to Borrowers pursuant to Article II.

"Advance Request Form" means a certificate, in a form approved by Lender, properly completed and signed by Borrowers requesting a Revolving Credit Advance.

"*Affiliate*" means, as to any Person, any other Person (a) that directly or indirectly, through one or more intermediaries, Controls or is Controlled by, or is under common Control with, such Person; (b) that directly or indirectly beneficially owns or holds five percent (5%) or more of any class of voting stock of such Person; or (c) five percent (5%) or more of the voting stock of which is directly or indirectly beneficially owned or held by the Person in question; provided, however, in no event shall Lender be deemed an Affiliate of Borrowers, any of their Affiliates or the Subsidiaries.

"Agreement" has the meaning set forth in the Introductory Paragraph hereto, as the same may, from time to time, be amended, modified, restated, renewed, waived, supplemented, or otherwise changed, and includes all schedules, exhibits and appendices attached or otherwise identified therewith.

4864-5069-8851 v.9

"Beneficial Ownership Certification" means a certification regarding beneficial ownership required by the Beneficial Ownership Regulation, which certification shall be substantially in form and substance satisfactory to Lender.

### "Beneficial Ownership Regulation" means 31 C.F.R. §1010.230.

"Borrowers" means the Persons identified as such in the Introductory Paragraph hereof, and their successors and assigns, and "Borrower" means any one of the Borrowers.

"Business Day" means any day other than a Saturday or a Sunday or any day on which commercial banks in Los Angeles, California, are authorized or required to close.

"*Capital Expenditure*" shall mean any expenditure by a Person for (a) an asset which will be used in a year or years subsequent to the year in which the expenditure is made and which asset is properly classified in relevant financial statements of such Person as equipment, real property, a fixed asset or a similar type of capitalized asset in accordance with GAAP or (b) an asset relating to or acquired in connection with an acquired business, and any and all acquisition costs related to (a) or (b) above.

"Cash and Cash Equivalents" shall mean, with respect to any Person, an unrestricted or unencumbered (A) cash and (B) any of the following: (x) marketable direct obligations issued or unconditionally guaranteed by the United States Government or issued by an agency thereof and backed by the full faith and credit of the United States: (v) marketable direct obligations issued by any state of the United States of America or any political subdivision of any such state or any public instrumentality thereof which, at the time of acquisition, has one of the two highest ratings obtainable from any two of S&P Global Ratings, Moody's Investors Service. Inc. or Fitch Investors (or, if at any time no two of the foregoing shall be rating such obligations, then from such other nationally recognized rating services as may be acceptable to Lender) and is not listed for possible down-grade in any publication of any of the foregoing rating services; (z) domestic certificates of deposit or domestic time deposits or repurchase agreements issued by any commercial bank organized under the laws of the United States of America or any state thereof or the District of Columbia having combined capital and surplus of not less than \$500,000,000,000, which commercial bank has a rating of at least either "AA" or such comparable rating from S&P Global Ratings or Moody's Investors Service, Inc., respectively; (aa) money market funds having assets under management in excess of \$1,000,000,000; (bb) any unrestricted stock, shares, certificates, bonds, debentures, notes or other instrument which constitutes a "security" under the Security Act of 1993, which are freely tradable on any nationally recognized securities exchange and are not otherwise encumbered by such transferee; (cc) lines of credit; (dd) unfunded, but unconditionally committed and unencumbered capital commitments of the direct or indirect members or limited partners of such Person (or, to the extent encumbered by a subscription facility (or the like), such capital commitments, less any amounts drawn and outstanding under a subscription facility (or the like)) or such other assets or properties as Lender may (in its sole discretion) deem acceptable as evidenced by Lender's written confirmation, excluding any and all retirement accounts and deferred profit-sharing accounts. To constitute "Cash and Cash Equivalents" the foregoing items described in (A) and (B) above must be: (i) owned solely in the name of such Person or its wholly owned direct or indirect subsidiaries, as set forth on such Person's balance sheet (and not jointly with any other person or entity) in a non-margin account identified as being owned by solely in the name of such Person or its wholly owned direct or indirect subsidiaries, as set forth on such Person's balance sheet; and (ii) free and clear of any lien, security interest, assignment, right of setoff or other encumbrance of any kind except Permitted Liens hereof.

<del>2</del> 4864-<del>5069-</del> 8851 v.9 "*Cash Flow Available for Debt Service*" means, for Borrowers and the Subsidiaries, on a Consolidated Basis, for any period, (a) EBITDA, *minus* (b) cash taxes paid or payable or Permitted Tax Distributions made during such period, *minus* (c) all Capital Expenditures not financed with Debt for borrowed money permitted hereunder during such period, *minus* (d) distributions.

### "Change in Control" shall mean:

(a) any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act) other than Direct Digital Management, LLC or its Control Affiliates shall become, or obtain rights (whether by means of warrants, options or otherwise) to become, the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Exchange Act), directly or indirectly, of forty percent (40%) or more of the voting power of the total outstanding Equity Interests of DDH Holdings entitled to vote for members of the board of directors or equivalent governing body of DDH Holdings on a fully-diluted basis (and taking into account all such Equity Interests that such "person" or "group" has the right to acquire pursuant to any option right); or

(b) during any period of 12 consecutive months, a majority of the members of the board of directors or other equivalent governing body of DDH Holdings cease to be composed of individuals (i) who were members of that board or equivalent governing body on the first (1<sup>st</sup>) day of such period, (ii) whose election or nomination for election to that board or equivalent governing body was approved by individuals referred to in clause (i) above constituting at the time of such election or nomination for election at least a majority of that board or equivalent governing body, in clauses (i) and (ii) above constituting at the time of such election or nomination at least a majority of that board or equivalent governing body, in each case with respect to clauses (i), (ii) and (iii), as a result of hostile proxy contest; or

(c) the occurrence of any event (whether in one or more transactions) which results in DDH Holdings, Direct Digital Management, LLC and its Control Affiliates, collectively, failing to own, directly or indirectly, legally and beneficially (free and clear of Liens, other than Permitted Liens), 100% of the Equity Interests (on a fully diluted basis) of Direct Digital;

(d) except as otherwise permitted by Section 8.03 or Section 8.08, the occurrence of any event (whether in one or more transactions) which results in Direct Digital failing to own, directly or indirectly, legally and beneficially (free and clear of Liens, other than Permitted Liens), 100% of the Equity Interests (on a fully diluted basis) of each of its Subsidiaries.

"Code" means the Internal Revenue Code of 1986, as amended, and the regulations promulgated and rulings issued thereunder.

"Collateral" means all property and assets granted as collateral security for the Obligations, whether real or personal property, whether granted directly or indirectly, whether granted now or in the future, and whether granted in the form of a security interest, mortgage, collateral mortgage, deed of trust, assignment, pledge, crop pledge, chattel mortgage, collateral chattel mortgage, chattel trust, factor's lien, equipment trust, conditional sale, trust receipt, lien, charge, lien or title retention contract, lease or consignment intended as a security device, or any other security or lien interest or Hedge Agreement whatsoever, whether created by law, contract, or otherwise.

<del>3</del> 4<del>864-</del> <del>5069-</del> <del>8851 v.!</del> "*Commitment*" means the obligation of Lender to make Revolving Credit Advances pursuant to *Section 2.01* in an aggregate principal amount up to but not exceeding Ten Million Dollars (\$10,000,000), subject to termination pursuant to *Section 10.02*.

"Commitment Fee" means Fifty Thousand and No/100 Dollars (\$50,000).

"Commodity Exchange Act" means the Commodity Exchange Act (7 U.S.C. §1 et seq.), and any successor statute.

"Compliance Certificate" means a certificate, substantially in the form of Exhibit B attached hereto, prepared by and executed by an officer of Borrowers.

"Conforming Changes" has the meaning specified in Section 2.02(b).

"*Consolidated Basis*" shall mean, with respect to DDH Holdings and its Subsidiaries as the context may require, the consolidation in accordance with GAAP of the accounts and other items of DDH Holdings and its Subsidiaries, it being understood that any reference to Borrowers and their Subsidiaries in any financial definitions or calculations herein shall be deemed to refer to DDH Holdings and its Subsidiaries.

"Constituent Documents" means (i) in the case of a corporation, its articles or certificate of incorporation and bylaws; (ii) in the case of a general partnership, its partnership agreement and certificate of formation or other instrument filed in connection with its formation; (iii) in the case of a limited partnership, its certificate of limited partnership agreement; (iv) in the case of a trust, its trust agreement; (v) in the case of a joint venture, its joint venture agreement; (vi) in the case of a limited liability company, its articles of organization and operating agreement or regulations; and (vii) in the case of any other entity, its organizational and governance documents and agreements.

"*Control*" the possession, directly or indirectly, of the power to direct or cause direction of the management and policies of a Person, whether through the ownership of voting securities, by contract, or otherwise. "*Controlling*" and "*Controlled*" have meanings correlative thereto.

"*Control Affiliate*" shall mean, with respect to any Person, each other Person that owns or controls directly or indirectly the Person, any Person that controls or is controlled by or is under common control with the Person, and each of that Person's senior executive officers, directors, partners and, for any Person that is a limited liability company, that Person's managers and members.

"Debt" means as to any Person at any time (without duplication): (a) all obligations of such Person for borrowed money, (b) all obligations of such Person evidenced by bonds, notes, debentures, or other similar instruments, (c) all obligations of such Person to pay the deferred purchase price of property or services, except trade accounts payable of such Person arising in the ordinary course of business that are not past due by more than ninety (90) days, (d) all Finance Lease Obligations of such Person, (e) all Debt or other obligations of others Guaranteed by such Person, (f) all obligations secured by a Lien existing on property owned by such Person, whether or not the obligations secured thereby have been assumed by such Person or are non-recourse to the credit of such Person, (g) any other obligation for borrowed money or other financial accommodations which in accordance with GAAP would be shown as a liability on the balance sheet of such Person, (h) any repurchase obligation or liability of a Person with respect to accounts, chattel paper or notes receivable sold by such Person, (i) any liability under a sale and leaseback transaction that is not a Finance Lease Obligation, (j) any obligation under any so-called "synthetic leases", (k) any obligation arising with respect to any other transaction that is the functional

<del>4</del> <del>4864-</del> <del>5069-</del> <del>8851 v.9</del> equivalent of borrowing but which does not constitute a liability on the balance sheets of a Person, (l) all reimbursement obligations of such Person (whether contingent or otherwise) in respect of letters of credit, bankers' acceptances, surety or other bonds and similar instruments, (m) all liabilities of such Person in respect of unfunded vested benefits under any Plan, and (n) all obligations of such Person in respect of Disqualified Equity Interests. The term "*Debt*" shall exclude obligations under an operating lease.

"Debt Service" means, for any Person as of any date of determination, the sum of (a) all regularly scheduled principal payments that are paid in cash with respect of all Debt, (b) all interest expense that is paid in cash with respect to all Debt, (c) payments with respect to Finance Lease Obligations, and (d) all other regularly scheduled payments, in each case, for such Person during the trailing twelve-month period ending on the date of determination.

"Default" means an Event of Default or the occurrence of an event or condition which with notice or lapse of time or both would become an Event of Default.

"Default Interest Rate" means a rate per annum equal to the Loan Rate plus five percent (5%), but in no event in excess of the Maximum Lawful Rate.

"Delaware LLC" means any limited liability company organized or formed under the laws of the State of Delaware.

"Delaware LLC Division" means the statutory division of any Delaware LLC into two or more Delaware LLCs pursuant to Section 18-217 of the Delaware Limited Liability Company Act.

"Designated Jurisdiction" means any country or territory to the extent that such country or territory itself is the subject of any Sanction.

"Direct Digital Management" means Direct Digital Management, LLC, a Delaware limited liability company.

"Disqualified Equity Interest" means any Equity Interest that, by its terms (or the terms of any security or other Equity Interests into which it is convertible or for which it is exchangeable), or upon the happening of any event or condition (a) matures or is mandatorily redeemable (other than solely for Equity Interests that are not Disqualified Equity Interests), pursuant to a sinking fund obligation or otherwise (except as a result of a change of control or asset sale so long as any rights of the holders thereof upon the occurrence of a change of control or asset sale event shall be subject to the prior repayment in full of the Advances and all other Obligations that are accrued and payable and the termination of the Commitment), (b) is redeemable at the option of the holder thereof, in whole or in part, (c) provides for scheduled payments of dividends in cash, or (d) is or becomes convertible into or exchangeable for Debt or any other Equity Interests that would constitute Disqualified Equity Interests, in each case, prior to the date that is ninety one (91) days after the Maturity Date; provided that if such Equity Interests are issued pursuant to a plan for the benefit of employees of any Obligated Party or any Subsidiary of an Obligated Party or by any such plan to such employees, such Equity Interests shall not constitute Disqualified Equity Interests solely because they may be required to be repurchased by any Obligated Party or any of its Subsidiaries in order to satisfy applicable statutory or regulatory obligations or as a result of such employee's termination, death or disability.

"Dollars" and "\$" mean lawful money of the United States of America.

<del>5</del> 4<del>864-</del> <del>5069-</del> <del>8851 v.9</del> "Domestic Subsidiary" means any Subsidiary that is organized under the laws of any political subdivision of the United States of America.

"EBITDA" means, for any period, for DDH Holdings and its Subsidiaries on a Consolidated Basis, an amount equal to (a) Net Income for such period plus, (b) to the extent deducted in determining such Net Income, the sum, without duplication, of (i) Interest Expense during such period, (ii) all federal, state, local and/or foreign income taxes payable by DDH Holdings and its Subsidiaries during such period, (iii) depreciation expenses of DDH Holdings and its Subsidiaries during such period, (iv) amortization expenses of DDH Holdings and its Subsidiaries during such period, (v) any non-cash loss or expense resulting from any impairment charge or asset write-off or write-down related to intangible assets, long-lived assets and other assets that occurs during such period, (vi) one-time loss associated with debt refinancing, (vii) without duplication, non-recurring actual, documented legal or consulting expenses or retention bonuses paid in an amount up to \$500,000750,000 during any 12 month period, (viii) any cash payments (including all premiums) made with respect to the Key Executive Policies required pursuant to Section 4.21 of the Term Loan Agreement, and (ivix) reasonable and documented one-time, non-recurring severance and retention expenses in an aggregate amount not to exceed \$750,000 for employees and management of the Borrowers, in each case to the extent paid on or before June 30, 2024, and (x) with respect to the fiscal year ended on December 31, 2023, accounting, audit, auditor and/or legal costs and expenses associated with any audits, regulatory compliance, investigations, remediation of audit financings, and the implementation of required changes as a result of audit outcomes or the change in auditor due to the resignation and replacement of the Borrowers' Accountants, provided that such add-backs shall not exceed \$1,500,000 with respect to such fiscal year, (xi) any amounts paid out to employees in stock, including the 2023 bonus accrual which the compensation committee of the board of directors of DDH Holdings approved to pay out in stock in February 2024. (xii) the gross amount of the profit on the disputed shortpay which impacted December 2023 income in the amount up to a total of \$1,781,000 and (xiii) any amounts paid to publishers associated with the disputed short pay when recorded as expense in the financial statements up to a total of \$8,800,000 minus (c) any extraordinary, non-recurring and/or non-cash gains or income during such period as reported in the monthly and annual financials of DDH Holdings and its Subsidiaries.

*"Eligible Account Report"* means, as of any date of preparation, a certificate setting forth the Eligible Accounts (in a form acceptable to Lender in substantially the form of *Exhibit A* attached hereto) prepared by and certified by a Responsible Officer of Borrower.

"*Eligible Accounts*" means, at any time, all accounts receivable of Borrowers and their Subsidiaries that are Guarantors created in the ordinary course of business that are acceptable to Lender in its Permitted Discretion and satisfy the following conditions:

(a) The account complies with all applicable laws, rules, and regulations, including, without limitation, usury laws, the Federal Truth in Lending Act, and Regulation Z of the Board of Governors of the Federal Reserve System;

(e) The account has not been outstanding for more than ninety (90) days past the original date of invoice;

(f) The account does not represent a commission and the account was created in connection with (i) the sale of goods by a Borrower or a Subsidiary in the ordinary course of business and such sale has been consummated and such goods have been shipped and delivered and received by the account debtor, or (ii) the performance of services by a Borrower or a

<del>6</del> 4864-5069-8851 v.9 Subsidiary in the ordinary course of business and such services have been completed and accepted by the account debtor;

(g) The account arises from an enforceable contract, the performance of which has been completed by a Borrower or a Subsidiary;

(h) The account does not arise from the sale of any good that is on a bill-and-hold, guaranteed sale, sale-or-return, sale on approval, consignment, or any other repurchase or return basis;

(i) A Borrower or a Subsidiary has good and indefeasible title to the account and the account is not subject to any Lien except Liens in favor of Lender and Term Loan Lender;

(j) The account does not arise out of a contract with or order from, an account debtor that, by its terms, prohibits or makes void or unenforceable the grant of a security interest by a Borrower or a Subsidiary to Lender in and to such account;

(k) The account is not subject to any setoff, counterclaim, defense, dispute, recoupment, or adjustment other than normal discounts for prompt payment;

(1) The account debtor is not insolvent or the subject of any bankruptcy or insolvency proceeding and has not made an assignment for the benefit of creditors, suspended normal business operations, dissolved, liquidated, terminated its existence, ceased to pay its debts as they become due, or suffered a receiver or trustee to be appointed for any of its assets or affairs;

# (m) The account is not evidenced by chattel paper or an instrument;

(n) No default exists under the account by any party thereto beyond any applicable notice and cure period;

(o) The account debtor has not returned or refused to retain, or otherwise notified a Borrower or a Subsidiary of any dispute concerning, or claimed nonconformity of, any of the goods from the sale of which the account arose;

(p) The account is not owed by an Affiliate, employee, officer, director or shareholder of any Borrower or a Subsidiary;

## (q) The account is payable in Dollars by the account debtor;

(r) The account is not owed by an account debtor whose accounts Lender in its Permitted Discretion has chosen to exclude from Eligible Accounts;

(s) Except with respect to BidSwitch GmbH, the account shall be ineligible if the account debtor is domiciled in any country other than the United States of America and/or Canada;

(t) The account shall be ineligible if more than twenty-five percent (25%) of the aggregate balances then outstanding on accounts owed by such account debtor and its Affiliates

<del>7</del> <del>4864-</del> <del>5069-</del> <del>8851 v.9</del> to any Borrower or a Subsidiary are more than ninety (90) days past the dates of their original invoices;

(u) The account shall be ineligible if the account debtor is the United States of America or any department, agency, or instrumentality thereof, and the Federal Assignment of Claims Act of 1940, as amended, shall not have been complied with;

(v) Except with respect to BidSwitch GmbH, the account shall be ineligible to the extent the aggregate of all accounts owed by the account debtor and its Affiliates to which the account relates exceeds twenty-five percent (25%) of all accounts owed by all of Borrower's and its Subsidiaries' account debtors; and

(w) The Account is otherwise acceptable in the Permitted Discretion of Lender; provided that Lender shall have the right to create and adjust eligibility standards and related reserves from time to time in its Permitted Discretion.

The amount of the Eligible Accounts owed by an account debtor to Borrower or a Subsidiary shall be reduced by the amount of all "contra accounts" and other obligations owed by Borrower or a Subsidiary to such account debtor.

"Environmental Laws" means any and all federal, state, and local laws, regulations, judicial decisions, orders, decrees, plans, rules, permits, licenses, and other governmental restrictions and requirements pertaining to health, safety, or the environment, including, without limitation, the Comprehensive Environmental Response, Compensation and Liability Act of 1980, 42 U.S.C. §9601 *et seq.*, the Resource Conservation and Recovery Act of 1976, 42 U.S.C. §6901 *et seq.*, the Clean Air Act, 42 U.S.C. §7401 *et seq.*, the Clean Water Act, 33 U.S.C. §1251 *et seq.*, and the Toxic Substances Control Act, 15 U.S.C. §2601 *et seq.*, as the same may be amended or supplemented from time to time.

"Environmental Liabilities" means, as to any Person, all liabilities, obligations, responsibilities, Remedial Actions, losses, damages, punitive damages, consequential damages, treble damages, costs, and expenses, (including, without limitation, all reasonable fees, disbursements and expenses of counsel, expert and consulting fees and costs of investigation and feasibility studies), fines, penalties, sanctions, and interest incurred as a result of any claim or demand, by any Person, whether based in contract, tort, implied or express warranty, strict liability, criminal or civil statute, including any Environmental Law, permit, order or agreement with any Governmental Authority or other Person, arising from environmental, health or safety conditions or the Release or threatened Release of a Hazardous Material into the environment, resulting from the past, present, or future operations of such Person or its Affiliates.

"*Equity Cure*" means the cash contributions made to Direct Digital in immediately available funds by one or more holders of Equity Interests therein as additional common equity contributions to Direct Digital (that do not constitute Disqualified Equity Interests) and which are designated an "Equity Cure" by Direct Digital under *Section 10.04* at the time contributed.

"*Equity Interests*" means, with respect to any Person, all of the shares of capital stock of (or other ownership or profit interests in) such Person, all of the warrants, options or other rights for the purchase or acquisition from such Person of shares of capital stock of (or other ownership or profit interests in) such Person, all of the securities convertible into or exchangeable for shares of capital stock of (or other ownership or profit interests in) such Person, rights or options for the purchase or acquisition from such Person of such shares (or such other interests), and all of the other ownership or profit interests

<del>8</del> 4<del>864-</del> <del>5069-</del> <del>8851 v.</del> in such Person (including partnership, member or trust interests therein), whether voting or nonvoting, and whether or not such shares, warrants, options, rights or other interests are outstanding on any date of determination.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended from time to time, and the regulations and published interpretations thereunder.

*"ERISA Affiliate"* means any corporation or trade or business which is a member of the same controlled group of corporations (within the meaning of *Section 414(b)* of the Code) as any Borrower or is under common control (within the meaning of *Section 414(c)* of the Code) with any Borrower.

"Event of Default" has the meaning specified in Section 10.01.

"Exchange Act" means the Securities Exchange Act of 1934, as amended from time to time, and any successor statute.

*"Excluded Account"* means any deposit account (including, for the avoidance of doubt, any cash, cash equivalents or other property contained therein): (i) solely to the extent, and for so long as, such deposit account is pledged to secure performance of obligations arising under clause (vi) of the defined term "Permitted Liens", and whether such pledge is by escrow or otherwise, in all cases with a balance no greater than such obligations under clause (vi) of the defined term "Permitted Liens"; (ii) used exclusively for payroll, payroll taxes and other employee wage and benefit payments with a balance no greater than such payroll, payroll taxes and other employee wage and benefit payments obligations that are to be paid within any two-week period; (iii) constituting a "zero balance" deposit account; or (iv) consisting of a disbursement account established with a payment processor to process vendor payments so long as the average monthly balance in such account does not exceed \$250,000 at any one time.

"Excluded Hedge Obligation" means, with respect to any Obligated Party, any Hedge Obligations if, and to the extent that, all or a portion of such Obligated Party's Guarantee of (whether such Guarantee arises pursuant to a Guaranty, by such Obligated Party's being jointly and severally liable for such Hedge Obligations, or otherwise (any such Guarantee, an "Applicable Guarantee")), or the grant by such Obligated Party of a security interest to secure, such Hedge Obligations (or any Applicable Guarantee thereof) is or becomes illegal or unlawful under the Commodity Exchange Act or any rule, regulation or order of the Commodity Futures Trading Commission (or the application or official interpretation of any thereof) by virtue of such Obligated Party's failure for any reason not to constitute an Eligible Contract Participant (as defined in the Commodity Exchange Act), and any and all Guarantees of such Obligated Parties' Hedge Obligations by other Obligated Parties at the time the Applicable Guarantee of such Obligated Party or the grant of such security interest becomes effective with respect to such related Hedge Obligations. If any Hedge Obligations arise under a Master Agreement governing more than one Hedge Agreement, then such exclusion shall apply only to the portion of such Hedge Obligations that is attributable to Hedge Agreements for which such Applicable Guarantee or security interest is or becomes illegal.

"*Excluded Taxes*" means (a) backup withholding taxes, (b) franchise taxes, (c) taxes imposed on or measured by net income (however denominated), in each case, (i) imposed on (or measured by) Lender's net income by the jurisdiction under the laws of which Lender is organized or in which its principal office is located or (ii) that are taxes imposed as a result of a present or former connection between Lender and the jurisdiction imposing such tax (other than connections arising from Lender having executed, delivered, become a party to, performed its

<del>9</del> 4<del>864-</del> <del>5069-</del> <del>8851 v.!</del> obligations under, received payments under, received or perfected a security interest under, or engaged in any other transaction pursuant to or enforced any Loan Document or sold or assigned an interest in any Advance), and (d) taxes attributable to Lender's failure to provide Borrowers with any forms or other documentation required by applicable law or reasonably requested by Borrowers in order for Borrowers to determine whether or not payments pursuant to any Loan Document are subject to withholding or information requirements.

"Finance Lease Obligation" shall mean the amount of Debt under a lease of Property by a Person that would be shown as a liability on a balance sheet of such Person prepared for financial reporting purposes in accordance with GAAP.

*"Fixed Charge Coverage Ratio"* means the ratio of (a) Cash Flow Available for Debt Service for the trailing twelve-month period ending on the date of determination, to (b) Debt Service, in each case for Borrowers and the Subsidiaries on a Consolidated Basis.

"*Funding Loss*" means the amount (which shall be payable on demand by Lender) necessary to promptly compensate Lender for, and hold it harmless from, any loss, cost or expense incurred by Lender as a result of:

(a) (u) any payment or prepayment of any Term SOFR Amount on a day other than the last day of the relevant Interest Period (whether voluntary, mandatory, automatic, by reason of acceleration, or otherwise); or

(b) (v)-any failure by Borrowers to borrow a Term SOFR Amount bearing or selected to bear interest based upon the Term SOFR Rate on the date or in the amount selected by Borrowers;

including any loss of anticipated profits and any loss or expense arising from the liquidation or reemployment of funds obtained by it to maintain such Term SOFR Amount or from fees payable to terminate the deposits from which such funds were obtained. Borrowers shall also pay any customary administrative fees charged by Lender in connection with the foregoing. For purposes of calculating amounts payable by Borrowers to Lender hereunder, Lender shall be deemed to have funded the Term SOFR Amount based upon the Term SOFR Rate by a matching deposit or other borrowing in the London inter-bank market for a comparable amount and for a comparable period, whether or not such Term SOFR Amount was in fact so funded.

"GAAP" means generally accepted accounting principles, applied on a consistent basis, as set forth in Opinions of the Accounting Principles Board of the American Institute of Certified Public Accountants and/or in statements of the Financial Accounting Standards Board and/or their respective successors and which are applicable in the circumstances as of the date in question. Accounting principles are applied on a "consistent basis" when the accounting principles applied in a current period are comparable in all material respects to those accounting principles applied in a preceding period.

"Governmental Authority" means any nation or government, any state or political subdivision thereof and any entity exercising executive, legislative, judicial, regulatory, or administrative functions of or pertaining to government.

"Guarantee" by any Person means any obligation, contingent or otherwise, of such Person directly or indirectly guaranteeing any Debt or other obligation of any other Person as well as any obligation or liability, direct or indirect, contingent or otherwise, of such Person (a) to purchase or pay (or

<del>10</del> 4<del>864-</del> <del>5069-</del> <del>8851 v.</del> advance or supply funds for the purchase or payment of) such Debt or other obligation or liability (whether arising by virtue of partnership arrangements, by agreement to keep-well, to purchase assets, goods, securities or services, to operate Property, to take-or-pay, or to maintain net worth or working capital or other financial statement conditions or otherwise) or (b) entered into for the purpose of indemnifying or assuring in any other manner the obligee of such Debt or other obligation or liability of the payment thereof or to protect the obligee against loss in respect thereof (in whole or in part), provided that the term Guarantee shall not include endorsements for collection or deposit in the ordinary course of business. The term "*Guarantee*" used as a verb has a corresponding meaning.

"Guarantor" means any Person who from time to time guarantees all or any part of the Obligations, including, the Subsidiary Guarantors.

"Guaranty" means a written guaranty of each Guarantor in favor of Lender, in form and substance satisfactory to Lender, as the same may be amended, modified, restated, renewed, replaced, extended, supplemented or otherwise changed from time to time.

"Hazardous Material" means any substance, product, waste, pollutant, material, chemical, contaminant, constituent, or other material which is or becomes listed, regulated, or addressed under any Environmental Law, including, without limitation, asbestos, petroleum, and polychlorinated biphenyls.

"Hedge Agreement" means (a) any and all interest rate swap transactions, basis swaps, credit derivative transactions, forward rate transactions, commodity swaps, commodity options, forward commodity contracts, equity or equity index swaps or options, bond or bond price or bond index swaps or options or forward bond or forward bond price or forward bond index transactions, interest rate options, forward foreign exchange transactions, cap transactions, floor transactions, collar transactions, currency swap transactions, cross-currency rate swap transactions, currency options, spot contracts, or any other similar transactions or any combination of any of the foregoing (including any options to enter into any of the foregoing), whether or not any such transaction is governed by or subject to any master agreement, (b) any and all transactions of any kind, and the related confirmations, which are subject to the terms and conditions of, or governed by, any form of master agreement published by the International Swaps and Derivatives Association, Inc., any International Foreign Exchange Master Agreement, or any other master agreement (any such master agreement, together with any related schedules and annexes, a "Master Agreement") and (c) any and all Master Agreements and any and all related confirmations.

"Hedge Bank" means any Person that, at the time it enters into an interest rate Hedge Agreement permitted under this Agreement, is Lender or an Affiliate of Lender, in its capacity as a party to such Hedge Agreement.

"Hedge Obligations" means, for any Person, any and all obligations (whether absolute or contingent and howsoever and whensoever created) of such Person to pay or perform under any agreement, contract or transaction that constitutes a "swap" within the meaning of Section 1a(47) of the Commodity Exchange Act arising, evidenced or acquired under (a) any and all Hedging Agreements, (b) any and all cancellations, buy backs, reversals, terminations or assignments of any Hedging Agreements, and (c) any and all renewals, extensions and modifications of any Hedging Agreements and any and all substitutions of any Hedging Agreements.

"*Immaterial Subsidiary*" means any Subsidiary that does not (i) conduct any business operations, (ii) have any assets in excess of \$25,000, (iii) incur any liabilities in excess of \$25,000, (iv) engage in any transactions with the Borrowers, or (v) have any Subsidiaries.

<del>11</del> 4<del>864-</del> <del>5069-</del> <del>8851 v</del> "Intercreditor Agreement" means that certain Intercreditor Agreement of even date herewith, by and among Term Loan Lender, as amended, restated, supplemented or otherwise modified from time to time.

"*Interest Expense*" means, for any period, the interest expense of Borrowers and their Subsidiaries for the period in question, determined on a Consolidated Basis and consistent with practices as of the date hereof or otherwise in accordance with GAAP.

"Interest Payment Date" means (a) with respect to any principal amount bearing interest based upon the Prime Rate, the first day of each and every calendar month during the term of the Notes and (b) with respect to each Term SOFR Amount, the last day of each Interest Period applicable to such Term SOFR Amount.

"Interest Period" means, for any Revolving Credit Advance, the period commencing on the date of such Revolving Credit Advance and ending on the numerically corresponding day in the calendar month that is one month thereafter; provided, however, that: (i) the first day of an Interest Period must be a U.S. Government Securities Business Day; (ii) no Interest Period shall extend beyond the Maturity Date; (iii) no Interest Period shall extend beyond the Scheduled payment date of any principal payment required with respect to such Revolving Credit Advance; (iv) any Interest Period that would otherwise expire on a day that is not a U.S. Government Securities Business Day shall be extended to the next succeeding U.S. Government Securities Business Day; and (v) any Interest Period that begins on the last U.S. Government Securities Business Day; and (v) any Interest Period that begins on the last U.S. Government Securities Business Day of a calendar month (or on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period) shall end on the last U.S. Government Securities Business Day of a calendar month.

"*Investment*" means any beneficial ownership (including stock, partnership or limited liability company interests) of or in any Person, or any loan, advance or capital contribution to any Person or the acquisition of any material assets of another Person, other than equipment purchases made by the Borrower as part of the operation of its business.

"Letter of Credit" means any letter of credit issued by Lender for the account of or at the direction of Borrowers pursuant to Article II of this Agreement.

*"Letter of Credit Liabilities"* means, at any time, the aggregate face amounts of all outstanding Letters of Credit, plus any amounts drawn under any Letters of Credit for which Lender has not been fully reimbursed by Borrowers (unless Lender, in its sole discretion, has cleared the drawn amount by means of an Advance under the Revolving Credit Note, in which case the drawn amount would not constitute a Letter of Credit Liability).

"Letter of Credit Request Form or Application" means a certificate or agreement, in a form acceptable to Lender, properly completed and signed by Borrowers requesting issuance of a Letter of Credit and containing provisions for fees for the issuance of Letters of Credit, repayment of drawn letters of credit, the interest rate applicable to drawn and unpaid Letters of Credit, and such other matters as Lender may require.

"Liabilities" means, at any particular time, all amounts which, in conformity with GAAP, would be included as liabilities on a balance sheet of a Person.

<del>12</del> 4<del>864-</del> <del>5069-</del> <del>8851 v.</del> "*Lien*" means any lien, mortgage, security interest, tax lien, pledge, charge, hypothecation, assignment, preference, priority, or other encumbrance of any kind or nature whatsoever (including, without limitation, any conditional sale or title retention agreement), whether arising by contract, operation of law, or otherwise.

"*Liquid Assets*" shall mean, with respect to any Person, (a) unencumbered Cash and Cash Equivalents (as defined below) and (b) marketable securities, each valued in accordance with GAAP, consistently applied (or other principles acceptable to Lender), as reasonably determined by such Person and reasonably approved by Lender.

"Loan Documents" means this Agreement, the Intercreditor Agreement, and all promissory notes, security agreements, pledge agreements, deeds of trust, assignments, letters of credit, guaranties, Hedge Agreements and other instruments, documents, and agreements executed and delivered pursuant to or in connection with this Agreement, as such instruments, documents, and agreements may be amended, modified, renewed, restated, extended, supplemented, replaced, consolidated, substituted, or otherwise changed from time to time.

"Loan Rate" means the Term SOFR Rate, plus the Term SOFR Adjustment, plus 3.00% per annum; provided, that, in no event shall the Loan Rate be less than 0.50% of the Loan Rate effective on the date of this Agreement.

"Master Agreement" has the meaning set forth in the definition of "Hedge Agreement."

"*Material Adverse Event*" means any act, event, condition, or circumstance which could materially and adversely affect: (a) the operations, business, properties, liabilities (actual or contingent), or condition (financial or otherwise) of Borrowers or Borrowers and the Subsidiaries, taken as a whole; (b) the ability of any Obligated Party to perform its obligations under any Loan Document to which it is a party; or (c) the legality, validity, binding effect or enforceability against any Obligated Party of any Loan Document to which it is a party.

"Maturity Date" means 3:00 P.M. Dallas, Texas time on July 7, 2025, or such earlier date on which the Commitment terminates as provided in this Agreement.

"Maximum Lawful Rate" means, at any time, the maximum rate of interest which may be charged, contracted for, taken, received or reserved by Lender in accordance with applicable Texas law (or applicable United States federal law to the extent that such law permits Lender to charge, contract for, receive or reserve a greater amount of interest than under Texas law). The Maximum Lawful Rate shall be calculated in a manner that takes into account any and all fees, payments, and other charges in respect of the Loan Documents that constitute interest under applicable law. Each change in any interest rate provided for herein based upon the Maximum Lawful Rate resulting from a change in the Maximum Lawful Rate shall take effect without notice to Borrowers at the time of such change in the Maximum Lawful Rate.

"*Multiemployer Plan*" means a multiemployer plan defined as such in *Section 3(37)* of ERISA to which contributions have been made by any Borrower or any ERISA Affiliate and which is covered by Title IV of ERISA.

"*Net Income*" means the net income (or loss) of Borrowers and their Subsidiaries for the period in question, determined on a Consolidated Basis and consistent with practices as of the date hereof or otherwise in accordance with GAAP.

<del>13</del> 4864-5069-8851 v.9 "*Notes*" means, collectively, all promissory notes (and "*Note*" means any of such Notes) executed at any time by Borrowers and payable to the order of Lender, as amended, renewed, replaced, extended, supplemented, consolidated, restated, modified, otherwise changed and/or increased from time to time.

"Obligated Party" means each Borrower, each Guarantor and any other Person who is or becomes party to any agreement that guarantees or secures payment and performance of the Obligations or any part thereof.

"Obligations" means all obligations, indebtedness, and liabilities of Borrowers, each Guarantor and any other Obligated Party to Lender or any Affiliate of Lender, or both, now existing or hereafter arising, whether direct, indirect, related, unrelated, fixed, contingent, liquidated, unliquidated, joint, several, or joint and several, including, without limitation, the obligations, indebtedness, and liabilities under this Agreement, all Hedge Obligations under any Secured Hedge Agreements, the other Loan Documents, any cash management or treasury services agreements and all interest accruing thereon (whether a claim for post-filing or post-petition interest is allowed in any bankruptcy, insolvency, reorganization or similar proceeding) and all attorneys' fees and other expenses incurred in the enforcement or collection thereof; *provided, however*, that any other term or provision of this Agreement or any other Loan Document to the contrary notwithstanding, the "*Obligations*" of any Obligated Party shall exclude, as to such Obligated Party, Excluded Hedge Obligations of such Obligated Party.

"Operating Agreement of Direct Digital" means the Second Amended and Restated Limited Liability Company Agreement of Direct Digital dated February 15, 2022.

"Patriot Act" means the USA PATRIOT Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)).

"PBGC" means the Pension Benefit Guaranty Corporation or any entity succeeding to all or any of its functions under ERISA.

"*Permitted Discretion*" means a determination in good faith and in the exercise of reasonable (from the perspective of a secured asset-based lender) business judgment.

"Permitted Indebtedness" means: (i) Debt to Lender arising under this Agreement or any other Loan Document; (ii) Term Loan Debt in accordance with the Intercreditor Agreement that matures after the Maturity Date; (iii) Debt existing on the date hereof which is disclosed in *Schedule 8.01*; (iv) Debt of up to \$200,000 outstanding at any time secured by a Lien described in clause (viii) of the defined term "Permitted Liens," provided such Debt does not exceed the cost of the equipment or intellectual property financed with such Debt; (v) amounts billed to a Borrower by its suppliers for goods delivered to or services performed for such Borrower in the ordinary course of business; (vi) reimbursement obligations in connection with trade letters of credit entered into in the ordinary course of business and, to the extent not subject to an Excluded Account, cash management services (including credit cards, debit cards and other similar instruments) that are secured by cash and issued on behalf of any Borrower or a Subsidiary thereof in an amount not to exceed \$200,000 at any time outstanding; (vii) Debt secured by a Lien described in clause (xi) of the defined term "Permitted Liens"; (viii) Debt; extensions, refinancings and renewals of any items of Permitted Indebtedness; provided that the principal amount is not increased or the terms modified to impose materially more burdensome terms upon any Borrower or its Subsidiary, as the case may be; and (ix) other unsecured Debt in an amount not to exceed \$200,000 in the aggregate.

<del>14</del> <del>4864-</del> <del>5069-</del> <del>8851 v.</del>

"Permitted Investment" means: (i) Investments existing as of the date hereof which are disclosed on Schedule 8.05; (ii) (a) marketable direct obligations issued or unconditionally guaranteed by the United States of America or any agency or any State thereof maturing within one year from the date of acquisition thereof currently having a rating of at least A-2 or P-2 from either Standard & Poor's Corporation or Moody's Investors Services, (b) commercial paper maturing no more than one year from the date of creation thereof and currently having a rating of at least A-2 or P-2 from either Standard & Poor's Corporation or Moody's Investors Service, (c) certificates of deposit issued by any bank with assets of at least \$250,000,000 maturing no more than one year from the date of investment therein, and (d) money market accounts; (iii) repurchases of stock from current or former employees, directors, or consultants of the Borrower under the terms of applicable repurchase agreements at the original issuance price of such securities in an aggregate amount not to exceed \$250,000 in any fiscal year; provided that no Event of Default has occurred, is continuing or could exist after giving effect to the repurchases; (iv) Investments (including debt obligations) received in connection with the bankruptcy or reorganization of customers or suppliers and in settlement of delinquent obligations of, and other disputes with, customers or suppliers arising in the ordinary course of Borrowers' business; (v) Investments consisting of notes receivable of, or prepaid royalties and other credit extensions to, customers and suppliers who are not Affiliates, in the ordinary course of business; provided that this subparagraph (vi) shall not apply to Investments of any Borrower in any Obligated Party; (vii) Investments consisting of loans not involving the net transfer on a substantially contemporaneous basis of cash proceeds to employees, officers or directors relating to the purchase of capital stock or other Equity Interests of a Borrower pursuant to employee stock purchase plans or other similar agreements approved by such Borrower's Board of Directors (or, if not a corporation, its equivalent authorizing body); (viii) Investments consisting of travel advances in the ordinary course of business; (ix) Investments in newly-formed Subsidiaries; provided that any such Subsidiary that is or is expected to become an After-Acquired Subsidiary complies with Section 7.13 hereof; and (x) additional Investments that do not exceed \$200,000 in the aggregate in any fiscal year if, at the time of such Investment and after giving effect thereto, the Borrower is in compliance with the Financial Covenants in Article IX.

"Permitted Liens" means any and all of the following: (i) Liens in favor of the Lender; (ii) Liens in favor of the Term Loan Lender securing the Debt under the Term Loan Documents, subject to the Intercreditor Agreement; (iii) Liens existing as of the date hereof which are disclosed in Schedule 8.02\_hereto; (iv) Liens for taxes, fees, assessments or other governmental charges or levies, either not delinquent or being contested in good faith by appropriate proceedings; provided that Borrowers maintain adequate reserves therefor in accordance with GAAP; (v) Liens securing claims or demands of materialmen, artisans, mechanics, carriers, warehousemen, landlords and other like Persons arising in the ordinary course of any Borrower's business and imposed without action of such parties; (vi) Liens arising from judgments, decrees or attachments in circumstances which do not constitute an Event of Default hereunder; (vii) Liens on deposits held in an Excluded Account; (viii) Liens on equipment or software or other intellectual property constituting purchase money Liens and Liens in connection with Finance Lease Obligations securing Indebtedness permitted in clause (iv) of "Permitted Indebtedness"; (ix) Liens incurred in connection with Subordinated Debt; (x) leasehold interests in leases or subleases and licenses granted in the ordinary course of business and not interfering in any material respect with the business of the licensor; (xi) Liens in favor of customs and revenue authorities arising as a matter of law to secure payment of custom duties that are promptly paid on or before the date they become due; (xii) Liens on insurance proceeds securing the payment of financed insurance premiums that are promptly paid on or before the date they become due (provided that such Liens extend only to such insurance proceeds and not to any other property or assets); (xiii) statutory and common law rights of set-off and other similar rights as to deposits of cash and securities in favor of banks, other depository institutions and

<del>15</del> 4864-5069-8851imposed by law or arising in the ordinary course of business so long as they do not materially impair the value or marketability of the related property; (xv) (A) Liens on cash securing obligations permitted under clause (vi) of the definition of Permitted Indebtedness and (B) security deposits in connection with real property leases, the combination of (A) and (B) in an aggregate amount not to exceed \$300,000 at any time; (xvi) sales, transfers or other dispositions of assets permitted by *Section 8.08* and, in connection therewith, customary rights and restrictions contained in agreements relating to such transactions pending the completion thereof or during the term thereof, and any option or other agreement to sell, transfer, license, sublicense, lease, sublease or dispose of an asset permitted by *Section 8.08*, in each case, such terms being agreed to and such transactions entered into in the ordinary course of business; and (xvii) Liens incurred in connection with the extension, renewal or refinancing of the Debt secured by Liens of the type described in clauses (i) through (xvi) above; provided that any extension, renewal or replacement Lien shall be limited to the property encumbered by the existing Lien and the principal amount of the Debt being extended, renewed or refinanced (as may have been reduced by any payment thereon) does not increase.

"*Permitted Tax Distributions*" means, quarterly tax distributions by Direct Digital to its direct or indirect constituent members in the amount necessary to satisfy the U.S. federal, state and local income tax obligations of such members based on the taxable income of Direct Digital and its Subsidiaries on a Consolidated Basis for such taxable year, in an aggregate amount determined in accordance with the terms of the organizational documents of Direct Digital. Direct Digital may make such distributions after the end of the taxable year, or make such distributions on a quarterly basis during the taxable year to reflect estimated tax obligations of the members and their direct or indirect equityholders. For the avoidance of doubt, Permitted Tax Distributions based on estimates shall be made on a "rolling basis" and will be trued-up at least annually.

"Permitted Warrant Transaction" means the repurchase by Direct Digital Holdings of the warrants issued in February 2022 for cash in a single transaction or series of transactions.

"*Person*" means any individual, corporation, limited liability company, business trust, association, company, partnership, joint venture, Governmental Authority, or other entity, and shall include such Person's heirs, administrators, personal representatives, executors, successors and assigns.

"Plan" means any employee benefit or other plan established or maintained by any Borrower or any ERISA Affiliate and which is covered by Title IV of ERISA.

"Prime Rate" means, for any day, the rate of interest announced from time to time by Lender as its "base" or "prime" rate of interest, which Borrowers hereby acknowledge and agree may not be the lowest interest rate charged by Lender and is set by Lender in its sole discretion, changing when and as said prime rate changes.

"Principal Office" means the principal office of Lender, presently located at 5001 Spring Valley Road, Suite 825W, Dallas, Texas 75224.

"Priority Collateral" means the Collateral in which the Lender has a first priority security interest under the Security Agreement, subject to the Intercreditor Agreement.

"Prohibited Transaction" means any non-exempt transaction set forth in Section 406 of ERISA or Section 4975 of the Code.

<del>16</del> 4<del>864-</del> <del>5069-</del> <del>8851 v</del> "*Property*" of a Person means any and all property, whether real, personal, tangible, intangible or mixed, of such Person, or any other assets owned, operated or leased by such Person.

"Public Company Costs" shall mean actual documented costs relating to compliance with the Sarbanes-Oxley Act of 2002, as amended, and other actual documented expenses arising out of or incidental to DDH Holdings' status as a reporting company, including actual documented costs, fees and expenses (including reasonable legal, accounting and other professional fees) relating to compliance with provisions of the Securities Act of 1933, as amended, and the Exchange Act, registration and reporting obligations, permitted securities offerings, the rules of securities exchange companies with listed equity securities, directors' compensation, fees and expense reimbursement, shareholder meetings and reports to shareholders, indemnification and reimbursement of directors, officers and employees in respect of liabilities relating to their serving in any such capacity, directors' and officers' insurance and other executive costs, legal and other professional fees.

"*Redemption/Exchange Transactions*" means the transactions pursuant to which Direct Digital redeems Class A Common Units for either (A)(1) a stock exchange payment or (2) a cash exchange payment, in each case in accordance with the Operating Agreement of Direct Digital or (B) the direct purchase by DDH Holdings of vested Class A Common Units and paired voting stock pursuant to any call right, in accordance with the Operating Agreement of Direct Digital.

#### "Related Indebtedness" has the meaning set forth in Section 11.20 of this Agreement.

"*Release*" means, as to any Person, any release, spill, emission, leaking, pumping, injection, deposit, disposal, disbursement, leaching, or migration of Hazardous Materials into the indoor or outdoor environment or into or out of property owned by such Person, including, without limitation, the movement of Hazardous Materials through or in the air, soil, surface water, ground water, or property.

"*Relevant Governmental Body*" means the Board of Governors of the Federal Reserve System or the Federal Reserve Bank of New York, or a committee officially endorsed or convened by the Board of Governors of the Federal Reserve System or the Federal Reserve Bank of New York, or any successor thereto.

"*Remedial Action*" means all actions required to (a) clean up, remove, treat, or otherwise address Hazardous Materials in the indoor or outdoor environment, (b) prevent the Release or threat of Release or minimize the further Release of Hazardous Materials so that they do not migrate or endanger or threaten to endanger public health or welfare or the indoor or outdoor environment, or (c) perform pre-remedial studies and investigations and post-remedial monitoring and care.

"*Reportable Event*" means any of the events set forth in *Section 4043* of ERISA that requires the Borrower or Subsidiary to notify the PBGC of such event, and the reporting of which is not otherwise waived.

"*Responsible Officer*" means (a) for any Borrower, the chief executive officer, president, chief financial officer, or treasurer of such Borrower or any Person designated by a Responsible Officer to act on behalf of a Responsible Officer; *provided that* such designated Person may not designate any other Person to be a Responsible Officer. Any document delivered hereunder that is signed by a Responsible Officer of Borrowers shall be conclusively presumed to have been authorized by all necessary corporate, partnership and/or other action on the part of Borrowers and such Responsible Officer, president, chief financial officer, treasurer, assistant treasurer

<del>17</del> 4<del>864-</del> <del>5069-</del> 8851-v or controller, and a secretary or assistant secretary for the purposes of delivering incumbency certificates, or as a second Responsible Officer in any case where two Responsible Officers are acting on behalf of such corporation; (ii) in the case of a limited partnership, the Responsible Officer of the general partner, acting on behalf of such general partner in its capacity as general partner; or (iii) in the case of a limited liability company, the Responsible Officer of the managing member, acting on behalf of such managing member in its capacity as managing member.

"Revolving Credit Advance" means any Advance made by Lender to Borrowers pursuant to Section 2.01(a) of this Agreement.

"*Revolving Credit Availability*" an amount such that the ratio of the value of Eligible Accounts to the aggregate amount of all Revolving Credit Advances at such time outstanding shall not be less than  $\frac{2.0 \text{ to } 1.0}{2.0 \text{ to } 1.0}$  to 1.0 as of the calendar months ending November 30, 2024 and December 31, 2024, (b) 1.50 to 1.0 as of the calendar months ending January 31, 2025, February 28, 2025 and March 31, 2025, and (c) 2.0 to 1.0 as of April 15, 2025 and at all times thereafter.

"*Revolving Credit Note*" means the promissory note of Borrowers payable to the order of Lender, in substantially the form of *Exhibit C* hereto, and all amendments, extensions, renewals, replacements, and modifications thereof.

"RICO" means the Racketeer Influenced and Corrupt Organization Act of 1970.

"Sanction(s)" means any sanction administered or enforced by the United States Government (including without limitation, OFAC), the United Nations Security Council, the European Union, Her Majesty's Treasury ("HMT") or other relevant sanctions authority.

"Secured Hedge Agreement" means any Hedge Agreement permitted under this Agreement entered into by and between any Obligated Party and any Hedge Bank.

"Security Agreement" means the Security Agreement dated of even date herewith of Borrowers and the other Obligated Parties party thereto in favor of Lender, in form and substance satisfactory to Lender, as the same may be amended, restated, supplemented, modified, or changed from time to time.

"Security Documents" means the Security Agreement, each Guaranty, and each and every other security agreement, pledge agreement, mortgage or other collateral security agreement required by or delivered to Lender from time to time to secure the Obligations or any portion thereof.

"Specified Financial Covenants" has the meaning set forth in Section 10.04.

"Specified Obligated Party" means any Obligated Party that is not an Eligible Contract Participant (determined prior to giving effect to Section 7.14 hereof or any other "keepwell, support or other agreement" (as defined in the Commodity Exchange Act), or any similar provision contained in any Guaranty).

"Subordinated Debt" means any Debt of Borrowers (other than the Obligations) that has been subordinated to the Obligations by written agreement, in form and content satisfactory to Lender.

"*Subsidiary*" means (a) any corporation of which at least a majority of the outstanding shares of stock having by the terms thereof ordinary voting power to elect a majority of the board of directors of such corporation (irrespective of whether or not at the time stock of any other class or classes of such

<del>18</del> 4<del>864-</del> <del>5069-</del> <del>8851 v.9</del> corporation shall have or might have voting power by reason of the happening of any contingency) is at the time directly or indirectly owned or controlled by any Borrower or one or more of the Subsidiaries or by any Borrower and one or more of the Subsidiaries; and (b) any other entity (i) of which at least a majority of the ownership, equity or voting interest is at the time directly or indirectly owned or controlled by one or more of Borrowers and the Subsidiaries and (ii) which is treated as a subsidiary in accordance with GAAP.

"Subsidiary Guarantors" means each Domestic Subsidiary of each Borrower formed or acquired after the date hereof who from time to time guarantees all or any part of the Obligations, and "Subsidiary Guarantor" means any one of the Subsidiary Guarantors.

"*Tax Receivable Agreement*" means that certain tax receivable agreement to be entered into by and among DDH Holdings, Direct Digital and Direct Digital Management.

"*Term Loan Agreement*" means that certain Term Loan and Security Agreement dated as of December 3, 2021, by and between Term Loan Lender and Borrowers as amended, restated or modified as of the Closing Date and from time to time thereafter as permitted by the terms of the Intercreditor Agreement.

"*Term Loan Debt*" means the secured indebtedness of Borrowers under the Term Loan Agreement in a principal amount not to exceed Twenty-Five Million Five-Hundred-Twenty Thousand Dollars (\$25,520,000).

"*Term Loan Documents*" means the Term Loan Agreement and the Other Documents (as defined in the Term Loan Agreement) executed in connection therewith, in each case, as amended, restated or modified from time to time as permitted by the terms of the Intercreditor Agreement.

"Term Loan Lender" means Lafayette Square Loan Servicing, LLC and the other lenders and financial institutions from time to time party to the Term Loan Agreement and their successors and permitted assigns.

"Term Loan Priority Collateral" means all Collateral that is not Priority Collateral.

"Term SOFR Adjustment" means 0.10% (10 basis points).

"Term SOFR Administrator" means the CME Group Benchmark Administration Limited (CBA) (or a successor administrator of the secured overnight financing rate).

"Term SOFR Amount" means each principal amount for which the Term SOFR Rate applies.

"*Term SOFR Rate*" means the one (1) month Term Secured Overnight Financing Rate, as administered by the Term SOFR Administrator and displayed by Bloomberg LP (or any successor thereto, or replacement thereof, as approved by Lender) and as determined by Lender on the first day of the applicable Interest Period.

"Total Debt" means all Debt of Borrowers and the Subsidiaries at such time.

"Total Funded Debt" means Total Debt, other than Debt described in *clauses (c), (e), (h), (j), (k), (l)* (solely with respect to the undrawn portion of any letters of credit), and *(m)* of the definition of "Debt" herein.

<del>19</del> 4<del>864-</del> <del>5069-</del> <del>8851-v 9</del> "Total Funded Debt to EBITDA Ratio" means the ratio of (a) Total Funded Debt of Borrowers and the Subsidiaries, on a Consolidated Basis, as of such date, to (b) EBITDA for Borrowers and the Subsidiaries, on a Consolidated Basis, for the four (4) fiscal quarters ending on such date.

"UCC" means the Chapters 1 through 11 of the Texas Business and Commerce Code, as amended from time to time.

"Unfunded Pension Liability" means the excess, if any, of (a) the funding target as defined under Section 430(d) of the Code without regard to the special at risk rules of Section 430(i) of the Code, over (b) the value of plan assets as defined under Section 430(g)(3)(A) of the Code determined as of the last day of each calendar year, without regard to the averaging which may be allowed under Section 310(g)(3)(B) of the Code and reduced for any prefunding balance or funding standard carryover balance as defined and provided for in Section 430(f) of the Code.

"U.S. Government Securities Business Day" means any day except for (a) a Saturday, (b) a Sunday or (c) a day on which the Securities Industry and Financial Markets Association recommends that the fixed income departments of its members be closed for the entire day for purposes of trading in United States government securities.

I.2 Accounting Matters. Any accounting term used in this Agreement or the other Loan Documents shall have, unless otherwise specifically provided therein, the meaning customarily given such term in accordance with GAAP, and all financial computations thereunder shall be computed, unless otherwise specifically provided therein, in accordance with GAAP consistently applied; provided, that all financial covenants and calculations in the Loan Documents shall be made in accordance with GAAP as in effect on the date of this Agreement unless Borrowers and Lender shall otherwise specifically agree in writing; provided, further that all obligations of any Person that are or would have been treated as operating leases for purposes of GAAP prior to the effectiveness of FASB ASC 842 shall continue to be accounted for as operating leases for purposes of all financial definitions and calculations for purpose of this Agreement (whether or not such operating lease obligations were in effect on such date) notwithstanding the fact that such obligations are required in accordance with FASB ASC 842 (on a prospective or retroactive basis or otherwise) to be treated as capital lease obligations in the financial statements. That certain items or computations are explicitly modified by the phrase "in accordance with GAAP" shall in no way be construed to limit the foregoing.

I.3 **Other Definitional Provisions.** All definitions contained in this Agreement are equally applicable to the singular and plural forms of the terms defined. The words *"hereof"*, *"herein"*, and *"hereunder"* and words of similar import referring to this Agreement refer to this Agreement as a whole and not to any particular provision of this Agreement. Unless otherwise specified, all Article and Section references pertain to this Agreement. Terms used herein that are defined in the UCC, unless otherwise defined herein, shall have the meanings specified in the UCC.

# Article II. ADVANCES

## II.1 Advances.

(a) **Revolving Credit Advances**. Subject to the terms and conditions of this Agreement, Lender agrees to make one or more Revolving Credit Advances to Borrowers from time to time from the date hereof to and including the Maturity Date in an aggregate principal amount at any time outstanding up to but not exceeding the amount of the Commitment, provided

that the aggregate amount of all Revolving Credit Advances at any time outstanding shall not exceed the lesser of (i) the amount of the Commitment or (ii) Revolving Credit Availability. Subject to the foregoing limitations, and the other terms and provisions of this Agreement, Borrowers may borrow, repay, and reborrow hereunder.

(c) (i) The Revolving Credit Note. The obligation of Borrowers to repay the Revolving Credit Advances and interest thereon shall be evidenced by the Revolving Credit Note executed by Borrowers payable to the order of Lender, in the principal amount of the Commitment as originally in effect and dated the date hereof.

(d) (ii) Payments on Revolving Credit Advances. All accrued but unpaid interest on the Revolving Credit Advances outstanding from time to time shall be payable in monthly installments on each Interest Payment Date until the Maturity Date when the then outstanding principal balance of the Revolving Credit Note and all accrued but unpaid interest thereon shall be due and payable. Borrowers may from time to time during the term of this Agreement borrow, partially or wholly repay its outstanding borrowings, and reborrow, subject to all of the limitations, terms and conditions of this Agreement and of the Loan Documents; *provided, however*, that the unpaid principal of the Revolving Credit Note shall not at any time exceed the principal amount stated above.

(c) (iii) Borrowing Procedure. Borrowers shall give Lender notice of each Revolving Credit Advance by means of an Advance Request Form containing the information required therein and delivered (by hand or by mechanically confirmed facsimile) to Lender no later than 1:00 p.m. (Texas time) on the day on which the Revolving Credit Advance is desired to be funded. Advances shall be in a minimum amount of \$10,000. Lender at its option may accept telephonic requests for such Advances, provided that such acceptance shall not constitute a waiver of Lender's right to require delivery of an Advance Request Form in connection with subsequent Advances. Any telephonic request for a Revolving Credit Advance by Borrowers shall be promptly confirmed by submission of a properly completed Advance Request Form to Lender, but failure to deliver an Advance Request Form shall not be a defense to payment of the Advance. Lender shall have no liability to Borrowers for any loss or damage suffered by Borrowers as a result of Lender's honoring of any requests, execution of any instructions, authorizations or agreements or reliance on any reports communicated to it telephonically, by facsimile or electronically and purporting to have been sent to Lender by Borrowers and Lender shall have no duty to verify the origin of any such communication or the identity or authority of the Person sending it. Subject to the terms and conditions of this Agreement, each Revolving Credit Advance shall be made available to Borrowers by depositing the same, in immediately available funds, in an account of Borrowers designated by Borrowers maintained with Lender at the Principal Office. At no time shall there be more than five (5) Term SOFR Amounts outstanding under this Agreement.

## II.2 General Provisions Regarding Interest, Etc.

(a) **Repayment of Advances**. Borrowers shall repay the unpaid principal amount of all Advances on the Maturity Date, unless sooner due by reason of acceleration by Lender as provided in this Agreement.

(b) **Interest Rate**. The unpaid principal balance of the Notes shall bear interest from the date hereof through the Maturity Date at a per annum rate which shall be, except as otherwise

provided in this Agreement, the lesser of (A) the Loan Rate in effect from day to day, or (B) the Maximum Lawful Rate. The determination by Lender of the Loan Rate shall, in the absence of manifest error, be conclusive and binding in all respects. Notwithstanding anything herein to the contrary, in the event that (i) Term SOFR is permanently or indefinitely unavailable or unascertainable, or ceases to be published by the Term SOFR Administrator or its successor, (ii) the Term SOFR Administrator or its successor invokes its insufficient admissions policy, (iii) Term SOFR is determined to be no longer representative by the Term SOFR Administrator, (iv) Term SOFR can no longer be lawfully relied upon in contracts of this nature by one or both of the parties, or (v) Term SOFR does not accurately and fairly reflect the cost of making or maintaining the type of loans or advances under this Agreement and in any such case, such circumstances are unlikely to be temporary, then all references to the Loan Rate herein will instead be to a replacement rate determined by Lender in its reasonable judgment, including any adjustment to the replacement rate to reflect a different credit spread, term or other mathematical adjustment deemed necessary by the Lender in its reasonable judgment. Lender will provide reasonable notice to Borrowers of such replacement rate, which will be effective on the date of the earliest event set forth in clauses (i)-(v) of this paragraph. If there is any ambiguity as to the date of occurrence of any such event, Lender's judgment will be dispositive. Lender may also from time to time, in Lender's sole discretion, make any technical, administrative or operational changes (including changes to the timing and frequency of determining rates and making payments of interest, timing of borrowing requests or prepayment, conversion or continuation notices, length or applicability of lookback periods, the applicability of breakage provisions, and other technical, administrative or operational matters) ("Conforming Changes") that Lender decides may be appropriate to reflect the adoption and implementation of such replacement rate and to permit the administration of the loans by Lender an administratively and operationally practicable manner. If there is any ambiguity as to the date of occurrence of any such event, Lender's judgment will be dispositive. Lender does not warrant or accept responsibility for, and shall not have any liability with respect to (a) the administration of, submission of, calculation of or any other matter related to any replacement rate, any component definition thereof or rates referenced in the definition thereof or any alternative, comparable or successor rate thereto, including whether the composition or characteristics of any such alternative, comparable or successor rate will be similar to, or produce the same value or economic equivalence of, or have the same volume or liquidity as the immediately preceding interest index rate or any other interest rate index, or (b) the effect, implementation or composition of any Conforming Changes.

(c) **Default Interest Rate**. Any outstanding principal of any Advance and (to the fullest extent permitted by law) any other amount payable by Borrowers under this Agreement or any other Loan Document that is not paid in full when due (whether at stated maturity, by acceleration, or otherwise) shall bear interest at the Default Interest Rate for the period from and including the due date thereof to but excluding the date the same is paid in full. Additionally, upon the occurrence and during the continuance of an Event of Default all outstanding and unpaid principal amounts of all of the Obligations shall, to the extent permitted by law, bear interest at the Default Interest Rate until such time as Lender shall waive in writing the application of the Default Interest Rate to such Event of Default situation. Interest payable at the Default Interest Rate shall be payable from time to time on demand.

(d) **Computation of Interest**. Interest on the Advances and all other amounts payable by Borrowers hereunder is computed on a 365/360 basis; that is, by applying the ratio of the interest rate over a year of 360 days, multiplied by the outstanding principal balance, multiplied by the actual number of days the principal balance is outstanding.

(e) Application of Payments. Except as expressly provided this Agreement to the contrary, all payments on the Notes shall be applied in the following order of priority: (a) the payment or reimbursement of any Funding Loss, expenses, costs or obligations (other than the outstanding principal balance hereof and interest hereon) for which Borrowers shall be obligated or Lender shall be entitled pursuant to the provisions of this Agreement or the other Loan Documents; (b) the payment of accrued but unpaid interest hereon; and (c) the payment of all or any portion of the principal balance of the Advances then outstanding hereunder. If an Event of Default exists, then Lender may, at the sole option of Lender, apply any such payments, at any time and from time to time, to any of the items specified in *clauses (a)*, (b) or (c) above without regard to the order of priority otherwise specified in this Section 2.02(e) and any application to the outstanding principal balance hereof may be made in either direct or inverse order of maturity. Payments by check or draft shall not constitute payment in immediately available funds until the required amount is actually received by Lender. Payments in immediately available funds received by Lender in the place designated for payment on a Business Day prior to 3:00 p.m. Dallas, Texas time at said place of payment shall be credited prior to the close of business on the Business Day received, while payments received by Lender on a day other than a Business Day or after 3:00 p.m. Dallas, Texas time on a Business Day shall not be credited until the next succeeding Business Day. If any payment of principal or interest on the Notes shall become due and payable on a day other than a Business Day, such payment shall be made on the next succeeding Business Day. Acceptance by Lender of any payment in an amount less than the full amount then due shall be deemed an acceptance on account only, and the failure to pay the entire amount then due may become an Event of Default. Borrowers agree that all payments of any Obligation due hereunder shall be final, and if any such payment is recovered in any bankruptcy, insolvency or similar proceedings instituted by or against Borrowers, all obligations due hereunder shall be automatically reinstated in respect of the obligation as to which payment is so recovered.

II.3 Unused Facility Fee. Borrowers agree to pay to Lender an unused facility fee on the daily unused amount of the Commitment for the period from and including the date of this Agreement to and including the Maturity Date, at the rate of 0.40% per annum based on a 360-day year and the actual number of days elapsed. For the purpose of calculating the unused facility fee hereunder, the Commitment shall be deemed utilized by the amount of all outstanding Revolving Credit Advances. Accrued unused facility fees shall be payable quarterly in arrears on the first (1st) Business Day of each April, July, October, and January during the term of this Agreement and on the Maturity Date.

II.4 <u>Use of Proceeds</u>. The proceeds of the Revolving Credit Advances shall be used by Borrowers for working capital in the ordinary course of business and other general corporate purposes.

II.5 Late Charges. If a payment required by this Agreement is more than ten (10) days late, Borrowers will be charged 6.000% of the unpaid portion of the regularly scheduled payment or \$5.00, whichever is greater.

II.6 Funding Loss. Borrowers shall pay to Lender any amounts required to compensate Lender for any Funding Loss.

<del>23</del> 4864-5069-8851 v.9

### II.7 Uncommitted Increase in Commitment-

(a) **Request for Increase.** So long as no Default or Event of Default has occurred and is continuing, upon notice to Lender, Borrower may from time to time, request an increase in the aggregate Commitment (a "*Revolving Facility Increase*") by an amount (for all such requests) not exceeding \$5,000,000; *provided* that (i) any such request for an increase shall be in a minimum amount of \$1,000,000, (ii) Borrower may make a maximum of three such requests, and (iii) Lender shall not be required or otherwise obligated to provide any portion of such Revolving Facility Increase.

(b) **Conditions to Effectiveness of Increase.** As a condition precedent to any such increase, (i) Borrowers shall deliver to Lender a certificate of the Borrowers dated as of the effective date of the Revolving Facility Increase (the "*Increase Effective Date*") signed by a Responsible Officer of Borrowers, in each case in form and substance satisfactory to Lender, (A) certifying and attaching the resolutions adopted by Borrowers approving or consenting to such increase, and (B) certifying that, before and after giving effect to such increase, (1) the representations and warranties contained in *Article VI* and the other Loan Documents are true and correct on and as of the Increase Effective Date, except to the extent that such representations and warranties specifically refer to an earlier date, in which case they are true and correct as of such earlier date, and (2) no Default exists, (ii) in Lender's discretion, Borrowers shall pay a Commitment Fee in the amount of 1.0% of the amount of the Revolving Facility Increase, and (iii) Borrowers shall submit additional information and documentation as the Lender may reasonably request from Borrowers.

## II.8 Letters of Credit.

(a) Subject to the terms and conditions of this Agreement, upon request of Borrowers, Lender in its sole discretion agrees to issue one or more Letters of Credit for the account of Borrowers from time to time from the date hereof to and including the Maturity Date; provided, however, that the outstanding Letter of Credit Liabilities shall not at any time exceed the lesser of (a) Three Million Dollars (\$3,000,000) and (b) an amount equal to the amount of the Commitment minus the outstanding Revolving Credit Advances. Each Letter of Credit shall have an expiration date not to exceed three hundred sixty-five (365) days, shall not have an expiration date beyond the Maturity Date, shall be payable in Dollars, shall have a minimum face amount of Twenty-Five Thousand Dollars (\$25,000), must support a transaction that is entered into in the ordinary course of Borrowers' business, must be satisfactory in form and substance to Lender, will be subject to the payment of such Letter of Credit fees as Lender may require, and shall be issued pursuant to such documents and instruments executed by Borrowers (including, without limitation, a Letter of Credit Application as then in effect) as Lender may require. Notwithstanding that a Letter of Credit issued or outstanding hereunder is in support of any obligations of, or is for the account of, a Subsidiary of a Borrower, Borrowers shall be obligated to reimburse Lender hereunder for any and all drawings under such Letter of Credit. Each Borrowers' business derives substantial benefits from the businesses of such Subsidiaries.

(b) **Fees.** Borrowers agree to pay to Lender, as a condition precedent to the issuance (including the extension) of each Letter of Credit, an issuance fee payable on the date of issuance in an amount to be agreed between Borrowers and Lender prior to the date of such issuance.

(c) **Reimbursement.** Upon receipt from the beneficiary of any Letter of Credit of any notice of a drawing under such Letter of Credit, Lender shall notify the Borrowers thereof. Each payment by Lender pursuant to a drawing under a Letter of Credit is required to be reimbursed by Borrowers to Lender and payable immediately upon receipt of such notice and, if Borrowers fail to so reimburse Lender, shall be charged by Lender as (and in such event will be deemed to be) a Revolving Credit Advance by Lender to Borrowers under the Revolving Credit Note and this Agreement as of the day and time such payment is made by Lender and in the amount of such payment.

## Article III. PAYMENTS

III.1 **Method of Payment**. All payments of principal, interest, and other amounts to be made by Borrowers under this Agreement and the other Loan Documents shall be made in immediately available funds in Dollars at the Principal Office (or at such other place as Lender, in Lender's sole discretion, may have established by delivery of written notice thereof to Borrowers from time to time), without offset, in lawful money of the United States of America, which shall at the time of payment be legal tender in payment of all debts and dues, public and private. Payments by check or draft shall not constitute payment in immediately available funds until the required amount is actually received by Lender in full. Payments in immediately available funds received by Lender in the place designated for payment on a Business Day prior to 11:00 a.m. (Dallas, Texas time) at such place of payment shall be credited prior to the close of business on the Business Day received, while payments received by Lender on a day other than a Business Day or after 11:00 a.m. (Dallas, Texas time) on a Business Day shall not be credited until the next succeeding Business Day. If any payment of principal or interest on the Notes shall become due and payable on a day other than a Business Day, then such payment shall be made on the next succeeding Business Day. Any such extension of time for payment shall be included in computing interest which has accrued and shall be payable in connection with such payment.

### III.2 Prepayments.

(a) **Voluntary Prepayments**. Borrowers may prepay all or any portion of the Notes at any time without fee, premium or penalty, all or any portion of the outstanding principal balance hereof; *provided, that*, (i) such prepayment shall also include any and all accrued but unpaid interest on the amount of principal being so prepaid through and including the date of prepayment, plus any other sums which have become due to Lender under the other Loan Documents on or before the date of prepayment, but which have not been fully paid and (ii) such prepayment shall also include any Funding Loss. Prepayments shall be in a minimum of \$10,000. Notwithstanding the provisions of this paragraph, Borrowers must consult with Lender prior to making any prepayments when a Hedge Agreement has been executed between Borrowers and Lender in connection with the Notes. Borrowers acknowledge that partial prepayments of the Notes may require the Hedge Agreement to be amended, and full prepayment will terminate the Hedge Agreement. Full and partial prepayments will trigger an early termination valuation under the Hedge Agreement. Thus, an early termination fee may occur under the Hedge Agreement upon partial and full prepayment of the Notes. Notwithstanding the provisions of this paragraph, Borrowers shall remain obligated to pay any fee due and owing under the Hedge Agreement, including but not limited to any fee owed upon early termination of the Hedge Agreement.

(b) **Mandatory Prepayment of Revolving Credit Advances**. (i) On and after December 31, 2024, Borrowers must pay on DEMAND the amount by which at any time the

unpaid principal balance of the Revolving Credit Note exceeds the lesser of ( $\underline{iA}$ ) the amount of the Commitment or ( $\underline{iiB}$ ) Revolving Credit Availability:, (ii) Borrowers shall make principal payments on the Notes equal to (A) \$1,000,000 on or before January 15, 2025 and (B) \$2,000,000 on or before April 15, 2025, and (iii) Borrowers shall make the payments as required pursuant to *Section 7.16*.

# Article IV. SECURITY

IV.1 **Collateral**. To secure full and complete payment and performance of the Obligations, Borrowers and each Guarantor shall execute and deliver or cause to be executed and delivered all of the Security Documents required by Lender covering the Property and Collateral described in such Security Documents. Each Obligated Party shall execute and cause to be executed such further documents and instruments, including without limitation, Uniform Commercial Code financing statements, as Lender, in its sole discretion, deems necessary or desirable to create, evidence, preserve, and perfect its liens and security interests in the Collateral.

IV.2 Setoff. If an Event of Default shall have occurred and be continuing, Lender shall have the right to set off and apply against the Obligations in such manner as Lender may determine, at any time and without notice to Borrowers, any and all deposits (general or special, time or demand, provisional or final) or other sums at any time credited by or owing from Lender to Borrowers whether or not the Obligations are then due. As further security for the Obligations, Borrowers hereby grant to Lender a security interest in all money, instruments, and other property of Borrowers now or hereafter held by Lender, including, without limitation, property held in safekeeping. In addition to Lender's right of setoff and as further security for the Obligations, Borrowers now or hereafter on deposits (general or special, time or demand, provisional or final) and other accounts of Borrowers now or hereafter on deposit with or held by Lender and all other sums at any time credited by or owing from Lender to Borrowers. The rights and remedies of Lender hereunder are in addition to other rights of setoff) which Lender may have.

## Article V. CONDITIONS PRECEDENT

V.1 **Initial Extension of Credit**. The obligation of Lender to make the initial Advance under any Note is subject to the condition precedent that Lender shall have received on or before the day of such Advance all of the following, each dated (unless otherwise indicated) the date hereof, in form and substance satisfactory to Lender:

(a) **Resolutions**. Resolutions of the applicable governing body of each Borrower and each other Obligated Party which authorize the execution, delivery, and performance by such Person of this Agreement and the other Loan Documents to which such Person is or is to be a party;

(b) **Incumbency Certificate**. A certificate of incumbency certified by a Responsible Officer certifying the names of the individuals or other Persons authorized to sign this Agreement and each of the other Loan Documents to which each Borrower and each other Obligated Party is or is to be a party (including the certificates contemplated herein) on behalf of such Person together with specimen signatures of such individual Persons;

<del>26</del> 4<del>864-</del> <del>5069-</del> <del>8851 v</del> (c) **Constituent Documents**. The Constituent Documents for each Borrower and each other Obligated Party certified as of a date acceptable to Lender by the appropriate government officials of the state of incorporation or organization of each such party;

(d) **Governmental Certificates**. Certificates of the appropriate government officials of the state of incorporation or organization of each Borrower, each other Obligated Party and each Pledgor as to the existence and good standing of each such party, each dated within fifteen (15) days (or such longer period acceptable to Lender) prior to the date of the initial Advance;

- (e) **Notes**. The Notes executed by Borrowers;
- (f) Intercreditor Agreement. A fully executed copy of the Intercreditor Agreement;
- (g) Term Loan Documents. Executed copies of the Term Loan Documents in form reasonably acceptable to Lender;
- (h) Security Documents. The Security Documents executed by Borrowers, the other Obligated Parties and the Pledgors;

(i) **Financing Statements.** UCC financing statements reflecting each of the Obligated Parties, as debtors, and Lender, as secured party, which are required to grant a Lien which secures the Obligations and covering such Collateral as Lender may request;

(j) **Insurance Matters**. Copies of insurance certificates describing all insurance policies required by *Section 7.05* together with loss payable and lender endorsements in favor of Lender with respect to all insurance policies covering Collateral;

(k) UCC Search. The results of a Uniform Commercial Code search showing all financing statements and other documents or instruments on file against each Borrower, and each other Obligated Party in the appropriate filing offices, such search to be as of a date no more than fifteen (15) days (or such longer period acceptable to Lender) prior to the date of the initial Advance;

(1) Attorneys' Fees and Expenses. Evidence that the costs and expenses (including reasonable attorneys' fees) referred to in *Section 11.01*, to the extent incurred, shall have been paid in full by Borrowers;

# (m) KYC Information.

(i) Upon the reasonable request of Lender made at least five (5) days prior to the date hereof, Borrowers shall have provided to Lender, and Lender shall be reasonably satisfied with, the documentation and other information so requested in connection with applicable "know your customer" and anti-money-laundering rules and regulations, including, without limitation, the Patriot Act; and

(ii) At least five (5) days prior to the date hereof, if any Borrower qualifies as a "legal entity customer" under the Beneficial Ownership Regulation,

such Borrower shall deliver, to Lender, a Beneficial Ownership Certification in relation to such Borrower;

(n) **Existing Debt of Obligated Parties**. All of the existing Debt of each Obligated Party (*other than* Debt permitted to exist pursuant to *Section 8.01*) will be repaid in full and all security interests related thereto shall be terminated on or prior to the date hereof;

(o) **Opinions of Counsel.** Customary opinions of legal counsel to Borrowers, each Obligated Party and each Pledgor as to such other matters as Lender may reasonably request;

(p) **Closing Fees.** Evidence that the Commitment Fee and any other fees due at closing have been paid;

(q) **Minimum EBITDA**. Evidence that Borrowers had EBITDA of not less than \$8,000,000 for the trailing twelve-month period ending December 31, 2022, in form and substance acceptable to Lender; and

V.2 All Extensions of Credit. The obligation of Lender to make any Advance or issue any Letter of Credit (including the initial Advance and the initial Letter of Credit) is subject to the following additional conditions precedent:

(a) **Request for Advance or Letter of Credit**. Lender shall have received in accordance with this Agreement, as the case may be, an Advance Request Form or Letter of Credit Request Form pursuant to Lender's requirements dated the date of such Advance or Letter of Credit and executed by a Responsible Officer of Borrowers;

(b) No Default. No Event of Default shall have occurred and be continuing, or would result from or after giving effect to such Advance or Letter of Credit as evidenced by a Compliance Certificate;

(c) No Material Adverse Event. No Material Adverse Event has occurred, and no circumstance exists that could be a Material Adverse Event;

(d) **Representations and Warranties**. All of the representations and warranties contained in *Article VI* hereof and in the other Loan Documents shall be true and correct in all material respects on and as of the date of such Advance with the same force and effect as if such representations and warranties had been made on and as of such date, except that for purposes of this *Section 5.02(d)*, the representations and warranties contained in *Section 6.02* shall be deemed to refer to (i) Borrowers and the Subsidiaries and (ii) the most recent financial statements furnished pursuant to *clauses (a)* and 7.01(c) of *Section 7.01* and any representations and warranties made as of an earlier date shall be true and correct in all material respects as of such earlier date; and

(e) Additional Documentation. Lender shall have received such additional approvals, or documents as Lender or its legal counsel may reasonably request consistent with the transaction contemplated in this Agreement.

## Article VI. REPRESENTATIONS AND WARRANTIES

To induce Lender to enter into this Agreement, and except as set forth on the Schedules, Borrowers represents and warrants to Lender that:

VI.1 **Corporate Existence**. Each Borrower and each Subsidiary (a) is duly incorporated or organized, as the case may be, validly existing, and in good standing under the laws of the jurisdiction of its incorporation or organization; (b) has all requisite power and authority to own its assets and carry on its business as now being or as proposed to be conducted; and (c) is qualified to do business in all jurisdictions in which the nature of its business makes such qualification necessary and where failure to so qualify could result in a Material Adverse Event. Each Borrower and each of the other Obligated Parties has the power and authority to execute, deliver, and perform its obligations under this Agreement and the other Loan Documents to which it is or may become a party.

VI.2 **Financial Statements, Etc.** Borrowers have delivered to Lender audited financial statements of Borrowers as at and for the fiscal year ended December 31, 2022 and unaudited financial statements of Borrowers as of March 31, 2023. Such financial statements are true and correct, have been prepared in accordance with GAAP, and fairly and accurately present, on a Consolidated Basis, the financial condition of Borrowers as of the respective dates indicated therein and the results of operations for the respective periods indicated therein. No Borrower nor any Subsidiary nor any other Obligated Party has any material contingent liabilities, liabilities for taxes, unusual forward or long-term commitments, or unrealized or anticipated losses from any unfavorable commitments referred to in this *Section 6.02*. All projections delivered by Borrowers to Lender have been prepared in good faith, with care and diligence and use assumptions that are reasonable under the circumstances at the time such projections were prepared and delivered to Lender and all such assumptions are disclosed in the projections (it being understood for purposes that such projections are subject to significant uncertainties and contingencies, many of which are beyond the control of the Borrowers, that no assurance is given that any particular projections will be realized, and that actual results may differ from the projected results). No Borrower nor any Subsidiary has any material Guarantees, contingent liabilities, liabilities for taxes, or any long-term leases or unusual forward or long-term commitments, or any Hedge Agreement or other transaction or obligation in respect of derivatives, that are not reflected in the most-recent financial statements referred to in this *Section 6.02*. Other than the Debt listed on *Schedule 8.01*, Debt reflected on the financial statements delivered pursuant to *Sections 1.01(c)*, and Debt otherwise permitted by *Section 8.01*, each Borrower and each Subsidiary has no Debt.

VI.3 Action; No Breach. The execution, delivery, and performance by each Borrower and each other Obligated Party of this Agreement and the other Loan Documents to which such Person is or may become a party and compliance with the terms and provisions hereof and thereof have been duly authorized by all requisite action on the part of such Person and do not and will not (a) violate or conflict with, or result in a breach of, or require any consent under (i) the Constituent Documents of such Person, (ii) any applicable law, rule, or regulation or any order, writ, injunction, or decree of any Governmental Authority or arbitrator, or (iii) any agreement or instrument to which such Person is a party or by which it or any of its Properties is bound or subject, or (b) constitute a default under any such agreement or instrument, or result in the creation or imposition of any Lien upon any of the revenues or assets of such Person.

<del>29</del> <del>4864-</del> <del>5069-</del> <del>8851 v.</del> VI.4 **Operation of Business**. Borrowers and each of the Subsidiaries possess all licenses, permits, franchises, patents, copyrights, trademarks, and tradenames, or rights thereto, necessary to conduct their respective businesses substantially as now conducted and as presently proposed to be conducted, and Borrowers and each of the Subsidiaries are not in violation of any valid rights of others with respect to any of the foregoing.

VI.5 **Litigation and Judgments**. There is no action, suit, investigation, or proceeding before or by any Governmental Authority or arbitrator pending, or to the knowledge of Borrowers, threatened against or affecting any Borrower or any of the Subsidiaries, that would, if adversely determined, would constitute a Material Adverse Event on the business, condition (financial or otherwise), operations, or properties of any Borrower or any of the Subsidiaries or the ability of any Borrower to pay and perform the Obligations. There are no outstanding judgments against any Borrower or any Subsidiary.

VI.6 **Rights in Properties; Liens**. Borrowers and each of the Subsidiaries have good and indefeasible title to or valid leasehold interests in their respective Properties, including the Properties reflected in the financial statements described in *Section 6.02*, and none of the Properties of Borrowers or any Subsidiary is subject to any Lien, except Permitted Liens.

VI.7 **Enforceability**. This Agreement constitutes, and the other Loan Documents to which Borrowers or any other Obligated Party is a party, when delivered, shall constitute legal, valid, and binding obligations of such Person, enforceable against such Person in accordance with their respective terms, except as limited by bankruptcy, insolvency, or other laws of general application relating to the enforcement of creditors' rights.

VI.8 **Approvals**. No authorization, approval, or consent of, and no filing or registration with, any Governmental Authority or third party is or will be necessary for the execution, delivery, or performance by Borrowers of this Agreement and the other Loan Documents to which each Borrower is or may become a party or the validity or enforceability thereof.

VI.9 **Taxes.** Each Borrower and each Subsidiary have filed all tax returns (federal, state, and local) required to be filed, including all income, franchise, employment, Property, and sales tax returns, and have paid all of their respective liabilities for taxes, assessments, governmental charges, and other levies that are due and payable, except where the failure to file such tax returns or to pay such taxes would not reasonably be expected to result in a Material Adverse Event. No Borrower knows of any pending investigation of any Borrower or any of the Subsidiaries by any taxing authority or of any pending but unassessed tax liability of Borrowers or any of the Subsidiaries.

VI.10 Use of Proceeds; Margin Securities. No Borrower nor any Subsidiary is engaged principally, or as one of its important activities, in the business of extending credit for the purpose of purchasing or carrying margin stock (within the meaning of Regulations G, T, U, or X of the Board of Governors of the Federal Reserve System), and no part of the proceeds of any Advance will be used to purchase or carry any margin stock or to extend credit to others for the purpose of purchasing or carrying margin stock.

VI.11 ERISA. Each Plan that is intended to qualify under *Section 401(a)* of the Code has received a favorable determination letter from the IRS or an application for such a letter is currently being processed by the IRS with respect thereto and, to the knowledge of Borrowers, nothing has occurred which would prevent, or cause the loss of, such qualification. No application for a funding waiver or an extension of any amortization period pursuant to *Section 412* of the Code has been made with respect to

<del>30</del> 4<del>864-</del> <del>5069-</del> <del>8851 v:</del> any Plan. There are no pending or, to the knowledge of Borrowers, threatened claims, actions or lawsuits, or action by any Governmental Authority, with respect to any Plan. There has been no Prohibited Transaction or violation of the fiduciary responsibility rules with respect to any Plan. No ERISA Event has occurred or is reasonably expected to occur. No Plan has any Unfunded Pension Liability. No Obligated Party or ERISA Affiliate has incurred, or reasonably expects to incur, any liability under Title IV of ERISA with respect to any Plan (other than premiums due and not delinquent under *Section 4007* of ERISA). No Obligated Party or ERISA Affiliate has incurred, or reasonably expects to incur, any liability (and no event has occurred which, with the giving of notice under *Section 4219* of ERISA, would result in such liability) under *Section 4201* or *4243* of ERISA with respect to a Multiemployer Plan. No Obligated Party or ERISA Affiliate has engaged in a transaction that could be subject to *Section 4069* or *4212(c)* of ERISA.

VI.12 **Disclosure**. No written statement, information, report, representation, or warranty made by any Borrower or any other Obligated Party in this Agreement or in any other Loan Document or furnished to Lender in connection with this Agreement or any of the transactions contemplated hereby contains any untrue statement of a material fact or omits to state any material fact necessary to make the statements herein or therein not misleading. There is no fact known to any Borrower which is a Material Adverse Event, or which might in the future be a Material Adverse Event that has not been disclosed in writing to Lender.

VI.13 Subsidiaries, Ventures, Etc. Borrowers have no Subsidiaries, or joint ventures or partnerships other than those listed on *Schedule 6.13* and *Schedule 6.13* sets forth the jurisdiction of incorporation or organization of each such Person and, percentage of Borrowers' ownership interest in such Person and whether each such Person is an Immaterial Subsidiary. All of the outstanding capital stock or other ownership interest of each Person described in *Schedule 6.13* has been validly issued, is fully paid, and is non-assessable (except with respect to limited liability company interests). There are no outstanding subscriptions, options, warrants, calls, rights or other agreements or commitments (other than stock or similar options granted to employees or directors and directors' qualifying shares) of any nature relating to any Equity Interests of Borrowers or any Subsidiary, except as created by the Loan Documents. Each Domestic Subsidiary of Borrowers is a Subsidiary Guarantor (other than any Immaterial Subsidiary). Taken in the aggregate, all Immaterial Subsidiaries under this Agreement do not have any material assets, liabilities or business operations.

VI.14 Agreements. No Borrower nor any Subsidiary is a party to any indenture, loan, or credit agreement, or to any lease or other agreement or instrument, or subject to any charter or corporate or other organizational restriction which could create or cause a Material Adverse Event on the business, condition (financial or otherwise), operations, or properties of Borrowers or any Subsidiary, or the ability of Borrowers to pay and perform its obligations under the Loan Documents to which it is a party. No Borrower nor any Subsidiary is in default in any material respect in the performance, observance, or fulfillment of any of the obligations, covenants, or conditions contained in any agreement or instrument material to its business to which it is a party.

VI.15 **Compliance with Laws**. No Borrower nor any Subsidiary is in violation in any material respect of any law, rule, regulation, order, or decree of any Governmental Authority or arbitrator.

VI.16 **Regulated Entities**. No Borrower nor any Subsidiary is (a) an *"investment company"* or a company *"controlled"* by an *"investment company"* within the meaning of the Investment Company Act of 1940, as amended, or (b) subject to regulation under the Federal Power Act, the Interstate Commerce

<del>31</del> <del>4864-</del> <del>5069-</del> <del>8851 v.</del> Act, any state public utilities code, or any other federal or state statute, rule or regulation limiting its ability to incur Debt, pledge its assets or perform its obligations under the Loan Documents.

### VI.17 Environmental Matters.

(a) Each Borrower, each Subsidiary, and all of their respective properties, assets, and operations are in compliance in all material respects with all Environmental Laws. No Borrower is aware of, nor has Borrower received notice of, any past, present, or future conditions, events, activities, practices, or incidents which may interfere with or prevent the compliance or continued compliance of Borrowers and the Subsidiaries with all Environmental Laws;

(b) Each Borrower and each Subsidiary have obtained all permits, licenses, and authorizations that are required under applicable Environmental Laws, and all such permits are in good standing and each Borrower and the Subsidiaries are in compliance with all of the terms and conditions of such permits except to the extent that it would not cause a Material Adverse Event;

(c) No Hazardous Materials exist on, about, or within or have been used, generated, stored, transported, disposed of on, or Released from any of the properties or assets of any Borrower or any Subsidiary except in accordance with Environmental Laws. The use which Borrowers and the Subsidiaries make and intend to make of their respective properties and assets will not result in the use, generation, storage, transportation, accumulation, disposal, or Release of any Hazardous Material on, in, or from any of their properties or assets except in accordance with Environmental Laws;

(d) No Borrower nor any Subsidiary nor any of their respective currently or previously owned or, to any Borrower's knowledge, leased properties or operations is subject to any outstanding or threatened order from or agreement with any Governmental Authority or other Person or subject to any judicial or docketed administrative proceeding with respect to (i) failure to comply with Environmental Laws, (ii) Remedial Action, or (iii) any Environmental Liabilities arising from a Release or threatened Release;

(e) There are no conditions or circumstances associated with the currently or previously owned or, to any Borrower's knowledge, leased properties or operations of any Borrower or any Subsidiary that could reasonably be expected to give rise to any Environmental Liabilities;

(f) No Borrower nor any of the Subsidiaries is a treatment, storage, or disposal facility requiring a permit under the Resource Conservation and Recovery Act, 42 U.S.C. §6901 *et seq.*, regulations thereunder or any comparable provision of state law. Borrowers and the Subsidiaries are in compliance in all material respects with all applicable financial responsibility requirements of all Environmental Laws;

(g) No Borrower nor any Subsidiary has filed or failed to file any notice required under applicable Environmental Law reporting a Release except to the extent that it would not cause a Material Adverse Event; and

(h) No Lien arising under any Environmental Law has attached to any property or revenues of any Borrower or the Subsidiaries except to the extent that it would not cause a Material Adverse Event.

VI.18 Intellectual Property. All material copyrights, trademarks and patents owned or used by Borrowers and the Subsidiaries is listed, together with application or registration numbers, where applicable, in *Schedule 6.18*. Each Person identified on *Schedule 6.18* owns, or is licensed to use, all intellectual property necessary to conduct its business as currently conducted except for such Intellectual Property the failure of which to own or license could be a Material Adverse Event. Each Person identified on *Schedule 6.18* will maintain the patenting and registration of all copyrights, trademarks and patents with the United States Patent and Trademark Office, the United States Copyright Office, or other appropriate Governmental Authority, and each Person identified on *Schedule 6.18* will promptly patent or register, as the case may be, all new copyrights, trademarks and patents and notify Lender in writing five (5) Business Days prior to filing any such new patent or registration.

VI.19 Foreign Assets Control Regulations and Anti-Money Laundering. Each Obligated Party and each Subsidiary of each Obligated Party is and will remain in compliance in all material respects with all United States economic sanctions laws, Executive Orders and implementing regulations as promulgated by the United States Treasury Department's Office of Foreign Assets Control ("*OFAC*"), and all applicable anti-money laundering and counter-terrorism financing provisions of the Bank Secrecy Act and all regulations issued pursuant to it. No Obligated Party and no Subsidiary, and to Borrowers' knowledge, no Affiliate, or any director, officer, employee, agent, affiliate or representative of any Obligated Party, is an individual or entity that is (a) currently the subject or target of any Sanctions, (b) a Person designated by the United States government on the list of the Specially Designated Nationals and Blocked Persons (the "*SDN List*") with which a United States Person cannot deal with or otherwise engage in business transactions, or included on HMT's Consolidated List of Financial Sanctions Targets and the Investment Ban List or any similar list enforced by any other relevant sanctions authority, (c) a Person who is otherwise the target of United States economic sanction laws such that a United States Person cannot deal or otherwise the target of united States economic sanction laws such that a United States Person cannot deal or otherwise therewise the target of resident in a Designated Jurisdiction.

VI.20 **Patriot Act**. The Obligated Parties, each of their Subsidiaries, and, to Borrowers' knowledge, each of their Affiliates are in compliance with (a) the Trading with the Enemy Act, and each of the foreign assets control regulations of the United States Treasury Department (31 CFR, Subtitle B Chapter V, as amended), and all other enabling legislation or executive order relating thereto, (b) the Patriot Act, and (c) all other federal or state laws relating to "*know your customer*" and anti-money laundering rules and regulations. No part of the proceeds of any Revolving Loan will be used directly or indirectly for any payments to any government official or employee, political party, official of a political party, candidate for political office, or anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the United States Foreign Corrupt Practices Act of 1977.

VI.21 Solvency. Borrowers and their Subsidiaries, on a Consolidated Basis, are solvent and have not entered into any transaction with the intent to hinder, delay or defraud a creditor.

VI.22 Anti-Corruption Laws. Each Obligated Party and each Subsidiary of each Obligated Party has conducted their businesses in compliance with the United States Foreign Corrupt Practices Act of 1977, the UK Bribery Act 2010, and other similar anti-corruption legislation in other jurisdictions, and has instituted and maintained policies and procedures designed to promote and achieve compliance with such laws.

VI.23 Beneficial Ownership Regulation. The information included in the Beneficial Ownership Certification is true and correct in all respects.

<del>33</del> <del>4864-</del> <del>5069-</del> <del>8851 v.</del>

## Article VII. AFFIRMATIVE COVENANTS

Each Borrower covenants and agrees that, as long as the Obligations or any part thereof are outstanding, or Lender has any Commitment hereunder, such Borrower will perform and observe the following positive covenants, unless Lender shall otherwise consent in writing:

VII.1 Reporting Requirements. Borrowers will furnish to Lender:

(a) **Annual Financial Statements**. As soon as available, and in any event, within one hundred twenty (120) days after the fiscal year of Borrowers, a copy of the annual audit report of Borrowers and the Subsidiaries for such fiscal year containing, on a Consolidated and Consolidating Basis, balance sheets and statements of income, retained earnings, and cash flow as at the end of such fiscal year and for the twelve-month period then ended, in each case setting forth in comparative form the figures for the preceding fiscal year, all in reasonable detail and audited and certified by independent certified public accountants of recognized standing acceptable to Lender, to the effect that such report has been prepared in accordance with GAAP and containing no material qualifications or limitations on scope;

(b) **Compliance Certificate**. As soon as available, and in any event within forty-five (45) days after the end of each fiscal quarter of Borrowers, and together with the delivery of the financial statements required pursuant to *Section 7.01(a)* above, a Compliance Certificate executed by a Responsible Officer of Borrowers;

(c) **Monthly Financial Statements**. As soon as available, and in any event within thirty (30) days after the end of each calendar month, a copy of an unaudited financial report of Borrowers and the Subsidiaries as of the end of such month and for the portion of the fiscal year then ended, containing, on a consolidated and consolidating basis, balance sheets and statements of income, retained earnings, and cash flow, all in reasonable detail certified by a Responsible Officer of Borrowers to have been prepared in accordance with GAAP and to fairly and accurately present (subject to year-end audit adjustments) the financial condition and results of operations of Borrowers and the Subsidiaries, on a consolidated and consolidating basis, at the date and for the periods indicated therein;

(d) **Eligible Account Report**. As soon as available, and in any event within thirty (30) days after the end of each calendar month, an Eligible Account Report, in substantially the form of *Exhibit A* attached hereto, certified by a Responsible Officer of Borrowers;

(e) Accounts Receivable and Accounts Payable Aging. As soon as available, and in any event within thirty (30) days after the end of each calendar month, an account receivable aging, classifying Borrowers' and the Subsidiaries' accounts receivable in categories of 0-30, 31-60, 61-90 and over ninety (90) days from date of invoice, and in such form and detail as Lender shall require, and account payable aging classifying Borrowers' and the Subsidiaries' accounts payable by categories of 0-30, 31-60 and over sixty (60), from date of invoice, also in such detail as Lender shall reasonably require, and in each case certified by a Responsible Officer of Borrowers;

(f) **Management Letters**. Promptly upon receipt thereof, a copy of any management letter or written report submitted to Borrowers or any Subsidiary by independent certified public

<del>34</del> 4<del>864-</del> <del>5069-</del> <del>8851 v.</del> accountants with respect to the business, condition (financial or otherwise), operations, or properties of Borrowers or any Subsidiary;

(g) **Notice of Litigation**. Promptly after the commencement thereof, notice of all actions, suits, and proceedings before any Governmental Authority or arbitrator affecting any Borrower or any Subsidiary which, if determined adversely to such Borrower or such Subsidiary, could cause or create a Material Adverse Event on the business, condition (financial or otherwise), operations, or properties of such Borrower or such Subsidiary;

(h) **Notice of Material Adverse Event**. As soon as possible and in any event within five (5) Business Days after the occurrence thereof, written notice of any event or circumstance that could reasonably be expected to result in a Material Adverse Event;

(i) **ERISA Reports**. Promptly after the filing or receipt thereof, copies of all reports, including annual reports, and notices which any Borrower or any Subsidiary files with or receives from the PBGC or the U.S. Department of Labor under ERISA; and as soon as possible and in any event within ten (10) days after any Borrower or any Subsidiary knows or has reason to know that any Reportable Event or Prohibited Transaction has occurred with respect to any Plan or that the PBGC or any Borrower or any Subsidiary has instituted or will institute proceedings under Title IV of ERISA to terminate any Plan, a certificate of a Responsible Officer of such Borrower setting forth the details as to such Reportable Event or Prohibited Transaction or Plan termination and the action that such Borrower proposes to take with respect thereto;

(j) **Annual Projections**. As soon as available, but in any event not more than forty-five (45) days after the end of each fiscal year of Borrowers, forecasts prepared by management of Borrowers and approved by the members of Borrowers, in form and substance reasonably satisfactory to Lender, of financial projections of Borrowers and the Subsidiaries on a monthly basis for the current fiscal year;

(k) **KYC**. Promptly following any request therefor, Borrowers shall provide information and documentation reasonably requested by Lender for purposes of compliance with applicable "know your customer" requirements under the Patriot Act, the Beneficial Ownership Regulation or other applicable anti-money laundering laws, including but not limited to a Beneficial Ownership Certification form acceptable to Lender; and

(l) General Information. Promptly, such other information concerning Borrowers, or any Subsidiary or Obligated Party as Lender may from time to time reasonably request.

VII.2 Maintenance of Existence; Conduct of Business. Borrowers will preserve and maintain, and will cause each Subsidiary to preserve and maintain, its existence and all of its leases, privileges, licenses, permits, franchises, qualifications, and rights the failure of which to maintain would result in a Material Adverse Event. Borrowers will conduct, and will cause each Subsidiary to conduct, its business in an orderly and efficient manner in accordance with good business practices. Without limitation, Borrowers will not make (and will not permit any of the Subsidiaries to make) any material change in its credit collection policies if such change would materially impair the collectability of any account, nor will it rescind, cancel or modify any account except in the ordinary course of business.

VII.3 **Maintenance of Properties**. Borrowers will maintain, keep, and preserve, and cause each Subsidiary to maintain, keep, and preserve, all of its Properties (tangible and intangible) necessary or useful in the proper conduct of its business in good working order and condition.

<del>35</del> <del>4864-</del> <del>5069-</del> <del>8851 v:</del> VII.4 **Taxes and Claims**. Borrowers will pay or discharge, and will cause each Subsidiary to pay or discharge, at or before maturity or before becoming delinquent (a) all taxes, levies, assessments, and governmental charges imposed on it or its income or profits or any of its property, and (b) all lawful claims for labor, material, and supplies, which, if unpaid, might become a Lien upon any of its property; provided, however, that no Borrower nor any Subsidiary shall be required to pay or discharge (i) any tax, levy, assessment, or governmental charge or (ii) such Lien for labor, material or supplies, which is (y) being contested in good faith by appropriate proceedings diligently pursued, and for which adequate reserves have been established or (z) where such failure to pay or discharge would not reasonably be expected to result in a Material Adverse Event.

## VII.5 Insurance.

(a) Borrowers shall, and shall cause each of the Subsidiaries to, maintain insurance with financially sound and reputable insurance companies in such amounts and covering such risks as is usually carried by corporations engaged in similar businesses and owning similar Properties in the same general areas in which Borrowers and the Subsidiaries operate, *provided that* in any event Borrowers will maintain and cause each of the Subsidiaries to maintain workmen's compensation insurance, property insurance, comprehensive general liability insurance, reasonably satisfactory to Lender. Each insurance policy covering Collateral shall name Lender as loss payee and each insurance policy covering liabilities shall name Lender as additional insured, and each such insurance policy shall provide that such policy will not be cancelled or reduced without thirty (30) days prior written notice to Lender.

(b) During the continuance of an Event of Default, all proceeds of insurance shall be paid over to Lender for application to the Obligations. So long as no Event of Default is continuing, subject to *Section 7.05(c)*, all proceeds of insurance in excess of \$50,000 shall be paid over to Lender for application to the Obligations.

(c) Borrowers may apply the net proceeds of a casualty or condemnation (each a "Loss") to the repair, restoration, or replacement of the assets suffering such Loss, so long as (i) such repair, restoration, or replacement is completed within two hundred seventy (270) days after the date of such Loss (or such longer period of time agreed to in writing by Lender), (ii) while such repair, restoration, or replacement is underway, all of such net proceeds are on deposit with Lender in a separate deposit account over which Lender has exclusive control, and (iii) such Loss did not cause an Event of Default. If an Event of Default occurs pursuant to which Lender exercises its rights to accelerate the Obligations as provided in Section 10.02 or such repair, restoration, or replacement is not completed within two hundred seventy (270) days of the date of such Loss (or such longer period of time agreed to in writing by Lender), then Lender may immediately and without notice to any Person apply all of such net proceeds to the Obligations, regardless of any other prior agreement regarding the disposition of such net proceeds.

VII.6 **Inspection Rights**. Upon reasonable prior notice to Borrowers from Lender, and at any reasonable time and from time to time, Borrowers shall, and shall cause each of the Subsidiaries to, (a) permit representatives of Lender to examine, inspect, review, evaluate and make physical verifications and appraisals of the inventory and other Collateral in any manner and through any medium that Lender considers advisable, (b) to examine, copy, and make extracts from its books and records, (c) to visit and inspect its Properties, and (d) to discuss its business, operations, and financial condition with its officers, employees, and independent certified public accountants, in each instance, at Borrowers' expense;

<del>36</del> 4864-<del>5069-</del> 8851 v.9 provided, that so long as no Default or Event of Default has occurred and is continuing such inspection rights shall be limited to no more than twice per calendar year.

VII.7 Keeping Books and Records. Borrowers will maintain, and will cause each Subsidiary to maintain, proper books of record and account in which full, true, and correct entries in conformity with GAAP shall be made of all dealings and transactions in relation to its business and activities.

VII.8 **Compliance with Laws**. Borrowers will comply, and will cause each Subsidiary to comply, in all material respects with all applicable laws, rules, regulations, orders, and decrees of any Governmental Authority or arbitrator.

VII.9 **Compliance with Agreements**. Borrowers will comply, and will cause each Subsidiary to comply, in all material respects with all agreements, contracts, and instruments binding on it or affecting its properties or business.

VII.10 **Further Assurances**. Borrowers will, and will cause each Subsidiary to, execute and deliver such further agreements and instruments and take such further action (including promptly completing any registration or stamping of documents as may be applicable) as may be requested by Lender to carry out the provisions and purposes of this Agreement and the other Loan Documents and to create, preserve, and perfect the Liens of Lender in the Collateral.

VII.11 ERISA. Borrowers will comply, and will cause each Subsidiary to comply, with all minimum funding requirements, and all other material requirements, of ERISA, if applicable, so as not to give rise to any liability thereunder.

VII.12 **Depository Relationship**. To induce Lender to establish the interest rates provided for in the Notes, Borrowers shall, and shall cause each of the Subsidiaries to, use Lender as its principal depository bank and Borrowers shall, and shall cause each of the Subsidiaries to, maintain Lender as its principal depository bank, including for the maintenance of business, cash management, operating and administrative deposit accounts.

VII.13 Subsidiaries. Concurrently upon the formation or acquisition of any Subsidiary, after the date hereof (an "*After-Acquired Subsidiary*"), Borrowers shall cause the After-Acquired Subsidiary to deliver all of its Constituent Documents to Lender and (a) if such Subsidiary is a Domestic Subsidiary, other than an Immaterial Subsidiary, execute a Guaranty in favor of Lender and such Loan Documents as shall be required by Lender to create first priority Liens in the Priority Collateral and second priority Liens in the Term Loan Priority Collateral (in each case, subject to Liens permitted under *Section 8.02*) in favor of Lender in such After-Acquired Subsidiary's assets and such other documents as Lender deems reasonably necessary in connection with such actions and execute any other amendment to this Agreement as deemed necessary by Lender and (b) execute and deliver or cause to be delivered to Lender all Security Documents, stock certificates, stock powers and other agreements and instruments as may be required by the Security Agreement or otherwise requested by Lender to ensure that Lender has a perfected Lien on all Priority Collateral held by Borrowers or any other Obligated Party with respect to such Subsidiary.

VII.14 Keepwell. Borrowers hereby absolutely, unconditionally and irrevocably undertake to provide such funds or other support to each Specified Obligated Party with respect to such Hedge Obligations as may be needed by such Specified Obligated Party from time to time to honor all of its obligations under its Guaranty and the other Loan Documents in respect of such Hedge Obligations and to cause such Specified Obligated Party to be an Eligible Contract Participant (as defined in the Commodity

<del>37</del> <del>4864-</del> <del>5069-</del> <del>8851 v.!</del> Exchange Act) with respect to all Hedge Obligations (but, in each case, only up to the maximum amount of such liability that can be hereby incurred without rendering Borrowers' obligations and undertakings under this *Section 7.14* voidable under applicable law relating to fraudulent conveyance or fraudulent transfer, and not for any greater amount). The obligations and undertakings of Borrowers under this *Section 7.14* shall remain in full force and effect until the Obligations have been indefeasibly paid and performed in full. Borrowers intend this *Section 7.14* to constitute, and this *Section 7.14* shall be deemed to constitute, a Guarantee of the obligations of, and a "*keepwell, support, or other agreement*" (as defined in the Commodity Exchange Act) for the benefit of, each Specified Obligated Party for all purposes of the Commodity Exchange Act.

## VII.15 Deposit Account Control Agreements

. No Borrower shall maintain any deposit accounts, except with respect to which the Lender has notice, setting forth the information included for Accounts on *Schedule 3.10* to the Security Agreement. Each deposit account maintained by any Borrower shall be subject to an account control agreement satisfactory in form and substance satisfactory to the Lender.

1.1 Equity Line of Credit. On or before October 31, 2024, DDH Holdings will file a registration statement with the Securities and Exchange Commission to establish an "equity line of credit" offering with one or more institutional investors and will take all commercially reasonable efforts to cause such registration statement to be declared effective and to maintain its effectiveness and to comply with all applicable rules and regulations of the Securities and Exchange Commission and all applicable state securities "Blue Sky" laws. The net proceeds received by DDH Holdings from such offering will be applied by Borrowers, within one (1) Business Day after receipt, as a principal payment on the Notes in an amount that would cause, after giving effect to such payment, the ratio of the value of Eligible Accounts to the aggregate amount of all Revolving Credit Advances at such time outstanding to be not less than 1.0 to 1.0.

# Article VIII. NEGATIVE COVENANTS

Each Borrower covenants and agrees that, as long as the Obligations or any part thereof are outstanding, or Lender has any Commitment hereunder, Borrower will perform and observe the following negative covenants, unless Lender shall otherwise consent in writing:

VIII.1 Debt. No Borrower will incur, create, assume, or permit to exist, and will not permit any Subsidiary to incur, create, assume, or permit to exist, any Debt (other than Permitted Indebtedness), unless

- (a) There is not then an Event of Default or Default that has occurred and is continuing;
- (b) Such Debt constitutes Subordinated Debt and matures after the Maturity Date; and

(c) After giving effect to such additional Debt, such Borrower's Debt is less than 2.5 times its EBITDA over the preceding 12 month period as reported in its most recent quarterly or annual financial statements.

<del>38</del> 4<del>864-</del> <del>5069-</del> <del>8851 v.</del> VIII.2 Limitation on Liens. No Borrower will incur, create, assume, or permit to exist, and will not permit any Subsidiary to incur, create, assume, or permit to exist, any Lien upon any of its property, assets, or revenues, whether now owned or hereafter acquired, other than Permitted Liens.

VIII.3 Mergers, Etc. No Borrower will, nor will it permit any Subsidiary to, become a party to a merger or consolidation, or any Acquisition, or windup, dissolve, or liquidate, including, in each case, including pursuant to a Delaware LLC Division, other than as follows:

(a) As necessary in order to effect the Redemption/Exchange Transactions; and

(b) An Immaterial Subsidiary may (i) wind-up, dissolve or liquidate if Borrowers determine in good faith that such liquidation or dissolution is in the best interests of the Borrowers and not disadvantageous to Lender, (ii) merge into another Immaterial Subsidiary if such merger results in an Immaterial Subsidiary as the surviving entity, or (iii) merge into another Subsidiary if such merger results in a Subsidiary Guarantor as the surviving entity.

VIII.4 Restricted Payments. No Borrower shall, and nor shall it allow any Subsidiary to:

(a) repurchase or redeem any class of stock or other Equity Interest other than:

(i) (i) pursuant to employee, director or consultant repurchase plans or other similar agreements; provided, however, in each case (other than any such repurchase or redemption in the ordinary course of business in connection with an employee incentive plan or other transaction permitted by *Section 8.07*) the repurchase or redemption price does not exceed the original consideration paid for such stock or Equity Interest,

(ii) (ii) the conversion of any of its convertible securities into other securities of such Borrower pursuant to the terms of such convertible securities,

(iii) (iii) the payment of cash in lieu of fractional shares upon the conversion of any such convertible securities, not to exceed \$500,000 in the aggregate,

(iv) (iv) pursuant to the Redemption/Exchange Transactions, or

(v) (v) if both before and after giving effect to such Permitted Warrant Transaction (taken in the aggregate if consummated through a series of substantially simultaneous transactions), (x) there shall not then be an Event of Default or Default that has occurred and is continuing, and (y) each of the Financial Covenants in *Article IX* shall be satisfied, then a Permitted Warrant Transaction;

 declare or pay any cash dividend or make a cash distribution on any class of stock or other Equity Interest, except (i) that a Subsidiary may pay dividends or make distributions to any Borrower and any such Borrower may make Permitted Tax Distributions to its direct and indirect equity holders, (ii) with respect to the payment of Public Company Costs; (c) lend money to any employees, officers or directors or guarantee the payment of any such loans granted by a third party in excess of \$250,000 in the aggregate at any one time outstanding;

- (d) waive, release or forgive any Indebtedness owed by any employees, officers or directors in excess of \$250,000 in the aggregate;
- (e) make any payments on Subordinated Debt; or

(f) make any other any dividend or other distribution (whether in cash, securities or other Property) with respect to any capital stock or other Equity Interest of a Person, or any other payment (whether in cash, securities or other Property), including any sinking fund or similar deposit, on account of the purchase, redemption, retirement, acquisition, cancellation or termination of any capital stock or other Equity Interest or on account of any return of capital to such Person's stockholders, partners or members (or the equivalent Person thereof) (each of the foregoing *Sections 8.04(a) – (f)*, collectively, the "*Restricted Payments*"), unless: without the prior written consent of Lender.

- (i) There shall not then be an Event of Default or Default that has occurred and is continuing,
- (ii) The Restricted Payment is not with respect to Disqualified Equity Interests; and

(iii) Both before and after giving effect to such Restricted Payment, (x) each of the Financial Covenants in *Article IX* shall be satisfied, and (y) the Borrowers' Total Debt shall be less than 2.5 times their EBITDA over the preceding 12-month period reported in its most recent quarterly or annual financial statements.

VIII.5 Loans and Investments. No Borrower will make, and will not permit any Subsidiary to make, any advance, loan, extension of credit, or capital contribution to or investment in, or purchase, or permit any Subsidiary to purchase, any stock, bonds, notes, debentures, or other securities of, any Person, other than Permitted Investments; provided that no Event of Default has occurred and is continuing or would occur after giving effect to such Investments.

VIII.6 Limitation on Issuance of Equity. No Borrower will, nor will it permit any Subsidiary to, at any time issue, sell, assign, or otherwise dispose of Disqualified Equity Interests.

VIII.7 **Transactions with Affiliates**. No Borrower will enter into, nor will it permit any Subsidiary to enter into, any transaction, including, without limitation, the purchase, sale, or exchange of property or the rendering of any service, with any Affiliate of such Borrower or such Subsidiary, except (a) in the ordinary course of and pursuant to the reasonable requirements of such Borrower's or such Subsidiary's business and upon fair and reasonable terms no less favorable to such Borrower or such Subsidiary than would be obtained in a comparable arm's-length transaction with a Person not an Affiliate of Borrowers or such Subsidiary, (b) the payment of management fees as permitted pursuant to *Section 8.19*, (c) compensation and employee benefit, incentive and severance arrangements paid to, and indemnities provided for the benefit of, directors, officers or employees any Borrower or their respective Subsidiaries in the ordinary course of business, (d) any issuances of securities or other payments, awards or grants in cash, securities or otherwise pursuant to, or the funding of, employment agreements, stock options and stock ownership plans approved by the governing body of Direct Digital or DDH Holdings, as applicable, (e) transactions pursuant to the Tax Receivable Agreement, (f) transactions pursuant to the Redemption/Exchange Transactions, and (g) Borrower's Portower's repurchase of warrants in connection

40 4864-5069-8851 v.9 with a Permitted Warrant Transaction, provided that the repurchase is upon fair and reasonable terms no less favorable to Borrower than would be obtained in a comparable arm's-length transaction with a Person not an Affiliate of Borrowers or any Subsidiary.

VIII.8 **Disposition of Assets**. Except as permitted by Section 8.06, no Borrower will sell, lease, assign, transfer, or otherwise dispose of any of its assets, or permit any Subsidiary to do so with any of its assets, except (a) dispositions of inventory in the ordinary course of business, (b) dispositions, for fair value, of worn-out and obsolete equipment not necessary or useful to the conduct of business, or (c) for the avoidance of doubt, pursuant to the Redemption/Exchange Transactions.

VIII.9 Sale and Leaseback. No Borrower will enter into, nor will it permit any Subsidiary to enter into, any arrangement with any Person pursuant to which it leases from such Person real or personal property that has been or is to be sold or transferred, directly or indirectly, by it to such Person.

VIII.10 Nature of Business. No Borrower will, nor will it permit any Subsidiary to, engage in any business other than the businesses in which they are engaged as of the date hereof and businesses reasonably related thereto and logical extensions thereof.

VIII.11 Environmental Protection. No Borrower will, nor will it permit any Subsidiary to, (a) use (or permit any tenant to use) any of their respective properties or assets for the handling, processing, storage, transportation, or disposal of any Hazardous Material, (b) generate any Hazardous Material, (c) conduct any activity that is likely to cause a Release or threatened Release of any Hazardous Material, or (d) otherwise conduct any activity or use any of their respective properties or assets in any manner that is likely to violate any Environmental Law or create any Environmental Liabilities for which Borrowers or any Subsidiary would be responsible.

VIII.12 Accounting. No Borrower will, nor will it permit any Subsidiary to, change its fiscal year or make any change (a) in accounting treatment or reporting practices, except as required by GAAP and disclosed to Lender, or (b) in tax reporting treatment, except as required by law and disclosed to Lender.

VIII.13 No Negative Pledge. No Borrower will, nor will it permit any Subsidiary to, enter into or permit to exist any arrangement or agreement, other than pursuant to this Agreement, any Loan Document, or the Term Loan Documents, which directly or indirectly prohibits any Borrower or any Subsidiary from creating or incurring a Lien on any of its assets.

VIII.14 Subsidiaries. No Borrower will, directly or indirectly, form or acquire any Subsidiary unless such Subsidiary complies with the requirements of *Section 7.13*.

VIII.15 Hedge Agreements. No Borrower will, nor shall it permit any Subsidiary to, enter into any Hedge Agreement, except (a) Hedge Agreements entered into to hedge or mitigate risks to which Borrowers or any Subsidiary have actual exposure (other than those in respect of Equity Interests or any Term Loan Debt) which have terms and conditions reasonably acceptable to Lender, and (b) other Hedge Agreements entered into in order to effectively cap, collar or exchange interest rates (from fixed to floating rates, from one floating rate to another floating rate or otherwise) with respect to any Debt of Borrowers or any Subsidiary which have terms and conditions reasonably acceptable to Lender.

VIII.16 **OFAC**. No Borrower will, nor shall it permit any Subsidiary to, fail to comply with the laws, regulations and executive orders referred to in *Sections 6.19* and *6.20*. No Borrower shall, directly or indirectly, use the proceeds of any Loan, or lend, contribute or otherwise make available such proceeds to any Subsidiary, joint venture partner or other individual or entity, to fund any activities of or business

41 4864-5069-8851 v.9 with any individual or entity, or in any Designated Jurisdiction, that, at the time of such funding, is the subject of Sanctions, or in any other manner that will result in a violation by any individual or entity of Sanctions.

VIII.17 **Payments under Term Loan Agreement**. No Borrower will, nor will it permit any Subsidiary to, directly or indirectly, make any optional or voluntary payment, prepayment, repurchase or redemption of any Term Loan Debt not otherwise permitted under the terms of the Intercreditor Agreement.

# Article IX. FINANCIAL COVENANTS

Each Borrower covenants and agrees that, as long as the Obligations or any part thereof are outstanding, or Lender has any Commitment hereunder, such Borrower will, at all times, observe and perform the following financial covenants, unless Lender shall otherwise consent in writing.

IX.1 **Fixed Charge Coverage Ratio**. Borrowers will maintain a Fixed Charge Coverage Ratio of not less than 1.25 to <u>1.01.00</u>, beginning with the fiscal quarter of Borrowers ending June 30, 2023 through and including March 31, 2024. Beginning with the fiscal quarter of Borrowers ending June 30, 2025 and thereafter, Borrowers will maintain a Fixed Charge Coverage Ratio of not less than 1.25 to <u>1.00</u>. This ratio shall be calculated at the end of each fiscal quarter of Borrowers using the results of the twelve-month period ending with such fiscal quarter end.

IX.2 Total Funded Debt to EBITDA Ratio. Borrowers will maintain a Total Funded Debt to EBITDA Ratio not to exceed the ratios set forth in the following table:

June 30, 2023 and the last day of each fiscal	3.50:1.00
quarter through and including December 31,	
2023	
March 31, 2024 and the last day of each fiscal	3.25:1:00
quarter through and including March 31, 2025	
June 30, 2024 and the last day of each fiscal	<del>3:00:1.00<u>n/a</u></del>
quarter through and including June 30, 2025-and	
thereafter	

This ratio shall be calculated at the end of each fiscal quarter of Borrowers using the results of the twelve-month period ending with such fiscal quarter end.

## IX.3 Liquid Assets

. Borrowers and the Subsidiaries, on a Consolidated Basis, shall maintain minimum Liquid Assets at all times, in one or more accounts held with Lender *plus* Revolving Credit Availability (a) for the fiscal quarters ending September 30, 2024 and December 31, 2024, in the amount of \$1,000,0001,500,000, and (b) beginning with the fiscal quarter ending March 31, 2025 and thereafter, in the amount of \$2,000,000.

I.2 Minimum EBITDA. Borrowers shall maintain EBITDA, as at the last day of each of the following twelve-month periods, of not less than (a) \$5,000,000 for the twelve-month period ending

<del>42</del> 4<del>864-</del> 5069-8851 5 September 30, 2024, (b) \$3,500,000 for the twelve-month period ending December 31, 2024, (c) \$5,500,000 for the twelve-month period ending March 31, 2025, and (d) \$7,500,000 for the twelve-month period ending June 30, 2025.

### Article X. DEFAULT

X.1 **Events of Default**. Each of the following shall be deemed an "Event of Default":

(a) Borrowers shall fail to pay the Obligations, or any part thereof shall not be paid when due or declared due and, other than with respect to payments of principal, such failure shall continue unremedied for three (3) days after such payment became due.

(b) Borrowers shall breach any provision of Section 7.01, Article VIII or Article IX of this Agreement.

(c) Any representation or warranty made or deemed made by any Borrower, any other Obligated Party or any Pledgor (or any of their respective officers) in any Loan Document or in any certificate, report, notice, or financial statement furnished at any time in connection with this Agreement shall be false, misleading, or erroneous in any material respect (without duplication of any materiality qualifier contained therein) when made or deemed to have been made.

(d) Any Borrower or any Obligated Party shall fail to perform, observe, or comply with any covenant, agreement, or term contained in this Agreement or any other Loan Document (other than as covered by *Section 10.01(a)* and *(b)* above), and such failure continues for more than thirty (30) days following the date such failure first began.

(e) Any Borrower, any Subsidiary, any Obligated Party or any Pledgor shall commence a voluntary proceeding seeking liquidation, reorganization, or other relief with respect to itself or its debts under any bankruptcy, insolvency, or other similar law now or hereafter in effect or seeking the appointment of a trustee, receiver, liquidator, custodian, or other similar official of it or a substantial part of its property or shall consent to any such relief or to the appointment of or taking possession by any such official in an involuntary case or other proceeding commenced against it or shall make a general assignment for the benefit of creditors or shall generally fail to pay its debts as they become due or shall take any corporate action to authorize any of the foregoing.

(f) Any Borrower, any Subsidiary, or any Obligated Party shall fail to pay when due any principal of or interest on any Debt in excess of \$250,000 (other than the Obligations), or the maturity of any such Debt shall have been accelerated, or any such Debt shall have been required to be prepaid prior to the stated maturity thereof, or any event shall have occurred that permits (or, with the giving of notice or lapse of time or both, would permit) any holder or holders of such Debt or any Person acting on behalf of such holder or holders to accelerate the maturity thereof or require any such prepayment.

(g) This Agreement or any other Loan Document shall cease to be in full force and effect or shall be declared null and void or the validity or enforceability thereof shall be contested or challenged by Borrowers, any Subsidiary, any Obligated Party, any Pledgor or any of their respective shareholders, or Borrowers, any Obligated Party or any Pledgor shall deny that it has

any further liability or obligation under any of the Loan Documents, or any lien or security interest created by the Loan Documents shall for any reason cease to be a valid, first priority perfected security interest in and lien upon any Priority Collateral or a valid, a second priority perfected security interest in and lien upon the Term Loan Priority Collateral or any other Collateral purported to be covered thereby; provided that, Borrowers shall have fifteen (15) days to have this Agreement or any other Loan Document reinstated, or to have any such security interest perfected.

(h) Any of the following events shall occur or exist with respect to Borrowers: (i) any Prohibited Transaction involving any Plan; (ii) any Reportable Event with respect to any Plan; (iii) the filing under *Section 4041* of ERISA of a notice of intent to terminate any Plan or the termination of any Plan; (iv) any event or circumstance that might constitute grounds entitling the PBGC to institute proceedings under *Section 4042* of ERISA for the termination of, or for the appointment of a trustee to administer, any Plan, or the institution by the PBGC of any such proceedings; or (v) complete or partial withdrawal under *Section 4201* or *4204* of ERISA from a Multiemployer Plan or the reorganization, insolvency, or termination of any Multiemployer Plan; and in each case above, such event or condition, together with all other events or conditions, if any, have subjected or could in the reasonable opinion of Lender subject Borrowers to any tax, penalty, or other liability to a Plan, a Multiemployer Plan, the PBGC, or otherwise (or any combination thereof) which in the aggregate could reasonably be expected to result in a Material Adverse Event.

(i) If a Guarantor or any other Obligated Party is a corporation, partnership or other entity, such Person shall be the subject of a bankruptcy or receivership proceeding or shall have dissolved, liquidated or otherwise ceased doing business.

(j) Any Borrower, any of the Subsidiaries, or any Obligated Party, or any of their material properties, revenues, or assets, shall become subject to an order of forfeiture, seizure, or divestiture (whether under RICO or otherwise) and the same shall not have been discharged within sixty (60) days from the date of entry thereof.

#### (k) A Change in Control shall occur.

(1) An involuntary proceeding shall be commenced against any Borrower, any Subsidiary, any Obligated Party or any Pledgor seeking liquidation, reorganization, or other relief with respect to it or its debts under any bankruptcy, insolvency, or other similar law now or hereafter in effect or seeking the appointment of a trustee, receiver, liquidator, custodian, or other similar official for it or a substantial part of its property, and such involuntary proceeding shall remain undismissed and unstayed for a period of sixty (60) days.

(m) Any Borrower, any Subsidiary or any Obligated Party shall fail to discharge within a period of thirty (30) days after the commencement thereof any attachment, sequestration, or similar proceeding or proceedings involving an aggregate amount in excess of Two Hundred Fifty Thousand Dollars (\$250,000) against any of its assets or properties.

(n) A final judgment or judgments for the payment of money in excess of Two Hundred Fifty Thousand Dollars (\$250,000) in the aggregate shall be rendered by a court or courts against any Borrower, any of the Subsidiaries, or any Obligated Party and the same shall not be discharged (or provision shall not be made for such discharge), or a stay of execution

44 4<del>864-</del> <del>5069-</del> <del>8851 v.</del> thereof shall not be procured, within thirty (30) days from the date of entry thereof and such Borrower or the relevant Subsidiary or Obligated Party shall not, within said period of thirty (30) days, or such longer period during which execution of the same shall have been stayed, appeal therefrom and cause the execution thereof to be stayed during such appeal.

(o) An Event of Default (as defined in the Term Loan Agreement) under the Term Loan Documents shall occur and be continuing.

X.2 **Remedies Upon Default**. If any Event of Default shall occur and be continuing, Lender may without notice terminate the Commitment and declare the Obligations or any part thereof to be immediately due and payable, and the same shall thereupon become immediately due and payable, without notice, demand, presentment, notice of dishonor, notice of acceleration, notice of intent to accelerate, notice of intent to demand, protest, or other formalities of any kind, all of which are, to the maximum extent permitted by law, hereby expressly waived by Borrowers; provided, however, that upon the occurrence of an Event of Default under *Section 10.01(e)* or *Section 10.01(l)*, and so long as continuing, the Commitment shall automatically terminate, and the Obligations shall become immediately due and payable without notice, demand, protest, or other formalities of any kind, all of which are, to the maximum extent permitted by law, hereby expressly waived by Borrowers. If any Event of Default shall occur and be continuing, Lender may exercise all rights and remedies available to it in law or in equity, under the Loan Documents, or otherwise.

X.3 **Performance by Lender**. If Borrowers shall fail to perform any covenant or agreement contained in any of the Loan Documents, Lender may perform or attempt to perform such covenant or agreement on behalf of Borrowers. In such event, Borrowers shall, at the request of Lender, promptly pay any amount reasonably expended by Lender in connection with such performance or attempted performance to Lender, together with interest thereon at the Default Interest Rate from and including the date of such expenditure to but excluding the date such expenditure is paid in full. Notwithstanding the foregoing, it is expressly agreed that Lender shall not have any liability or responsibility for the performance of any obligation of Borrowers under this Agreement or any other Loan Document.

### X.4 Equity Cure

. Notwithstanding the foregoing, and subject to the last sentence of this *Section 10.04*, the Borrowers may cure (and shall be deemed to have cured) (any such cure, an "*Equity Cure*") an Event of Default arising out of a breach of any of the financial covenants set forth in *Sections 9.01*, *9.02* or *9.03* (the "*Specified Financial Covenants*") if (i) the Borrowers receive, within 10 Business Days after the date on which the Specified Financial Covenants are first required to be tested pursuant to the terms hereof, cash proceeds in an amount which, if treated as income for the preceding fiscal quarter, would result in compliance with such Specified Financial Covenants, and (ii) Lender receives written notice from the Borrowers that such payment has been made and that it is to be deemed an Equity Cure hereunder. Upon any Equity Cure of a Specified Financial Covenant, any Event of Default that occurred and is continuing from a breach of such Specified Financial Covenant shall be deemed cured with no further action required by Lender. An Equity Cure may not be used to cure an Event of Default more than twice in any calendar year (or be in an aggregate amount of such cash proceeds in any calendar year of more than \$2,000,000), or more than four times during the term of this Agreement (including any extension thereof), or be in an amount greater than necessary to cure the Specified Financial Covenants.

#### X.5 Cash Collateral

<del>45</del> 4<del>864-</del> <del>5069-</del> <del>8851 v.9</del> . If any Event of Default shall occur and be continuing or the Maturity Date shall have occurred, Borrowers shall, if requested by Lender, immediately deposit with and pledge to Lender cash or cash equivalent investments in an amount equal to the outstanding Letter of Credit Liabilities as security for the Obligations.

#### Article XI. MISCELLANEOUS

Expenses. Each Borrower hereby agrees, jointly and severally, to pay on demand: (a) all reasonable and documented costs and out-of-pocket XI1 expenses of Lender in connection with the preparation, negotiation, execution, and delivery of this Agreement and the other Loan Documents and any and all amendments, modifications, renewals, extensions, and supplements thereof and thereto, including, without limitation, the reasonable fees and expenses of outside legal counsel, advisors, consultants, and auditors for Lender; (b) all costs and expenses of Lender in connection with any Default and the enforcement of this Agreement or any other Loan Document, including, without limitation, the reasonable and documented fees and out-of-pocket expenses of legal counsel. advisors, consultants, and auditors for Lender; (c) all transfer, stamp, documentary, or other similar taxes, assessments, or charges levied by any Governmental Authority in respect of this Agreement or any of the other Loan Documents; (d) all reasonable and documented costs, expenses, assessments, and other charges incurred in connection with any filing, registration, recording, or perfection of any Lien contemplated by this Agreement or any other Loan Document; and (e) all other reasonable and documented costs and expenses incurred by Lender in connection with (i) this Agreement or any other Loan Document, (ii) the servicing and administration of the Obligations, (iii) any litigation, dispute, suit, proceeding or action arising from or related to the Obligations or any Loan Document, or (iv) the enforcement of its rights and remedies, and the protection of its interests in bankruptcy, insolvency or other legal proceedings, including, without limitation, all costs, expenses, and other charges incurred in connection with evaluating, observing, collecting, examining, auditing, appraising, selling, liquidating, or otherwise disposing of the Collateral or other assets of Borrowers. Each Borrower authorizes Lender, at its sole option, to (i) cause a Revolving Credit Advance on or after the date of this Agreement, (ii) debit any other Borrower account with Lender, or (iii) make demand upon Borrowers, for payment of all reasonable attorneys' fees and out-of-pocket expenses incurred by Lender in connection with the negotiation and documentation of this Agreement and the other Loan Documents by counsel retained by Lender, which attorney's fees and expenses become due through the date of this Agreement and/or after the date of this Agreement.

XI.2 **INDEMNIFICATION**. EACH BORROWER SHALL INDEMNIFY LENDER AND EACH AFFILIATE THEREOF AND THEIR RESPECTIVE OFFICERS, DIRECTORS, EMPLOYEES, ATTORNEYS, AND AGENTS (COLLECTIVELY, THE "*INDEMNIFIED PARTIES*" AND INDIVIDUALLY AN "*INDEMNIFIED PARTY*") FROM, AND HOLD EACH OF THEM HARMLESS AGAINST, ANY AND ALL LOSSES, LIABILITIES, CLAIMS, DAMAGES, PENALTIES, JUDGMENTS, DISBURSEMENTS, COSTS, AND EXPENSES (INCLUDING ATTORNEYS' FEES) TO WHICH ANY OF THEM MAY BECOME SUBJECT WHICH DIRECTLY OR INDIRECTLY ARISE FROM OR RELATE TO (a) ANY OF THE LOAN DOCUMENTS INCLUDING THE NEGOTIATION, EXECUTION, DELIVERY, PERFORMANCE, ADMINISTRATION, OR ENFORCEMENT OF ANY OF THE LOAN DOCUMENTS, (b) ANY OF THE TRANSACTIONS CONTEMPLATED BY THE LOAN DOCUMENTS, (c) ANY BREACH BY ANY BORROWER OF ANY REPRESENTATION, WARRANTY, COVENANT, OR OTHER AGREEMENT CONTAINED IN ANY OF THE LOAN DOCUMENTS, (d) ANY ACTION TAKEN OR NOT TAKEN BY LENDER (OR ANY TRUSTEE UNDER ANY SECURITY INSTRUMENT) THAT IS ALLOWED OR PERMITTED UNDER ANY OF THE LOAN DOCUMENTS, INCLUDING

46 4864-5069-8851 v.9 THE PROTECTION OR ENFORCEMENT OF ANY LIEN, SECURITY INTEREST, OR OTHER RIGHT, REMEDY, OR RECOURSE CREATED OR AFFORDED BY THE LOAN DOCUMENTS OR AT LAW OR IN EQUITY, (e) ANY DISPUTE AMONG OR BETWEEN ANY OF THE OBLIGATED PARTIES OR BETWEEN OR AMONG ANY PARTNERS, VENTURERS, EMPLOYEES, OFFICERS, DIRECTORS, SHAREHOLDERS, MEMBERS, MANAGERS, TRUSTEES, OR OTHER RESPONSIBLE PARTIES OF BORROWERS, (f) THE PRESENCE, RELEASE, THREATENED RELEASE, DISPOSAL, REMOVAL, OR CLEANUP OF ANY HAZARDOUS MATERIAL LOCATED ON, ABOUT, WITHIN, OR AFFECTING ANY OF THE PROPERTIES OR ASSETS OF BORROWERS OR ANY OF THE SUBSIDIARIES OR ANY OTHER OBLIGATED PARTY, (g) THE USE OR PROPOSED USE OF ANY LETTER OF CREDIT, (h) ANY AND ALL TAXES (OTHER THAN EXCLUDED TAXES), LEVIES, DEDUCTIONS, OR CHARGES IMPOSED ON LENDER OR ANY OF LENDER'S CORRESPONDENTS IN RESPECT OF ANY LETTER OF CREDIT, OR (i) ANY INVESTIGATION, LITIGATION, OR OTHER PROCEEDING, INCLUDING, WITHOUT LIMITATION, ANY THREATENED INVESTIGATION, LITIGATION, OR OTHER PROCEEDING, RELATING TO ANY OF THE FOREGOING INCLUDING THOSE BROUGHT OR INITIATED BY. WITHOUT LIMITING ANY PROVISION OF THIS AGREEMENT OR OF ANY OTHER LOAN DOCUMENT, IT IS THE EXPRESS INTENTION OF THE PARTIES HERETO THAT THE INDEMNIFIED PARTIES BE INDEMNIFIED FROM AND HELD HARMLESS AGAINST ANY AND ALL LOSSES, LIABILITIES, CLAIMS, DAMAGES, PENALTIES, JUDGMENTS, DISBURSEMENTS, COSTS, AND EXPENSES (INCLUDING REASONABLE ATTORNEYS' FEES) ARISING OUT OF OR RESULTING FROM THE STRICT LIABILITY, SOLE CONTRIBUTORY OR ORDINARY NEGLIGENCE OF ANY OF THE INDEMNIFIED PARTIES; PROVIDED, HOWEVER, THAT THE INDEMNITY SET FORTH IN THIS SECTION 11.02 WILL NOT APPLY TO CLAIMS CAUSED BY THE GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF LENDER OR ANY OF ITS OFFICERS, EMPLOYEES, AGENTS, ADVISORS, OR REPRESENTATIVES. AS DETERMINED BY A COURT OF COMPETENT JURISDICTION IN FINAL AND NON APPEALABLE JUDGMENT.

LENDER MAY EMPLOY AN ATTORNEY OR ATTORNEYS OF ITS OWN CHOOSING TO PROTECT OR ENFORCE ITS RIGHTS, REMEDIES, AND RECOURSES, AND TO ADVISE AND DEFEND THE INDEMNIFIED PARTIES WITH RESPECT TO THOSE ACTIONS AND OTHER MATTERS. EACH BORROWER SHALL REIMBURSE LENDER FOR THE ATTORNEYS' REASONABLE FEES AND OUT-OF-POCKET EXPENSES (INCLUDING EXPENSES AND COSTS FOR EXPERTS AND/OR CONSULTANTS) OF THE INDEMNIFIED PARTIES IMMEDIATELY ON RECEIPT OF WRITTEN DEMAND FROM LENDER, WHETHER ON A MONTHLY OR OTHER TIME INTERVAL, AND WHETHER OR NOT AN ACTION IS ACTUALLY COMMENCED OR CONCLUDED. ALL OTHER REIMBURSEMENT AND INDEMNITY OBLIGATIONS UNDER THIS AGREEMENT SHALL BECOME DUE AND PAYABLE WHEN ACTUALLY INCURRED BY LENDER OR ANY OF THE OTHER THE INDEMNIFIED PARTIES. ANY PAYMENTS NOT MADE WITHIN TEN (10) DAYS AFTER WRITTEN DEMAND FROM LENDER SHALL BEAR INTEREST AT THE DEFAULT INTEREST RATE FROM THE DATE OF THAT DEMAND UNTIL FULLY PAID. THE PROVISIONS OF THIS **SECTION 11.02** SHALL SURVIVE REPAYMENT AND PERFORMANCE OF THE OBLIGATIONS, THE RELEASE OF ANY LIENS SECURING THE OBLIGATIONS, ANY FORECLOSURE (OR ACTION IN LIEU OF FORECLOSURE), THE TRANSFER BY ANY BORROWER OF ANY OF ITS RIGHTS, TITLE, AND INTERESTS IN OR TO ANY COLLATERAL SECURING THE OBLIGATIONS, AND THE EXERCISE BY LENDER OF ANY OR ALL REMEDIES SET FORTH IN ANY LOAN DOCUMENT. NOTWITHSTANDING THE FOREGOING THIS SECTION 11.02 SHALL NOT APPLY WITH RESPECT TO EXCLUDED TAXES

<del>47</del>
<del>4864-</del>
<del>5069-</del>
<del>8851 v.9</del>

# OR TO ANY OTHER TAXES (OTHER THAN ANY TAXES THAT REPRESENT LOSSES, CLAIMS, DAMAGES, ETC. ARISING FROM ANY NON-TAX CLAIM.

XI.3 Limitation of Liability. Neither Lender nor any Affiliate, officer, director, employee, attorney, or agent of Lender shall have any liability with respect to, and, to the maximum extent permitted by law, each Borrower hereby waives, releases, and agrees not to sue any of them upon, any claim for any special, indirect, incidental, or consequential damages suffered or incurred by such Borrower in connection with, arising out of, or in any way related to, this Agreement or any of the other Loan Documents, or any of the transactions contemplated by this Agreement or any of Lender's Affiliates, officers, directors, employees, attorneys, or agents for punitive damages in respect of any claim in connection with, arising out of, or in any way related to, this Agreement or any of the other Loan Documents, or any of the transactions contemplated by this Agreement or, any way related to, this Agreement or any of the other Loan Documents, or any of the transactions contemplated by the other Loan Documents.

XI.4 **No Duty**. All attorneys, accountants, appraisers, and other professional Persons and consultants retained by Lender shall have the right to act exclusively in the interest of Lender and shall have no duty of disclosure, duty of loyalty, duty of care, or other duty or obligation of any type or nature whatsoever to Borrowers or any of Borrowers' shareholders or any other Person.

XI.5 **Lender Not Fiduciary**. The relationship between Borrowers and Lender is solely that of debtor and creditor, and Lender has no fiduciary or other special relationship with Borrowers, and no term or condition of any of the Loan Documents shall be construed so as to deem the relationship between Borrowers and Lender to be other than that of debtor and creditor.

XI.6 Equitable Relief. Borrowers recognizes that in the event Borrowers fail to pay, perform, observe, or discharge any or all of the Obligations, any remedy at law may prove to be inadequate relief to Lender. Borrowers therefore agree that Lender, if Lender so requests, shall be entitled to temporary and permanent injunctive relief in any such case without the necessity of proving actual damages.

XI.7 **No Waiver; Cumulative Remedies.** No failure on the part of Lender to exercise and no delay in exercising, and no course of dealing with respect to, any right, power, or privilege under this Agreement shall operate as a waiver thereof, nor shall any single or partial exercise of any right, power, or privilege under this Agreement preclude any other or further exercise thereof or the exercise of any other right, power, or privilege. The rights and remedies provided for in this Agreement and the other Loan Documents are cumulative and not exclusive of any rights and remedies provided by law.

XI.8 **Successors and Assigns**. This Agreement is binding upon and shall inure to the benefit of Lender and Borrowers and their respective successors and assigns, except that Borrowers may not assign or transfer any of its rights or obligations under this Agreement without the prior written consent of Lender.

XI.9 **Survival**. All representations and warranties made in this Agreement or any other Loan Document or in any document, statement, or certificate furnished in connection with this Agreement shall survive the execution and delivery of this Agreement and the other Loan Documents, and no investigation by Lender or any closing shall affect the representations and warranties or the right of Lender to rely upon them. Without prejudice to the survival of any other obligation of Borrowers hereunder, the obligations of Borrowers under *Sections 11.01* and *11.02* shall survive repayment of the Notes and termination of the Commitment.

<del>48</del>
<del>4864-</del>
<del>5069-</del>
<del>8851 v.9</del>

XI.10 ENTIRE AGREEMENT; AMENDMENT. THIS AGREEMENT, THE NOTES, AND THE OTHER LOAN DOCUMENTS REFERRED TO HEREIN EMBODY THE FINAL, ENTIRE AGREEMENT AMONG THE PARTIES HERETO AND SUPERSEDE ANY AND ALL PRIOR COMMITMENTS, AGREEMENTS, REPRESENTATIONS, AND UNDERSTANDINGS, WHETHER WRITTEN OR ORAL, RELATING TO THE SUBJECT MATTER HEREOF AND MAY NOT BE CONTRADICTED OR VARIED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OR DISCUSSIONS OF THE PARTIES HERETO. THERE ARE NO ORAL AGREEMENTS AMONG THE PARTIES HERETO. The provisions of this Agreement and the other Loan Documents to which Borrowers are parties may be amended or waived only by an instrument in writing signed by the parties hereto.

XI.11 Notices. Unless otherwise expressly provided herein, all notices and other communications provided for hereunder shall be in writing (including by facsimile transmission) and mailed, faxed or delivered, to the address, facsimile number or subject to the last sentence hereof electronic mail address specified for notices below the signatures hereon or to such other address as shall be designated by such party in a notice to the other parties. All such other notices and other communications shall be deemed to have been given or made upon the earliest to occur of (i) actual receipt by the intended recipient or (ii) (A) if delivered by hand or courier, when signed for by the designated recipient; (B) if delivered by mail, four (4) Business Days after deposit in the mail, postage prepaid; (C) if delivered by facsimile when sent and receipt has been confirmed by telephone; and (D) if delivered by electronic mail (which form of delivery is subject to the provisions of the last sentence below) when delivered; provided, however, that notices and other communications pursuant to *Article II* shall not be effective until actually received by Lender. Electronic mail and intranet websites may be used only to distribute only routine communications, such as financial statements and other information, and to distribute Loan Documents for execution by the parties thereto and may not be used for any other purpose.

XI.12 Governing Law; Venue; Service of Process. THIS AGREEMENT AND ANY CONTROVERSY, DISPUTE, CLAIM OR CAUSE OF ACTION ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE OTHER LOAN DOCUMENTS, ANY BREACH THEREOF, THE TRANSACTIONS CONTEMPLATED THEREBY, OR ANY OTHER DISPUTE BETWEEN OR AMONG LENDER AND ANY OF THE OBLIGATED PARTIES (WHETHER IN CONTRACT, TORT OR OTHERWISE) SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF TEXAS; *PROVIDED THAT* LENDER SHALL RETAIN ALL RIGHTS UNDER FEDERAL LAW. IF, FOR ANY REASON, A COURT OF COMPETENT JURISDICTION DETERMINES THAT TEXAS LAW SHOULD NOT APPLY TO THE PROVISIONS OF THE LOAN DOCUMENTS PERTAINING TO THE CREATION, PERFECTION, ENFORCEMENT, OR VALIDITY OF THE LIENS AND SECURITY INTERESTS CREATED PURSUANT TO THE APPLICABLE LOAN DOCUMENTS, THEN SUCH PROVISIONS (BUT ONLY THOSE PROVISIONS) SHALL BE GOVERNED BY, CONSTRUED, AND INTERPRETED IN ACCORDANCE WITH THE LAWS OF THE STATE WHERE THE APPLICABLE COLLATERAL IS LOCATED. THIS AGREEMENT HAS BEEN ENTERED INTO IN DALLAS COUNTY, TEXAS, AND IS PERFORMABLE FOR ALL PURPOSES IN DALLAS COUNTY, TEXAS. THE PARTIES HEREBY AGREE THAT ANY LAWSUIT, ACTION, OR PROCEEDING THAT IS BROUGHT (WHETHER IN CONTRACT, TORT OR OTHERWISE) ARISING OUT OF OR RELATING TO ANY OF THE LOAN DOCUMENTS, THE ASSIGNS OR OF ANY OF THE LOAN DOCUMENTS, THE NEGOTIATION, ADMINISTRATION OR ENFORCEMENT, SUCCESSORS OR ASSIGNS OR OF ANY OF THE OBLIGATED PARTIES IN THE NEGOTIATION, ADMINISTRATION OR ENFORCEMENT OF ANY OF THE LOAN DOCUMENTS SHALL BE

<del>49</del>
<del>4864 -</del>
<del>5069-</del>
8851 v.9

BROUGHT IN A STATE OR FEDERAL COURT OF COMPETENT JURISDICTION LOCATED IN DALLAS COUNTY, TEXAS. EACH BORROWER HEREBY IRREVOCABLY AND UNCONDITIONALLY (A) SUBMITS TO THE EXCLUSIVE JURISDICTION OF SUCH COURTS, (B) WAIVES ANY OBJECTION IT MAY NOW OR HEREAFTER HAVE AS TO THE VENUE OF ANY SUCH LAWSUIT, ACTION, OR PROCEEDING BROUGHT IN ANY SUCH COURT, AND (C) FURTHER WAIVES ANY CLAIM THAT IT MAY NOW OR HEREAFTER HAVE THAT ANY SUCH COURT IS AN INCONVENIENT FORUM. EACH OF THE PARTIES HERETO AGREE THAT SERVICE OF PROCESS UPON IT MAY BE MADE BY CERTIFIED OR REGISTERED MAIL, RETURN RECEIPT REQUESTED AT THE ADDRESS FOR NOTICES REFERENCED IN **SECTION 11.11** HEREOF.

XI.13 **Counterparts**. This Agreement may be executed in any number of counterparts and by different parties on separate counterparts, each of which, when executed and delivered, shall be deemed to be an original, and all of which, when taken together, shall constitute but one and the same agreement. Delivery of an executed signature page of this Agreement and all other documents executed in connection with the Loan by facsimile or other electronic mail transmission shall be effective as delivery of a manually executed counterpart hereof.

XI.14 Severability. Any provision of this Agreement held by a court of competent jurisdiction to be invalid or unenforceable shall not impair or invalidate the remainder of this Agreement and the effect thereof shall be confined to the provision held to be invalid or illegal.

XI.15 Headings. The headings, captions, and arrangements used in this Agreement are for convenience only and shall not affect the interpretation of this Agreement.

XI.16 **Participations, Etc.** Lender shall have the right at any time and from time to time to grant participations in, and sell and transfer, the Obligations and any Loan Documents; provided, that so long as no Default or Event of Default has occurred and is continuing, the Borrowers shall have the right to consent to any such sale or transfer of the Obligations. Each actual or proposed participant or assignee, as the case may be, shall be entitled to receive all information received by Lender regarding Borrowers and the Subsidiaries, including, without limitation, information required to be disclosed to a participant or assignee pursuant to Banking Circular 181 (Rev., August 2, 1984), issued by the Comptroller of the Currency (whether the actual or proposed participant or assignee is subject to the circular or not).

XI.17 **Construction**. Borrowers and Lender acknowledge that each of them has had the benefit of legal counsel of its own choice and has been afforded an opportunity to review this Agreement and the other Loan Documents with its legal counsel and that this Agreement and the other Loan Documents shall be construed as if jointly drafted by Borrowers and Lender.

XI.18 **Independence of Covenants**. All covenants hereunder shall be given independent effect so that if a particular action or condition is not permitted by any of such covenants, the fact that it would be permitted by an exception to, or be otherwise within the limitations of, another covenant shall not avoid the occurrence of a Default if such action is taken or such condition exists.

XI.19 **WAIVER OF JURY TRIAL**. TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, EACH BORROWER HEREBY IRREVOCABLY AND EXPRESSLY WAIVES ALL RIGHT TO A TRIAL BY JURY IN ANY ACTION, PROCEEDING, OR COUNTERCLAIM (WHETHER BASED UPON CONTRACT, TORT, OR OTHERWISE) ARISING OUT OF OR RELATING TO ANY OF THE LOAN DOCUMENTS OR THE TRANSACTIONS CONTEMPLATED

<del>50</del> 4864-<del>5069-</del> 8851 v.9

# THEREBY OR THE ACTIONS OF LENDER IN THE NEGOTIATION, ADMINISTRATION, OR ENFORCEMENT THEREOF. EACH PARTY HERETO (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PERSON HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PERSON WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HERETO HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS **SECTION 11.19**.

XI.20 Additional Interest Provision. It is expressly stipulated and agreed to be the intent of Borrowers and Lender at all times to comply strictly with the applicable Texas law governing the maximum rate or amount of interest payable on the indebtedness evidenced by any Note, any Loan Document, and the Related Indebtedness (or applicable United States federal law to the extent that it permits Lender to contract for, charge, take, reserve or receive a greater amount of interest than under Texas law). If the applicable law is ever judicially interpreted so as to render usurious any amount (i) contracted for, charged, taken, reserved or received pursuant to any Note, any of the other Loan Documents or any other communication or writing by or between Borrowers and Lender related to the transaction or transactions that are the subject matter of the Loan Documents, (ii) contracted for, charged, taken, reserved or received by reason of Lender's exercise of the option to accelerate the maturity of any Note and/or any and all indebtedness paid or payable by Borrowers to Lender pursuant to any Loan Document other than any Note (such other indebtedness being referred to in this Section as the "Related Indebtedness"), or (iii) Borrowers will have paid or Lender will have received by reason of any voluntary prepayment by Borrowers of any Note and/or the Related Indebtedness, then it is Borrowers' and Lender's express intent that all amounts charged in excess of the Maximum Lawful Rate shall be automatically canceled, ab initio, and all amounts in excess of the Maximum Lawful Rate theretofore collected by Lender shall be credited on the principal balance of any Note and/or the Related Indebtedness (or, if any Note and all Related Indebtedness have been or would thereby be paid in full, refunded to Borrowers), and the provisions of any Note and the other Loan Documents shall immediately be deemed reformed and the amounts thereafter collectible hereunder and thereunder reduced, without the necessity of the execution of any new document, so as to comply with the applicable law, but so as to permit the recovery of the fullest amount otherwise called for hereunder and thereunder; provided, however, if any Note has been paid in full before the end of the stated term of any such Note, then Borrowers and Lender agree that Lender shall, with reasonable promptness after Lender discovers or is advised by Borrowers that interest was received in an amount in excess of the Maximum Lawful Rate, either refund such excess interest to Borrowers and/or credit such excess interest against such Note and/or any Related Indebtedness then owing by Borrowers to Lender. Borrowers hereby agree that as a condition precedent to any claim seeking usury penalties against Lender, Borrowers will provide written notice to Lender, advising Lender in reasonable detail of the nature and amount of the violation, and Lender shall have sixty (60) days after receipt of such notice in which to correct such usury violation, if any, by either refunding such excess interest to Borrowers or crediting such excess interest against the Note to which the alleged violation relates and/or the Related Indebtedness then owing by Borrowers to Lender. All sums contracted for, charged, taken, reserved or received by Lender for the use, forbearance or detention of any debt evidenced by any Note and/or the Related Indebtedness shall, to the extent permitted by applicable law, be amortized or spread, using the actuarial method, throughout the stated term of such Note and/or the Related Indebtedness (including any and all renewal and extension periods) until payment in full so that the rate or amount of interest on account of any Note and/or the Related Indebtedness does not exceed the Maximum Lawful Rate from time to time in effect and applicable to such Note and/or the Related Indebtedness for so long as debt is outstanding. In no event shall the provisions of Chapter 346 of the Texas Finance Code (which regulates certain revolving credit loan accounts and revolving triparty

<del>51</del> 4864-5069-8851 v accounts) apply to this Note and/or any of the Related Indebtedness. Notwithstanding anything to the contrary contained herein or in any of the other Loan Documents, it is not the intention of Lender to accelerate the maturity of any interest that has not accrued at the time of such acceleration or to collect unearned interest at the time of such acceleration.

XI.21 Ceiling Election. To the extent that Lender is relying on Chapter 303 of the Texas Finance Code to determine the Maximum Lawful Rate payable on any such Note and/or any other portion of the Obligations, Lender will utilize the weekly ceiling from time to time in effect as provided in such Chapter 303, as amended. To the extent United States federal law permits Lender to contract for, charge, take, receive or reserve a greater amount of interest than under Texas law, Lender will rely on United States federal law instead of such Chapter 303 for the purpose of determining the Maximum Lawful Rate. Additionally, to the extent permitted by applicable law now or hereafter in effect, Lender may, at its option and from time to time, utilize any other method of establishing the Maximum Lawful Rate under such Chapter 303 or under other applicable law by giving notice, if required, to Borrowers as provided by applicable law now or hereafter in effect.

XI.22 USA Patriot Act Notice. Lender hereby notifies Borrowers that pursuant to the requirements of the Patriot Act, it is required to obtain, verify and record information that identifies Borrowers, which information includes the names and addresses of Borrowers and other information that will allow Lender to identify Borrowers in accordance with the Patriot Act. Borrowers shall, promptly following a request by Lender, provide all documentation and other information that Lender requests in order to comply with its ongoing obligations under applicable "know your customer" and anti-money laundering rules and regulations, including the Patriot Act. Lender will require the legal entity to provide identifying information about each beneficial owner and/or individuals who have significant responsibility to control, manage or direct the legal entity.

#### XI.23 Confidentiality

. LENDER ACKNOWLEDGES THAT INFORMATION RECEIVED FROM ANY BORROWER RELATING TO SUCH BORROWER OR ANY SUBSIDIARY OR THEIR BUSINESSES OR THE COLLATERAL FURNISHED TO SUCH PERSON PURSUANT TO THIS AGREEMENT MAY INCLUDE MATERIAL NON-PUBLIC INFORMATION CONCERNING SUCH BORROWER, ITS SUBSIDIARIES AND ITS OTHER AFFILIATES AND THEIR RESPECTIVE SECURITIES, AND CONFIRMS THAT IT HAS DEVELOPED COMPLIANCE PROCEDURES REGARDING THE USE OF MATERIAL NON-PUBLIC INFORMATION AND THAT IT WILL HANDLE SUCH MATERIAL NON-PUBLIC INFORMATION IN ACCORDANCE WITH THOSE PROCEDURES AND APPLICABLE LAW, INCLUDING FEDERAL AND STATE SECURITIES LAWS. ALL INFORMATION, INCLUDING REQUESTS FOR WAIVERS AND AMENDMENTS, FURNISHED BY ANY BORROWER PURSUANT TO, OR IN THE COURSE OF ADMINISTERING, THIS AGREEMENT MAY CONTAIN MATERIAL NON-PUBLIC INFORMATION CONCERNING THE BORROWERS, THEIR SUBSIDIARIES AND THEIR OTHER AFFILIATES AND THEIR RESPECTIVE SECURITIES. ACCORDINGLY, LENDER REPRESENTS TO THE BORROWER THAT SUCH PERSON HAS IDENTIFIED A CREDIT CONTACT WHO MAY RECEIVE INFORMATION THAT MAY CONTAIN SUCH MATERIAL NON-PUBLIC INFORMATION IN ACCORDANCE WITH ITS COMPLIANCE PROCEDURES AND APPLICABLE LAW, INCLUDING FEDERAL AND STATE SECURITIES LAWS.

XI.24 Intercreditor Legend. Anything herein to the contrary notwithstanding, the Liens and security interests securing the obligations evidenced by this agreement, the exercise of any right or

<del>52</del> 4<del>864-</del> <del>5069-</del> <del>8851 v.</del> remedy with respect hereto and certain of the rights of the holder hereof are subject to the provisions of the Intercreditor Agreement, dated as July 7, 2023 (as amended, restated, supplemented, substituted, replaced or otherwise modified from time to time, the "*Intercreditor Agreement*"), by and between Lafayette Square Loan Servicing, LLC (in its capacity as agent for the LS Facility Lenders and together with its successors and assigns, the "*LS Facility Agent*"), for and on behalf of the LS Facility Creditors and each other LS Facility Claimholder (each as defined in the Intercreditor Agreement) from time to time, and East West Bank, acting on behalf of each A/R Facility Claimholder (each as defined in the Intercreditor Agreement). In the event of any conflict between the terms of the Intercreditor Agreement and this agreement, the terms of the Intercreditor Agreement shall govern and control.

[Remainder of Page Intentionally Left Blank; Signature Page Follows.]

<del>53</del> 4864-5069-8851 v.9 IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement as of the day and year first above written.

**BORROWERS: DIRECT DIGITAL HOLDINGS, INC.** Title: **DIRECT DIGITAL HOLDINGS, LLC** Title: **COLOSSUS MEDIA, LLC** <del>By:</del> \_\_\_\_ <del>Name:</del> Title: HUDDLED MASSES LLC <del>By:</del> \_\_\_\_ Name: Title: ORANGE142, LLC <del>By:</del> \_\_\_\_ Name: Title: Address for Notices: e/o Direct Digital Holdings, Inc. 1233 West Loop South Suite 1170 Houston, Texas 77027 Email: Phone:

> Signature Page to Credit Agreement

LENDER:

EAST WEST BANK, a California state bank

By: \_\_\_\_ Hamilton LaRoc

First Vice President

Address for Notices:

5001 Spring Valley Road; Suite 825W Dallas, Texas 75244 Attention: Hamilton LaRoe Email: Hamilton.LaRoc@EastWestBank.com

Signature Page to Credit Agreement

# SCHEDULES

6.13 Subsidiaries, Ventures, Etc.

- 6.18 Intellectual Property
- 8.01 Existing Debt
- 8.02 Existing Liens
- 8.05 Existing Investments

<u>SCHEDULE 6.13 – Subsidiaries, Ventures, Etc.</u>

[TO BE ADDED ONCE FINAL]

SCHEDULE 6.18 - Intellectual Property

[TO BE ADDED ONCE FINAL]

SCHEDULE 8.01 – Existing Debt

[TO BE ADDED ONCE FINAL]

SCHEDULE 8.02 – Existing Liens

[TO BE ADDED ONCE FINAL]

SCHEDULE 8.05 – Existing Investments

[TO BE ADDED ONCE FINAL]

#### EXHIBIT A

#### **ELIGIBLE ACCOUNT REPORT**

FOR MONTH ENDED (THE "SUBJECT QUARTER")

BANK: EAST WEST BANK

BORROWERS: Direct Digital Holdings, Inc., a Delaware corporation ("DDH Holdings"), Direct Digital Holdings, LLC, a Texas limited liability company ("Direct Digital"), Colossus Media, LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware l

This Certificate is delivered under the Credit Agreement (the "*Agreement*") dated as of July 7, 2023, by and among Borrowers and Bank as such may have been amended, supplemented or replaced. Capitalized terms used in this Certificate shall, unless otherwise indicated, have the meanings set forth in the Agreement. On behalf of Borrowers, the undersigned certifies to Bank on the date hereof that (a) no Default or Event of Default has occurred and is continuing, (b) a review of the activities of Borrowers during the Subject Month has been made under my supervision with a view to determining the amount of the current Eligible Accounts, (c) the Accounts included in the Eligible Account Report below meet all conditions to qualify for inclusion therein as set forth in the Agreement, and all representations and warranties set forth in the Agreement with respect thereto are true and correct in all material respects, and (d) the information set forth below hereto is true and correct as of the last day of the Subject Month.

#### Accounts

1. Beginning Balance of Accounts Date: \$\_\_\_\_ 2. Plus Sales/Debits <del>(+)</del> <u></u> 3. Minus Collections (-) \$\_\_\_\_ 4. Minus Credit Memos/Other Credits (-) \$\_\_\_\_ 5. Ending Balance of Accounts Date: \_\_\_\_\_\$\_\_\_ 6. Ineligible Accounts (Per the A/R Ineligible Listing) (-) \$\_\_\_\_\_ 7. Eligible Accounts (Line 5 minus Line 6) <del>\$</del>\_\_\_\_ 8. Current Revolving Credit Availability (Line 7 divided by 2) <del>\$</del> 9. Ending Revolving Principal Balance Date: (-) <del>\$</del>\_\_\_ 10. TOTAL NET REVOLVING CREDIT AVAILABILITY <del>\$</del> (Line 8 minus Line 9) (not to exceed Commitment) Accounts Incligible Listing 11. Accounts over 90 days from invoice date \$\_\_\_\_ 12. 25% taint rule (Accounts 90 days or more past due) \$\_\_\_\_

13. Debtor concentration limit - 25% of total A/R \$\_\_\_\_\_

<del>Exhibit A</del> 4864-5069-8851 v.9

(Line 6 of EAR)

<del>A-3</del> 4<del>864-</del> <del>5069-</del> <del>8851 v.9</del> BORROWERS:

# **DIRECT DIGITAL HOLDINGS, INC.**

<del>By:</del> \_\_\_\_\_ <del>Name:</del> Titlo:

Title:

# DIRECT DIGITAL HOLDINGS, LLC

# **COLOSSUS MEDIA, LLC**

<del>By:</del> \_\_\_\_ <del>Name:</del> <del>Title:</del>

# HUDDLED MASSES LLC

<del>By:</del> \_\_\_\_ <del>Name:</del> <del>Title:</del>

# ORANGE142, LLC

<del>By:</del> \_\_\_\_ <del>Name:</del> <del>Title:</del>

<del>Exhibit A</del> 4864-5069-8851 v.9

#### EXHIBIT B

#### COMPLIANCE CERTIFICATE

FOR QUARTER ENDED (THE "SUBJECT QUARTER")

BANK: EAST WEST BANK

BORROWERS: Direct Digital Holdings, Inc., a Delaware corporation ("DDH Holdings"), Direct Digital Holdings, LLC, a Texas limited liability company ("Direct Digital"), Colossus Media, LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("HM") and Orange142, LLC, a Delaware limited liability company ("Orange" and together with DDH Holdings, Direct Digital, Colossus and HM "Borrowers" and each individually a "Borrower")

This Certificate is delivered under the Credit Agreement (the "Agreement") dated as of July 7, 2023, by and among Borrowers and Bank as such may have been amended, supplemented or replaced. Capitalized terms used in this Certificate shall, unless otherwise indicated, have the meanings set forth in the Agreement. On behalf of Borrowers, the undersigned certifies to Bank on the date hereof that (a) no Default or Event of Default has occurred and is continuing, (b) all representations and warranties of Borrowers contained in the Agreement and in the other Loan Documents are true and correct in all material respects, and (c) the information set forth below hereto is true and correct as of the last day of the Subject Quarter:

#### **DESCRIPTION OF COVENANT**

(1) Fixed Charge Coverage Ratio of not less than

#### (2) Total Funded Debt to EBITDA Ratio of not greater than (Section 9.02 of Agreement): to 1.0

June 30, 2023 and the last day of each fiseal quarter through and including December 31, 2023	<del>3.50:1.00</del>
March 31, 2024 and the last day of each fiseal quarter through and including March 31, 2025	<del>3.25:1:00</del>
June 30, 2025 and thereafter	<del>3:00:1.00</del>

(3) Liquid Assets plus Revolving Credit Availability in the minimum amount of \$1,000,000.

(Section 9.03 of Agreement) <del>\$</del>\_\_\_\_

Exhibit A 4864-5069-8851 v.9 BORROWERS:

# **DIRECT DIGITAL HOLDINGS, INC.**

<del>By:</del> \_\_\_\_ <del>Name:</del> <del>Title:</del>

# DIRECT DIGITAL HOLDINGS, LLC

# **COLOSSUS MEDIA, LLC**

<del>By:</del> \_\_\_\_ <del>Name:</del> <del>Title:</del>

# HUDDLED MASSES LLC

<del>By:</del> \_\_\_\_ <del>Name:</del> <del>Title:</del>

# ORANGE142, LLC

<del>By:</del> \_\_\_\_ <del>Name:</del> <del>Title:</del>

<del>Exhibit A</del> 4864-5069-8851 v.9

#### EXHIBIT C

Anything herein to the contrary notwithstanding, the Liens and security interests securing the obligations evidenced by this revolving eredit note, the exercise of any right or remedy with respect hereto and certain of the rights of the holder hereof are subject to the provisions of the Intercreditor Agreement, dated as of July 7, 2023 (as amended, restated, supplemented, substituted, replaced or otherwise modified from time to time, the "*Intercreditor Agreement*"), by and between Lafayette Square Loan Servicing, LLC (in its capacity as agent for the LS Facility Lenders and together with its successors and assigns, the "*LS Facility Agent*"), for and on behalf of the LS Facility Creditors and each other LS Facility Claimholder (each as defined in the Intercreditor Agreement) from time to time, and East West Bank ("*EWB*"), acting on behalf of each A/R Facility Claimholder (as defined in the Intercreditor Agreement). In the event of any eonfliet between the terms of the Intercreditor Agreement and this revolving credit note, the terms of the Intercreditor Agreement.

#### **REVOLVING CREDIT NOTE**

<del>\$5,000,000</del> <u>, 20</u>

FOR VALUE RECEIVED, Direct Digital Holdings, Inc., a Delaware corporation ("DDH Holdings"), Direct Digital Holdings, LLC, a Texas limited liability company ("Direct Digital"), Colossus Media, LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("HM") and Orange142, LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited liability company ("HM") and Orange142, LLC, a Delaware limited liability company ("Colossus"), Huddled Masses LLC, a Delaware limited and HM "Borrowers" and each individually a "Borrower"), hereby unconditionally, jointly and severally, promise to pay to the order of EAST WEST BANK, a California state bank ("Lender"), in accordance with the provisions of the Credit Agreement (as hereinafter defined), the principal sum of Five Million Dollars (\$5,000,000), or such other amount as may from time to time be advanced by Lender as Revolving Credit Advance to or for the benefit or account of Borrower pursuant to the terms of that certain Credit Agreement, dated as of the date hereof (as amended, restated, extended, supplemented or otherwise modified in writing from time to time, the "Credit Agreement;" the terms defined therein being used herein as therein defined), between Borrower and Lender.

Borrower promises to pay interest on the unpaid principal amount of this Note from the date hereof until the Revolving Credit Advances made by Lender are paid in full, at such interest rates and at such times as provided in the Credit Agreement. All payments of principal and interest shall be made to Lender in Dollars in immediately available funds at Lender's Principal Office. If any amount is not paid in full when due hereunder, then such unpaid amount shall bear interest, to be paid upon demand, from the due date thereof until the date of actual payment (and before as well as after judgment) computed at the per annum rate set forth in the Credit Agreement.

This Note is the Revolving Credit Note referred to in the Credit Agreement, is entitled to the benefits thereof and may be prepaid in whole or in part subject to the terms and conditions provided therein. This Note is also entitled to the benefits of the Guaranty. Upon the occurrence and continuation of one or more of the Events of Default specified in the Credit Agreement, all amounts then remaining unpaid on this Note shall become, or may be declared to be, immediately due and payable all as provided in the Credit Agreement. Lender may also attach schedules to this Note and endorse thereon the date, amount and maturity of its Revolving Credit Advances and payments with respect thereto.

Exhibit B

4<del>864-</del> <del>5069-</del> <del>8851 v.</del> Borrower, for itself, its successors and assigns, hereby waives diligence, presentment, protest and demand and notice of protest, demand, dishonor and non-payment of this Note.

THIS NOTE, AND ANY CLAIM, CONTROVERSY, OR DISPUTE ARISING OUT OF OR IN CONNECTION WITH THIS NOTE, SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF TEXAS.

[Remainder of Page Intentionally Left Blank; Signature Page Follows.]

Exhibit B

4<del>864-</del> <del>5069-</del> <del>8851 v.9</del> IN WITNESS WHEREOF, Borrower, intending to be legally bound hereby, has duly executed this Note as of the day and year first written above.

BORROWER:

**DIRECT DIGITAL HOLDINGS, INC.** <del>By:</del> \_\_\_ Name: Title: **DIRECT DIGITAL HOLDINGS, LLC** <del>By:</del> \_\_\_ Name: Title: **COLOSSUS MEDIA, LLC** <del>By:</del> \_\_\_\_ Name: Title: HUDDLED MASSES LLC <del>By:</del> \_\_\_\_ Name: Title: ORANGE142, LLC <del>By:</del> \_\_\_ Name: Title:

[Different first page setting changed from off in original to on in modified.]. Exhibit C 4864-5069-8851 v.9

Summary report: Litera Compare for Word 11.9.1.1 Document comparison done on 10/15/2024 4:09:50 PM		
Intelligent Table Comparison: Active		
Original DMS: iw://dmsus.mwllp.dom/ACTIVE/195349912/3		
Modified DMS: iw://dmsus.mwllp.dom/ACTIVE/195349912/7		
Changes:		
Add	181	
<del>Delete</del>	416	
Move From	19	
Move To	19	
Table Insert	0	
Table Delete	1	
Table moves to	0	
Table moves from	0	
Embedded Graphics (Visio, ChemDraw, Images etc.)	0	
Embedded Excel	0	
Format changes	0	
Total Changes:	636	

#### Certification Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002

I, Mark Walker, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Direct Digital Holdings, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 13, 2024

/s/ Mark Walker

Mark Walker, Chairman and Chief Executive Officer (Principal Executive Officer)

#### Exhibit 31.2

#### Certification Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002

#### I, Diana P. Diaz, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Direct Digital Holdings, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(f)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 13, 2024

/s/ Diana P. Diaz

Diana P. Diaz, Chief Financial Officer (Principal Financial and Accounting Officer)

#### Certification Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

In connection with the Quarterly Report on Form 10-Q of Direct Digital Holdings, Inc. (the "Company") for the period ended September 30, 2024, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Mark Walker, Chief Executive Officer of the Company, hereby certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: November 13, 2024

/s/ Mark Walker

Mark Walker, Chairman and Chief Executive Officer (Principal Executive Officer)

#### Certification Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

In connection with the Quarterly Report on Form 10-Q of Direct Digital Holdings, Inc. (the "Company") for the period ended September 30, 2024, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Diana P. Diaz, Chief Financial Officer of the Company, hereby certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: November 13, 2024

/s/ Diana P. Diaz Diana P. Diaz, Chief Financial Officer (Principal Financial and Accounting Officer)